

Maxillomandibular advancement surgery

New insight into its role in obstructive sleep apnea management

Ning Zhou



Maxillomandibular advancement surgery

New insight into its role in obstructive sleep apnea management

Ning Zhou

Publication of this thesis was generously supported by:



ISBN: 978-94-6419-714-3

Cover and layout design: Ferdinand van Nispen, my-thesis.nl Printing: Gildeprint, gildeprint.nl

[©]Copyright 2023: Ning Zhou, Amsterdam, The Netherlands

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without prior written permission of the author.

Maxillomandibular advancement surgery

New insight into its role in obstructive sleep apnea management

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Universiteit van Amsterdam op gezag van de Rector Magnificus prof. dr. ir. P.P.C.C. Verbeek ten overstaan van een door het College voor Promoties ingestelde commissie, in het openbaar te verdedigen in de Agnietenkapel op dinsdag 28 maart 2023, te 12.00 uur

> door Ning Zhou geboren te Shandong

PROMOTIECOMMISSIE

Promotores:	prof. dr. J. de Lange prof. dr. F. Lobbezoo	Universiteit van Amsterdam Universiteit van Amsterdam
Copromotores:	prof. dr. N. de Vries prof. dr. G. Aarab	Universiteit van Amsterdam Universiteit van Amsterdam
Overige leden:	prof. dr. A.G. Becking prof. dr. F.R. Rozema prof. dr. F.G. Dikkers prof. dr. T.J.J. Maal dr. S.Y. Liu dr. M.J.L. Ravesloot	Universiteit van Amsterdam Universiteit van Amsterdam Universiteit van Amsterdam Radboud Universiteit Stanford University OLVG

Faculteit der Tandheelkunde

TABLE OF CONTENTS

CHAPTER 1	General introduction	
CHAPTER 2	Effects of maxillomandibular advancement on respiratory function and facial esthetics in obstructive sleep apnea patients with versus without maxillomandibular deficiency	25
CHAPTER 3	Maxillomandibular advancement for obstructive sleep apnea: A retrospective prognostic factor study for surgical response	47
CHAPTER 4	Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement	73
CHAPTER 5	Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis	95
CHAPTER 6	Maxillomandibular advancement and upper airway stimulation for treatment of obstructive sleep apnea: A systematic review	145
CHAPTER 7	Intra-individual variation of upper airway measurements based on computed tomography	173
CHAPTER 8	Obstructive sleep apnea caused by acromegaly: Case report	195
CHAPTER 9	General discussion	203

CHAPTER 10	Summary in English and Dutch	217
	Summary	219
	Samenvatting	222
APPENDICES		227
	Author contributions	229
	About the author	231
	PhD portfolio	232
	Acknowledgements	235

CHAPTER

านอุณรรอรระ

General introduction

BACKGROUND

Obstructive sleep apnea (OSA) is the most common sleep-related breathing disorder¹. It is characterized by repetitive episodes of complete or partial upper airway obstruction during sleep, resulting in cessations (apneas) or reductions (hypopneas) in ventilation, with consequent hypoxia, hypercapnia, and/or related arousals^{2, 3}. In the third edition of the International Classification of Sleep Disorders (ICSD-3), OSA is categorized into adult OSA and pediatric OSA¹. In this thesis, we only focus on adult OSA.

OSA has been reported to be present in 9% to 38% of the general adult population: 13% to 33% of men, and 6% to 19% of women⁴. People with OSA may have complaints, such as loud or irregular snoring, daytime sleepiness, nocturia, chocking and gasping in sleep, and morning headache, but many are asymptomatic^{2, 3, 5}. When left untreated, OSA can increase the risk of the development of cardiovascular disease, metabolic disease, psychiatric disorders, neurocognitive impairment, and all-cause morbidity^{3, 6-9}. Additionally, the impairment in daytime function associated with untreated OSA is also a safety hazard^{3, 10}. The adverse consequences of OSA highlight the importance of early diagnosis and effective management of this disorder.

Pathogenesis and risk factors of OSA

OSA is a highly heterogeneous disorder. Currently, four major phenotypic traits are thought to contribute to the pathogenesis of OSA, which include anatomical factors (narrow or collapsible upper airway), impaired pharyngeal dilator muscle function, premature awakening to airway narrowing (a low respiratory arousal threshold), and an oversensitive ventilatory control system (high loop gain)¹¹⁻¹³. Other factors, like end-expiratory lung volume and redistribution of body fluid, may also play a role^{14, 15}. The relative contribution of these traits to OSA pathogenesis varies substantially between patients.

Upper airway anatomy/collapsibility

Although OSA is a multifactorial disorder, a certain level of upper airway anatomical impairment is a prerequisite cause of its development. Studies using different imaging techniques (e.g., computed tomography [CT]) have provided insight into the OSA pathogenesis. Compared to non-OSA control subjects, patients with OSA were found to have a narrower pharyngeal space and a longer upper airway^{16, 17}. A

systematic review showed that the most relevant anatomical characteristics of the upper airway that is linked to OSA pathogenesis is a small minimum cross-sectional area¹⁸.

The upper airway consists of a collapsible segment (the pharynx), extending from the hard palate to the larynx. The collapsibility of the pharynx is determined by the difference between the pressure of the surrounding tissue and the stability of pharyngeal wall, which can be quantified by pharyngeal critical closing pressure (Pcrit)¹⁹. A high Pcrit indicates a high collapsibility of the airway. Upper airway collapse occurs when Pcrit exceeds the intraluminal pressure.

Dilator muscle function

The pharyngeal dilator muscles play a crucial role in maintaining upper airway stability. The largest pharyngeal dilator muscle is the genioglossus muscle, of which the activity is related to respiration, upper airway negative pressure, and arousal state²⁰. During wakefulness, for both OSA patients and non-OSA subjects, the activation of pharyngeal dilator muscles is effective to oppose the negative upper airway pressure and hold the airway open. However, during sleep, when basal and compensatory dilator muscle activity cannot counteract the inherently impaired airway anatomy and the negative airway pressure, the upper airway patency cannot maintain in OSA patients¹².

Respiratory arousal threshold

A cortical arousal from sleep during a respiratory event occurs when negative intrathoracic pressure reaches a threshold (i.e., respiratory arousal threshold)²¹. Arousals have been considered crucial to reopen the upper airway following a respiratory event. However, recent evidence suggests that arousals may not be necessary to reopen the airway in many cases. Frequent arousals perpetuate blood-gas disturbances, breathing instability, sleep fragmentation, and subsequent upper airway collapse during sleep. A low respiratory arousal threshold may lead to, or worsen OSA^{21, 22}.

Ventilatory control system

During sleep, ventilation is primarily governed by the metabolic chemoreflex control system, where the partial pressure of carbon dioxide (P_{CO2}) tightly regulates the ventilatory rate^{23, 24}. The sensitivity of the negative feedback system controlling

ventilation is characterized by "loop gain". The "loop gain" is calculated as the ventilatory response/ventilator disturbance ratio²⁴. A high loop gain indicates an unstable control system. Individuals with a high loop gain have a larger ventilatory response to a small change in CO_2 . This hyperventilation may increase the magnitude of hypocapnia and consequently result in low ventilatory drive with subsequent upper airway collapse²⁵.

Risk Factors

The risk factors for OSA mainly include obesity, increased age, male gender, family history and genetics, cranial facial anatomy resulting in a narrow airway, nasal congestion, alcohol, and smoking^{26, 27}. Postmenopausal women are reported to have a higher risk of OSA than premenopausal women²⁸. In addition, some other factors have also been identified as risk factors for OSA, including the use of certain medications (e.g., opioids and benzodiazepines²⁹) and endocrine disorders (e.g., hypothyroidism³⁰, polycystic ovarian syndrome³¹, and acromegaly³²).

Diagnosis of OSA

The diagnosis of OSA is based on clinical presentation and physical examination findings suggestive of the disorder, coupled with the objective demonstration of abnormal breathing during sleep, by means of polysomnographic recording^{3, 13, 33}.

Clinical assessment

The clinical presentation in people with suspected OSA can vary among individuals. The most common symptoms suggestive of OSA are snoring and excessive daytime sleepiness. Other symptoms include, but are not restricted to, witnessed apneas, nocturnal choking or gasping, insomnia, nocturia, unrefreshed sleep, morning headaches, dry mouth, memory impairment, and fatigue^{3, 33}.

The physical examination of a patient with suspected OSA should include body mass index, neck circumference, nasal examination, pharyngeal anatomy including the lateral wall, soft palate, uvula, and tongue, facial skeletal characteristics of the maxilla and mandible, and dental status including occlusion^{33, 34}.

Polysomnography

The "gold standard" for the diagnosis of OSA is full polysomnography (PSG). Multiple physiologic signals are monitored by PSG during sleep, and generally include brain

waves (electroencephalogram), eye movements (electrooculogram), chin muscle activity (chin electromyogram), air flow, thoracic and abdominal movements, blood oxygen levels (oximetry), heart rate and rhythm (electrocardiogram), leg movements (leg electromyogram), body position, and audio recordings³⁵.

The collected data can be scored according to the American Academy of Sleep Medicine (AASM) scoring manual. Apnea is defined as a decrease of more than 90% in the nasal-oral airflow with a duration of at least 10 seconds. Hypopnea is defined as a decrease of more than 30% in the airflow for at least 10 seconds, combined with an at least 3% oxygen desaturation and/or arousal³⁶. The apnea-hypopnea index (AHI) is defined as the number of apneas and hypopneas per hour of sleep. ICSD-3 defines OSA as: 1) an AHI \geq 5 events/h combined with one or more OSA-related symptoms or associated medical or psychiatric disorders; or 2) an AHI \geq 15 events/h without OSA-related symptoms or comorbidities¹. Although some alternative measures of OSA severity, such as hypoxic burden, have been suggested, AHI has been the most widely used measure of OSA severity³⁷. Based on the AHI, the severity of OSA is defined as: mild (AHI 5-15 events/h), moderate (AHI 16-30 events/h), and severe (AHI > 30 events/h).

Upper airway assessment

Multiple imaging techniques have been used to assess the upper airway abnormalities, such as lateral cephalogram, CT, magnetic resonance imaging (MRI), traditional nasopharyngoscopy, and drug-induced sleep endoscopy (DISE)^{38, 39}. As this thesis involves the reliability of upper airway measurements on CT and the role of DISE in improving patient selection for OSA treatment, only CT and DISE are further introduced below.

Computed tomography CT is a fast, non-invasive, and commonly available technique allowing for assessing the upper airway three-dimensionally (3D). It can provide excellent imaging of the airway and its surrounding soft tissues and bone structure. A specific type of CT, dynamic 3D CT, can be performed to acquire dynamic 3D imaging of the upper airway over the respiration cycle⁴⁰. Radiation exposure is the main downside of CT scanning. CT analysis of the upper airway has helped gain more insight into the OSA pathogenesis.

Chapter 1

Drug-induced sleep endoscopy DISE is an endoscopic examination performed during pharmacologically induced sleep in order to identify the site(s), degree(s), and configuration(s) of upper airway collapse. DISE is mainly indicated when continuous positive airway pressure (CPAP; the gold standard of OSA treatment) fails or is not accepted by the patient, and alternative treatment modalities (e.g., upper airway surgery, hypoglossal nerve stimulation [HNS], mandibular advancement device [MAD], or a combination of different therapies) are considered³⁹. Absolute contraindications are American Society of Anesthesiologists (ASA) score IV, pregnancy, and allergy to DISE sedative agents³⁹.

During DISE, different passive maneuvers can be performed with the aim of predicting the response to some specific treatment modalities⁴¹. A jaw thrust maneuver is a gentle advancement of the mandible up to approximately 5 mm, which may mimic the effect of MAD.

The Velum Oropharynx Tongue base Epiglottis (VOTE) classification is widely used for documenting the DISE findings⁴² (**Figure 1**). It involves the four most common sites of collapse in the upper airway: velum, oropharynx, tongue base, and epiglottis. The degree of obstruction at the four sites can be: 0 (up to 50 % of obstruction), 1 (50-75 %), 2 (75-100 %), or X (not visualized). As for the collapse configuration(s), a distinction is made between anteroposterior, lateral, and concentric. The possible level(s), degree(s), and configuration(s) of collapse based on the VOTE classification system are shown in **Table 1**⁴².



Figure 1. VOTE classification for DISE

Table 1. The VOTE classification⁴²

Structure	Degree of	Configuration ^c		
	obstruction ^a –	AP	Lateral	Concentric
Velum				
Oropharynx ^b				
Tongue base				
Epiglottis				

AP, anteroposterior.

^b Oropharynx obstruction can be distinguished as related solely to the tonsils or including the lateral walls.

^c Configuration noted for structures with degree of obstruction > 0.

Management of OSA

The most common treatment options for OSA include behavior therapy, such as weight loss, reducing alcohol and sedative use, and positional therapy; CPAP, which is the current gold standard treatment for especially severe OSA; MAD; and surgical therapy, such as upper airway surgery (single- and multi-level surgery), maxillomandibular advancement (MMA), and hypoglossal nerve stimulation (HNS)^{43,44}.

In this thesis, we mainly focus on MMA surgery in the treatment of OSA. Additionally, the clinical efficacy and safety of MMA were compared with those of multilevel surgery (MLS) and HNS. Hence, only MMA, MLS, and HNS are further introduced below.

Maxillomandibular advancement

MMA, also known as bimaxillary advancement surgery, is a form of facial skeletal surgery. It involves a combination of a LeFort I osteotomy of the maxilla and a bilateral sagittal split osteotomy of the mandible to advance the maxillomandibular complex, with or without counterclockwise rotation of the complex^{45, 46} (**Figure 2**). It has been suggested that by altering the skeletal framework, MMA can enlarge the entire retropalatal and retrolingual airway and stabilize the pharyngeal dilator muscles, thereby reducing upper airway collapsibility^{47, 48}. The reported rate of surgical success of MMA ranges from 65% to 100%⁴⁸⁻⁵⁰. A meta-analysis suggested that surgical success of MMA is associated with younger age, lower preoperative weight and AHI, as well as greater degree of maxillary advancement⁵¹.

^a Degree of obstruction: 0 = no obstruction; 1 = partial obstruction; 2 = complete obstruction.

The most current American Academy of Sleep Medicine (AASM) practice guidelines recommends that "MMA is indicated for surgical treatment of severe OSA in patients who cannot tolerate or who are unwilling to adhere to positive airway pressure therapy, or in whom oral appliances, which are more often appropriate in mild and moderate OSA patients, have been considered and found ineffective or undesirable"⁵².

The relative contraindications for MMA mainly include significant medical comorbidities (e.g., severe heart failure), unstable psychological problems, morbid obesity (body mass index [BMI] > 35 kg/m²), older age, and alcohol and/or drug dependency^{53, 54}.



Figure 2. Maxillomandibular advancement surgery. LeFort I osteotomy, maxillary advancement, and rigid fixation; bilateral sagittal split osteotomy, mandibular advancement, and rigid fixation. From *Alex Mit/Shutterstock.com*. Usage with permission.

Multilevel surgery

MLS is a combined procedure (simultaneous surgery) or stepwise multiple operations (staged surgery), which involves velopharyngeal and hypopharyngeal regions. The surgical procedures involved in MLS are heterogeneous. The most commonly performed MLS includes a palatal surgery (e.g., uvulopalatopharyngoplasty [UPPP], expansion sphincter pharyngoplasty [ESP]) as a basic technique, with a second

procedure (e.g., radiofrequency thermotherapy of the tongue base, genioglossus advancement, hyoidthyroidpexia) designed to improve the hypopharynx⁵⁵⁻⁵⁷. The surgical success rate of MLS varies amongst studies and ranges from 47.5% to 100%⁵⁶⁻⁶⁰.

According to the most current AASM practice, "use of multi-level or stepwise surgery (MLS), as a combined procedure or as stepwise multiple operations, is acceptable in patients with narrowing of multiple sites in the upper airway, particularly if they have failed UPPP as a sole treatment"⁵².

The relative contraindications for MLS mainly include significant medical or psychiatric comorbidities, morbid obesity (BMI > 35 kg/m²), and specific anatomical contraindications to the upper airway surgery (e.g., severe retrognathia)⁶¹. Additionally, a complete concentric collapse of the soft palate (CCCp) and a lateral oropharyngeal collapse during DISE may be negative prognostic factors for MLS^{62, 63}.

Hypoglossal nerve stimulation

HNS is a novel therapy for patients with moderate to severe OSA². In contrast to traditional surgical approaches for OSA, HNS is a non-anatomical modifying surgery, which involves a surgical procedure for device implantation. HNS device works by electrically stimulating the branches of the hypoglossal nerve that innervate muscles responsible for protruding the tongue and thus maintaining upper airway patency during sleep⁶⁴. Current evidence suggests that HNS therapy can improve upper airway patency not only at retrolingual level but also at retropalatal level⁶⁵. Although it remains to be proven, mechanical palatoglossal coupling may explain the multilevel effect of HNS⁶⁶. Currently, the most commonly used HNS device is Inspire upper airway stimulation (UAS) system (Inspire Medical Systems, Maple Grove, MN, USA). Previous studies have shown that UAS therapy is successful in 50% to 77.8% of the patients⁶⁷⁻⁶⁹.

HNS therapy is currently indicated for age 22 years or older, moderate to severe OSA (AHI 15 events/h to 65 events/h), and difficulty accepting or adhering to CPAP⁶⁴.

This therapy is currently not considered appropriate for > 25% central and mixed apneas of the total AHI, a BMI > 32 kg/m², patients who are pregnant or plan to become pregnant, preexisting anatomic variants or neurologic disorders, and patients who require MRI⁶⁴. In addition, CCCp is an absolute contraindication for unilateral HNS therapy⁷⁰.

THESIS OUTLINE

Main research questions

- Are there differences in the effects of MMA on respiratory function and facial esthetics between OSA patients with and without anteroposterior maxillomandibular deficiency? (**Chapter 2**)
- Which clinical features are predictive of MMA surgical outcome (response versus non-response) in patients with OSA? (**Chapter 3** and **4**)
- Are there differences in the clinical efficacy and safety between MMA and other multilevel approaches (MLS and UAS) for the treatment of OSA? (**Chapter 5** and **6**)
- What is the degree of the natural intra-individual variation in the upper airway measurements on CT scans at two time points? (**Chapter 7**)

The overall aim of this thesis is to gain further insight into the role of MMA in treating OSA, which may contribute to the optimization of surgical management of OSA.

Thesis chapters

This chapter (**chapter 1**) presents a general introduction, including the background of OSA, upper airway imaging techniques, and surgical therapies for OSA described in this thesis.

Chapter 2 presents a comparison of the MMA outcome between OSA patients with and without anteroposterior maxillomandibular deficiency. More specifically, that study compares the effects of MMA on respiratory function between patients with and without maxillomandibular deficiency based on PSG variables and patient satisfaction in postoperative breathing; and compares the changes in facial esthetics after MMA between both groups based on cephalometric measurements and patient satisfaction in postoperative facial esthetics.

In **chapter 3**, we explore the existence of the predictors of MMA surgical outcome (response versus non-response), from the most commonly available clinical data including patient-related, polysomnographic, cephalometric, and surgical variables.

Chapter 4 focuses on the role of DISE in the prediction of MMA surgical outcome. The tested hypothesis is that the upper airway collapse site(s), configuration(s), and degree(s) during baseline DISE can predict MMA surgical outcome. Additionally, the value of jaw thrust maneuver during DISE in the prediction of MMA outcome is explored.

Chapter 5 presents a systematic review and meta-analysis, in which the clinical efficacy and safety are compared between MMA and MLS in the treatment of OSA.

In **chapter 6**, a systematic review is presented with the aim to comparatively evaluate the efficacy and safety of MMA and UAS in the treatment of OSA.

In **chapter 7**, we develop and validate a 3D method to characterize the upper airway on CT. Using this method, the natural intra-individual variation in the upper airway measurements on supine CT scans at two different time points (3 to 6 months interval) is quantified.

Chapter 8 reports a patient who was referred for consultation of MMA surgery for severe OSA but was subsequently diagnosed with acromegaly. After transsphenoidal resection of a pituitary adenoma, the patient's OSA was almost completely resolved.

Chapter 9 provides the main findings of the studies included in this thesis, general conclusions, and suggestions for future studies.

Chapter 10 presents a summary of this thesis in English and Dutch.

REFERENCES

- 1. American Academy of Sleep Medicine. International classification of sleep disorders, 3rd ed. Darien, IL: American Academy of Sleep Medicine; 2014.
- 2. Veasey SC, Rosen IM. Obstructive Sleep Apnea in Adults. N Engl J Med. 2019;380(15):1442-9.
- 3. Charibeh T, Mehra R. Obstructive sleep apnea syndrome: natural history, diagnosis, and emerging treatment options. *Nat Sci Sleep.* 2010;2:233-55.
- Senaratna CV, Perret JL, Lodge CJ, et al. Prevalence of obstructive sleep apnea in the general population: A systematic review. Sleep Med Rev. 2017;34:70-81.
- 5. Arnardottir ES, Bjornsdottir E, Olafsdottir KA, Benediktsdottir B, Gislason T. Obstructive sleep apnoea in the general population: highly prevalent but minimal symptoms. *Eur Respir J*. 2016;47:194-202.
- 6. Knauert M, Naik S, Gillespie MB, Kryger M. Clinical consequences and economic costs of untreated obstructive sleep apnea syndrome. *World J Otorhinolaryngol Head Neck Surg.* 2015;1:17-27.
- 7. Young T, Palta M, Dempsey J, Peppard PE, Nieto FJ, Hla KM. Burden of sleep apnea: rationale, design, and major findings of the Wisconsin Sleep Cohort study. *Wmj.* 2009;108:246-9.
- 8. Lal C, Strange C, Bachman D. Neurocognitive impairment in obstructive sleep apnea. *Chest.* 2012;141:1601-10.
- 9. Rich J, Raviv A, Raviv N, Brietzke SE. All-Cause Mortality and Obstructive Sleep Apnea Severity Revisited. Otolaryngol Head Neck Surg. 2012;147:583-7.
- 10. George CF. Sleep apnea, alertness, and motor vehicle crashes. Am J Respir Crit Care Med. 2007;176:954-56.
- 11. Eckert DJ, White DP, Jordan AS, Malhotra A, Wellman A. Defining phenotypic causes of obstructive sleep apnea. Identification of novel therapeutic targets. *Am J Respir Crit Care Med.* 2013;188:996-1004.
- 12. White DP. Pathogenesis of obstructive and central sleep apnea. Am J Respir Crit Care Med. 2005;172:1363-70.
- 13. Patil SP, Schneider H, Schwartz AR, Smith PL. Adult Obstructive Sleep Apnea: Pathophysiology and Diagnosis. *Chest.* 2007;132:325-37.
- 14. Redolfi S, Arnulf I, Pottier M, et al. Attenuation of obstructive sleep apnea by compression stockings in subjects with venous insufficiency. *Am J Respir Crit Care Med*. 2011;184:1062-6.
- 15. Bikov A, Losonczy G, Kunos L. Role of lung volume and airway inflammation in obstructive sleep apnea. *Respir Investig.* 2017;55:326-33.
- 16. Neelapu BC, Kharbanda OP, Sardana HK, et al. Craniofacial and upper airway morphology in adult obstructive sleep apnea patients: A systematic review and meta-analysis of cephalometric studies. *Sleep Med Rev.* 2017;31:79-90.
- 17. Abramson Z, Susarla S, August M, Troulis M, Kaban L. Three-dimensional computed tomographic analysis of airway anatomy in patients with obstructive sleep apnea. J Oral Maxillofac Surg. 2010;68:354-62.
- 18. Chen H, Aarab G, de Ruiter MH, de Lange J, Lobbezoo F, van der Stelt PF. Three-dimensional imaging of the upper airway anatomy in obstructive sleep apnea: a systematic review. *Sleep Med*. 2016;21:19-27.
- 19. Oliven A, Aspandiarov E, Gankin I, Gaitini L, Tov N. Collapsibility of the relaxed pharynx and risk of sleep apnoea. *Eur Respir J.* 2008;32:1309-15.
- 20. Jordan AS, White DP. Pharyngeal motor control and the pathogenesis of obstructive sleep apnea. *Respir Physiol Neurobiol.* 2008;160:1-7.
- 21. Eckert DJ, Younes MK. Arousal from sleep: implications for obstructive sleep apnea pathogenesis and treatment. J Appl Physiol. 2014;116:302-13.
- 22. Younes M. Role of arousals in the pathogenesis of obstructive sleep apnea. Am J Respir Crit Care Med. 2004;169:623-33.
- 23. Horner RL. Emerging principles and neural substrates underlying tonic sleep-state-dependent influences on respiratory motor activity. *Philos Trans R Soc Lond B Biol Sci.* 2009;364:2553-64.
- 24. Deacon-Diaz N, Malhotra A. Inherent vs. induced loop gain abnormalities in obstructive sleep apnea. *Front Neurol.* 2018;9:896.
- 25. Deacon NL, Catcheside PG. The role of high loop gain induced by intermittent hypoxia in the pathophysiology of obstructive sleep apnoea. *Sleep Med Rev.* 2015;22:3-14.
- 26. Young T, Skatrud J, Peppard PE. Risk factors for obstructive sleep apnea in adults. JAMA. 2004;291:2013-6.

- 27. Al Lawati NM, Patel SR, Ayas NT. Epidemiology, risk factors, and consequences of obstructive sleep apnea and short sleep duration. *Prog Cardiovasc Dis.* 2009;51:285-93.
- 28. Young T, Finn L, Austin D, Peterson A. Menopausal status and sleep-disordered breathing in the Wisconsin Sleep Cohort Study. *Am J Respir Crit Care Med.* 2003;167:1181-5.
- 29. Jullian-Desayes I, Revol B, Chareyre E, et al. Impact of concomitant medications on obstructive sleep apnoea. *Br J Clin Pharmacol.* 2017;83:688-708.
- 30. Winkelman JW, Goldman H, Piscatelli N, Lukas SE, Dorsey CM, Cunningham S. Are thyroid function tests necessary in patients with suspected sleep apnea? *Sleep*. 1996;1:790-3.
- Tasali E, Van Cauter E, Ehrmann DA. Polycystic ovary syndrome and obstructive sleep apnea. Sleep Med Clin. 2008;3:37-46.
- 32. Fatti LM, Scacchi M, Pincelli Al, Lavezzi E, Cavagnini F. Prevalence and pathogenesis of sleep apnea and lung disease in acromegaly. *Pituitary.* 2001;4:259-62.
- 33. McNicholas WT. Diagnosis of obstructive sleep apnea in adults. Proc Am Thorac Soc. 2008;5(2):154-160.
- Zonato AI, Bittencourt LR, Martinho FL, Júnior JFS, Gregório LC, Tufik S. Association of Systematic Head and Neck Physical Examination With Severity of Obstructive Sleep Apnea—Hypopnea Syndrome. *Laryngoscope*. 2003;113:973-80.
- Roebuck A, Monasterio V, Gederi E, et al. A review of signals used in sleep analysis. *Physiol Meas.* 2014;35:R1-R57.
- 36. Berry R, Quan S, Abreu A, et al. The AASM manual for the scoring of sleep and associated events: rules, terminology and technical specifications, version 2.6. Darien, IL: American Academy of Sleep Medicine; 2020.
- 37. Malhotra A, Ayappa I, Ayas N, et al. Metrics of sleep apnea severity: beyond the apnea-hypopnea index. *Sleep.* 2021;44:zsab030.
- 38. Schwab RJ. Upper airway imaging. Clin Chest Med. 1998;19:33-54.
- 39. De Vito A, Carrasco Llatas M, Ravesloot MJ, et al. European position paper on drug-induced sleep endoscopy: 2017 Update. *Clin Otolaryngol.* 2018;43:1541-52.
- 40. Fleck RJ, Ishman SL, Shott SR, et al. Dynamic Volume Computed Tomography Imaging of the Upper Airway in Obstructive Sleep Apnea. J Clin Sleep Med. 2017;13:189-96.
- 41. Beelen A, Vonk PE, de Vries N. Drug-induced sleep endoscopy: the effect of different passive maneuvers on the distribution of collapse patterns of the upper airway in obstructive sleep apnea patients. *Sleep Breath*. 2018;22:909-17.
- 42. Kezirian EJ, Hohenhorst W, de Vries N. Drug-induced sleep endoscopy: the VOTE classification. *Eur Arch Otorhinolaryngol*. 2011;268:1233-6.
- 43. Carberry JC, Amatoury J, Eckert DJ. Personalized Management Approach for OSA. Chest. 2018;153:74455.
- 44. Epstein LJ, Kristo D, Strollo PJ, Jr., et al. Clinical guideline for the evaluation, management and long-term care of obstructive sleep apnea in adults. J Clin Sleep Med. 2009;5:263-76.
- 45. Waite P, Wooten V, Lachner J, Guyette R. Maxillomandibular advancement surgery in 23 patients with obstructive sleep apnea syndrome. J Oral Maxillofac Surg. 1989;47:1256-61.
- 46. Liu SY-C, Awad M, Riley RW. Maxillomandibular advancement: contemporary approach at Stanford. *Atlas Oral Maxillofac Surg Clin North Am.* 2019;27:29-36.
- 47. Hsieh YJ, Liao YF, Chen NH, Chen YR. Changes in the calibre of the upper airway and the surrounding structures after maxillomandibular advancement for obstructive sleep apnoea. *Br J Oral Maxillofac Surg.* 2014;52:445-51.
- Liu SY, Huon LK, Iwasaki T, et al. Efficacy of Maxillomandibular Advancement Examined with Drug-Induced Sleep Endoscopy and Computational Fluid Dynamics Airflow Modeling. Otolaryngol Head Neck Surg. 2016;154:189-95.
- 49. Fairburn SC, Waite PD, Vilos G, et al. Three-dimensional changes in upper airways of patients with obstructive sleep apnea following maxillomandibular advancement. J Oral Maxillofac Surg. 2007;65:6-12.
- Rubio-Bueno P, Landete P, Ardanza B, et al. Maxillomandibular advancement as the initial treatment of obstructive sleep apnoea: Is the mandibular occlusal plane the key? Int J Oral Maxillofac Surg. 2017;46:1363-71.
- 51. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.

- 52. Aurora RN, Casey KR, Kristo D, et al. Practice parameters for the surgical modifications of the upper airway for obstructive sleep apnea in adults. *Sleep*. 2010;33:1408-13.
- 53. Prinsell JR. Maxillomandibular Advancement (MMA) in a Site-Specific Treatment Approach for Obstructive Sleep Apnea: A Surgical Algorithm. *Sleep Breath*. 2000;4:147-54.
- 54. Holty JE, Guilleminault C. Surgical options for the treatment of obstructive sleep apnea. *Med Clin North Am.* 2010;94:479-515.
- 55. Lin HC, Friedman M, Chang HW, Curpinar B. The efficacy of multilevel surgery of the upper airway in adults with obstructive sleep apnea/hypopnea syndrome. *Laryngoscope*. 2008;118:902-8.
- 56. Cammaroto G, Montevecchi F, D'Agostino G, Zeccardo E, Bellini C, Meccariello G, Vicini C. Palatal surgery in a transoral robotic setting (TORS): preliminary results of a retrospective comparison between uvulopalatopharyngoplasty (UPPP), expansion sphincter pharyngoplasty (ESP) and barbed repositioning pharyngoplasty (BRP). Acta Otorhinolaryngol Ital. 2017;37:406-9.
- 57. Binar M, Karakoc O, Akcam T, Asik B, Gerek M. Various combinations of velopharyngeal and hypopharyngeal surgical procedures for treatment of obstructive sleep apnea: Single-stage, multilevel surgery. *Am J Otolaryngol.* 2017;38:571-5.
- 58. Friedman M, Lin HC, Gurpinar B, Joseph NJ. Minimally invasive single-stage multilevel treatment for obstructive sleep apnea/hypopnea syndrome. *Laryngoscope*. 2007;117:1859e63.
- 59. Turhan M, Bostanci A, Bozkurt S. Predicting the outcome of modified tongue base suspension combined with uvulopalatopharyngoplasty. *Eur Arch Otorhinolaryngol.* 2015;272:3411e6.
- 60. Sorrenti G, Piccin O, Mondini S, Ceroni AR. One-phase management of severe obstructive sleep apnea: tongue base reduction with hyoepiglottoplasty plus uvulopalatopharyngoplasty. *Otolaryngol Head Neck Surg.* 2006;135:906e10.
- 61. MacKay S, Carney AS, Catcheside PG, et al. Effect of Multilevel Upper Airway Surgery vs Medical Management on the Apnea-Hypopnea Index and Patient-Reported Daytime Sleepiness Among Patients With Moderate or Severe Obstructive Sleep Apnea: The SAMS Randomized Clinical Trial. JAMA. 2020;324:1168-79.
- Meraj TS, Muenz DG, Glazer TA, Harvey RS, Spector ME, Hoff PT. Does drug-induced sleep endoscopy predict surgical success in transoral robotic multilevel surgery in obstructive sleep apnea? *Laryngoscope*. 2017;127:971-6.
- van den Boer C, van Maanen JP, Ravesloot MJL, Benoist L, Vries N. Multilevel surgery with or without genioglossal advancement. In: Sleep Apnea and Snoring: Surgical and Non-Surgical Therapy. Elsevier; 2019. p. 319-327.
- 64. Yu MS, Ibrahim B, Riley RW, Liu SY. Maxillomandibular Advancement and Upper Airway Stimulation: Extrapharyngeal Surgery for Obstructive Sleep Apnea. *Clin Exp Otorhinolaryngol.* 2020;13:225-33.
- 65. Ong AA, Murphey AW, Nguyen SA, et al. Efficacy of upper airway stimulation on collapse patterns observed during drug-induced sedation endoscopy. *Otolaryngol Head Neck Surg.* 2016;154:970-7.
- 66. Heiser C, Edenharter G, Bas M, Wirth M, Hofauer B. Palatoglossus coupling in selective upper airway stimulation. *Laryngoscope*. 2017;127:E378-E383.
- 67. Steffen A, Abrams N, Suurna MV, Wollenberg B, Hasselbacher K. Upper-Airway Stimulation Before, After, or Without Uvulopalatopharyngoplasty: A Two-Year Perspective. *Laryngoscope*. 2019;129:514-8.
- 68. Steffen A, Sommer UJ, Maurer JT, Abrams N, Hofauer B, Heiser C. Long-term follow-up of the German postmarket study for upper airway stimulation for obstructive sleep apnea. *Sleep Breath*. 2020;24:979-84.
- 69. Van de Heyning PH, Badr MS, Baskin JZ, et al. Implanted upper airway stimulation device for obstructive sleep apnea. *Laryngoscope*. 2012;122:1626-33.
- 70. Heiser C, Sommer JU, Hofauer B, et al. Bilateral vs Unilateral Hypoglossal Nerve Stimulation in Patients With Obstructive Sleep Apnea. OTO Open. 2022;6:2473974X221109794.



CHAPTER

cephalometi

Effects of maxillomandibular advancement on respiratory function and facial esthetics in obstructive sleep apnea patients with versus without maxillomandibular deficiency

Ning Zhou, Jean-Pierre T.F. Ho, Frank Lobbezoo, Ghizlane Aarab, Nico de Vries, Jan de Lange

Published in: Int J Oral Maxillofac Surg, 2022. doi: 10.1016/j.ijom.2022.08.012

ABSTRACT

The aim of this study was to compare the effects of maxillomandibular advancement (MMA) on respiratory function between obstructive sleep apnea (OSA) patients with and without maxillomandibular deficiency, and to compare the changes in facial esthetics after MMA between the two groups. MMA-treated patients who had both baseline and follow-up polysomnography (PSG) data and lateral cephalograms were enrolled in this retrospective study. In addition to PSG and cephalometric data, patient satisfaction with postoperative breathing and facial esthetics, and overall satisfaction with the treatment were assessed. Twenty-one patients were classified as not having maxillomandibular deficiency (without-deficiency group) and 40 patients as having maxillomandibular deficiency (withdeficiency group). The improvements in respiratory parameters (e.g., apneahypopnea index) and patient satisfaction with postoperative breathing were comparable in the two groups (P = 0.094-0.713). The changes in facial profile measurements (e.g., nasal prominence, nasolabial angel, and lip positions relative to the true vertical line) and patient satisfaction with postoperative facial esthetics were also comparable in the two groups (P = 0.148-0.983). In conclusion, no significant difference in the effects of MMA on respiratory function and facial esthetics between OSA patients with and without maxillomandibular deficiency was observed.

Keywords: Obstructive sleep apnea; Maxillo-mandibular surgery; Maxillofacial abnormalities; Treatment outcome; Cephalometry

INTRODUCTION

Obstructive sleep apnea (OSA) is increasingly recognized as a public health threat, with a prevalence of 9-38% in the general adult population^{1, 2}. Continuous positive airway pressure (CPAP) was introduced in 1981, and since that time it has become the gold standard therapy for moderate to severe OSA³. However, the efficacy of CPAP is often hampered by poor compliance and low tolerance, which has prompted the search for alternative treatments^{4, 5}.

Riley and Powell pioneered the use of maxillomandibular advancement (MMA) for the treatment of OSA in the mid-1980 s, due to the recognition of the aetiology of OSA, which often involves concomitant maxillary and mandibular deficiencies⁶. MMA consists of advancement of the maxillomandibular complex by osteotomies of the maxilla and mandible, thus leading to enlargement of the pharyngeal space and reduction of pharyngeal collapsibility^{7.8}.

Since the advancement of both jaws is functionally and esthetically beneficial to patients with maxillomandibular deficiency (maxillary and mandibular retrognathia), MMA has been primarily employed as the first-line treatment for OSA patients with this deficiency⁹. Nevertheless, MMA is also used to treat OSA patients without this deficiency but with other specific indications, for example failure or intolerance of other forms of therapy, or complete concentric collapse at the velum level as observed with drug-induced sleep endoscopy (DISE)^{10, 11}. Although MMA is generally thought to be a highly effective surgical therapy for moderate to severe OSA^{12, 13}, some reported rates of surgical success are not that high¹⁴⁻¹⁶, and there is still room for improvement. Besides, due to the limited evidence on the clinical efficacy of MMA in OSA patients without maxillomandibular deficiency^{10, 17}, in clinical practice some sleep specialists are of the opinion that MMA should preferably be performed for OSA patients with significant mandibular deficiency. More evidence on the efficacy of MMA in OSA patients without deficiency is therefore needed.

The unacceptable alteration in facial profile following MMA is also of great concern to OSA patients, especially for those without maxillomandibular deficiency, which may dissuade OSA patients from considering MMA as a treatment option¹⁸. It appears that the esthetic results of MMA in OSA patients without such deficiency have only been evaluated subjectively in two previous studies^{10, 17}.

Therefore, the objectives of this study were (1) to compare the effects of MMA on respiratory function between OSA patients with and without maxillomandibular deficiency based on respiratory parameters measured by polysomnography (PSG) and patient satisfaction with postoperative breathing, and (2) to compare the changes in facial esthetics after MMA between the two groups based on cephalometric measurements and patient satisfaction with postoperative facial esthetics.

METHODS

This retrospective study was deemed not to be subject to the Medical Research Human Subjects Act by the Medical Ethics Committee of the Amsterdam UMC (location AMC) and a formal approval was therefore waived (Reference number W19_170#19.209).

Participants

Participants were recruited from a consecutive series of patients with OSA undergoing MMA in the Department of Oral and Maxillofacial Surgery, Amsterdam UMC (location AMC), between November 2010 and March 2020. The following inclusion criteria were applied: age ≥ 18 years; presence of OSA diagnosed by PSG preoperatively; CPAP failure or intolerance; patients with a follow-up PSG at least 3 months after MMA; and patients with a preoperative cephalogram and a follow-up cephalogram at least 6 months after MMA. The exclusion criteria were as follows: patients who declined the use of their data for research purposes; edentulous individuals; previous history of LeFort I osteotomy and/or bilateral sagittal split osteotomy (BSSO); and syndromic patients.

All of the patients were classified into one of two groups, based on the maxillofacial skeletal criteria of the Steiner analysis¹⁹: those without maxillomandibular deficiency (without-deficiency group), i.e. patients with sella–nasion–A-point angle (SNA) > 80.5° and sella–nasion–B-point angle (SNB) > 78.5° ; those with maxillomandibular deficiency (with-deficiency group), i.e. patients with SNA $\leq 80.5^{\circ}$ and/or SNB $\leq 78.5^{\circ}$.

Polysomnography

All patients included in this study underwent an overnight PSG at baseline and at least 3 months after surgery (mean 5.4 ± 2.8 months). The PSG recordings were scored manually according to the American Academy of Sleep Medicine (AASM)

criteria²⁰. The collected PSG parameters included preoperative and postoperative apnea-hypopnea index (AHI), oxygen desaturation index (ODI), and lowest oxygen saturation (LSAT). Based on Sher's criteria, surgical success was defined as a postoperative AHI of less than 20 events/h and at least 50% reduction in AHI following surgery²¹. Surgical cure was defined as a postoperative AHI of less than 5 events/h²².

Cephalometric measurements

A standard lateral cephalogram was taken before and at least 6 months after surgery (mean 12.8 \pm 7.7 months). Each radiograph was taken in centric occlusion and with the lips in relaxed position. All of the cephalograms were traced by one observer using Viewbox 4 software (dHAL Software, Kifissia, Greece). The landmarks and reference planes are shown in **Figure 1**. The variables were classified into hard tissue variables, upper airway variable (**Figure 2**), and soft tissue variables (**Figure 3**). To assess the reliability of the cephalometric analysis, the same observer randomly selected 10 lateral cephalograms and repeated the measurements 1 month later.

Maxillomandibular advancement surgery

All patients underwent a MMA procedure (LeFort I osteotomy of the maxilla and BSSO of the mandible) with or without counterclockwise rotation of the maxillomandibular complex, performed by two dedicated surgeons. Rigid fixation with titanium miniplates and screws was used to stabilize the maxillary and mandibular osteotomies. Additional procedures, including genioplasty and genioglossus advancement, were performed in certain cases. The patients treated during the earlier years of the study period had a two-dimensionally planned operation, using a standard surgical protocol with the goal of 8-10 mm advancement of the maxillomandibular complex. The patients treated later during the study period had a three-dimensionally planned operation, using a personalized surgical protocol. In the personalized protocol, the final position of the bony segments was determined comprehensively by taking into account multiple patient-related factors, i.e. the severity of the OSA, skeletal pattern, dental occlusion, and facial characteristics. In addition, given that scar tissue resulting from prior upper airway surgery could restrict the MMA surgical movement, when patients had received extensive prior airway surgery, the planned degree of advancement was appropriately reduced. Upper airway collapse patterns were also taken into account when preoperative DISE was available. For example, a sufficient degree of mandibular advancement was planned when there was significant collapse of the tongue base and/or the epiglottis during DISE.

Subjective evaluation

At least 6 months after MMA, a self-assessment questionnaire was mailed to the patients to subjectively evaluate their perceptions of the MMA surgery for OSA. The patients were requested to use an 11-point VAS to separately indicate the level of satisfaction with postoperative breathing, satisfaction with postoperative facial esthetics, and overall satisfaction with the MMA treatment, with o representing "not satisfied at all" and 10 representing "completely satisfied".

Statistical analysis

Data were analysed using IBM SPSS Statistics version 26 (IBM Corp., Armonk, NY, USA). Quantitative data were reported as the median and interguartile range (IQR). Categorical data were reported as the frequency and percentage. To determine intra-observer reliability of the cephalometric analysis, the intra-class correlation coefficient (ICC) was determined for the repeated measurements. Normality was tested using the Shapiro-Wilk test. To compare quantitative variables between the without-deficiency and with-deficiency groups, the independent-samples t-test was used when the data were normally distributed and the Mann-Whitney U-test was used when the data were not normally distributed. Differences between the two groups in categorical variables (sex and presence or absence of counterclockwise rotation, genioglossus advancement, and genioplasty) were assessed by χ_2 test or Fisher's exact test as appropriate. For the comparison of the preoperative and postoperative values, the paired-samples t-test was applied in the case of normally distributed data and the Wilcoxon signed-rank test in the case of non-normally distributed data. Spearman correlation analysis was used to assess the correlation between the reduction in AHI and facial esthetics satisfaction score. A P-value < 0.05 was considered statistically significant.

RESULTS

Patient characteristics

In total, 104 patients underwent MMA for OSA during the study period. Forty-three patients were excluded from the study for the following reasons: declined the

use of their data for research (n=3), edentulous individuals (n=17), and absence of a preoperative or a follow-up cephalogram (n=23). Therefore, 61 patients were included in this study (78.7% male, 21.3% female; median age 50.0 (IQR 44.0, 58.5) years; median body mass index (BMI) 29.0 (IQR 26.4, 31.3) kg/m²; median AHI 49.6 (IQR 35.1, 67.4) events/h).

The ICC of the cephalometric analysis ranged from 0.914 to 0.996, indicating excellent intra-observer reliability²³. According to the skeletal criteria of the Steiner analysis, 21 out of the 61 patients did not have maxillomandibular deficiency preoperatively (median SNA 83.3° (IQR 81.9°, 85.4°), median SNB 79.4° (IQR 78.6°, 82.2°)). Among the 40 patients with maxillary and/or mandibular deficiency (median SNA 79.8° (IQR 76.8°, 81.3°), median SNB 73.3° (IQR 71.6°, 75.9°)), 23 (57.5%) had concomitant maxillary and mandibular deficiency, 16 (40%) had only mandibular deficiency, and one (2.5%) had only maxillary deficiency.

Baseline characteristics of the two study groups, without-deficiency versus with-deficiency

When comparing the baseline characteristics between the without-deficiency and with-deficiency groups, no significant difference was found in the baseline demographic and PSG variables. For baseline soft tissue measurements, a more protrusive position of the upper lip (UL–TVL) (P=0.017), lower lip (LL–TVL) (P<0.001), and soft tissue pogonion (Pog'–TVL) (P<0.001) relative to the true vertical line (TVL) was observed in the without-deficiency group, while a significantly larger facial convexity (P=0.011) was observed in the with-deficiency group. In contrast, the nasal prominence, nasolabial angle, position of the upper lip relative to the E-line (UL–Eline), and position of the lower lip relative to the E-line (LL–E-line) did not differ significantly between the two groups (**Table 1**).

Clinical efficacy of maxillomandibular advancement

The surgical characteristics and airway space in the two study groups are summarized in **Table 2**. The degree of advancement of A-point and B-point did not differ significantly between the two groups, while the degree of advancement of pogonion (Pog) was significantly greater in the with-deficiency group (P=0.046). The increase in posterior airway space (PAS) following MMA did not differ significantly between the two groups (P=0.264) (**Table 2**). Chapter 2

An overview of the preoperative and postoperative PSG values in the two study groups can be found in **Table 3**. A significant reduction in median AHI from 41.6 (IQR 32.1, 62.6) events/h to 11.1 (IQR 6.2, 27.1) events/h in the without-deficiency group (P<0.001) and from 52.2 (IQR 35.3, 69.6) events/h to 10.3 (IQR 4.9, 21.9) events/h in the with-deficiency group (P<0.001) was observed. There was no significant difference between the two groups in the improvements in AHI, ODI, and LSAT. Surgical success was achieved in 57.1% of the without-deficiency group compared to 67.5% of the with-deficiency group (P=0.423), while surgical cure was achieved in 14.3% of the without-deficiency group compared to 27.5% of the with-deficiency group (P=0.398).

Change in facial esthetics after maxillomandibular advancement

After MMA, significant decreases in nasal prominence and nasolabial angle, as well as significant increases in UL–TVL, LL–TVL, Pog'–TVL, UL–E-line, and LL–E-line were observed in both groups (P=0.046 to P<0.001). A significant decrease in facial convexity was found in the with-deficiency group (P<0.001), but not in the without-deficiency group (P=0.070).

The changes in soft tissue measurements were comparable in the two groups. Postoperatively, UL–TVL (P=0.002), LL–TVL (P<0.001), and Pog'–TVL (P<0.001) were more protrusive in the without-deficiency group than in the with-deficiency group and the facial convexity was significantly lower in the without-deficiency group (P=0.012), while the nasal prominence, nasolabial angle, UL–E-line, and LL–E-line were similar in the two groups (**Table 4**).

Subjective assessment of patient satisfaction

Thirty (49.2%) questionnaires were completed and returned: 10 by patients without maxillomandibular deficiency and 20 by patients with deficiency. In the without-deficiency group, the number of patients reporting a satisfaction score \geq 7 in terms of postoperative breathing, facial esthetics, and overall satisfaction was six (60%), five (50%), and four (40%), respectively; in the with-deficiency group, it was 10 (50%), 13 (65%), and 11 (55%), respectively. The number of patients in the without-deficiency group reporting a satisfaction score<3 in terms of breathing, facial esthetics, and overall satisfaction was one (10%), two (20%), and four (40%), respectively; in the with-deficiency group, it was four (20%) for all.

The median VAS scores for satisfaction for both groups are shown in **Table 5**. The without-deficiency group reported the highest level of satisfaction with breathing, followed in descending order by facial esthetics and overall satisfaction, while the with-deficiency group reported the highest level of satisfaction with facial esthetics and overall satisfaction, followed by satisfaction with breathing. On comparison of the median VAS satisfaction scores between the two groups, the degree of satisfaction with breathing (P=0.713), satisfaction with facial esthetics (P=0.983), and overall satisfaction (P=0.681) did not differ significantly.

DISCUSSION

This study compared the treatment efficacy and changes in facial esthetics after MMA between OSA patients with and without maxillomandibular deficiency. The main findings were as follows: (1) MMA surgery was equally effective in improving respiratory parameters for patients with and without such deficiency; (2) the changes in soft tissue profile measurements following MMA did not differ significantly between the two groups; and (3) the two groups had similar levels of satisfaction with postoperative breathing and facial esthetics, and overall satisfaction with treatment.

The finding that the effect of MMA on respiratory parameters did not differ significantly between patients with and without deficiency is in line with a previous study by Ronchi et al.¹⁷, even though the two studies used different definitions of maxillomandibular deficiency. Ronchi et al. concluded that the improvements in AHI and Epworth Sleepiness Scale (ESS) after MMA were comparable in the OSA patients with and without skeletal anomalies¹⁷. The present study also found that patient perception of breathing after MMA was mainly positive and similar in both groups, which further supports MMA as an effective treatment option for patients with OSA, even in those without a skeletal deficiency. MMA surgery is generally thought to enlarge the airway space and stiffen the pharyngeal soft tissues by expanding the facial skeletal framework, thereby preventing airway collapse during sleep²⁴. The present study found that after MMA, the increase in PAS was comparable in patients with and without deficiency, which may partially explain the equal efficacy in the two groups. Additionally, it was found that neither baseline AHI nor baseline PAS differed between patients with and without deficiency. This may support the notion that the choice of MMA as the primary treatment for OSA

should depend mainly on the disease severity and restriction of PAS rather than on the dentofacial skeletal characteristics¹⁷.

It is interesting to note that although a surgical success rate of 57.1% and 67.5% was observed in the without-deficiency group and with-deficiency group, respectively, the surgical cure rate was only 14.3% for the without-deficiency group and 27.5% for the with-deficiency group. This difference between the surgical success and cure rates has also been observed in other studies on MMA^{7, 25}. For the patients whose OSA is improved but not cured after MMA, the authors suggest a collaboration between the surgeon and a sleep specialist to find the potential causes of the residual sleep apnea, and to evaluate the necessity for adjunctive therapy based on the severity of the residual OSA, patient symptoms, and patient preferences.

The patients treated earlier in the study period had a two-dimensionally planned operation, using a standard surgical protocol with the goal of 8-10 mm advancement; those treated later in the study period had a three-dimensionally planned operation in which the degree of advancement was personalized according to multiple patient-related factors, such as the severity of the OSA, skeletal pattern, and facial characteristics. It was anticipated that the degree of MMA advancement would be greater in patients with deficiency than in those without deficiency, however there was no significant difference between the two groups in the degree of advancement of A-point and B-point. This was because approximately 70% of the study population were treated with a standard surgical protocol. To further optimize the OSA treatment with MMA, future research should compare the surgical outcomes between the standard and personalized planned MMA.

According to the literature, the facial soft tissue should be evaluated 6 months after orthognathic surgery, in order to allow it to heal nearly completely²⁶. In this study, the facial profile was assessed at least 6 months after surgery (mean 12.8 months); the role of residual oedema in the observed soft tissue changes is thus likely to be negligible. After MMA, the protrusion of the upper lip, lower lip, and chin relative to TVL increased significantly, accompanied by a decrease in nasal prominence, nasolabial angle, and facial convexity. These findings are consistent with those of previous studies^{27, 28}. A finding of interest is that when examining the lip position to E-line²⁹, the protrusion of the upper lip and lower lip increased significantly after MMA, but the increase was less than the increase relative to
TVL. This is because the increased prominence of the chin can balance the lip protrusion relative to the E-line³⁰. According to Ricketts' analysis, the upper lip and lower lip in patients of White European descent should be estimated 4 mm and 2 mm behind the E-line respectively²⁹, with variations among different ethnicities. For both groups, the median of the postoperative UL–E-line (without-deficiency group –3.8 mm; with-deficiency group –3.0 mm) and postoperative LL–E-line (without-deficiency group –1.7 mm; with-deficiency group –2.0 mm) were similar to the norms reported by Ricketts. However, due to the unknown ethnicities of the present study population, this conclusion should be considered with care. Taken together, the findings suggest that although MMA can significantly alter the soft tissue facial profile, the balance between the nose, lips, and chin is acceptable for patients with and without deficiency.

Another point to be noted is that no significant difference was found between patients with and without deficiency with regard to the changes in facial profile measurements. Conley and Boyd²⁷ evaluated the facial soft tissue changes following MMA for the treatment of OSA, and concluded that the changes in soft tissue corresponded to nearly 90% of the underlying skeletal movements for most anatomical sites of the upper lip, lower lip, and chin. In the present study, the magnitude of the skeletal advancement did not differ significantly between the two groups. It is therefore not surprising that the corresponding changes in facial profile were comparable in the two groups.

Interestingly, despite the significant differences observed between the two groups in postoperative facial profile measurements, there was no significant difference between the two groups in perception of facial esthetics. This suggests that these objective soft tissue measurements may not play an important role in patient satisfaction with facial esthetics. This is further supported by the results of the post-hoc Spearman correlation analysis on the correlation between the facial esthetics satisfaction score on the one hand and soft tissue changes and post-surgical soft tissue variables on the other hand, in which only the change in LL–E-line was negatively associated with the degree of satisfaction with facial esthetics (r=-0.542, P=0.002). Thus, it can be advocated that the position of the lower lip in relation to the E-line should be integrated into the MMA surgery plan for OSA treatment. It is important to note that most people do not look at themselves in profile but rather look straight in a mirror, and while there are some soft tissue changes that can be observed by a discerning eye from frontal

Chapter 2

view, they are far less obvious than profile changes. Another point to be noted is that the patients might have imposed their own cultural bias during the subjective evaluation³¹. Additionally, it is likely that OSA patients, especially those without a baseline maxillomandibular deficiency, accept their alteration in facial esthetics due to the improvement in OSA, as the main motivation for MMA in these patients is treatment of the OSA. Nevertheless, no significant correlation was found between the facial esthetics satisfaction score and the improvement in AHI in this study population.

The study results should be interpreted with caution due to certain limitations. Firstly, as with any retrospective analysis, a weakness of the study was the inability to control the data. There is also a potential concern for selection bias, as only 60% of the total MMA cohort were enrolled in this study. However, no significant differences in baseline characteristics (age, sex distribution, BMI, neck circumference, and baseline AHI) were observed between the patients who were included in the study and those who were not. Furthermore, half of the study population did not respond to the questionnaire, which might have caused a non-response bias³². In addition, the incorporation of genioplasty or genioglossus advancement as an additional procedure in MMA should be considered as a confounding factor. Nevertheless, given that the additional procedure was only performed in six patients, it might not have played a leading role in the results observed. Counterclockwise rotation involved in MMA may also have an impact on respiratory function and facial esthetics. However, since counterclockwise rotation was not a main focus of interest in this study and was performed equally in both groups (47.6% vs 52.5%, P=0.717), it was decided not to take it into consideration in the analyses. Lastly, the study cohort comprised predominantly middle-aged and elderly male patients with a relatively high BMI (overweight) and of unknown ethnicity. This limits the generalizability of the findings. Larger, prospective multicentre studies are needed to further confirm the current findings. Additionally, a validated questionnaire would be preferable for the subjective assessments in future research.

Within the limitations of this study, it is concluded that there is no significant difference in the effects of MMA on respiratory function and facial esthetics between OSA patients with and without maxillomandibular deficiency. This supports the view that MMA can also be considered as an appropriate treatment for OSA patients without maxillomandibular deficiency.



Figure 1. Cephalometric landmarks and reference lines. Landmarks: S, sella; N, nasion; A, A-point; B, B-point; Go, gonion; Pog, pogonion; G', soft tissue glabella; Pn, pronasale; Cm, columella; Sn, subnasale; UL, upper lip; LL, lower lip; Pog', soft tissue pogonion. Reference lines: SN, a plane running through S and N; HRL, horizontal reference line, a line through S at 7° from SN; VRL, vertical reference line, a perpendicular line dropping from HRL and passing through S; TVL, true vertical line, a line perpendicular to HRL and passing through Sn; E-line, a line running through Pn and Pog'.

2



Figure 2. Hard tissue and upper airway cephalometric measurements. 1, S–N–A-point angle (SNA); 2, S–N–Bpoint angle (SNB); 3, A-point–N–B-point angle (ANB); 4, distance from A-point to VRL (A–VRL); 5, distance from B-point to VRL (B–VRL); 6, distance from Pog to VRL (Pog–VRL); 7, posterior airway space, width of the airway along Go–B-point line (PAS).



Figure 3. Soft tissue cephalometric measurements. 1, nasal prominence, distance from Pn to a line perpendicular to HRL and passing through UL; 2, nasolabial angle, Cm–Sn–UL angle; 3, distance from UL to TVL (UL–TVL); 4, distance from LL to TVL (LL–TVL); 5, distance from Pog' to TVL (Pog'–TVL); 6, distance from UL to E-line (UL–E-line); 7, distance from LL to E-line (LL–E-line); 8, facial convexity, G'–Sn–Pog' angle.

	Without deficiency (n = 21)	With deficiency (n = 40)	<i>P</i> -value ^a
Demographic variables			
Male:female	19:2	29:11	0.194
Age, years	50.0 (46.0, 57.5)	51.0 (43.3, 59.0)	0.992
BMI, kg/m²	29.6 (26.7, 31.5)	28.9 (25.9, 31.2)	0.976
Neck circumference, cm	42.0 (39.8, 44.3)	43.0 (29.0, 46.0)	0.546
Polysomnographic variable	S		
AHI, events/h	41.6 (32.1, 62.6)	52.2 (35.3, 69.6)	0.168
ODI, events/h	44.9 (28.7, 62.1)	51.0 (29.5, 70.0)	0.438
LSAT, %	81.0 (79.0, 86.0)	79.0 (73.0, 84.0)	0.221
Cephalometric variables – u	ıpper airway		
PAS, mm	8.5 (6.5, 10.5)	8.7 (6.9, 11.7)	0.690
Cephalometric variables – s	oft tissue ^b		
Nasolabial prominence, mm	15.8 (13.3, 20.0)	18.3 (14.4, 20.7)	0.065
Nasolabial angle, degree	119.7 (113.2, 125.2)	121.0 (113.8, 129.1)	0.347
UL–TVL, mm	1.2 (-0.6, 3.4)	-0.1 (-2.4, 2.1)	0.017
LL–TVL, mm	0.2 (-2.0, 1.2)	-4.4 (-7.0, -2.6)	<0.001
Pog'–TVL, mm	-5.1 (-8.0, 0.2)	-13.5 (-18.1, -9.6)	<0.001
UL–E-line, mm	-6.0 (-9.8, -3.1)	-4.0 (-6.7, -1.4)	0.132
LL–E-line, mm	-4.3 (-6.9, -2.7)	-2.6 (-7.1, -0.6)	0.397
Facial convexity, degree	7.1 (3.2, 12.3)	13.4 (5.8, 18.8)	0.011

Table 1. Baseline characteristics in without deficiency group and with deficiency group

AHI, apnea hypopnea index; BMI, body mass index; LL–E-line, distance of lower lip to E-line; LL–TVL, distance of lower lip to true vertical line; LSAT, lowest oxygen desaturation; ODI, oxygen desaturation index; PAS, posterior airway space; Pog'–TVL, distance of soft tissue pogonion to true vertical line; UL–E-line, distance of upper lip to E-line; UL–TVL, distance of upper lip to true vertical line.

Data presented as median (interquartile range).

^a *P*-value for the comparison of the without-deficiency and with-deficiency groups; *P*<0.05 was considered statistically significant.

^b For UL-TVL, LL-TVL, Pog'-TVL, UL-E-line, and LL-E-line, a positive value is for a position in front of the TVL or E-line, and a negative value is for a posterior position.

	Without deficiency (n = 21)	With deficiency (n = 40)	<i>P</i> -value ^a
Adv. A, mm	5.9 (4.8, 9.2)	7.7 (5.9, 9.1)	0.536
Adv. B, mm	7.9 (6.4, 10.6)	9.9 (7.7, 13.4)	0.064
Adv. Pog, mm	9.0 (4.8, 11.4)	10.4 (8.3, 13.1)	0.046
Counterclockwise rotation, n (%)	10 (47.6)	21 (52.5)	0.717
Genioglossus advancement, n (%)	O (O)	1 (2.5)	1.000
Genioplasty, n (%)	1 (4.8)	4 (10)	0.828
To PAS, mm	8.5 (6.5, 10.5)	8.7 (6.9, 11.7)	0.690
T1 PAS, mm	14.2 (12.2, 16.0)	13.3 (10.8, 17.0)	0.643
ΔPAS, mm	6.2 (2.3, 8.0)	4.6 (3.3, 5.7)	0.264

Table 2. Surgical characteristics and airway space in without deficiency group and with deficiency group

Adv. A, advancement degree of A-point; Adv. B, advancement degree of B-point; Adv. Pog, advancement degree of pogonion; PAS, posterior airway space; To, preoperative; T1, postoperative; Δ , postoperative and preoperative change.

Cephalometric data presented as median (interquartile range), additional surgical techniques presented as number with percentage.

^a P-value for the comparison of the without-deficiency and with-deficiency groups; P<0.05 was considered statistically significant.

-	-	-			-	-				
		Without deficie	ncy (n = 21)			With deficiency	y (n=40)		Without o vs With d	leficiency eficiency
	Preoperative (To)	Postoperative (T1)	P-value To vs T1	Δ	Preoperative (To)	Postoperative (T1)	P-value To vs T1	Δ	P-value ^a	P-value ^b
AHI, events/h	41.6 (32.1, 62.6)	11.1 (6.2, 27.1)	<0.001	-28.0 (-44.4, -10.8)	52.2 (35.3, 69.6)	10.3 (4.9, 21.9)	<0.001	-34.1 (-52.9, -21.8)	0.094	0.490
ODI, events/h	44.9 (28.7, 62.1)	19.1 (9.9, 28.1)	0.005	-27.0 (-34.4, -5.8)	51.0 (29.5, 70.0)	20.0 (7.7, 33.3)	<0.001	-26.0 (-46.6, -12.3)	0.284	0.851
LSAT, %	81.0 (79.0, 86.0)	86.0 (84.8, 89.0)	<0.001	5.0 (2.5, 9.0)	79.0 (73.0, 84.0)	87.0 (81.3, 89.0)	<0.001	6.0 (2.0, 9.0)	0.612	0.876
AHI, apnea hypopn	iea index; LSAT, lov	west oxygen desatu	tration; ODI,	oxygen desaturatio	n index; ∆, postope	rative and preoper	rative chanξ	že.		
^a P-value for the co	imparison of the p	rune range). reoperative to post	operative ch	ange (Δ) between tl	he without-deficien	icy and with-defici	ency group.	S.		
^b <i>P</i> -value for the co.	mparison of the p	ostoperative values	between th	e without-deficienc	y and with-deficien	icy groups. For all t	ests, P<0.0	5 was considered st	atistically s	ignificant.
Table 4. Preoperati [,]	ve and postoperat	tive soft tissue mea:	surements in	without deficiency	' group and with de	ficiency group				
		Without defici	ency (n = 21)			With deficiend	cy (n = 40)		Wit deficien defic	hout cy vs With ciency
	Preoperative (To)	e Postoperativ (T1)	re p-valu TovsT	e 1	Preoperative (To)	Postoperative (T1)	e P-valu Tovs T	e `1	p-value ^a	P-value ^b
Nasolabial prominence, mm	15.8 (13.3, 20.C	0) 11.0 (7.0, 15.5	<0.00	1 -4.6 (-6.7, -3.7)	18.3 (14.4, 20.7)	13.5 (9.3, 16.1)	00 [.] 00	1 -5.1 (-6.6, -3.9)	0.649	0.167
Nasolabial angle degree	, 119.7 (113.2, 125.	.2) 111.9 (108.4, 118	3.5) 0.005	-5.0 (-9.8, 0.3)	121.0 (113.8, 129.1)	115.3 (109.9, 125.	.6) <0.00	1 -6.4 (-10.8, 0.2)	0.909	0.200
UL–TVLc, mm	1.2 (-0.6, 3.4)	5.2 (3.4, 7.1)	<0.00	1 2.8 (2.2, 5.8)	-0.1 (-2.4, 2.1)	2.4 (0.6, 5.1)	<0.00	1 3.1 (1.9, 4.3)	0.195	0.002
LL–TVLc, mm	0.2 (-2.0, 1.2)	4.5 (3.1, 7.8)	<0.00	1 4.0 (2.5, 7.0)	-4.4 (-7.0, -2.6)	0.5 (-3.0, 2.3)	<0.00	1 4.3 (2.5, 7.1)	0.738	<0.001
Pog'–TVLc, mm	-5.1 (-8.0, 0.2)	1.6 (-3.5, 4.5)	<0.00	1 5.8 (0.8, 8.2)	-13.5 (-18.1, -9.6)	-6.4 (-11.4, -4.2	00.00	1 6.8 (3.2, 9.6)	0.148	<0.001
UL–E-linec, mm	-6.0 (-9.8, -3.1	1) -3.8 (-6.8, -1.2	2) 0.003	2.6 (-0.6, 3.8)	-4.0 (-6.7, -1.4)	-3.0 (-5.3, -5.1)	0.046	0.8 (-0.9, 2.5)	0.173	0.456
LL–E-linec, mm	-4.3 (-6.9, -2.7	-1.7 (-4.8, 1.2	0.001	2.4 (-0.1, 4.1)	-2.6 (-7.1, -0.6)	-2.0 (-3.7, 1.5)	0.002	: 1.1 (-0.4, 3.2)	0.403	0.735

Table 3. Preoperative and postoperative polysomnographic values in without deficiency group and with deficiency group

LL-E-line, distance of lower lip to E-line; LL-TVL, distance of lower lip to true vertical line; Pog-TVL, distance of soft tissue pogonion to true vertical line; UL-E-line, distance of upper 0.502 -2.7 (-6.8, -0.3) 11.1 (4.6, 14.3) 13.4 (5.8, 18.8) ip to E-line; UL–TVL, distance of upper lip to true vertical line; Δ , postoperative and preoperative change. -2.2 (-5.5, 2.8) 4.0 (1.0, 9.7) 7.1 (3.2, 12.3) convexity, degree

0.070

Facial

0.012

<0.001

Data presented as median (interquartile range).

^a P-value for the comparison of the preoperative to postoperative change (Δ) between the without-deficiency and with-deficiency groups. ^b P-value for the comparison of the postoperative values between the without-deficiency and with-deficiency groups. For all tests, P < 0.05 was considered statistically significant. For UL-TVL, Pog'-TVL, Og'-TVL, UL-E-line, a positive value is for a position in front of the TVL or E-line, and a negative value is for a posterior position.

	Without deficiency (n = 10)	With deficiency (n = 20)	<i>P</i> -value ^a
Breathing	7.0 (2.8, 9.0)	6.5 (5.0, 8.0)	0.713
Facial esthetics	6.5 (4.8, 9.0)	7.0 (5.0, 8.0)	0.983
Overall satisfaction	6.0 (1.8, 8.3)	7.0 (3.0, 8.8)	0.681

Table 5. Patients' satisfaction in without deficiency group and with deficiency group

Data presented as median (interquartile range).

^a *P*-value for the comparison of the without-deficiency and with-deficiency groups; *P*<0.05 was considered statistically significant.

REFERENCES

- Senaratna CV, Perret JL, Lodge CJ, Lowe AJ, Campbell BE, Matheson MC, Hamilton GS, Dharmage SC. Prevalence of obstructive sleep apnea in the general population: A systematic review. *Sleep Med Rev.* 2017;34:70-81.
- 2. Jordan AS, McSharry DG, Malhotra A. Adult obstructive sleep apnoea. Lancet. 2014;383:736-47.
- 3. Sullivan CE, Issa FG, Berthon-Jones M, Eves L. Reversal of obstructive sleep apnoea by continuous positive airway pressure applied through the nares. *Lancet.* 1981;1:862-5.
- 4. Lee CHK, Leow LC, Song PR, Li H, Ong TH. Acceptance and Adherence to Continuous Positive Airway Pressure Therapy in patients with Obstructive Sleep Apnea (OSA) in a Southeast Asian privately funded healthcare system. *Sleep Sci.* 2017;10:57-63.
- 5. Shelgikar AV, Aronovich S, Stanley JJ. Multidisciplinary Alternatives to CPAP Program for CPAP-Intolerant Patients. J Clin Sleep Med. 2017;13:505-10.
- Riley RW, Powell NB, Guilleminault C, Nino-Murcia G. Maxillary, mandibular, and hyoid advancement: an alternative to tracheostomy in obstructive sleep apnea syndrome. *Otolaryngol Head Neck Surg.* 1986;94:584-8.
- 7. Liu SY, Huon LK, Iwasaki T, Yoon A, Riley R, Powell N, Torre C, Capasso R. Efficacy of Maxillomandibular Advancement Examined with Drug-Induced Sleep Endoscopy and Computational Fluid Dynamics Airflow Modeling. *Otolaryngol Head Neck Surg.* 2016;154:189-95.
- Li KK, Guilleminault C, Riley RW, Powell NB. Obstructive sleep apnea and maxillomandibular advancement: an assessment of airway changes using radiographic and nasopharyngoscopic examinations. J Oral Maxillofac Surg. 2002;60:526-30; discussion 531.
- 9. Smatt Y, Ferri J. Retrospective study of 18 patients treated by maxillomandibular advancement with adjunctive procedures for obstructive sleep apnea syndrome. J Craniofac Surg. 2005;16:770-7.
- 10. Li KK, Riley RW, Powell NB, Guilleminault C. Maxillomandibular advancement for persistent obstructive sleep apnea after phase I surgery in patients without maxillomandibular deficiency. *Laryngoscope*. 2000;110:1684-8.
- 11. Liu SY, Awad M, Riley RW. Maxillomandibular Advancement: Contemporary Approach at Stanford. Atlas Oral Maxillofac Surg Clin North Am. 2019;27:29-36.
- 12. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.
- Zhou N, Ho JTF, Huang Z, Spijker R, de Vries N, Aarab G, Lobbezoo F, Ravesloot MJL, de Lange J. Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis. *Sleep Med Rev.* 2021;57:101471.
- 14. Zhou N, Ho JTF, de Vries N, Bosschieter PFN, Ravesloot MJL, de Lange J. Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement. J Clin Sleep Med. 2022;18:1073-81.
- 15. Veys B, Pottel L, Mollemans W, Abeloos J, Swennen G, Neyt N. Three-dimensional volumetric changes in the upper airway after maxillomandibular advancement in obstructive sleep apnoea patients and the impact on quality of life. *Int J Oral Maxillofac Surg.* 2017;46:1525-32.
- 16. Bettega G, Pepin JL, Veale D, Deschaux C, Raphael B, Levy P. Obstructive sleep apnea syndrome. fifty-one consecutive patients treated by maxillofacial surgery. *Am J Respir Crit Care Med*. 2000;162:641-9.
- 17. Ronchi P, Novelli G, Colombo L, Valsecchi S, Oldani A, Zucconi M, Paddeu A. Effectiveness of maxillomandibular advancement in obstructive sleep apnea patients with and without skeletal anomalies. *Int J Oral Maxillofac Surg.* 2010;39:541-7.
- Blumen MB, Buchet I, Meulien P, Hausser Hauw C, Neveu H, Chabolle F. Complications/adverse effects of maxillomandibular advancement for the treatment of OSA in regard to outcome. *Otolaryngol Head Neck* Surg. 2009;141:591-7.
- 19. Dantas JF, Carvalho SH, Oliveira LS, Barbosa DB, de Souza RF, Sarmento VA. Accuracy of two cephalometric analyses in the treatment of patients with skeletal class III malocclusion. *Braz Dent J.* 2015; 26:186-92.
- 20. Berry RB, Budhiraja R, Gottlieb DJ, Gozal D, Iber C, Kapur VK, Marcus CL, Mehra R, Parthasarathy S, Quan SF, Redline S, Strohl KP, Davidson Ward SL, Tangredi MM; American Academy of Sleep Medicine. Rules for scoring respiratory events in sleep: update of the 2007 AASM Manual for the Scoring of Sleep and

Associated Events. Deliberations of the Sleep Apnea Definitions Task Force of the American Academy of Sleep Medicine. J Clin Sleep Med. 2012;8:597-619.

- 21. Sher AE, Schechtman KB, Piccirillo JF. The efficacy of surgical modifications of the upper airway in adults with obstructive sleep apnea syndrome. *Sleep*. 1996;19:156-77.
- 22. Elshaug AG, Moss JR, Southcott AM, Hiller JE. Redefining Success in Airway Surgery for Obstructive Sleep Apnea: A Meta Analysis and Synthesis of the Evidence. *Sleep.* 2007;30:461-7.
- 23. Fleiss JL. Analysis of Covariance and the Study of Change. In: *The Design and Analysis of Clinical Experiments*. John Wiley & Sons, Inc.; 1999. p. 186-219.
- 24. Li KK. Maxillomandibular advancement for obstructive sleep apnea. J Oral Maxillofac Surg. 2011;69: 687-94.
- 25. Vicini C, Dallan I, Campanini A, De Vito A, Barbanti F, Giorgiomarrano G, Bosi M, Plazzi G, Provini F, Lugaresi E. Surgery vs ventilation in adult severe obstructive sleep apnea syndrome. *Am J Otolaryngol.* 2010;31:14-20.
- 26. Paredes de Sousa Gil A, Guijarro-Martínez R, Haas OL Jr, Hernández-Alfaro F. Three-dimensional analysis of nasolabial soft tissue changes after Le Fort I osteotomy: a systematic review of the literature. *Int J Oral Maxillofac Surg.* 2019;48:1185-200.
- 27. Conley RS, Boyd SB. Facial soft tissue changes following maxillomandibular advancement for treatment of obstructive sleep apnea. J Oral Maxillofac Surg. 2007;65:1332-40.
- Cifuentes J, Teuber C, Gantz A, Barrera A, Danesh G, Yanine N, Lippold C. Facial soft tissue response to maxillo-mandibular advancement in obstructive sleep apnea syndrome patients. *Head Face Med.* 2017; 13:15.
- 29. Ricketts RM. Esthetics, environment, and the law of lip relation. Am J Orthod. 1968;54:272-89.
- 30. Czarnecki ST, Nanda RS, Currier GF. Perceptions of a balanced facial profile. *Am J Orthod Dentofacial Orthop.* 1993;104:180-7.
- 31. Arnett GW, Bergman RT. Facial keys to orthodontic diagnosis and treatment planning. Part I. Am J Orthod Dentofacial Orthop. 1993;103:299-312.
- 32. Johnson TP, Wislar JS. Response rates and nonresponse errors in surveys. JAMA. 2012;307:1805-6.



Maxillomandibular advancement for obstructive sleep apnea: A retrospective prognostic factor study for surgical response

Ning Zhou, Jean-Pierre T.F. Ho, Wouter P. Visscher, Naichuan Su, Frank Lobbezoo, Jan de Lange

Published in: Sleep Breath, 2022. doi: 10.1007/s11325-022-02731-x

ABSTRACT

Objective: To identify potential predictors of surgical response to maxillomandibular advancement (MMA) in patients with obstructive sleep apnea (OSA) from the most common clinically available data (patient-related, polysomnographic, cephalometric, and surgical variables).

Methods: This was a retrospective study comprising of consecutive patients who underwent MMA for moderate to severe OSA. Relevant clinical, polysomnographic, cephalometric, and surgical variables were collected as independent variables (predictors). The association of the independent variables with a favorable surgical response to MMA was assessed in univariate and multivariate analyses.

Results: One hundred patients were included (82% male; mean age of 50.5 years; mean apnea hypopnea index [AHI] of 53.1 events/h). The rate of favorable surgical response was 67.0%. Based on multivariate analysis, patients with cardiovascular disease (CVD) had 0.140 times lower odds to favorably respond to MMA (OR:0.140 [0.038, 0.513], P = 0.003). For each 1-unit increase in central apnea index (CAI) and superior posterior airway space (SPAS), there were 0.828 and 0.724 times lower odds to favorably respond to MMA (OR: 0.124 times lower odds to favorably respond to MMA (OR: 0.828 [0.687, 0.997], P = 0.047; and 0.724 [0.576, 0.910], P = 0.006), respectively.

Conclusion: Within the limitations of this study, it is suggested that the surgical outcome of MMA may be less favorable when OSA patients have certain phenotypic characteristics: the presence of CVD, higher CAI, and larger SPAS. If confirmed in future studies, these variables may guide patient selection for MMA.

Keywords: Obstructive sleep apnea; Maxillomandibular advancement; Surgical response; Predictor

INTRODUCTION

Maxillomandibular advancement (MMA) is a skeletal surgery for treatment of obstructive sleep apnea (OSA), which enlarges the upper airway space and reduces the upper airway collapsibility by displacing the maxilla and mandible anteriorly^{1, 2}. Despite the fact that MMA has been demonstrated to be a highly effective therapy for moderate to severe OSA, with a surgical success rate of approximately 85%^{3, 4}, there are still patients who do not respond as favorably as others to MMA. In order to improve preoperative counselling of patients regarding the chance of surgical response, and also to avoid ineffective therapy and unnecessary burden on nonresponders to MMA, it is essential and clinically meaningful to identify the potential responders and nonresponders to MMA prior to the surgery.

Some factors have been reported to correlate with increased surgical response to MMA, mainly in terms of patient-related characteristics, polysomnographic variables, and surgical characteristics. For example, a meta-analysis suggested that younger age, lower baseline weight, lower baseline apnea hypopnea index [AHI], and greater degree of maxillary advancement were associated with increased surgical response⁴. In addition, a few studies also identified radiographic or drug-induced sleep endoscopy (DISE) predictors of surgical response to MMA⁵⁻⁷, such as cephalometric minimum retrolingual space⁶ and complete anteroposterior epiglottic collapse during DISE⁷. However, the evidence on predictors of MMA surgical outcome is still incomplete. Consequently, the clinicians' ability to predict MMA outcome and preselect suitable candidates for MMA is still limited and mainly based on the clinician's expertise.

For patients undergoing MMA for OSA, a preoperative assessment in daily clinical practice mainly involves medical and sleep history, physical and radiographic examination, a polysomnography (PSG), and sometimes a DISE. Therefore, the aim of this study was to identify the potential predictors of surgical response to MMA in OSA patients, from the most common clinically available data (patient-related, polysomnographic, cephalometric, and surgical variables).

METHODS

Patient selection

This study recruited consecutive patients who underwent MMA for OSA at the Department of Oral and Maxillofacial Surgery, Amsterdam UMC (location AMC), from September 2011 to July 2021. The further inclusion criteria were the following: (1) age 18 years or older; (2) presence of moderate to severe OSA diagnosed by an overnight PSG; (3) continuous positive airway pressure (CPAP) failure, intolerance, or refusal; and (4) patients with a follow-up PSG recording at least three months after MMA. The exclusion criteria were as follows: (1) patients who declined their data to be used for research purposes; (2) previous history of a LeFort I osteotomy and/or a bilateral sagittal split osteotomy (BSSO); and (3) craniofacial and/or syndromic patients.

Variables

All data were retrospectively collected from patients' electronic files. Recorded baseline characteristics included patient-related variables, respiratory variables as measured by PSG, and cephalometric variables. Postoperative PSG variables and cephalometric measurements were also recorded. The surgical characteristics were determined by preoperative and postoperative cephalograms. The potential predictors of MMA surgical response included the recorded baseline characteristics and surgical characteristics.

Patient-related variables

The collected patient-related variables included age, gender, body mass index (BMI), preoperative physical status represented by the ASA (American Society of Anesthesiology) classification system score⁸, specific comorbidities (i.e., hypertension, cardiovascular diseases [CVD]⁹, diabetes mellitus, and chronic obstructive pulmonary disease), previous history of upper airway surgery for OSA, and the number of lost teeth. The tooth loss was categorized as the following: 0-4 lost teeth, 5-8 lost teeth, 9-31 lost teeth, and 32 lost teeth, i.e., being edentulous¹⁰.

Polysomnography

An overnight PSG was performed preoperatively and at least 3 months postoperatively. All respiratory events were scored according to the American Academy of Sleep Medicine (AASM) criteria¹¹. The collected baseline PSG variables included AHI, central apnea index (CAI), mixed apnea index (MAI), positional OSA

or non-positional OSA (positional OSA was defined as an AHI at least twice as high in supine position as in non-supine position¹²), 3% oxygen desaturation index (3% ODI), and lowest oxygen saturation (LSAT).

Postoperative AHI, 3% ODI, and LSAT were collected to assess the surgical outcome. According to Sher's criteria, surgical response was defined as "at least 50% AHI reduction following MMA and a postoperative AHI < 20 events/h"¹³.

Cephalometry

All patients underwent a standardized lateral cephalogram preoperatively and at least one week postoperatively. All radiographs were taken with the subjects in natural head position with centric occlusion and lips at rest. Cephalometric analysis was performed by one observer using Viewbox software (Viewbox 4, dHAL Software, Kifissia, Greece). Twenty-two cephalometric variables for skeletal and soft tissue, including the cranial base, face height, maxilla and mandible, soft palate, tongue, hyoid, and upper airway, were measured (**Table 1**; Supplementary **Fig. S1** and **Fig. S2**).

To quantify the reliability of the measurements, the same observer repeated the tracings in 20 randomly selected radiographs one month later.

Maxillomandibular advancement

The MMA procedures were completed by two dedicated OSA surgeons and consisted of a LeFort I osteotomy of the maxilla and a BSSO of the mandible. The maxillomandibular complex was advanced and counterclockwise rotation was performed for selected cases. The surgical variables used in this study included degrees of A-point, B-point and pogonion (Pog) advancement, and presence or absence of anticlockwise rotation. The degrees of A-point, B-point, and Pog advancement were determined by comparing preoperative and postoperative distance between A-point to the true vertical plane (TVP), B-point to TVP, and Pog to TVP, respectively. After MMA, cases with a mandibular plane angle change of \leq -2 degrees were classified as counterclockwise rotation cases¹⁴.

Statistical analysis

All collected data were analyzed with SPSS (IBM SPSS Statistical version 26, IBM Corp., Armonk, NY, USA). Normality was tested using the Shapiro-Wilk test. Continuous variables were reported as mean and standard deviation when normally distributed Chapter 3

or as median and interquartile range when not normally distributed. Categorical variables were reported as frequency and percentage. To compare the preoperative and postoperative continuous variables, the paired-samples t-test or Wilcoxon signed-rank test was applied in cases of normally or non-normally distributed data, respectively. To compare the continuous variables between responders and nonresponders, the independent-samples t-test or Mann-Whitney U test was used in cases of normally or non-normally distributed data, respectively. Chi-square test was used to compare the categorical variables between responders and nonresponders. The intra-observer reliability of the cephalometric measurements was evaluated using intraclass correlation coefficient (ICC).

Logistic regression was used to identify the variable(s) that was (were) predictive of a favorable response to MMA. First, univariate logistic regression analyses were used to assess the association between each independent variable (predictor) and the surgical response, separately. Multivariate logistic regression with backward selection (P < 0.05 for removal) was then used to identify the variables that were independently associated with the surgical response. The independent variables included in the multivariate model were those with a *P*-value of < 0.10 in univariate logistic regression. For variables including age, gender, BMI, baseline AHI, and degrees of maxillary and mandibular advancement, they were forced into the multivariate model regardless of their *P*-values in univariate logistic regression because of their potential importance for MMA surgical outcome⁴. Collinearity diagnostics test was performed using the variance inflation factors (VIF) cutoff value of 5; a variable(s) with VIF greater than 5 was excluded from the multivariate model. Complete case analysis was used to handle the missing values for logistic analysis. A *P*-value < 0.05 was considered statistically significant.

RESULTS

Patient characteristics

A total of 111 patients underwent MMA for obstructive sleep apnea (OSA). Of these, 100 patients (82% male) were included in this study. The reasons for exclusion from the study were as follows: no follow-up PSG available (n = 4), rejected their data to be used for research (n = 3), mild OSA (n = 3), and craniofacial and/or syndromic patient

(n = 1). Participants were middle aged (50.5 \pm 9.9 years) and overweight (BMI = 29.8 \pm 4.2 kg/m²), with a mean baseline AHI of 53.1 \pm 21.2 events/h.

Surgical outcome

The mean degrees of A-point, B-point, and Pog advancement were 7.2 ± 2.3 mm, 9.8 ± 4.2 mm, and 9.8 ± 5.1 mm, respectively. The postoperative PSGs were performed 4.0 (3.0-6.0) months after MMA. At the time of postoperative PSG, the mean BMI of the patients was 29.1 ± 4.5 kg/m². The major outcomes of the MMA surgery in the total population are shown in **Table 2**. The median AHI was significantly reduced from 51.7 (36.8-68.5) events/h to 12.9 (5.9-23.1) events/h (P < 0.001). A favorable surgical response was achieved in 67 of 100 patients (67%), and 19 patients (19%) had an AHI of < 5 events/h postoperatively. The preoperative and postoperative PSG values and upper airway measurements in responders and nonresponders are presented in Supplementary **Table S1**.

Baseline and surgical characteristics and surgical response

Compared to responders, the occurrences of hypertension and CVD were significantly higher in nonresponders (P = 0.003 and 0.001, respectively). Preoperative CAI was significantly higher in nonresponders (P = 0.011) (**Table 3**). ICC of the cephalometric analysis ranged from 0.859-0.998, which indicated an excellent intra-observer reliability¹⁵. Of the cephalometric variables, nonresponders had a significantly larger superior posterior airway space (SPAS; P = 0.002) than responders (**Table 4**). There were no significant differences between responders and nonresponders in the other baseline characteristics. In terms of surgical characteristics, no significant difference was found between responders and nonresponders (**Table 4**).

Prediction of surgical response

The univariate analyses revealed six independent variables with a *P*-value < 0.1 (Supplementary **Table S2**). After collinearity diagnostics test, all the six variables were included in the multivariate model, including age, hypertension, CVD, CAI, ANB, and SPAS (**Table 5**).

After adjusting for the covariables (gender, BMI, AHI, and degrees of maxillary and mandibular advancement), the multivariate model revealed that the independent factors associated with surgical response were CVD, CAI, and SPAS. Patients with the presence of CVD had 0.140 times lower odds to respond favorably to MMA (OR: 0.140

[0.038, 0.513]; P = 0.003) compared with those without. For each 1-unit increase in CAI, there was 0.828 times lower odds to respond favorably to MMA (OR: 0.828 [0.687, 0.997]; P = 0.047). For each 1-unit increase in SPAS, there was 0.724 times lower odds to respond favorably to MMA (OR: 0.724 [0.576, 0.910]; P = 0.006).

DISCUSSION

The present study aimed to investigate if the most common clinically available data, i.e., patient-related, polysomnographic, cephalometric, and surgical variables, have predictive value on MMA surgical outcome. Our main finding was that among baseline and surgical characteristics, cardiovascular disease (CVD), central apnea index (CAI), and superior posterior airway space (SPAS) were the independent predictors of response to MMA: the presence of CVD is indicative of non-response, and CAI and SPAS are inversely related to a favorable response.

Notably, in the present study, the overall success rate of MMA – 67.0% – was lower than that reported in previous studies³, which ranged from 70 to 100%. One probable reason for this difference in the success rate between the present study and previous studies is that patients recruited in our institute for MMA have been refractory to multiple therapies (e.g., CPAP, mandibular advancement device, upper airway surgery), or were considered poor candidates for upper airway surgery for various reasons (e.g., central and mixed apneas > 25% of the total AHI¹⁶, multilevel complete collapse during DISE¹⁷). Thus, for some of our patients, there could be a complex interplay between anatomical and non-anatomical traits in OSA pathogenesis, which might have led to the relatively low success rate in our study. In addition, although baseline DISE was not performed in all the patients, over half of the study population (65/100) received DISE, 52 of whom presented with epiglottic collapse. A recent study from Kastoer et al. suggested that MMA surgery may not be an effective therapy for epiglottic collapse¹⁸.

Prior work has suggested that OSA is associated with CVD^{19, 20}. In a recent study consisting of 1717 patients with moderate to severe OSA, the prevalence of CVD was 52%²⁰. In the present study, CVD also affects 35% of our study population (26 patients with coronary heart disease, six patients with cerebrovascular disease, and three patients with both coronary heart disease and cerebrovascular disease; seven

of these patients had heart failure), which further supports the notion that CVD is highly prevalent in patients with OSA. Notably, our study is the first to show that the presence of CVD in OSA patients is independently associated with non-response to MMA. We inferred that OSA with coexisting CVD may represent a subtype involving a complex interaction between anatomical and non-anatomical causes of OSA that cannot be fully resolved by MMA. Currently, only very limited evidence can partially support our inference. It has been suggested that chronic hypoxemia and/or high left atrial pressure in heart failure could yield an elevated loop gain via increases in chemosensitivity²¹. Additionally, the increased fluid retention and nocturnal rostral fluid shift in heart failure could narrow the upper airway and increase the extraluminal tissue pressure²². In this study population, however, the post-hoc chisquare test showed that there is no significant difference in the percentage of heart failure between responders and nonresponders (6.0% (4/67) vs 9.1% (3/33), P = 0.874). Further work should be performed to investigate the underlying pathophysiological mechanism of OSA with coexisting CVD for personalized treatment. Additionally, it is important to take into account the duration of CVD for its severity and to use such severity as an element for subgrouping in order to investigate the contribution of CVD to the surgical outcome of MMA. However, among the 35 patients with CVD, the duration of CVD is only available in 7 patients (10.1 \pm 3.9 years, range 6-16 years), which prevents us from further analysis of those patients. Future investigations are necessary to confirm our finding and to explore the association between duration of CVD and MMA surgical response.

In clinical practice, it is not uncommon that individuals with OSA exhibit some proportion of central and/or mixed events, leading to a dilemma in the selection of the most appropriate OSA treatment. Our study demonstrated that a higher preoperative CAI was independently associated with non-response to MMA. This finding is supported by a previous study by Makovey et al.⁵, which found that the mean pre-MMA CAI in their failure group was significantly higher than that in their success group (5.7 events/h vs 0.6 events/h; P = 0.005). The heterogeneity of pure OSA (i.e., 100% of apneas are obstructive) and predominant OSA (i.e., coexisting obstructive and central apneas, and 50% < obstructive apneas < 100%) has been investigated previously²³. It was suggested that the pure OSA patient group and predominant OSA patient group have equally elevated upper airway collapsibility (i.e., critical closing pressure [Pcrit]); however, the predominant OSA patients differed from the pure OSA patients in showing less breathing control stability²³. The finding

Chapter 3

that OSA patients with relatively higher baseline CAI are less likely respond favorably to MMA also indicates that in these patients breathing control instability may play a significant role in the development of obstructive events. Recently some studies have suggested that breathing control instability (high loop gain) promotes treatment failure on oral appliance or upper airway stimulation for patients with OSA²⁴⁻²⁶. Future research is required to determine whether treatment for central respiratory instability in predominant OSA patients may help relieve the obstructive events.

So far, little evidence is available on the predictive value of cephalometric variables in terms of surgical response to MMA in OSA patients. In this study, we have included parameters of craniofacial and upper airway morphology such as maxillary and mandibular position, face height, soft palate, and tongue, which have not been assessed together in previous studies on surgical response to MMA. This patient cohort presented only one cephalometric variable that is independently related to MMA surgical response, i.e., SPAS. We found that larger SPAS was independently associated with non-response to MMA. This finding is in line with that in a study by Teitelbaum et al.⁶. Their study showed that the minimal SPAS in their MMA success group was significantly narrower than that in their MMA failure group $(4.6 \pm 1.3 \text{ mm})$ vs 7.2 + 1.7mm, P = 0.009). There are several possible explanations for our finding. First and foremost, in this study cephalograms were taken with the patients awake in upright position. Most of skeletal cephalometric parameters such as cranial base and mandibular length could completely reflected the condition during sleep as they are stable and independent of posture and sleep state, whereas the skeletal parameters that could be affected by mandibular movement (e.g., SNB, ANB) and soft tissue parameters (e.g., soft palate, pharyngeal space) might not. As a consequence, the value of SPAS, as well as some other cephalometric measures, in predicting surgical response to MMA might have been over- or underestimated. Secondly, it has been suggested that airway shape may be a predisposing factor for the development of OSA; patients with OSA are likely to have an elliptical airway with the long axis oriented anteroposteriorly (A-P), and this A-P orientation may adversely affect the airway muscle function which results in airway collapse during sleep²⁷. We hypothesize that the OSA patients with larger SPAS are more likely to present with A-P airway orientation. Several previous studies have shown that after MMA there were significant increases in both lateral and A-P airway diameters, and the ratio of A-P and lateral airway dimension tended to be higher^{28, 29}. This indicates that MMA surgery may actually exacerbate the A-P airway orientation in some patients,

leading to a less beneficial surgical outcome. Of note, MMA can not only alter the upper airway morphology, but also increase the pharyngeal wall tension³⁰. The latter element, i.e., pharyngeal wall tension, was not evaluated and therefore not weighed in this study. Lastly, for OSA patients with a larger pharyngeal airway space, there is a higher possibility that non-anatomical contributors play a more prominent role in the pathogenesis of OSA, which may not be treated with MMA. The predictive value of SPAS for MMA surgical outcome needs further investigation. Furthermore, the predictive value of 3D upper airway parameters (e.g., volume, cross-sectional area) should be also explored.

It is interesting to note that several other predictors recognized previously were found not to be predictive of surgical response in our study, mainly including lower baseline AHI, lower baseline BMI, and larger degree of maxillary advancement⁴. Currently, there is still a question as whether these factors could predict MMA surgical response. In a study from Goodday et al.³¹, the efficacy of MMA was evaluated in 13OSA patients with an AHI higher than 100 events/h, and a favorable surgical response was achieved in 10 of those patients. The authors concluded that MMA was highly effective for patients with extremely severe OSA. Of note, although AHI is currently the most widely used measure of OSA severity, there is a growing recognition in its limitation to predict clinical consequences of OSA and response to OSA treatment³². Recently some other alternative measures of OSA severity have been proposed, such as apnea-hypopnea event duration³³ and hypoxic burden³⁴. However, our study did not analyze such PSG parameters because these relatively novel measures were not available in the PSG reports of our patients. Future research should explore the value of these alternative measures in predicting response to MMA. Besides, due to the fact that in the study by Goodday et al.³¹, eight of nine patients with available BMI values were obese (BMI range 31.2-61.3 kg/m²) before surgery, and all but one remained obese (BMI range 29-53.9 kg/m²) after surgery, they assumed that BMI did not appear to influence changes in AHI. With regard to the maxillary advancement, multiple studies have found no correlation between degree of maxillary advancement and a reduction in AHI^{35, 36}. Increased airway volume following MMA has been considered to be necessary for improving OSA^{28, 37}, while Chang et al. reported that there was a plateau effect for the airway volume increase as a result of maxillary advancement³⁸. In addition to the potential predictors mentioned above, some other factors of interest to clinicians were also investigated in terms of predicting MMA outcome. For example, tooth loss may be an independent risk factor for OSA¹⁰, but few evidence is available on the association between the number of lost teeth and treatment outcome for OSA³⁹. This study is the first to suggest that MMA outcome is not significantly related to number of lost teeth. Taken together, more research is required to recognize which parameters can reliably predict the surgical response, and thus should be included in the patient selection procedure of MMA for OSA.

The study results should be interpreted with caution due to certain limitations. First, it was a retrospective study, whereas a prospective study would be preferred allowing for better control of the data. Second, our cohort consisted predominantly of middle-aged, overweight males with severe OSA, thus the results may be limited to this patient profile. Furthermore, as we have stated before, given those relatively novel PSG measures of OSA severity (e.g., hypoxic burden) were absent in PSG reports of our patients, such parameters were not included in the analysis. This may also limit the generalizability of our findings. Lastly, the cephalograms were obtained with the patient awake in a standard upright position. Some measurement results, especially the soft tissue measurements, may thus not represent the condition during sleep. This may explain why most of the measurements of upper airway structures cannot be implicated in the surgical outcome. However, from the aspects of cost and/or convenience, an upright cephalogram remains an important imaging technique to evaluate the craniofacial and upper airway anatomy.

CONCLUSION

Within the limitations of the study, it is suggested that the presence of cardiovascular disease, higher central apnea index, and larger superior posterior airway space are independently associated with non-response to MMA for OSA. Our results may further support the concept that OSA is a heterogeneous disorder with multifactorial pathophysiological causes, which highlights the importance of evolving different OSA phenotypes and thereby developing personalized treatment.

	Variable	Definition
Cranial base	S-N	Distance between S and N
	N-S-Ba	Angle from N to S to Ba
Face height	ATFH	Distance between N and Me
	ALFH	Distance between ANS and Me
	PTFH	Distance between S and Go
	MP-SN	Inclination of the mandibular plane in relation to the SN plane
Maxilla and mandible	SNA	Angle from S to N to A
	SNB	Angle from S to N to B
	ANB	Angle from A to N to B
	Maxillary length	Distance between ANS and PNS
	Mandibular corpus length	Distance between Go and Me
Soft palate	SPL	Distance between PNS and UT
	SPT	Maximal diameter of soft palate perpendicular to PNS-UT line
Tongue	TGL	Tongue length as the distance between TT and Eb
	TGH	Maximum tongue height perpendicular to TT-Eb line
Hyoid bone	H-S	Distance between H and S
	H-MP	Distance between H and MP
	H-C3	Distance between H and C3
Upper airway	UAL	Upper airway length as distance between PNS to Eb
	SPAS	Width of airway along parallel line to Go-B line at the level of the midpoint of UT and PNS
	MAS	Width of airway along parallel line to Go-B line through UT
	IAS	Width of airway along Go-B line
Surgical movement	A-TVP	Distance between A to TVP
	B-TVP	Distance between B to TVP
	Pog-TVP	Distance between Pog to TVP

 Table 1. Overview of cephalometric variables and definitions

A, A-point (subspinale); ALFH, anterior lower face height; ANS, anterior nasal spine; ATFH, anterior total face height; B, B-point (supramentale); Ba, basion; C3, the most anterior-inferior point of the third cervical vertebra; Eb, epiglottis base; Co, gonion; H, hyoid point; IAS, inferior airway space; MAS, middle airway space; Me, menton; MP, mandibular plane; N, nasion; PNS, posterior nasal spine; Pog, pogonion; PTFH, posterior total face height; S, sella; SN, sella-nasion line; SPAS, superior posterior airway space; SPL, soft palate length; SPT, soft palate thickness; TGH, tongue height; TGL, tongue length; UAL, upper airway length; UT, uvula tip; THP, true horizontal plane; TT, tongue tip; TVP, true vertical plane.

Variable	Preoperative (n = 100)	Postoperative (n = 100)	P-value
AHI, events/h	51.7 (36.8-68.5)	12.9 (5.9-23.1)	<0.001
ODI 3%, events/h	51.0 (34.3-66.6)	21.2 (10.5-30.2)	<0.001
LSAT, %	79.5 (73.0-84.0)	86.0 (82.0-89.0)	<0.001

Table 2. Treatment outcome of maxillomandibular advancement in the total population

AHI, apnea hypopnea index; LSAT, lowest oxygen saturation; n, number of patients; ODI 3%, 3% oxygen desaturation index.

Data presented as median (interquartile range).

P-value for the comparison of the preoperative versus postoperative values; P < 0.05 was considered statistically significant.

Variable	Responder (n = 67)	Nonresponder (n = 33)	P-value
Patient-related variables			
Age, years	49.0 (41.0-59.0)	54.0 (45.5-58.0)	0.162
Male, n (%)	54 (80.6)	28 (84.8)	0.603
BMI, kg/m²	29.7 (27.4-32.4)	29.8 (28.2-32.0)	0.652
ASA-score, n (%)			
I	17 (25.4%)	6 (18.2%)	0.487
11	38 (56.7%)	18 (54.5%)	
111	12 (17.9%)	9 (27.3%)	
Hypertension, n (%)			
Absence	49 (73.1)	14 (42.4)	0.003
Presence	18 (26.9)	19 (57.6)	
CVD, n (%)			
Absence	51 (76.1)	14 (42.4)	0.001
Presence	16 (23.9)	19 (57.6)	
DM, n (%)			
Absence	58 (86.6)	29 (87.9)	1.000
Presence	9 (13.4)	4 (12.1)	
COPD, n (%)			
Absence	64 (95.5)	31 (94.0)	1.000
Presence	3 (4.5)	2 (6.0)	
Previous upper airway surgery, n (%)			
Absence	40 (59.7)	20 (60.6)	0.931
Presence	27 (40.3)	13 (39.4)	
Lost teeth, n (%)			
0-4 lost teeth	15 (22.4)	4 (12.1)	0.527
5-8 lost teeth	28 (41.8)	13 (39.4)	
9-31 lost teeth	16 (23.9)	10 (30.3)	
32 lost teeth	8 (11.9)	6 (18.2)	

Table 3. Patient-related variables and polysomnographic variables in responders and nonresponders

Variable	Responder (n = 67)	Nonresponder (n = 33)	P-value
Polysomnographic variables			
AHI, events/h	54.2 <u>+</u> 20.9	50.9 <u>+</u> 21.9	0.474
CAI, events/h	0.4 (0.2-1.4) ^a	1.5 (0.4-6.3) ^b	0.011
MAI, events/h	1.9 (0.2-9.1) ^a	5.6 (0.8-14.6) ^b	0.129
Positional/non-positional OSA, n (%)			
Positional OSA	22 (43.1)	11 (37.9)	0.649
Non-positional OSA	29 (56.9)	18 (62.1)	
ODI 3%, events/h	52.4 ± 22.3	51.5 <u>+</u> 21.0	0.866
LSAT, %	79 (71.0-84.0)	80 (76.0-85.0)	0.236

Table 3. continued

AHI, apnea hypopnea index; ASA, American Society of Anesthesiology; BMI, body mass index; CAI, central apnea index; COPD, chronic obstructive pulmonary disease; CVD, cardiovascular disease; DM, diabetes mellitus; LSAT, lowest oxygen saturation; MAI, mixed apnea index; n, number of patients; ODI 3%, 3% oxygen desaturation index; OSA, obstructive sleep apnea.

Continuous data presented as mean \pm standard deviation or median (interquartile range), categorical data presented as number with percentage.

P-value for the comparison of the responders versus nonresponders; P<0.05 was considered statistically significant.

^a Number of patients = 55; ^b Number of patients = 29.

Variable	Responder	Nonresponder	P-value
Cephalometric variables (responder	: n = 64; nonresponder: n = 31)	
Cranial base			
S-N, mm	70.0 <u>±</u> 3.6	70.6 ± 4.0	0.408
N-S-Ba, degree	130.0 (126.5-132.2)	131.2 (128.2-134.5)	0.076
Face height			
ATFH, mm	122.0 ± 7.9	124.5 ± 9.7	0.197
ALFH, mm	71.6 ± 7.0	73.7 ± 8.3	0.213
PTFH, mm	80.0 <u>±</u> 7.9	82.2 <u>+</u> 8.1	0.222
MP-SN, degree	36.7 ± 8.4	36.7 <u>+</u> 10.9	0.979
Maxilla and mandible			
SNA, degree	80.3 ± 3.6	80.0 <u>+</u> 4.0	0.760
SNB, degree	75.1 ± 4.1	76.5 ± 4.7	0.149
ANB, degree	5.2 <u>+</u> 2.6	3.7 ± 4.2	0.093
ANS-PNS, mm	53.0 ± 3.4	52.8 ± 4.2	0.745
Go-Me, mm	65.0 <u>+</u> 6.4	66.7 <u>+</u> 5.9	0.226
Soft palate			
SPL, mm	39.6 ± 7.0	40.4 ± 5.9	0.561
SPT, mm	9.9 (8.6-11.4)	11.0 (9.4-11.9)	0.096
Топяце			

Table 4. Cephalometric variables and surgical variables in responders and nonresponders

Variable	Responder	Nonresponder	P-value
TGL, mm	84.0 (79.8-87.3)	83.7 (79.7-88.1)	0.795
TGH, mm	36.5 ± 3.9	35.2 ± 4.6	0.156
Pharyngeal dimensions and hyoid bone positio	n		
UAL, mm	76.8 ± 6.4	78.8 ± 7.7	0.249
SPAS, mm	7.3 (5.5-9.2)	8.8 (7.6-11.0)	0.002
MAS, mm	9.9 <u>+</u> 2.6	10.8 ± 3.4	0.172
IAS, mm	8.9 <u>+</u> 3.1	9.2 <u>+</u> 3.0	0.625
H-S, mm	118.0 ± 9.5	120.7 ± 9.6	0.197
MP-H, mm	25.4 ± 5.5	25.9 ± 5.9	0.682
H-C3, mm	39.4 ± 4.8	41.4 ± 6.7	0.105
Surgical variables (responder: n = 63; non	responder: n = 29)		
Advancement degree of A-point, mm	7.0 ± 2.5	7.4±1.9	0.485
Advancement degree of B-point, mm	10.0 ± 4.3	9.6 ± 4.0	0.678
Advancement degree of Pog, mm	9.8 ± 5.2	9.9 ± 5.2	0.909
Counterclockwise rotation, n (%)			
Absence	29 (46.0)	10 (34.5)	0.298
Presence	34 (54.0)	19 (65.6)	

Table 4. continued

A, A-point; ALFH, anterior lower face height; ANS, anterior nasal spine; ATFH, anterior total face height; B, B-point; Ba, basion; C3, the most anterior-inferior point of the third cervical vertebra; Go, gonion; H, hyoid bone; IAS, inferior airway space; MAS, middle airway space; Me, menton; mm, millimeter; MP, mandibular plane; N, nasion; n, number of patients; PNS, posterior nasal spine; PTFH, posterior total face height; S, sella; SPL, soft palate length; SPAS, superior posterior airway space; SPT, soft palate thickness; TGL, tongue length; TGH, tongue height; UAL, upper airway length.

Continuous data presented as mean \pm standard deviation or median (interquartile range), categorical data presented as number with percentage.

P-value for the comparison of the responders versus nonresponders; P<0.05 was considered statistically significant.

Independent variable		Univa	riate analysis		Multivariate analysi: advancement	s (adjusted for t of A-point, an	the covariables: gender, B d advancement of B-point	MI, AHI, t)
	Coefficient B	SE	OR (95%CI)	P-value	Coefficient B	SE	OR (95%CI)	P-value
Age	-0.041	0.023	0.959 (0.917-1.003)	0.070				
Hypertension								
Absence	Ref.							
Presence	-1.307	0.447	0.271 (0.113-0.650)	0.003				
CVD								
Absence	Ref.				Ref.			0.003
Presence	-1.465	0.454	0.231 (0.095-0.563)	0.001	-1.964	0.662	0.140 (0.038-0.513)	
CAI	-0.191	0.080	0.826 (0.707-0.966)	0.017	-0.189	0.095	0.828 (0.687-0.997)	0.047
ANB	0.144	0.074	1.155 (1.000-1.334)	0.051				
SPAS	-0.242	0.083	0.785 (0.666-0.924)	0.004	-0.323	0.117	0.724 (0.576-0.910)	0.006

SUPPLEMENTARY MATERIAL



Fig. S1. Landmarks, reference lines, and the corresponding hard tissue variables used in the study. Landmarks: A, A-point (subspinale); ANS, anterior nasal spine; B, B-point (supramentale); Ba, basion; Go, gonion; Me, menton; N, nasion; PNS, posterior nasal spine; Pog, pogonion; S, sella. Reference lines: MP, mandibular plane; SN, sellanasion line; THP, true horizontal plane, plane through point S at 7° clockwise from SN plane; TVP, true vertical plane, plane through point S perpendicular to THP. Hard tissue variables: 1, S-N; 2, N-S-Ba; 3, ATFH (anterior total face height, N-Me); 4, ALFH (anterior lower face height, ANS-Me); 5, PTFH (posterior total face height, S-Go); 6, MP-SN; 7, SNA; 8, SNB; 9, ANB; 10, maxillary length (ANS-PNS); 11, mandibular corpus length (Go-Me); 12, A-TVP; 13, B-TVP; 14, Pog-TVP.



Fig. S2. Landmarks, reference lines, and the corresponding soft tissue variables used in the study. Landmarks: B, B-point (supramentale); C3, the most anterior-inferior point of the third cervical vertebra; Eb, epiglottis base; Go, gonion; H, hyoid point; Me, menton; UT, uvula tip; PNS, posterior nasal spine; TT, tongue tip. Reference lines: Go-B, plane between Go and B; MP, mandibular plane. Soft tissue variables: 1, SPL (soft palate length); 2, SPT (soft palate thickness); 3, TGL (tongue length); 4, TGH (tongue height); 5, H-S; 6, H-MP; 7, H-C3; 8, UAL (upper airway length); 9, SPAS (superior posterior airway space); 10, MAS (middle airway space); 11, IAS (inferior airway space).

Variables	Responder (n = 67)	Nonresponder (n = 33)	P-value
Polysomnographic variables			
Preop AHI, events/h	51.8 (37.1-68.6)	51.6 (35.2-69.3)	0.474
Postop AHI, events/h	8.3 (4.5-13.0)	33.0 (23.0-42.9)	<0.001
Preop ODI 3%, events/h	48.7 (35.3-68.9)	57.0 (29.5-66.0)	0.866
Postop ODI 3%, events/h	11.2 (9.2-20.7)	33.6 (25.8-50.3)	<0.001
Preop LSAT, %	79 (71.0-84.0)	80 (76.0-85.0)	0.236
Postop LSAT, %	87.5 (82.0-89.3)	85.0 (82.0-87.0)	0.019
Upper airway measurements			
Preop UAL, mm	76.8 <u>+</u> 6.4	78.8 <u>+</u> 7.7	0.249
Postop UAL, mm	75.1 ± 7.5	77.5 ± 9.5	0.189
Preop SPAS, mm	7.3 (5.5-9.2)	8.8 (7.6-11.0)	0.002
Postop SPAS, mm	12.5 (10.4-15.3)	14.0 (11.7-16.1)	0.143
Preop MAS, mm	10.0 (7.9-12.0)	10.8 (8.0-13.5)	0.172
Postop MAS, mm	14.9 (12.8-18.4)	17.4 (12.8-20.0)	0.202
Preop IAS, mm	8.4 (6.7-11.5)	8.9 (6.5-11.3)	0.625
Postop IAS, mm	13.6 (10.9-16.0)	15.3 (11.5-17.5)	0.266

Table S1. Preoperative and postoperative polysomnographic values and upper airway measurements in responders and nonresponders

AHI, apnea hypopnea index; IAS, inferior airway space; LSAT, lowest oxygen saturation; MAS, middle airway space; n, number of patients; ODI 3%, 3% oxygen desaturation index; Postop, postoperative; Preop, preoperative; SPAS, superior posterior airway space; UAL, upper airway length.

Data presented as mean ± standard deviation or median (interquartile range).

P-value for the comparison of the responders versus nonresponders; P<0.05 was considered statistically significant.

Table S2. Univariate analysis of patient-related, polysomnographic, cephalometric, and surgical variables for
predicting surgical response to maxillomandibular advancement

Variable	Coefficient B	SE	OR (95%CI)	P-value
Patient-related variables	S			
Age, years	-0.041	0.023	0.959 (0.917-1.003)	0.070
Gender				
Female	Ref.			
Male	-0.299	0.575	0.742 (0.240-2.291)	0.604
BMI, kg/m²	-0.005	0.051	0.996 (0.901-1.100)	0.930
ASA-score				
I	Ref.			
II	-0.294	0.554	0.745 (0.251-2.209)	0.596
III	-0.754	0.648	0.471 (0.132-1.676)	0.245
Hypertension				
Absence	Ref.			
Presence	-1.307	0.447	0.271 (0.113-0.650)	0.003
CVD				
Absence	Ref.			
Presence	-1.465	0.454	0.231 (0.095-0.563)	0.001

Variable	Coefficient B	SE	OR (95%CI)	P-value
DM				
Absence	Ref.			
Presence	0.118	0.643	1.125 (0.319-3.963)	0.855
COPD				
Absence	Ref.			
Presence	-0.319	0.939	0.727 (0.115-4.574)	0.734
Previous upper airway surge	ery			
Absence	Ref.			
Presence	0.038	0.435	1.038 (0.443-2.434)	0.931
Lost teeth				
0-4 lost teeth	Ref.			
5-8 lost teeth	-0.555	0.655	0.574 (0.159-2.074)	0.397
9-31 lost teeth	-0.852	0.692	0.427 (0.110-1.657)	0.219
32 lost teeth	-1.034	0.780	0.356 (0.077-1.640)	0.185
Polysomnographic variable	es			
AHI, events/h	0.007	0.010	1.007 (0.987-1.028)	0.470
CAI, events/h	-0.191	0.080	0.826 (0.707-0.966)	0.017
MAI, events/h	-0.013	0.016	0.987 (0.957-1.018)	0.408
Positional/non-positional O	SA			
Non-positional OSA	Ref.			
Positional OSA	0.216	0.476	1.241 (0.489-3.154)	0.650
ODI 3%, events/h	0.002	0.011	1.002 (0.980-1.025)	0.864
LSAT, %	-0.033	0.026	0.967 (0.919-1.018)	0.967
Cephalometric variables				
Cranial base				
S-N, mm	-0.049	0.059	0.952 (0.848-1.069)	0.404
N-S-Ba, degree	0.002	0.005	1.002 (0.993-1.012)	0.627
Face height				
ATFH, mm	-0.035	0.027	0.966 (0.917-1.018)	0.197
ALFH, mm	-0.038	0.030	0.963 (0.907-1.022)	0.212
PTFH, mm	-0.036	0.030	0.964 (0.910-1.022)	0.221
MP-SN, degree	0.001	0.025	1.001 (0.954-1.050)	0.979
Maxilla and mandible				
SNA, degree	0.018	0.060	1.019 (0.906-1.145)	0.757
SNB, degree	-0.076	0.053	0.927 (0.836-1.028)	0.150
ANB, degree	0.144	0.074	1.155 (1.000-1.334)	0.051
ANS-PNS, mm	0.020	0.060	1.020 (0.907-1.147)	0.742
Go-Me, mm	-0.044	0.036	0.957 (0.892-1.027)	0.225
Soft palate				
SPL, mm	-0.020	0.033	0.981 (0.919-1.047)	0.557
SPT, mm	-0.086	0.086	0.918 (0.775-1.086)	0.318

Table S2. continued

Variable	Coefficient B	SE	OR (95%CI)	P-value			
Tongue							
TGL, mm	-0.029	0.033	0.971 (0.911-1.035)	0.371			
TGH, mm	0.076	0.054	1.079 (0.971-1.199)	0.158			
Pharyngeal dimensions and hyoid bone position							
UAL, mm	-0.038	0.033	0.963 (0.903-1.027)	0.248			
SPAS, mm	-0.242	0.083	0.785 (0.666-0.924)	0.004			
MAS, mm	-0.105	0.077	0.900 (0.773-1.047)	0.173			
IAS, mm	-0.036	0.072	0.965 (0.837-1.112)	0.621			
H-S, mm	-0.031	0.024	0.969 (0.924-1.016)	0.196			
MP-H, mm	-0.016	0.040	0.984 (0.910-1.063)	0.678			
H-C3, mm	-0.066	0.042	0.936 (0.863-1.015)	0.110			
Surgical variables							
Advancement degree of A-point, mm	-0.068	0.097	0.934 (0.772-1.130)	0.481			
Advancement degree of B-point, mm	0.024	0.057	1.024 (0.916-1.146)	0.675			
Advancement degree of Pog, mm	-0.005	0.046	0.995 (0.910-1.088)	0.908			
Counterclockwise rotation							
Absence	Ref.						
Presence	-0.311	0.476	0.733 (0.288-1.862)	0.513			

Table S2. continued

A, A-point; AHI, apnea hypopnea index; ALFH, anterior lower face height; ANS, anterior nasal spine; ASA, American Society of Anesthesiology; ATFH, anterior total face height; B, B-point; Ba, basion; BMI, body mass index; C3, the most anterior-inferior point of the third cervical vertebra; CAI, central apnea index; CI, confidence interval; COPD, chronic obstructive pulmonary disease; CVD, cardiovascular disease; DM, diabetes mellitus; Go, gonion; H, hyoid bone; IAS, inferior airway space; LSAT, lowest oxygen saturation; MAI, mixed apnea index; MAS, middle airway space; Me, menton; mm, millimeter; MP, mandibular plane; N, nasion; ODI 3%, 3% oxygen desaturation index; OR, odds ratio; OSA, obstructive sleep apnea; PNS, posterior nasal spine; PTFH, posterior total face height; Ref., reference category; S, sella; SE, standard error; SPAS, superior posterior airway space; SPL, soft palate length; SPT, soft palate thickness; TGL, tongue length; TGH, tongue height; UAL, upper airway length.

REFERENCES

- 1. Schendel SA, Broujerdi JA, Jacobson RL. Three-dimensional upper-airway changes with maxillomandibular advancement for obstructive sleep apnea treatment. *Am J Orthod Dentofacial Orthop*. 2014;146:385-93.
- Liu SY, Huon LK, Iwasaki T, et al. Efficacy of Maxillomandibular Advancement Examined with Drug-Induced Sleep Endoscopy and Computational Fluid Dynamics Airflow Modeling. Otolaryngol Head Neck Surg. 2016;154:189-95.
- 3. Zhou N, Ho JTF, Huang Z, et al. Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis. *Sleep Med Rev.* 2021;57:101471.
- 4. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.
- Makovey I, Shelgikar AV, Stanley JJ, Robinson A, Aronovich S. Maxillomandibular Advancement Surgery for Patients Who Are Refractory to Continuous Positive Airway Pressure: Are There Predictors of Success? J Oral Maxillofac Surg. 2017;75:363-70.
- Teitelbaum J, Diminutto M, Comiti S, et al. [Lateral cephalometric radiography of the upper airways for evaluation of surgical treatment of obstructive sleep apnea syndrome]. *Rev Stomatol Chir Maxillofac.* 2007;108:13-20.
- Zhou N, Ho JTF, de Vries N, Bosschieter PFN, Ravesloot MJL, de Lange J. Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement. J Clin Sleep Med. 2022;18:1073-81.
- Horvath B, Kloesel B, Todd MM, Cole DJ, Prielipp RC. The Evolution, Current Value, and Future of the American Society of Anesthesiologists Physical Status Classification System. *Anesthesiology*. 2021;135:904-19.
- 9. World Health Organization. Definition of cardiovascular diseases. https://www.euro.who.int/en/healthtopics/noncommunicable-diseases/cardiovascular diseases/cardiovascular-diseases2/definition-ofcardiovascular-diseases. Accessed 1 June 2022.
- 10. Sanders AE, Akinkugbe AA, Slade GD, Essick GK. Tooth loss and obstructive sleep apnea signs and symptoms in the US population. *Sleep Breath*. 2016;20:1095-102.
- Iber C, Ancoli-Israel S, Chesson A, Quan SF for the American Academy of Sleep Medicine. The AASM manual for the scoring of sleep and associated events: rules, terminology and technical specifications, 1st ed. Westchester, IL: American Academy of Sleep Medicine; 2007.
- 12. Oksenberg A, Silverberg DS, Arons E, Radwan H. Positional vs nonpositional obstructive sleep apnea patients: anthropomorphic, nocturnal polysomnographic, and multiple sleep latency test data. *Chest.* 1997;112:629-39.
- 13. Sher AE, Schechtman KB, Piccirillo JF. The efficacy of surgical modifications of the upper airway in adults with obstructive sleep apnea syndrome. *Sleep*. 1996;19:156-77.
- 14. Reyneke JP, Bryant RS, Suuronen R, Becker PJ. Postoperative skeletal stability following clockwise and counter-clockwise rotation of the maxillomandibular complex compared to conventional orthognathic treatment. *Br J Oral Maxillofac Surg.* 2007;45:56-64.
- 15. Fleiss JL. Analysis of Covariance and the Study of Change. In: *The Design and Analysis of Clinical Experiments*. John Wiley & Sons Inc; 1999. p. 186-219.
- Strollo PJ Jr, Soose RJ, Maurer JT, et al. Upper-airway stimulation for obstructive sleep apnea. N Engl J Med. 2014;370:139-49.
- 17. Koutsourelakis I, Safiruddin F, Ravesloot M, Zakynthinos S, de Vries N. Surgery for obstructive sleep apnea: sleep endoscopy determinants of outcome. *Laryngoscope*. 2012;122:2587-91.
- 18. Kastoer C, Op de Beeck S, Dom M, et al. Drug-Induced Sleep Endoscopy Upper Airway Collapse Patterns and Maxillomandibular Advancement. *Laryngoscope*. 2020;130:E268-E274.
- 19. Collen J, Lettieri C, Wickwire E, Holley A. Obstructive sleep apnea and cardiovascular disease, a story of confounders!. *Sleep Breath*. 2020;24:1299-313.
- 20. André S, Andreozzi F, Van Overstraeten C, et al. Cardiometabolic comorbidities in obstructive sleep apnea patients are related to disease severity, nocturnal hypoxemia, and decreased sleep quality. *Respir Res.* 2020;21:35.

- 21. Dempsey JA, Veasey SC, Morgan BJ, O'Donnell CP. Pathophysiology of sleep apnea [published correction appears in Physiol Rev. 2010;90:797-8]. *Physiol Rev.* 2010;90:47-112.
- 22. Yumino D, Redolfi S, Ruttanaumpawan P, et al. Nocturnal rostral fluid shift: a unifying concept for the pathogenesis of obstructive and central sleep apnea in men with heart failure. *Circulation*. 2010;121:1598-605.
- 23. Xie A, Bedekar A, Skatrud JB, Teodorescu M, Gong Y, Dempsey JA. The heterogeneity of obstructive sleep apnea (predominant obstructive vs pure obstructive apnea). *Sleep*. 2011;34:745-50.
- 24. Edwards BA, Andara C, Landry S, et al. Upper-Airway Collapsibility and Loop Gain Predict the Response to Oral Appliance Therapy in Patients with Obstructive Sleep Apnea. *Am J Respir Crit Care Med* 2016;194:1413-22.
- 25. Op de Beeck S, Dieltjens M, Azarbarzin A, et al. Mandibular Advancement Device Treatment Efficacy Is Associated with Polysomnographic Endotypes. *Ann Am Thorac Soc.* 2021;18:511-8.
- 26. Op de Beeck S, Wellman A, Dieltjens M, et al. Endotypic Mechanisms of Successful Hypoglossal Nerve Stimulation for Obstructive Sleep Apnea. *Am J Respir Crit Care Med*. 2021;203:746-55.
- 27. Leiter JC. Upper airway shape: Is it important in the pathogenesis of obstructive sleep apnea? *Am J Respir Crit Care Med.* 1996;153:894-8.
- 28. Abramson Z, Susarla SM, Lawler M, Bouchard C, Troulis M, Kaban LB. Three-dimensional computed tomographic airway analysis of patients with obstructive sleep apnea treated by maxillomandibular advancement. J Oral Maxillofac Surg. 2011;69:677-86.
- 29. Gonçalves JR, Gomes LC, Vianna AP, Rodrigues DB, Gonçalves DA, Wolford LM. Airway space changes after maxillomandibular counterclockwise rotation and mandibular advancement with TMJ Concepts® total joint prostheses: three-dimensional assessment. *Int J Oral Maxillofac Surg.* 2013;42:1014-22.
- Liu SY, Huon LK, Powell NB, et al. Lateral Pharyngeal Wall Tension After Maxillomandibular Advancement for Obstructive Sleep Apnea Is a Marker for Surgical Success: Observations From Drug-Induced Sleep Endoscopy. J Oral Maxillofac Surg. 2015;73:1575-82.
- Goodday RH, Bourque SE, Edwards PB. Objective and Subjective Outcomes Following Maxillomandibular Advancement Surgery for Treatment of Patients With Extremely Severe Obstructive Sleep Apnea (Apnea-Hypopnea Index >100). J Oral Maxillofac Surg. 2016;74:583-9.
- 32. Malhotra A, Ayappa I, Ayas N, et al. Metrics of sleep apnea severity: beyond the apnea-hypopnea index. *Sleep.* 2021;44:zsab030.
- 33. Saraç S, Afsar GC. Effect of mean apnea-hypopnea duration in patients with obstructive sleep apnea on clinical and polysomnography parameter. *Sleep Breath.* 2020;24:77-81.
- 34. Azarbarzin A, Sands SA, Stone KL, et al. The hypoxic burden of sleep apnoea predicts cardiovascular disease-related mortality: the Osteoporotic Fractures in Men Study and the Sleep Heart Health Study [published correction appears in Eur Heart]. 2019;40:1157]. *Eur Heart*]. 2019;40:1149-57.
- 35. Smatt Y, Ferri J. Retrospective study of 18 patients treated by maxillomandibular advancement with adjunctive procedures for obstructive sleep apnea syndrome. J Craniofac Surg. 2005;16:770-7.
- 36. Ubaldo ED, Greenlee GM, Moore J, Sommers E, Bollen AM. Cephalometric analysis and long-term outcomes of orthognathic surgical treatment for obstructive sleep apnoea. *Int J Oral Maxillofac Surg.* 2015;44:752-9.
- 37. Niskanen I, Kurimo J, Järnstedt J, Himanen SL, Helminen M, Peltomäki T. Effect of Maxillomandibular Advancement Surgery on Pharyngeal Airway Volume and Polysomnography Data in Obstructive Sleep Apnea Patients. J Oral Maxillofac Surg. 2019;77:1695-702.
- Chang MK, Sears C, Huang JC, Miller AJ, Kushner HW, Lee JS. Correlation of Airway Volume With Orthognathic Surgical Movement Using Cone-Beam Computed Tomography. J Oral Maxillofac Surg. 2015;73:S67-S76.
- 39. Heidsieck DS, de Ruiter MH, de Lange J. Management of obstructive sleep apnea in edentulous patients: an overview of the literature. *Sleep Breath.* 2016;20:395-404.
CHAPTER

Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement

> Ning Zhou, Jean-Pierre T.F. Ho, Nico de Vries, Pien F.N. Bosschieter, Madeline J.L. Ravesloot, Jan de Lange

> > Published in: J Clin Sleep Med, 2022. doi: 10.5664/jcsm.9802

ABSTRACT

Objectives: (1) To investigate if drug-induced sleep endoscopy (DISE) findings are predictive of surgical response for patients undergoing maxillomandibular advancement (MMA) for obstructive sleep apnea (OSA) and (2) to investigate the predictive value of the jaw thrust maneuver during DISE in terms of surgical response to MMA.

Methods: A retrospective cohort study was conducted in patients with OSA who underwent a baseline polysomnography (PSG) and DISE followed by MMA and a 3- to 6-month follow-up PSG between September 1, 2011, and September 30, 2020.

Results: Sixty-four patients with OSA (50 males [78.1%]; mean \pm SD age = 51.7 \pm 9.5 years; mean \pm SD apnea-hypopnea index = 49.0 \pm 20.8 events/h) were included. Thirty-nine patients were responders, and 25 were nonresponders. Adjusting for baseline characteristics and surgical characteristics (e.g., age, baseline apnea-hypopnea index, degree of maxillary advancement), patients with complete anteroposterior epiglottic collapse had 0.239 times lower odds for response to MMA (95% confidence interval, 0.059-0.979; *P* = 0.047). No significant relationship was found between complete concentric velum collapse and MMA response. There was no statistically significant association between effect of jaw thrust maneuver during DISE on upper airway patency and treatment outcome of MMA.

Conclusions: This study indicates that DISE is a promising tool to identify patients who will or will not respond to MMA for treating OSA. Patients with complete anteroposterior epiglottic collapse may be less suitable candidates for MMA.

Keywords: Maxillomandibular advancement; Drug-induced sleep endoscopy; Obstructive sleep apnea; Surgical response

INTRODUCTION

Of the surgical options available to patients with obstructive sleep apnea (OSA), maxillomandibular advancement (MMA), a combination of a LeFort I osteotomy with a bilateral sagittal split osteotomy (BSSO), has been shown to be the most effective surgical option for OSA, with the exception of tracheostomy¹. The reported surgical success rate for MMA is 85.0%². Previous studies have recognized some patient characteristics as predictors of surgical response to MMA – mainly age, weight, and baseline AHI^{3, 4} – but there is room for improvement.

Drug-induced sleep endoscopy (DISE), proposed by Croft and Pringle⁵ in 1991, is an endoscopic examination performed during pharmacologically induced sleep in order to determine the exact site(s) of upper airway collapse. DISE is a unique and dynamic method for upper airway evaluation in patients with OSA, which can facilitate the treatment decision-making process for OSA. Since DISE plays a substantial role in otolaryngologic upper airway surgery and upper airway stimulation for OSA⁶⁻⁸, it is therefore of interest to explore its role in identifying suitable candidates for MMA.

Only a few studies have evaluated upper airway collapse patterns using DISE before and after MMA^{9, 10}. Of interest, it is suggested that MMA may not be effective in correcting the collapse at the level of epiglottis¹⁰. All in all, the association between baseline DISE findings and surgical outcome of MMA remains debatable.

Therefore, the aim of this study was to investigate if the sites, patterns, and degrees of upper airway collapse during DISE, along with the individual characteristics, results of other diagnostic modalities, and surgical characteristics, were predictive of surgical response for patients undergoing MMA for treating OSA. We hypothesized that DISE findings could predict the surgical outcome of MMA in OSA patients, and that the presence of epiglottic collapse and complete concentric collapse at the level of the palate (CCCp) may be associated with surgical failure of MMA. In addition, the predictive value of jaw thrust during DISE was also investigated in terms of surgical outcome of MMA.

METHODS

Study participants

This retrospective clinical trial was approved by the Medical Ethical Committee of the Amsterdam University Medical Centers (Amsterdam UMC) (location Academic Medical Center [AMC]) (reference number W19_171 #19.210). All patients were given the option to decline the use of their data in this study.

Consecutive patients with OSA were enrolled in the study if they underwent DISE at the Department of Otolaryngology of OLVG in Amsterdam prior to MMA at the Department of Oral and Maxillofacial Surgery of Amsterdam UMC (location AMC), between September 1, 2011 and September 30, 2020. The further inclusion criteria were as follows: (1) age 18 years or older, (2) presence of OSA (AHI \ge 5 events/h) diagnosed by polysomnography (PSG) preoperatively, (3) continuous positive airway pressure therapy failure or intolerance, (4) patients who underwent baseline DISE, and (5) patients with follow-up PSG at least 3 months after MMA. The exclusion criteria were (1) patients who declined the use of their data for this study and (2) patients with incomplete data.

Variables

All data were retrospectively collected and recorded in the patients' electronic file. Recorded baseline characteristics were age, sex, body mass index, neck circumference, previous OSA upper airway surgery, respiratory variables as measured by PSG, DISE findings, and cephalometric measurements. Postoperative respiratory variables determined using PSG and postoperative cephalometric measurements were recorded. Degrees of maxillary and mandibular advancement were obtained by preoperative and postoperative cephalometric analysis. Primary outcomes were DISE findings (independent variables) and surgical response (dependent variables). Secondary outcomes of interest were the other baseline characteristics, postoperative PSG variables, and postoperative cephalometric measurements. We adhered to the STROBE (Strengthening the Reporting of Observational Studies in Epidemiology) guidelines for reporting in observational studies¹¹.

Polysomnography

All patients underwent a full-night PSG preoperatively and at least 3 months postoperatively to assess the surgical outcome. The PSG method has been described

in a previous study¹². All respiratory events were recorded according to the American Academy of Sleep Medicine 2007 criteria¹³. Collected PSG data consisted of AHI, oxygen desaturation index (ODI), and lowest oxygen saturation (LSAT). Surgical response was defined as a postoperative AHI < 20 events/h with > 50% reduction of AHI based on Sher's criteria¹⁴.

Drug-induced sleep endoscopy

A standard DISE procedure was performed by a specific physician experienced in DISE with a trained nurse anesthetist, in a semi-dark and silent outpatient endoscopy room with standard anesthetic equipment¹⁵. Sedation was induced by intravenous administration of propofol using a target-controlled infusion pump. Proper sedation level was achieved when the patient showed hyporesponsiveness to verbal and tactile stimuli or when the patient began to snore. A fiberoptic laryngoscope was used to visualize the patients' upper airway. The upper airway was assessed starting in a supine position without jaw thrust and subsequently with manually performed jaw thrust. During the jaw thrust, the trained nurse anesthetist gently advanced the patient's mandible. The advancement was estimated to be 5 mm.

The Velum Oropharynx Tongue base Epiglottis (VOTE) system was used to document the DISE findings. Patterns of collapse (anteroposterior, lateral, and concentric) and degrees of collapse (0: no obstruction; 1: partial obstruction; and 2: complete obstruction) were recorded at the most common levels of upper airway obstruction (velum, oropharynx, base of tongue, and epiglottis)¹⁶. The total VOTE score was calculated as the sum of the collapse degree at each site of obstruction for a maximum score of 8.

During the jaw thrust maneuver, patients were categorized as "total resolution" when upper airway collapse was completely resolved (i.e., total VOTE score was reduced to zero), and as "partial resolution" when collapse was partially resolved (i.e., VOTE score was reduced but larger than zero).

Collected DISE data included levels, patterns and degrees of upper airway collapse in supine position and its corresponding total VOTE score, and categorization of patients during the jaw thrust maneuver.

Maxillomandibular advancement surgery

All patients underwent a standardized MMA procedure by 2 dedicated OSA surgeons, consisting of a LeFort I osteotomy and a BSSO with rigid internal fixation⁴. Additional procedures included genioplasty in five cases and genioplossal advancement in 1 case.

Cephalometric measurements

Preoperative and postoperative cephalometric measurements were performed using the following skeletal landmarks: sella (S), nasion (N), A-point (A) and B-point (B). A true horizontal plane (HP) (plane through point S at 7° clockwise from the SN plane) and vertical plane (plane through point S perpendicular to the HP) were constructed. The measured parameters included SNA, SNB, ANB, advancement degree of the maxilla (i.e., advancement degree of A-point), and advancement degree of the mandible (i.e., advancement degree of B-point) (**Figure 1**)⁴.

Data collection and statistical analysis

Statistical analysis was performed using SPSS version 26 (IBM Corp, Armonk, NY, USA). Descriptive statistical analysis was performed for all demographic and outcome variables and reported as mean and standard deviation (SD) for continuous variables and frequency and percentage for categorical variables. Normality was tested using the Shapiro-Wilk test. To compare quantitative variables between responders and nonresponders to MMA, the independent-samples t-test or Mann-Whitney U test was used in cases of normally or nonnormally distributed data, respectively. To compare the preoperative and postoperative values, the paired-samples t-test or Wilcoxon signed-rank test was applied in cases of normally or nonnormally distributed data, respectively. Collapse patterns in DISE findings were compared between responders and nonresponders with the chi-square test or Fisher's exact test.

The association between collapse patterns and surgical response was investigated using logistic regression. In cases of insufficient numbers of events in certain collapse patterns, the DISE findings regarding degree of collapse were recategorized (i.e., "no collapse" vs "collapse"; and "non-complete collapse" vs "complete collapse") to ensure the power of the logistic regression analysis and accuracy of the regression coefficients. When the numbers of events were still small after recategorization, the specific collapse patterns were not included as independent variables. Univariate logistic regression analyses were used to assess the association between each independent variable and the response to MMA separately. Multivariate logistic regression was used to identify the variables that were independently associated with the response to MMA. The independent variables included in the model were those with *P*-value of < 0.10 in univariate logistic regression. Additionally, the potential confounders consisting of age, gender, BMI, baseline AHI, degree of maxillary advancement, and degree of mandibular advancement were also included in the models³. A forward stepwise procedure was used to select the best logistic regression model, and the goodness-of-fit of the model was assessed using the Hosmer-Lemeshow test. A *P*-value < 0.05 was considered statistically significant.

RESULTS

Patient characteristics

In total, 101 patients underwent MMA for OSA. Thirty-seven patients who did not undergo baseline DISE were excluded, among whom 2 patients had no postoperative PSG data. Therefore, 64 patients were included. There was no significant difference in baseline characteristics and surgical outcome between patients with and without baseline DISE (see **Table S1** in the supplemental material).

Among the included 64 patients, sixty-two patients (96.9%) had CPAP failure or intolerance, two patients (3.1%) refused to try CPAP, and 29 patients (45.3%) underwent other upper airway surgeries (e.g., uvulopalatopharyngoplasty) prior to MMA. A detailed overview of baseline characteristics is summarized in **Table 1**.

Surgical outcome

After MMA, the mean degree of advancement was 7.1 \pm 2.4 mm in maxilla and 9.3 \pm 4.3 mm in mandible (**Table 1**). **Table 2** shows the preoperative and postoperative anteroposterior skeletal pattern based on SNA, SNB, and ANB, and PSG variables. Postoperatively, the total mean AHI was significantly decreased from 49.0 \pm 20.8 events/h to 16.4 \pm 13.3 events/h (*P* < 0.001). In total, 39 patients (60.9%) were responders, and 25 patients (39.1%) were nonresponders.

The baseline characteristics and surgical characteristics in responders and nonresponders are summarized in **Table 1**. No statistically significant difference in any parameter was found between the 2 groups.

Upper airway collapse pattern

An overview of the distribution of levels of upper airway collapse in the present cohort is shown in a Venn diagram (**Figure 2**). As shown, velum collapse was the most commonly present (98.4%), followed by base of tongue (90.6%), epiglottis (84.4%), and oropharynx collapse (51.6%). All patients had multilevel collapse of the upper airway, predominantly a combination of velum, base of tongue, and epiglottic collapse (43.8%), and a combination of velum, oropharyngeal, base of tongue, and epiglottic collapse (35.9%). Fifty-four (84.4%) patients had a complete collapse at multiple levels in the upper airway.

In 64 patients, velum collapse was present either as anteroposterior collapse (n = 42/64; 65.6%) or concentric collapse (n = 21/64; 32.8%). Oropharynx collapse only occurred in lateral configuration, and base of tongue collapse only occurred in anteroposterior pattern. Epiglottis collapse was present either as anteroposterior collapse (n = 53/64; 82.8%) or lateral collapse (n = 1/64; 1.6%).

The mean total degree of obstruction – based on total VOTE score – was 5.8 \pm 1.5 in the total population. No significant difference was found between responders and nonresponders in the mean total VOTE score (5.6 \pm 1.4 vs 6.2 \pm 1.5; *P* = 0.079).

Table 3 presents the upper airway collapse patterns in responders and nonresponders. As shown, responders had a significantly higher occurrence of anteroposterior velum collapse. In contrast, nonresponders had a significantly higher occurrence of concentric velum collapse and complete anteroposterior epiglottic collapse.

Prediction of surgical outcome

In the first model ("no collapse" vs "collapse"), univariate logistic regression showed a significant relationship between response to MMA and anteroposterior velum collapse (odds ratio [OR], 3.611; 95% confidence interval [CI], 1.224-10.654; P = 0.020), indicating that patients with the presence of anteroposterior velum collapse had 3.611 times higher odds to respond to MMA; and a significant relationship between nonresponse to MMA and concentric velum collapse (OR, 0.325; 95%CI, 0.110-0.959; P= 0.042), indicating that patients with the presence of concentric velum collapse had 0.325 times lower odds to respond to MMA (**Table 4**). No predictor of surgical response was found in multivariate logistic regression when controlling for the confounders. In the second model ("non-complete collapse" vs "complete collapse"), univariate analysis revealed a significant relationship between MMA response and complete anteroposterior epiglottic collapse (OR, 0.246; 95%Cl, 0.027-0.854; P = 0.027), indicating that patients with the presence of complete anteroposterior epiglottic collapse had 0.246 times lower odds to respond to MMA (**Table 4**). The association remained significant in multivariate logistic regression (OR, 0.239; 95% Cl, 0.059-0.979; P = 0.047) when controlling for the confounders, indicating that patients with the presence of complete anteroposterior epiglottic collapse had 0.239 times lower odds to respond to MMA. No other predictor regarding collapse patterns, baseline characteristics, and surgical characteristics was found. The Hosmer-Lemeshow test indicated that the fit of the model was good (P > 0.999).

In this study population, patients were categorized as "total resolution" (n = 16) and as "partial resolution" (n = 48) with jaw thrust. The rates of surgical response in "total resolution" group and "partial resolution" group were 75% and 56.3% respectively, while logistic analysis revealed that there was no significant association between change of upper airway collapse with jaw thrust and response to MMA (P = 0.190).

Treatment outcome in patients with and without complete anteroposterior epiglottic collapse

In patients with complete anteroposterior epiglottic collapse (n = 43), the mean AHI decreased from 47.1 \pm 20.8 events/h to 19.2 \pm 14.7 events/h compared with 52.9 \pm 20.7 events/h to 10.6 \pm 6.8 events/h in patients without complete anteroposterior epiglottic collapse (n = 21). The mean reduction in AHI following MMA in patients with complete anteroposterior epiglottic collapse was 27.9 \pm 20.9 events/h vs 42.3 \pm 23.1 events/h in patients without complete anteroposterior epiglottic collapse (*P* = 0.015). The response rates were 51.1% and 81.0% in patients with and without complete anteroposterior epiglottic collapse, respectively.

DISCUSSION

This study is the first to determine whether patients' response to MMA in OSA treatment is predicted by the levels, patterns, and degrees of upper airway collapse identified during DISE along with baseline characteristics and surgical characteristics. In addition, we assessed the value of the jaw thrust maneuver during DISE in the

Chapter 4

prediction of the surgical outcome of MMA. Our key findings were as follows: 1) among baseline characteristics, surgical characteristics, and DISE findings, only the presence of complete anteroposterior epiglottic collapse was independently associated with failure of MMA in treating OSA, and 2) while the surgical response rate in patients without upper airway obstruction during jaw thrust tends to be higher than that in those with upper airway obstruction during jaw thrust, no significant association was found between jaw thrust maneuver and response to MMA.

In this study, epiglottic collapse was observed in 84.4% of the OSA patients, which was higher than the 9.7-73.5% previously described^{17, 18}. First, this may be due to the fact that our patients presented with more severe OSA than the patients recruited in other studies¹⁸⁻²⁰. Second, secondary epiglottic collapse caused by tongue compression was not separated from primary epiglottic collapse. To date, knowledge about treatment for epiglottic collapse is still limited. The available evidence shows that continuous positive airway pressure²¹ and mandibular advancement device²² may be ineffective for dealing with epiglottic collapse, while positional therapy has been proven to provide favorable results for these patients due to the association between epiglottic collapse and positional OSA^{23, 24}. In addition, epiglottic surgery has been suggested to be a good option for correcting epiglottic collapse²⁵. There is currently lack of evidence on the effectiveness of MMA in addressing epiglottic collapse. Our study demonstrated that the presence of complete epiglottic collapse on preoperative DISE was independently associated with surgical failure. This finding is supported by a prospective study by Kastoer et al.¹⁰, who used DISE to assess upper airway collapse pattern in 14 OSA patients before and after MMA, and found that six of eight patients (75%) who had preoperative epiglottic collapse exhibited residual collapse postoperatively¹⁰. They assumed that MMA seemed to be ineffective in treating epiglottic collapse. It should be noted that epiglottic collapse can be classified as primary or secondary collapse, these two types of collapse may have different predictive value of surgical outcome of MMA. However, our study did not distinguish them because this information was absent in some DISE reports based on VOTE score. Furthermore, in the present study, only a few patients presented with no collapse or partial collapse of the tongue base. For this reason, a subgroup analysis of the patients with complete anteroposterior epiglottic collapse based on degrees of the tongue base collapse was difficult to conduct. Due to the limited availability of data, future investigations aimed at this specific DISE finding (i.e., complete anteroposterior epiglottic collapse [primary or secondary collapse]) is necessary. If our finding is confirmed in future research, this DISE phenotype could be a useful tool for patient selection for MMA surgery.

Another specific DISE phenotype, CCCp, has been defined as an absolute contraindication to upper airway stimulation⁸. CCCp was also found to be a negative predictor for a mandibular advancement device²⁶. In the past few years, evidence has been accumulating that MMA is a solution for OSA patients presenting with CCCp. In 2 published studies by Liu et al.⁹ and Kastoer et al.¹⁰, it was demonstrated that all CCCp observed during baseline DISE was eliminated after MMA. Kastoer et al.¹⁰ also concluded that the reduction in AHI after MMA was equal in patients with and without CCCp. In this study, no relationship was found between CCCp and MMA response in multivariate logistic regression. Hence, our study suggests that CCCp may not be an adverse DISE finding toward MMA response. In other words, patients with CCCp, who are denied upper airway surgery, upper airway stimulation, or mandibular advancement device therapy for OSA might still be candidates for MMA.

Different passive maneuvers can be performed during DISE with the aim of predicting the response to some specific treatment modalities. Jaw thrust is often used to predict the effect of MAD on OSA because of the same mechanism, i.e., mandibular protrusion²⁷. Although jaw thrust can stimulate only mandibular advancement, one would intuitively think that, if the obstruction disappears after jaw thrust, the patient would respond favorably to MMA surgery. However, this has not yet been confirmed by scientific evidence. The present study indicates that there is no significant correlation between this maneuver and response to MMA. This is probably due to the fact that the degrees of mandibular advancement are inconsistent between jaw thrust and MMA surgery. More importantly, the effect on upper airway patency of maxillary advancement cannot be mimicked during DISE. Nevertheless, it is interesting to note that the surgical response rate in patients whose upper airway collapse is totally resolved by jaw thrust tends to be higher than that in other patients undergoing MMA. The predictive value of jaw thrust during DISE for MMA surgical response needs further investigation.

In our study population, the surgical success rate of MMA was 60.9%, which was lower than that in published studies^{2, 3, 28, 29}. This may be explained by differences in patient populations between our institute and other centers. First, multiple complete obstructions were observed in the majority of our patients during DISE, and the mean Chapter 4

total VOTE score in our study population was higher than in some other studies³⁰⁻³², which may hint at the treatment complexity for these patients¹². Moreover, 43 of the 64 patients had complete anteroposterior epiglottic collapse, which may be a negative DISE finding for MMA. Second, half of the patients had failure of single-level or multilevel surgery prior to MMA. In some of these nonresponders to upper airway surgery, nonanatomical traits may play a prominent role in the etiology of OSA³³, making them poorer candidates for MMA. Hence, further improving patient selection for MMA therapy – with the use of screening and diagnostic tools – is crucial in increasing its efficacy and cost-effectiveness, and in reducing patient morbidity.

Unlike clinical routine applications for OSA patients before MMA (e.g., PSG, lateral cephalogram, cone beam computed tomography), the indication for DISE has not been fully settled. It was suggested that DISE might be indicated either before any type of surgery, or after surgical correction of volume abnormalities observed on clinical examination, or after failure of primary surgery⁷. Among the routine diagnostic modalities prior to MMA, only PSG has been widely proven to play a role in patient selection for MMA, for which a lower AHI is predictive of increased surgical success^{3, 34}. Our study suggests that patients with complete anteroposterior epiglottic collapse observed during DISE may be at increased risk of surgical failure. Therefore, if the predictive value of DISE for MMA response is confirmed in future research, DISE may be considered as the standard application before MMA surgery.

The authors are aware of the limitations of this study. First, DISE was not performed preoperatively in one-third of patients undergoing MMA for OSA in our institution; however, the baseline characteristics and surgical outcome were comparable between the patients with and without baseline DISE. Second, the interrater and intrarater reliability of DISE cannot be judged and/or enhanced due to the nature of a retrospective study. Third, the VOTE classification is simplistic and may overlook interaction between the upper airway structure³⁵. Furthermore, as stated before, our study did not distinguish primary and secondary epiglottic collapse because this information was absent in some DISE reports based on VOTE classification. Additionally, the numbers of events for some specific DISE variables were limited in our study population and 2 DISE phenotypes (lateral velum collapse and lateral epiglottic collapse) were therefore not included as independent factors in logistic analysis. However, given the relative low incidence of these 2 patterns of obstruction in patients with OSA³⁶, the generalizability of our findings may not be diminished.

Last, in addition to having a small sample size, it is important to note that the study was limited by the study population being predominantly males of relatively low BMI and unknown race/ethnicity. Thus, the results may be limited to this patient profile.

Future prospective studies with larger population are certainly needed to confirm our findings and to further explore the value of DISE in predicting surgical response to MMA for OSA. In this way, primary and secondary epiglottic collapse will be differentiated and numbers of each DISE phenotype will be sufficient to perform the statistical analysis. Additionally, multilevel obstruction is prevalent in patients with moderate to severe OSA³⁷. It may be the case that certain combinations of collapse levels are associated with the surgical outcome of MMA for OSA. Future research in this field is also necessary to answer this question.

CONCLUSION

This study suggests that DISE can be a promising tool in order to identify less suitable candidates for MMA in OSA treatment. The presence of complete anteroposterior epiglottic collapse is associated with a higher possibility of MMA treatment failure. Prospective, larger-scale studies are required to further evaluate the use of DISE in predicting response to MMA.



Figure 1. Cephalometric measurements. 1, SNA (degree); 2, SNB (degree); 3, ANB (degree); 4, A-point to vertical plane (mm); 5, B-point to vertical plane (mm).



Figure 2. A Venn diagram showing the percentages of collapse in each single level and multi levels

Variable	Total (n = 64)	Responders (n = 39)	Nonresponders (n = 25)	P-value
Age, years	51.7 ± 9.5	51.1 <u>+</u> 9.6	52.7 ± 9.3	0.506
Male:female	50:14	28:11	22:3	0.126
BMI, kg/m²	28.7 ± 4.0	28.4 <u>+</u> 3.6	29.2 ± 4.5	0.413
Neck circumference, cm	41.7 ± 3.8	41.1 ± 3.2	42.6±4.4	0.140
Previous OSA upper airway surgery, n (%)	29 (45.3)	16 (41.0)	13 (52.0)	0.390
AHI, events/h	49.0 <u>+</u> 20.8	52.3 <u>+</u> 18.4	43.7±23.5	0.107
ODI, events/h	46.4 <u>+</u> 21.2	48.0 <u>+</u> 19.6	44.0 ± 23.8	0.501
LSAT, %	77.8 ± 10.5	76.0 <u>+</u> 12.5	80.6 ± 5.3	0.395
SNA, degree	80.1 <u>+</u> 3.5	79.9 ± 3.1	80.5 ± 4.2	0.536
SNB, degree	75.8 <u>+</u> 4.1	75.4 ± 4.0	76.6 <u>+</u> 4.1	0.292
ANB, degree	4.4 ± 3.1	4.7 ± 2.5	3.9 ± 3.8	0.416
Adv A, mm	7.1 <u>+</u> 2.4	7.9 ± 2.5	6.7 ± 2.3	0.275
Adv B, mm	9.3 ± 4.3	9.9±4.9	7.8 ± 3.1	0.067

 Table 1. Baseline characteristics and surgical characteristics of the total population, responders and nonresponders

Adv A, advancement degree of A-point; Adv B, advancement of degree of B-point; AHI, apnea-hypopnea index; ANB, A-point–Nasion–B-point angle; BMI, body mass index; LSAT, lowest oxygen saturation; ODI, oxygen desaturation index; OSA, obstructive sleep apnea; SNA, sella–nasion–A-point angle; SNB, sella–nasion–B-point angle.

Continuous data presented as mean \pm standard deviation, categorical data presented as number with percentage.

P-value for the comparison of the responders and nonresponders; P < 0.05 was considered statistically significant.

Variable	Total population (n = 64)	Responders (n = 39)	Nonresponders (n = 25)
Cephalometric variables			
Pre-op SNA, degree	80.1 <u>+</u> 3.5	79.9 ± 3.1	80.5 ± 4.2
Post-op SNA, degree	87.0 ± 5.1	87.1 ± 4.4	87.0 <u>+</u> 6.2
Pre-op SNB, degree	75.8 ± 4.1	75.4 ± 4.0	76.6 <u>+</u> 4.1
Post-op SNB, degree	81.6 ± 4.8	81.5 ± 4.3	81.7 ± 5.6
Pre-op ANB, degree	4.4 ± 3.1	4.7 ± 2.5	3.9 ± 3.8
Post-op ANB, degree	5.7 ± 3.1	6.0 <u>±</u> 2.1	5.3 ± 4.1
Polysomnographic variables			
Pre-op AHI, events/h	49.0 <u>+</u> 20.8	52.3 ± 18.4	43.7±23.5
Post-op AHI, events/h	16.4 ± 13.3	8.8 <u>+</u> 5.1	28.2 ± 13.5
Pre-op ODI, events/h	46.4 <u>+</u> 21.2	48.0 <u>+</u> 19.6	44.0 <u>±</u> 23.8
Post-op ODI, events/h	22.0 <u>+</u> 14.6	14.1 ± 7.8	33.0 ± 14.9
Pre-op LSAT, %	77.8 ± 10.5	76.0 ± 12.5	80.6 ± 5.3
Post-op LSAT, %	85.6 <u>+</u> 4.6	86.4 ± 5.4	84.6 ± 2.9

 Table 2. Cephalometric analysis and polysomnographic variables before and after maxillomandibular advancement in total population, responders and nonresponders

AHI, apnea-hypopnea index; ANB, A-point–Nasion–B-point angle; LSAT, lowest oxygen saturation; ODI, oxygen desaturation index; SNA, sella–nasion–A-point angle; SNB, sella–nasion–B-point angle. Data presented as mean ± standard deviation.

Level: pattern of	Responders (n = 39)			Nonresponders (n = 25)		
collapse	No collapse (%)	Partial collapse (%)	Complete collapse (%)	No collapse (%)	Partial collapse (%)	Complete collapse (%)
Velum						
Anteroposterior	9 (23.1) [*]	4 (10.3)	26 (66.7)	13 (52.0) [*]	1 (4.0)	11 (44.0)
Lateral	39 (100.0)	0 (0)	0 (0)	25 (100.0)	0 (0)	0 (0)
Concentric	30 (76.9) [*]	0 (0)	9 (23.1)	13 (52.0)*	1 (4.0)	11 (44.0)
Oropharynx: lateral	18 (46.2)	9 (23.1)	12 (30.8)	13 (52.0)	3 (12.0)	9 (36.0)
Base of tongue: anteroposterior	5 (12.8)	10 (25.6)	24 (61.5)	1 (4.0)	5 (20.0)	19 (76.0)
Epiglottis						
Anteroposterior	10 (25.6)	7 (17.9)	22 (56.4) [†]	1 (4.0)	3 (12.0)	21 (84.0) [†]
Lateral	38 (97.4)	0 (0)	1 (2.6)	25 (100.0)	0 (0.0)	0 (0.0)

Table 3. Drug-induced sleep endoscopy findings in responders and nonresponders

Data presented as number of patients with percentage. * Significance accepted at P-value < 0.05 for no collapse: responders vs nonresponders.

[†] Significance accepted at *P*-value < 0.05 for complete collapse: responders vs nonresponders.

Collapse site and pattern	OR [95% CI]	P-value
Collapse degree: no collapse vs collapse		
Velum anteroposterior		
No collapse	Ref.	Ref.
Collapse	3.611 [1.224-10.654]	0.020
Velum lateral	NA	NA
Velum concentric		
No collapse	Ref.	Ref.
Collapse	0.325 [0.110-0.959]	0.042
Oropharynx lateral		
No collapse	Ref.	Ref.
Collapse	1.264 [0.462-3.456]	0.648
Base of tongue anteroposterior		
No collapse	Ref.	Ref.
Collapse	0.283 [0.031-2.582]	0.263
Epiglottis anteroposterior		
No collapse	Ref.	Ref.
Collapse	0.106 [0.014-1.012]	0.051
Epiglottis lateral	NA	NA
Total VOTE score	0.740 [0.514-1.067]	0.107
Jaw thrust maneuver effect		
Partial resolution	Ref.	Ref.
Total resolution	2.333 [0.657-8.285]	0.190
Collapse degree: non-complete collapse vs con	nplete collapse	
Velum anteroposterior		
Non-complete collapse	Ref.	Ref.
Complete collapse	2.545 [0.906-7.151]	0.076
Velum lateral	NA	NA
Velum concentric		
Non-complete collapse	Ref.	Ref.
Complete collapse	0.382 [0.129-1.131]	0.082
Oropharynx lateral		
Non-complete collapse	Ref.	Ref.
Complete collapse	0.790 [0.273-2.287]	0.664
Base of tongue anteroposterior		
Non-complete collapse	ref.	ref.
Complete collapse	0.505 [0.165-1.551]	0.233
Epiglottis anteroposterior		
Non-complete collapse	Ref.	Ref.
Complete collapse	0.246 [0.027-0.854]	0.027
Epiglottis lateral	NA	NA
Total VOTE score	0.740 [0.514-1.067]	0.107
Jaw thrust maneuver effect		
Partial resolution	Ref.	Ref.
Total resolution	2.333 [0.657-8.285]	0.190

Table 4. Univariate logistic regression for DISE variables

CI, confidence interval; DISE, drug-induced sleep endoscopy; NA, not applicable; OR, odds ratio; Ref, reference category; VOTE, Velum Oropharynx Tongue base Epiglottis.

SUPPLEMENTARY MATERIAL

Variable	Patients with baseline DISE (n = 64)	Patients without baseline DISE (n = 37)	P-value
Age, years	51.7 ± 9.5	49.2 ± 9.5	0.200
Male:female	50:14	31:6	0.492
BMI, kg/m²	28.7 ± 4.0	30.4 ± 5.3	0.075
Neck circumference, cm	41.7 ± 3.8	43.4 ± 4.4	0.141
Previous OSA upper airway surgery, n (%)	29 (45.3)	13 (35.1)	0.317
AHI, events/h	49.0 <u>+</u> 20.8	50.6 <u>+</u> 26.1	0.738
ODI, events/h	46.4 <u>+</u> 21.2	53.1 ± 29.3	0.303
LSAT, %	77.8 <u>+</u> 10.5	76.3 <u>+</u> 10.8	0.552
SNA, degree	80.1 ± 3.5	80.5±5.4	0.700
SNB, degree	75.8 ± 4.1	74.8 ± 6.7	0.392
ANB, degree	4.4 ± 3.1	6.0 <u>±</u> 3.2	0.089
Responder:nonresponder	39:25	23:14	0.639

Table S1. Baseline characteristics of patients with and without baseline drug-induced sleep endoscopy

AHI, apnea-hypopnea index; ANB, A-point–Nasion–B-point angle; BMI, body mass index; DISE, drug-induced sleep endoscopy; LSAT, lowest oxygen saturation; ODI, oxygen desaturation index; SNA, sella–nasion–A-point angle; SNB, sella–nasion–B-point angle.

Continuous data presented as mean \pm standard deviation, categorical data presented as number with percentage.

P-value for the comparison of the patients with baseline DISE and patients without baseline DISE.

REFERENCES

- 1. Caples SM, Rowley JA, Prinsell JR, et al. Surgical modifications of the upper airway for obstructive sleep apnea in adults: a systematic review and meta-analysis. *Sleep.* 2010;33:1396-407.
- 2. Zhou N, Ho JTF, Huang Z, et al. Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis. *Sleep Med Rev.* 2021;57:101471.
- 3. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.
- 4. de Ruiter MHT, Apperloo RC, Milstein DMJ, de Lange J. Assessment of obstructive sleep apnoea treatment success or failure after maxillomandibular advancement. *Int J Oral Maxillofac Surg.* 2017;46:1357-62.
- 5. Croft CB, Pringle M. Sleep nasendoscopy: a technique of assessment in snoring and obstructive sleep apnoea. *Clin Otolaryngol Allied Sci.* 1991;16:504-9.
- Kim JW, Kim DS, Kim SD, Mun SJ, Koo SK, Cho KS. Does drug-induced sleep endoscopy predict surgical success of limited palatal muscle resection in patients with obstructive sleep apnea? *Auris Nasus Larynx*. 2018;45:1027-32.
- 7. Blumen M, Bequignon E, Chabolle F. Drug-induced sleep endoscopy: A new gold standard for evaluating OSAS? Part II: Results. *Eur Ann Otorhinolaryngol Head Neck Dis.* 2017;134:109-15.
- Vanderveken OM, Maurer JT, Hohenhorst W, et al. Evaluation of drug-induced sleep endoscopy as a patient selection tool for implanted upper airway stimulation for obstructive sleep apnea. J Clin Sleep Med. 2013;9:433-8.
- 9. Liu SY, Huon LK, Iwasaki T, et al. Efficacy of Maxillomandibular Advancement Examined with Drug-Induced Sleep Endoscopy and Computational Fluid Dynamics Airflow Modeling. *Otolaryngol Head Neck Surg.* 2016;154:189-95.
- 10. Kastoer C, Op de Beeck S, Dom M, et al. Drug-Induced Sleep Endoscopy Upper Airway Collapse Patterns and Maxillomandibular Advancement. *Laryngoscope*. 2020;130:E268-E274.
- 11. Von Elm E, Altman DG, Egger M, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) Statement: guidelines for reporting observational studies. *Int] Surg.* 2014;12:1495-9.
- 12. Koutsourelakis I, Safiruddin F, Ravesloot M, Zakynthinos S, de Vries N. Surgery for obstructive sleep apnea: sleep endoscopy determinants of outcome. *Laryngoscope*. 2012;122:2587-91.
- Berry RB, Brooks R, Gamaldo CE, et al; for the American Academy of Sleep Medicine. The AASM Manual for the Scoring of Sleep and Associated Events: Rules, Terminology and Technical Specifications, version 2.0. Darien, IL: American Academy of Sleep Medicine; 2012.
- 14. Sher AE, Schechtman KB, Piccirillo JF. The efficacy of surgical modifications of the upper airway in adults with obstructive sleep apnea syndrome. *Sleep*. 1996;19:156-77.
- 15. De Vito A, Carrasco Llatas M, Vanni A, et al. European position paper on drug-induced sedation endoscopy (DISE). *Sleep Breath.* 2014;18:453-65.
- 16. Kezirian EJ, Hohenhorst W, de Vries N. Drug-induced sleep endoscopy: the VOTE classification. *Eur Arch Otorhinolaryngol.* 2011;268:1233-6.
- 17. Torre C, Camacho M, Liu SY, Huon LK, Capasso R. Epiglottis collapse in adult obstructive sleep apnea: A systematic review. *Laryngoscope*. 2016;126:515-23.
- 18. Vroegop AV, Vanderveken OM, Boudewyns AN, et al. Drug-induced sleep endoscopy in sleep-disordered breathing: report on 1,249 cases. *Laryngoscope*. 2014;124:797-802.
- 19. Fernández-Julián E, García-Pérez M, García-Callejo J, Ferrer F, Martí F, Marco J. Surgical planning after sleep versus awake techniques in patients with obstructive sleep apnea. *Laryngoscope*. 2014;124:1970-4.
- 20. Lan MC, Liu SY, Lan MY, Modi R, Capasso R. Lateral pharyngeal wall collapse associated with hypoxemia in obstructive sleep apnea. *Laryngoscope*. 2015;125:2408-12.
- 21. Dedhia RC, Rosen CA, Soose RJ. What is the role of the larynx in adult obstructive sleep apnea? *Laryngoscope*. 2014;124:1029-34.
- 22. Kent DT, Rogers R, Soose RJ. Drug-Induced Sedation Endoscopy in the Evaluation of OSA Patients with Incomplete Oral Appliance Therapy Response. *Otolaryngol Head Neck Surg.* 2015;153:302-7.

- 23. Marques M, Genta PR, Sands SA, et al. Effect of Sleeping Position on Upper Airway Patency in Obstructive Sleep Apnea Is Determined by the Pharyngeal Structure Causing Collapse. *Sleep*. 2017;40:zsx005
- 24. Victores AJ, Hamblin J, Gilbert J, Switzer C, Takashima M. Usefulness of sleep endoscopy in predicting positional obstructive sleep apnea. *Otolaryngol Head Neck Surg.* 2014;150:487-93.
- Salamanca F, Leone F, Bianchi A, Bellotto RGS, Costantini F, Salvatori P. Surgical treatment of epiglottis collapse in obstructive sleep apnoea syndrome: epiglottis stiffening operation. *Acta Otorhinolaryngol Ital.* 2019;39:404-8.
- 26. Op de Beeck S, Dieltjens M, Verbruggen AE, et al. Phenotypic Labelling Using Drug-Induced Sleep Endoscopy Improves Patient Selection for Mandibular Advancement Device Outcome: A Prospective Study. J Clin Sleep Med. 2019;15:1089-99.
- 27. Park D, Kim JS, Heo SJ. The Effect of the Modified Jaw-Thrust Maneuver on the Depth of Sedation During Drug-Induced Sleep Endoscopy. J Clin Sleep Med. 2019;15:1503-8.
- 28. Goh YH, Lim KA. Modified maxillomandibular advancement for the treatment of obstructive sleep apnea: a preliminary report. *Laryngoscope*. 2003;113:1577-82.
- 29. Vicini C, Dallan I, Campanini A, et al. Surgery vs ventilation in adult severe obstructive sleep apnea syndrome. *Am J Otolaryngol.* 2010;31:14-20.
- Meraj TS, Muenz DG, Glazer TA, Harvey RS, Spector ME, Hoff PT. Does drug-induced sleep endoscopy predict surgical success in transoral robotic multilevel surgery in obstructive sleep apnea? *Laryngoscope*. 2017;127:971-6.
- Yui MS, Tominaga Q, Lopes BCP, et al. Nasal vs. oronasal mask during PAP treatment: a comparative DISE study. Sleep Breath. 2020;24:1129-36.
- 32. Ong AA, Murphey AW, Nguyen SA, et al. Efficacy of Upper Airway Stimulation on Collapse Patterns Observed during Drug-Induced Sedation Endoscopy. *Otolaryngol Head Neck Surg.* 2016;154:970-7.
- 33. Carberry JC, Amatoury J, Eckert DJ. Personalized Management Approach for OSA. Chest. 2018;153:744-55.
- 34. Zaghi S, Holty JE, Certal V, et al. Maxillomandibular Advancement for Treatment of Obstructive Sleep Apnea: A Meta-analysis. JAMA Otolaryngol Head Neck Surg. 2016;142:58-66.
- Hohenhorst W, Ravesloot MJL, Kezirian EJ, de Vries N. Drug-induced sleep endoscopy in adults with sleepdisordered breathing: Technique and the VOTE Classification system. Oper Tech Otolayngol Head Neck Surg. 2012;23:11-8.
- 36. Cavaliere M, Russo F, Iemma M. Awake versus drug-induced sleep endoscopy: evaluation of airway obstruction in obstructive sleep apnea/hypopnoea syndrome. *Laryngoscope*. 2013;123:2315-8.
- 37. Sharifian MR, Zarrinkamar M, Alimardani MS, et al. Drug Induced Sleep Endoscopy in Obstructive Sleep Apnea. *Tanaffos.* 2018;17:122-6.



Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis

Ning Zhou, Jean-Pierre T.F. Ho, Zhengfei Huang, René Spijker, Nico de Vries, Ghizlane Aarab, Frank Lobbezoo, Madeline J.L. Ravesloot, Jan de Lange

Published in: Sleep Med Rev, 2021. doi: 10.1016/j.smrv.2021.101471

ABSTRACT

Multilevel surgery (MLS) and maxillomandibular advancement surgery (MMA) are two established options in surgical management of obstructive sleep apnea (OSA), which target different levels of airway obstruction. The objective of this review was to comparatively evaluate the clinical efficacy and safety of MMA and MLS in the treatment of OSA. MEDLINE and Embase databases were searched for studies on MMA and/or MLS in OSA patients. Twenty MMA studies and 39 MLS studies were identified. OSA patients who underwent MMA showed significant improvements in apnea-hypopnea index (AHI), lowest oxygen saturation (LSAT), oxygen desaturation index (ODI), and Epworth sleepiness scale (ESS) by -46.2 events/h, 13.5%, -30.3 events/h, and -8.5, respectively. The pooled rates of surgical success and cure for MMA were 85.0% and 46.3%, respectively. Patients who underwent MLS showed significant improvements in AHI, LSAT, ODI, and ESS by -24.7 events/h, 8.7%, -19.1 events/h, and -5.8, respectively. The pooled surgical success and cure rates for MLS were 65.1% and 28.1%, respectively. The rates of major complication of MMA and MLS were 3.2% and 1.1%, respectively, and the rate of minor complication of MMA was higher than that of MLS. We conclude that both MMA and MLS are effective treatment options for OSA. Compared to MLS, MMA may be more effective in improving OSA. However, the complication rate of MMA is higher.

Keywords: Obstructive sleep apnea; Maxillomandibular advancement; Multilevel surgery; Surgery; Systematic review; Meta-analysis

INTRODUCTION

Obstructive sleep apnea (OSA), a potentially life-threatening sleep-related breathing disorder, is characterized by repetitive partial or complete obstruction of the upper airway during sleep, causing hypoxemia and sleep fragmentation¹. A recent systematic review reported that the overall prevalence of OSA ranges from 9% to 38% in the general adult population².

Continuous positive airway pressure (CPAP) is generally accepted as a first-line therapy for patients with moderate to severe OSA³. However, the clinical efficacy of CPAP can be hampered by its often low compliance rate, prompting a substantial proportion of OSA patients to seek therapeutic alternatives, such as a mandibular advancement device (MAD) and surgical treatment⁴. Surgical treatment is a viable alternative for patients who have specific surgically correctable anatomical abnormalities, which play an important role in upper airway obstruction⁵.

Moderate to severe OSA is usually characterized by multilevel obstructions⁶, hence the surgical interventions aimed to correct only one region cannot eliminate all obstructions in the upper airway. In 1986, Riley et al.⁷ have first proposed multilevel surgery (MLS) for OSA patients with multiple obstructions. Today, MLS for OSA is widely accepted as treatment modality in case of multilevel obstruction.

MLS however, is not suitable for all OSA patients. Another commonly employed surgical procedure that targets multiple levels is maxillomandibular advancement (MMA), which has been demonstrated to be the most effective surgical option for OSA⁸. The reported surgical success rate for MMA is 86.0%⁹.

Currently, there is still no universally accepted guideline of surgical procedures for OSA given the variations in anatomy, disease severity, patient comorbidities, and patient preference. For OSA cases with diffusely complex or multiple sites of obstruction, the indications and staged protocols of surgical treatment remain unclear. When there is no generally accepted indicative results of clinical, laboratory, or endoscopic examination in patients with moderate to severe OSA (e.g., significant skeletal-dental deformity, complete concentric collapse at velum observed with drug-induced sleep endoscopy [DISE]), some surgeons are inclined to start with MLS and keep MMA as a reserve therapeutic option in case of surgical failure, while others prefer to start with MMA as the primary treatment option. Thus, further definition of the role of MMA and MLS in the treatment protocol for OSA is called for, which is vital for both patients and physicians in final decision-making regarding the choice of surgery type. To our knowledge, only one systematic review¹⁰ published in 2010 has compared MMA and MLS for OSA treatment, but only regarding the aspect of clinical efficacy, which places emphasis on the need for an updated and thorough assessment and comparison of the two types of surgical interventions. Thus, the aim of this systematic review was to comprehensively evaluate and compare the treatment outcome of MMA and MLS for OSA treatment, through the assessment of apnea-hypopnea index (AHI) and Epworth sleepiness scale (ESS) as primary outcomes. The secondary objective was to investigate the differences in complication rates for both treatment options.

METHODS

In accordance with the preferred reporting items for systematic reviews and metaanalyses (PRISMA) statement, the protocol for the systematic review was registered (PROSPERO ID: CRD42020152077;https://www.crd.york.ac.uk/PROSPERO/display_ record.php?ID=CRD42020152077).

Selection criteria

The inclusion criteria were as follows: (1) adult patients (> 18 years old) with OSA diagnosed by means of polysomnography (PSG; AHI \geq 5 events/h); (2) patients that underwent MMA or one-phase MLS (at least one velopharyngeal and one hypopharyngeal surgery in single stage); (3) studies that reported pre- and postoperative PSG data; (4) studies with a follow-up \geq 6 months; (5) studies with the following designs: randomized controlled trials (RCTs), quasi-experimental studies, and cohort studies; and (6) English language.

Studies were excluded from the review if: (1) sample size < 10 patients; (2) studies with patients who underwent other adjunctive procedures at the time of MMA (e.g., tonsillectomy, uvulopalatopharyngoplasty, partial glossectomy); and (3) preliminary studies in which the findings had been nested in other studies with larger sample size and/or longer follow-up.

Literature Search

With the assistance of an information specialist, a literature search was performed using the MEDLINE and Embase database on May 6, 2020. Search terms and full search strategies used for each database utilized are available as supplementary information (Supplementary **Table S1a** and **S1b**).

Study selection

Two reviewers (NZ and ZH) independently selected studies for further assessment by title and abstract review. All potentially eligible studies were retrieved in full texts for further evaluation. In case of disagreement, a third reviewer (JH) was consulted. The reference lists of the retrieved papers were manually checked by NZ and ZH.

Data extraction

A specially designed data-extraction form was used to extract data from the included studies. Extracted information included:

- General information: article title, year of publication, and first author.
- Study characteristics: study design and length of follow-up.
- Participant characteristics: sample size, age, gender, and body mass index (BMI) (kg/m²).
- Intervention and setting: specific surgical technique.
- Outcome data: results of pre- and postoperative PSG, including apneahypopnea index (AHI), respiratory disturbance index (RDI), lowest saturation of oxygen (LSAT), and oxygen desaturation index (ODI); pre- and postoperative Epworth sleepiness scale (ESS) score; surgical success rate and cure rate; postoperative complications; and duration of hospital stay.

Data were extracted by NZ and ZH independently. Discrepancies were resolved through discussion with JH. If RDI was reported in a study, it would be extracted as AHI, since these two respiratory parameters have been consolidated based on the 2013 American Academy of Sleep Medicine's manual for the scoring of sleep and associated events¹¹. We defined "surgical success" as "at least 50% reduction in AHI following surgery accompanied by a postoperative AHI of < 20 events/h"¹², and "surgical cure" as "a postoperative AHI < 5 events/h"¹³. If there were multiple follow-up data in the results, the data with the longest follow-up time were selected.

Quality assessment

Methodologic quality assessment of each study was performed by NZ and ZH independently, and any discrepancies were resolved through discussion with JH.

The risk of bias of included RCTs were assessed using the Cochrane Collaboration "Risk of bias" tool¹⁴. Six domains of bias, including selection, attribution, detection, performance, reporting, and other bias, were classified as "low risk", "high risk" or "unclear risk". The total quality of each study was considered as good (low risk of bias for at least 3 items), fair (low risk of bias for 2 items), or low (low risk for no items or 1 item)¹⁵.

The quality assessment of non-randomized studies was based on the Methodological Index for Non-Randomized Studies (MINORS), which is a validated tool for the methodological assessment of non-randomized surgical studies¹⁶. The MINORS tool includes 12 items for comparative studies, the first eight being specifically for non-comparative studies. Each item was scored as 0 (not reported), 1 (reported but inadequate), or 2 (reported and adequate). The global ideal score was 24 for comparative studies and 16 for non-comparative studies. The categorization of comparative studies was as follows: 0-6 "very low quality", 7-10 "low quality", 11-15 "fair quality", and \geq 16 "high quality". For non-comparative studies, the total score of 0-4 indicates very low quality, 5-7 indicates low quality, 8-12 indicates fair quality, and \geq 13 indicates high quality¹⁷.

The studies categorized as "high risk of bias" or "low/very low quality" were excluded from the meta-analysis.

Statistical analysis

The weighted mean () and weighted standard deviation () of parameters (age, BMI, AHI, LSAT, and ESS) were calculated using the following equations, respectively¹⁸:

$$\overline{x}^* = \frac{\sum_{i=1}^N w_i x_i}{\sum_{i=1}^N w_i}$$
$$\overline{SD}^* = \sqrt{\frac{\sum_{i=1}^N w_i (x_i - \overline{x}^*)^2}{\frac{(M-1)}{M} \sum_{i=1}^N w_i}}$$

N is the number of observations; M is the number of nonzero weights; W_i are the weights; and x_i are the observations.

The inverse variance methods for meta-analysis was conducted to pool the results of AHI, LSAT, and ESS, respectively, and rendered a weighted mean difference (WMD) and its associated 95% confidence interval (CI). The magnitude of the effect was interpreted through the value of standardized mean difference (SMD); small = 0.2, medium = 0.5 and large = 0.8^{19} . The random effects model and fixed effects model were used depending on the presence of heterogeneity. Heterogeneity between studies was evaluated by Cochran Q statistic, with a statistical heterogeneity cutoff of $P < 0.10^{20}$, as well as I^2 statistic with cutoff of 25% (low), 50% (moderate), and 75% (high)²¹. Pooled surgical success and cure rates were generated in the meta-analysis by using the DerSimonian-Laird random effects pooling method.

Given the inconsistency of surgical interventions utilized in MLS, the subgroup analysis was done for the subsets of study groups according to the combination of different target levels of surgery (surgery addressing obstruction at the levels of soft palate and tongue base – subgroup 1; soft palate and hyoid – subgroup 2; and soft palate, tongue base, and hyoid – subgroup 3). Based on current literature, it is suggested that increasing preoperative severity of OSA is likely an important predictor of treatment failure^{9, 22}, combined with the heterogeneity of patients' baseline AHI in the analyzed studies. Therefore, we calculated separate pooled estimates for studies with different range of mean baseline AHI (AHI < 40 events/h; 40 events/h \leq AHI \leq 70 events/h; AHI > 70 events/h). These cut-off values were determined based on the range of average baseline AHI of all included studies. A subgroup analysis was also conducted in the studies with long follow-up periods (\geq 2 years). The comparison of the estimates for each outcome between MMA and MLS was performed by using Z test, as proposed by Altman and Bland²³.

Risk of publication bias across studies was assessed by Begg's test and Egger's test, with *P*-value of <0.05 suggesting the presence of bias. Sensitivity analyses were conducted to assess the stability of the results. Statistical analyses were conducted using Review Manager version 5.3 (Cochrane Centre, Copenhagen, Denmark) and Stata version 16.0 (StataCorp LLC, College Station, USA).

RESULTS

The PRISMA flow diagram of study selection progress is described in Supplementary **Fig. S1**. The search in the electronic database resulted in 3051 publications after deduplication, from which 172 full articles were retrieved for further full-text evaluation.

MMA group Twenty studies were identified²⁴⁻⁴³. One of these was an RCT, one was a retrospective quasi-experimental study, nine were prospective cohort studies, and nine were retrospective cohort studies. Their characteristics are shown in **Table 1**. The mean follow-up period from surgery to postoperative PSG was 25.4 months (range, 6.0 months-12.5 years).

MLS group Thirty-nine articles fulfilled the inclusion criteria, including one article added from hand searching of included articles' reference lists^{22, 44-81}. One was a randomized controlled trial, five were prospective quasi-experimental studies, six were retrospective quasi-experimental studies, eleven were prospective cohort studies, and 17 were retrospective cohort studies. Their characteristics are shown in **Table 2**. The mean follow-up period from surgery to postoperative PSG was 9.9 months (range, 6.0 months-3.3 years).

Quality assessment of individual studies

MMA group The only RCT⁴¹ was considered of good quality (Supplementary **Fig. S2**). Of the non-randomized studies, two studies were classified as "high quality", and the other 17 studies as "fair quality" (Supplementary **Table S2a**).

MLS group The only RCT⁵³ was considered of fair quality (Supplementary **Fig. S2**). Of the non-randomized studies, seven studies were classified as "high quality", twenty-nine studies as "fair quality", and two studies as "low quality" (Supplementary **Table S2b**).

Demographic data

MMA group Twenty studies on MMA were reviewed. Excluding duplication of data yielded a total of 528 distinct patients, most of whom were overweight (weighted BMI: $28.6 \pm 6.6 \text{ kg/m}^2$) males (78.9%) with a weighed mean age of 42.9 years (**Table 3**).

MLS group As shown in **Table 3**, the identified studies produced a pooled data set of 1712 OSA patients who underwent MLS. The majority of the patients were obese (weighted BMI: 29.1 ± 4.2 kg/m²) males (85.0%) with a weighted mean age of 45.5 years.

Respiratory parameters

MMA group One study³⁵ was excluded from the meta-analysis, because the data of a small subset of the patients with longer follow-up time were reported in another study³⁴. As shown in **Table 3**, nineteen studies, describing 393 patients with weighted preoperative AHI of 57.3 \pm 26.6 events/h, reported a statistically significant improvement in AHI of -46.2 events/h (95%CI, -52.4 to -39.9, *P* < 0.001), LSAT of 13.5% (95%CI, 10.5 to 16.5, *P* < 0.001), and ODI of -30.3 events/h (95%CI, -46.3 to -14.2, *P* < 0.001) (Supplementary **Fig. S3**). The SMDs of AHI, LSAT, and ODI were -2.90 (95%CI, -3.40 to -2.40) (large effect), 1.49 (95% CI, 1.21 to 1.76) (large effect), and -2.61 (95%CI, -4.23 to -1.00) (large effect), respectively.

MLS group Two studies^{44, 62} were excluded from the meta-analysis because of the low methodological quality. As shown in **Table 3**, thirty-seven studies, totaling 1639 patients with weighted preoperative AHI of 42.2 \pm 21.0 events/h, reported a statistically significant improvement in AHI of -24.7 events/h (95%Cl, -28.1 to -21.4, *P* < 0.001), LSAT of 8.7% (95%Cl, 6.2 to 11.1, *P* < 0.001), and ODI of -19.1 events/h (95%Cl, -34.2 to -4.0, *P* = 0.010) (Supplementary **Fig. S4**). The SMDs of AHI, LSAT, and ODI were -1.79 (95%Cl, -2.06 to -1.52) (large effect), 1.06 (95%Cl, 0.79 to 1.34) (large effect), and -1.18 (95%Cl, -1.74 to -0.62) (large effect), respectively. The results of weighted data for three subgroups according to the different target levels of obstructive sites addressed by surgery were summarized in **Table 4** (Supplementary **Fig. S5**).

The improvements of AHI and LSAT after MMA were significantly higher than after MLS, with *P*-value of <0.001 and 0.014, respectively. No significant difference in the improvement of ODI between MMA and MLS was found.

Subjective outcomes

MMA group Seven studies, totaling 164 patients with weighted preoperative ESS of 14.1 \pm 5.4, reported a significant decrease of 8.5 (95%Cl, -12.2 to -4.9, *P* < 0.001) (**Table 3**; Supplementary **Fig. S3**). The ESS SMD was -2.15 (95%Cl, -3.06 to -1.24) (large effect).

MLS group Twenty-nine studies, totaling 1309 patients with weighted preoperative ESS of 12.6 \pm 4.4, reported a significant reduction of -5.8 (95%Cl, -6.6 to -5.0, *P* < 0.001) (**Table 3**; Supplementary **Fig. S4**). The ESS SMD was -1.51 (95%Cl, -1.78 to -1.25) (large effect). The results of subgroup analysis based on surgical technique were shown in **Table 4** (Supplementary **Fig. S5**)

Chapter 5

No significant difference in the improvement of ESS between MMA and MLS was found.

Surgical success and cure

MMA group The pooled rate of surgical success reported in 15 studies (n = 340) was 85.0% (95%Cl, 76.4% to 91.9%), and the pooled rate of surgical cure reported in five studies (n = 130) was 46.3% (95%Cl, 38.0% to 54.7%).

MLS group The overall pooled rate of surgical success reported in 31 studies (35 MLS groups, n = 1339) was 65.1% (95%Cl, 60.6% to 69.5%), and the overall pooled rate of surgical cure was 28.1% (95%Cl, 13.2% to 46.1%) in five studies (5 MLS groups, n = 221). The pooled surgical success and cure rates for each subgroup with regard to surgical technique were listed in **Table 4**.

The overall pooled surgical success rate of MMA was significantly higher than that of MLS (P < 0.001), and no significant difference was found in the pooled surgical cure rate between these two therapies.

Severity of OSA: impact on results

All MMA study groups were divided into the following three cohorts with respect to the mean baseline AHI: less than 40 events/h, from 40 events/h to 70 events/h, and greater than 70 events/h. For MLS groups, they were only divided into two cohorts according to the mean baseline AHI, due to the absence of included MLS studies with mean baseline AHI > 70 events/h.

Baseline AHI less than 40 events/h

MMA group In **Table 5**, three studies, totaling 60 patients with weighted preoperative AHI of 35.7 \pm 13.7 events/h, reported a significant improvement in AHI of -27.1 events/h (P < 0.001), and ESS of -12.7 (P = 0.002) (Supplementary **Fig. S3**). No study described LSAT. Only one study with 34 patients reported data concerning the preoperative and postoperative ODI (34.7 \pm 12.5 events/h and 5.4 \pm 4.1 events/h (P < 0.001), respectively). The pooled rates of success and cure were 94.0% (95%CI, 74.3% to 99.9%) and 50.0% (95%CI, 35.7% to 64.2%), respectively.

MLS group In **Table 5**, fifteen studies, comprising 706 patients with weighted preoperative AHI of 30.7 ± 15.6 events/h, showed a significant improvement in AHI of -16.7 events/h (P < 0.001), LAST of 4.4% (P = 0.001), and ESS of -5.4 (P < 0.001) (Supplementary **Fig. S4**). No significant improvement of ODI was found. The pooled

rates of success and cure were 57.1% (95%Cl, 51.7% to 62.5%) and 44.7% (95%Cl, 33.2% to 56.4%), respectively.

Compared to the MLS, the AHI reductions after MMA was significantly higher, with *P*-values of 0.030. The pooled surgical success rate of MMA was significantly higher than MLS (P < 0.001), while there is no difference in the surgical cure rates between these two types of therapies.

Baseline AHI from 40 events/h to 70 events/h

MMA group In **Table 5**, twelve studies, comprising 257 patients with weighted preoperative AHI of 55.7 \pm 23.0 events/h, reported a significant improvement in AHI of -44.1 events/h (*P* < 0.001), LSAT of 11.6% (*P* < 0.001), ODI of -30.4 events/h (*P* = 0.030), and ESS of -7.0 (*P* < 0.001) (Supplementary **Fig. S3**). The pooled rates of success and cure were 82.3% (95%Cl, 69.1% to 92.5%) and 44.0% (95%Cl, 33.1% to 55.3%), respectively.

MLS group In **Table 5**, twenty-two studies, comprising 933 patients with weighted preoperative AHI of 51.0 ± 20.3 events/h, showed a significant improvement in AHI of -30.7 events/h (*P* < 0.001), LAST of 9.9% (*P* < 0.001), ODI of -28.6 events/h (*P* < 0.001), and ESS of -6.1 (*P* < 0.001) (Supplementary **Fig. S4**). The pooled rates of success and cure were 70.5% (95%Cl, 65.4% to 75.3%) and 17.4% (95%Cl, 7.1% to 31.0%), respectively.

The reduction in AHI after MMA was significantly higher than that after MLS (P < 0.001), and no difference was found in the improvement of LSAT, ODI, and ESS postoperatively between these two therapies. The pooled surgical cure rate of MMA was significantly higher than that of MLS (P = 0.020), while there was no difference in the surgical success rates between these two therapies.

Baseline AHI greater than 70 events/h

MMA group As shown in **Table 5**, four studies, totaling 76 patients with weighted preoperative AHI of 79.8 \pm 28.9 events/h, reported a significant improvement in AHI of -71.8 events/h (*P* < 0.001), LSAT of 18.7% (*P* < 0.001), and ESS of -7.9 (*P* < 0.001) (Supplementary **Fig. S3**). No study described ODI. The pooled rate of success was 84.2% (95%CI, 75.5% to 91.3%). One study reported a surgical cure rate of 46.2%.

Long-term follow-up outcomes

MMA group Four studies^{26, 27, 34, 42} reported long-term follow-up (\geq 2 years) in 98 OSA patients treated by MMA. At a mean follow-up of 8.9 years, a reduction of AHI was shown from 60.8 ± 25.2 to 13.1 ± 15.1 events/h. The meta-analysis showed a statistically significant improvement of -45.2 events/h (95%CI, -59.6 to -30.9, *P* < 0.001). Only one study with 40 patients presented long-term follow-up LSAT, reporting preoperative LSAT of 67.5 ± 14.8% and postoperative LSAT of 86.3 ± 3.9%. Surgical success rates were available for only two studies (90% and 41.4 %, respectively).

MLS group Three studies^{67, 77, 81} with 114 patients presented long-term follow-up $(\geq 2 \text{ years})$ data. In two of these studies, totaling 68 patients who had undergone uvulopalatopharyngoplasty (UPPP) and tongue base suspension with a mean follow-up of 2.8 years, AHI and ESS score decreased from 48.8 ± 17.8 events/h to 14.9 \pm 21.5 events/h, 12.1 \pm 4.3 to 7.5 \pm 5.9, respectively. The WMD between pre- and postsurgery were -27.4 events/h (95%Cl, -50.4 to -4.4, P = 0.020) and -4.5 (95%Cl, -6.2 to -2.8, P < 0.001), respectively. One of the two studies with 54 patients presented long-term follow-up LSAT increasing from 76.2 ± 12.4% preoperatively to 82.2 ± 11.2% postoperatively (P = 0.009). Another study with 14 patients reported longterm follow-up ODI from 30.3 ± 16.9 events/h preoperatively to 15.5 ± 13.2 events/h postoperatively (P < 0.001). Surgical success rates were 78% and 57.1%, respectively. In the third study consisting of 46 patients who had undergone uvulopalatal flap, genioglossus advancement, and hyoid suspension with a mean follow-up of 3.3 years, AHI and ESS score decreased from 47.9 ± 8.4 events/h to 18.6 ± 4.1 events/h, 15.9 ± 10.0 2.7 to 7.3 \pm 2.7, respectively; the LAST increased from 81.2 \pm 2.9% to 87.2 \pm 3.1%. The surgical success rate was 65.2%.

Surgical morbidity and mortality

MMA group The average length of hospitalization for OSA patients who underwent MMA was 3.5 days (range 2.3 days to 8 days). Among studies reporting participants' complications (n = 346)^{24, 26, 29, 35, 39-43}, no death was encountered. The rate of major complication was 3.2%, including ten re-operations for removal of osteosynthesis screws and plates (n = 8)^{26, 29, 42} and maxillary non-union (n = 2)^{24, 42}, and one sudden dyspnea⁴¹.The most frequent minor complication was facial paresthesia caused by the impairment of inferior alveolar nerve and/or maxillary nerve. In total, 76.9% of patients (n = 266) had transient facial paresthesia in mandibular and/or infraorbital areas, and 18.5% of patients (n = 64) reported persistent symptoms (mean follow-up of 6.0 years).
Excluding facial paresthesia, the rate of other minor complications was 10.1%, consisting of developed malocclusion (n = 13), temporomandibular disorders (n = 11), local infection (n = 5), minor postoperative wound pain (n = 2), unfavorable split (n = 1), loss of an interdental gingiva(n = 1), a perforation of the palate (n = 1), and transient unilateral angulus oris deviation (n = 1). Besides, only 9 of 206 patients perceived worsening of their facial appearance after MMA^{24, 26, 28, 37, 39-43}.

MLS group After surgery, patients required 4.1 days (range 1.25 ± 0.44 days to 16 ± 2 days) of hospitalization. No death was reported in 1386 patients^{22, 44-46, 48-51, 53, 56-60, 62, 63, 65-71, 73-75, 77-81}. The rate of major complications was 1.1%, including nine postoperative bleedings necessitating surgical exploration or surgical treatment^{51, 53, 64, 74}, five pillar extrusion requiring removal and replacement⁶⁰, and one pneumonia⁷⁸.

The minor complications included postoperative pain (n = 160), tongue discomfort (n = 74), velopharyngeal insufficiency (n = 70), dysphagia (n = 65), dysarthria (n = 25), odynophagia (n = 22), ulceration (n = 21), taste change (n = 14), and others (n = 112), which yield the minor complication rate of 40.6%. The majority of these complications were self-limited or could be cured by conservative treatment, with the exception of nine persistent complications: taste disturbance (n = 1)⁶⁴, dysphonia and dysphagia (n = 1)⁵³, oropharyngeal globus sensation (n = 2)⁴⁸, and dysphagia (n = 5)⁵¹.

Publication bias and sensitivity analysis

Both Begg's test and Egger's test suggested no significant publication bias for the included MMA and MLS studies (Supplementary **Fig. S6** and **S7**). The sensitivity analysis indicating high stability and robustness of the results (Supplementary **Fig. S8** and **S9**).

DISCUSSION

Respiratory parameters, and surgical success and cure

Although there are no comparative trials between MMA and MLS, greater improvement of OSA was found in MMA studies by pooling results from both surgical options, in terms of surgical success rate and improvement in the respiratory parameters. The observed superiority of MMA over MLS in treating OSA is explained by enlargement of the entire retropalatal and retrolingual airway by expanding the skeletal framework, while MLS cannot. Currently, there are a few studies^{25, 31, 40} reporting the significant Chapter 5

increases in pharyngeal airway volume (PAV) in OSA patients treated with MMA, by 60.5%, 35.7% and 35.4%, respectively. However, to our knowledge, only Chiffer et al.⁵⁴ quantitatively measured the volumetric changes in upper airway before and after MLS for treating OSA. They found a significant increase in PAV by 19.4%. Therefore, we inferred that the extent of the enlargement of the pharyngeal space could be associated with the therapeutic efficacy of upper airway surgery. Further investigation is essential to fully understand the treatment mechanisms of MMA and MLS, which may partly clarify the reason of differences in surgical outcome between them.

The discrepancy of surgical results between MMA and MLS varies with the different preoperative OSA severity. For example, there are benefits of MMA over MLS for the success rate in patients with baseline AHI < 40 events/h, and for the cure rate in patients with baseline AHI from 40 events/h to 70 events/h. The current evidence suggests that the pathophysiological causes of OSA are multifactorial and likely varies considerably between individuals, which puts an emphasis on personalized management for OSA based on its underlying causes⁵. Given the variable efficacy of these two types of surgeries, especially of MLS, careful selection of patients is needed. Therefore, one important objective in future research should be the identification of the factors that determine the success or failure in OSA patients treated by MMA or MLS. For the nonresponders to upper airway surgery, non-anatomical traits may play a prominent role as well in the etiology of OSA.

In MLS, precise identification of sites of airway collapse is imperative for favorable surgical outcome^{82, 83}, rather than only the severity of OSA. Among all the identified MLS studies, nasopharyngoscopy with Muller maneuver or DISE were performed preoperatively, except in four studies^{55, 59, 60, 78}. The significant improvement in OSA was noted in the three MLS subgroups with regard to surgical technique and the largest improvement in AHI was seen in subgroup 3. In one study⁵³, it was also demonstrated that compared with combined UPPP and tongue base radiofrequency ablation, combined hyoid suspension, UPPP, and tongue base radiofrequency ablation obtained better treatment outcome. However, due to the limited studies on subgroups 2 and 3, it is not possible to match each subgroup for baseline characteristics, which lead to the difficultly in comparing the clinical outcome between them in our study. Of interest is that in OSA surgery, palatal resection techniques such as UPPP are presently regarded as obsolete and are being replaced by modern reconstructive techniques, such as expansion sphincter pharyngoplasty,

because of better clinical outcome and less side effects⁸⁴. These better results are reported in both single level surgery and MLS^{49, 78}. In addition, upper airway stimulation⁸⁵, an emerging treatment option for moderate to severe OSA, has been found to be an effective therapy able to achieve success rate of 75% in patients with OSA⁸⁶. Interest in this emerging treatment modality has been increasing during the past decade. In the premise of precisely identifying anatomical abnormalities of the upper airway, the development of surgical techniques may further optimize the surgical outcome for well-selected patients with OSA. The comparison of clinical efficacy and safety between contemporary approaches and older ones for OSA is called for in future studies.

Subjective outcomes

Of note, not only the improvement in AHI but also the patients' subjective feeling should be taken into consideration when evaluating the efficacy of surgical interventions for OSA. Regrettably, ESS score was the only overlapping subjective index which was frequently reported in both MMA and MLS studies, leading to the impossibility of comprehensive comparison of other subjective outcomes (e.g., quality of life outcomes). There are studies that have assessed the improvement brought by MMA and MLS in patient's subjective feelings, such as snoring^{40, 58, 59, 87}, and bodily pain^{87, 88}. Both surgery modalities can significantly improve patient's subjective feeling. However, the comparison of improvement in quality of life between them should be addressed in future studies.

Long-term follow-up outcomes

The follow-up period of the included MMA studies ranges from 6 months to 12.5 years, and that of the included MLS studies ranges from 6 months to 3.3 years. Most of the retrieved studies reported short-term surgical outcomes at 6 months after surgery. In our study, a significant decrease in AHI of 45.23 events/h was shown, at a mean follow-up of 8.9 years after MMA. In a meta-analysis by Camacho et al.⁸⁹, it was demonstrated that OSA patients who were treated with MMA maintained improvements in AHI, sleepiness, and LSAT in the long term (4 years to < 8 years). However, the mean AHI increased to moderate OSA (mean AHI = 23.1 events/h) in the very long term (\geq 8 years). The longest follow-up result in MMA was reported by Pottel et al.⁸⁷, the long-term (range 14-20 years) success rate of nine patients performed MMA was 44.44%, and the short-term (within 2 years) success rate of MMA was 100% in young patients (age < 45) with BMI < 25kg/m², AHI < 45 events/h, SNB < 75°, narrow retrolingual space

(< 8 mm), and preoperative orthodontics. Marked weight gain and significant skeletal relapse can counterbalance the positive effect of MMA in the long-term, while there is no consensus on the effect of aging in long-term outcome of MMA^{34, 87}. Compared with the studies on MMA, currently, there are less studies on MLS evaluating the long-term surgical outcome. Hou et al.⁹⁰ performed combined midline glossectomy and UPPP in 34 patients and reported short-term (6 months) and long-term (5 years) outcome. At 6 months, the surgical success and cure rate were 79.41% and 17.65%, respectively; at 5 years, the surgical success and cure rate were 20.59% and 50%, respectively. The longest follow-up result of MLS was reported by Andsberg et al.⁹¹. In this study, 16 patients had undergone UPPP combined with midline glossectomy and followed up 1 year and 8.4 years after surgery. The success rates were 59% and 56%, respectively; and the cure rates were 32% and 25%, respectively. The weight of these patients did not change during the follow-up period, which may explain the long-term stable outcome. Neruntarat et al.⁶⁷ also found that patients with significant weight gain were at risk of recurrence of OSA. Based on the current literature, we concluded that the benefits of MMA and MLS persist for most patients with moderate-to-severe OSA over a long-term follow-up time. Marked weight gain after surgery and significant skeletal relapse after MMA may negatively influence the stability of clinical outcome. Thus, a recommendation regarding weight control and regular follow-up postoperatively are crucial for OSA patients. Moreover, due to the limited availability of data, the long-term outcome and the factors related to relapse require further investigation.

Surgical morbidity and mortality

Despite the apparent benefits, concerns about the safety and complications of surgical therapy for OSA still exist. In our study, both MMA and MLS were noted to be generally safe surgical therapies for OSA. Riley et al.⁹² concluded that OSA patients with apnea index higher than 70 events/h and LSAT less than 80% were at high risk of postoperative complication. Sensory disturbance in the territory of the inferior alveolar nerve was the most common complication of MMA, and the main predisposing factors were the degree of mandibular advancement, the patient's advanced age, and addition of a genioplasty⁹³. One study²⁴ demonstrated that the complication rate of MMA increased with increasing age, in particular after 45 years old. In a study of 487 consecutive OSA patients treated by MLS, Pang et al.⁹⁴ concluded that the overall complication rate was 7.1%, which is lower than our result. Besides, they pointed out that patients with severe OSA (AHI > 60 events/h and

LSAT < 80%) might be at high risk of postoperative oxygen desaturation. Although the major postoperative complication rate was low, patients who underwent MMA or MLS for treating OSA were recommended to be closely monitored after surgery^{95,} ⁹⁶. According to the available evidence, generally, more attention should be paid to the patients with highly severe OSA, who could be vulnerable to the postoperative complication, no matter after MMA or MLS.

Limitations

The results presented here should be considered in the context of several limitations. Firstly, the majority of the included studies are non-randomized studies, thus the level of evidence is limited inherently by the study design. Moreover, the overall quality of evidence was fair, with moderate risk of bias in the majority of studies included in the analysis, as evidenced by the Cochrane Collaboration "Risk of bias" tool and MINORS tool. However, unlike other medical areas, the randomized evaluations of surgical interventions are difficult to conduct. Secondly, there was high heterogeneity in most of the parameters pooled by meta-analysis, which may be attributed to a variety of potential confounding factors, i.e., patient characteristics, surgical techniques, follow-up time, and techniques of PSG scoring. Thirdly, only articles in English were included in our study, which may result in the language bias²⁰. Fourthly, since the comparison between MMA and MLS was clarified by separately pooling results from studies on these two types of surgery, it was not possible to quantify the differences in surgical outcomes between MMA and MLS for treatment of OSA. By the means of guasi-experimental studies or comparative cohort studies, the lack of comparative studies between MMA and MLS for treating OSA should be addressed in the future.

CONCLUSION

This systematic review and meta-analysis demonstrate that both MMA and MLS are effective treatment options for OSA with an acceptable rate of morbidity. However, regardless of disease severity, MMA may offer greater improvements in AHI compared to MLS, although this conclusion is based on separate analysis of MMA and MLS studies. The rates of major complication and minor complication of MMA are both higher than those of MLS.

Study	Design	N	Age (years) (mean <u>+</u>	% Male	Degree advanceme (mean <u>+</u>	Degree of advancement (mm) (mean <u>±</u> SD)		BA (mear	ИI 1±SD)
			SD)		Max	Mand	-	Pre-op	Post- op
Bettega et al. 2000	Retro	20	44.4± 10.6	90	11.8 <u>+</u> 0.5	11.8 <u>+</u> 0.5	6m	26.9 <u>+</u> 4.3	25.4 <u>+</u> 3.3
Bianchi et al. 2014	Retro	10	45±14	100	10	10	6m		
Boyd et al. 2015	Pro	14			7.0 <u>+</u> 2.3	9.2 <u>+</u> 3.3	6.6 <u>+</u> 2.8y		
Conradt et al. 1997	Retro	15	44 <u>+</u> 12	93.3			>2y	28.3 <u>+</u> 3.4	
Gerbino et al. 2014	Pro	10	44.9		9.2 <u>+</u> 1.2	10.4 <u>+</u> 2.2	6m	31.6 <u>+</u> 5.5	28 <u>+</u> 1.4
Goh et al. 2003	Pro	11	42.8 <u>+</u> 8.2	100	10	10	7.7m	29.4± 4.6	27.2 <u>+</u> 3.3
Goodday et al. 2016	Retro	13	37.8 <u>+</u> 8.6	84.6			9.6m	38.8 <u>+</u> 10.9	37.3± 8.0
Hsieh et al. 2014	Pro	16	33±7.9	75			12 <u>+</u> 8m	22.0 <u>+</u> 3.3	
Kastoer et al. 2019	Pro	14	51.1 <u>+</u> 7.3	57.1			6m	25.7± 3.7	
Li et al. 1999	Retro	175	43.5± 11.5	83			6m		
Li et al. 2000	Retro	40	45.6 <u>+</u> 20.7	82.5	10.8 <u>+</u> 2.7	10.8 <u>+</u> 2.7	4.2 <u>+</u> 2.7y	31.4± 6.7	32.2 <u>+</u> 6.3
Li et al. 2001	Retro	52	46.6 <u>+</u> 6.7	82.7	10.5 <u>+</u> 1.5		6m	32.0 <u>+</u> 6.0	
Li et al. 2002	Pro	12	47.3±9.8	75	10.5 <u>+</u> 1.2	10.5 <u>+</u> 1.2	6m	33.5± 6.2	32.3 <u>+</u> 4.1
Liao et al. 2015	Pro	20	33.4 <u>+</u> 6.5	85			14 <u>+</u> 9.3m	22.4 <u>+</u> 3.4	
Liu et al. 2016	Retro	20	44 <u>+</u> 12	85	7±1.4		6m	27 <u>+</u> 4.6	27.4 <u>+</u> 4.6
Rubio-Bueno et al. 2017	Pro	34	40.8 <u>+</u> 13.9	41.2	4.9± 3.2	10.4 <u>+</u> 3.9	6m	27.6 <u>+</u> 4.5	25.5 <u>+</u> 4.3
Veys et al. 2017	Pro	10	44.7±9.5	80	4.8 <u>+</u> 2.8	8.3 <u>+</u> 2.3	6m		
Vicini et al. 2010	RCT	25	49.1 <u>+</u> 9.1	92		11	1 <u>3±</u> 2.5m	32.7 <u>+</u> 5.8	31.4 <u>+</u> 6.5
Vigneron et al. 2017	Retro	29	40.7 <u>+</u> 12.6		8.4 <u>+</u> 4.1	11.7 <u>+</u> 5.1	12.5 <u>+</u> 3.5V	24.6 <u>+</u> 4	
Wu et al. 2019	Retro	28	37.2 <u>+</u> 11.8	53.6	2.0 <u>+</u> 3.1	8.8 <u>+</u> 3.7	>1y	24.2 <u>+</u> 5.1	

Table 1. Characteristics of studies on maxillomandibular advancement surgery

AHI, apnea-hypopnea index (events/h); BMI, body mass index (kg/m²); Day, days in hospital; ESS, Epworth sleepiness scale; LSAT, lowest oxygen saturation (%); m, months; Max, maxilla; Mand, mandible; N, number of patients; Post-op, postoperative; Pre-op, preoperative; Pro, prospective; RCT, randomized controlled trial; Retro, retrospective; y, years.

^a Respiratory disturbance index (RDI) in this study was extracted as AHI.

^b The number of patients was 9.

^c This study defined surgical success as an AHI < 15 events/h with \geq 50% reduction in postoperative AHI.

^d This study did not define the criteria of surgical success.

^e This study defined surgical success as a RDI < 15 events/h with \ge 50% reduction in postoperative RDI.

^f This study defined surgical success as a postoperative RDI < 20 events/h.

A (mea	HI n ±SD)	LS (mea	SAT n <u>+</u> SD)	O (mear	DI 1 <u>±</u> SD)	ESS (mean <u>+</u> SD)		% Success	% Cure	Day
Pre-op	Post- op	Pre-op	Post- op	Pre-op	Post-op	Pre-op	Post- op			
59.3 <u>+</u> 29.0	11.1 <u>+</u> 8.9	82 <u>+</u> 11	90 <u>+</u> 7					75 ^c		7
56.8 <u>+</u>	12.3 <u>+</u>									
5.2	5.5									
50.0 <u>+</u>	8.0 <u>+</u>									2.3
20.0	10.7									
51.4±	8.5±									
16.9	9.4									
69.8 <u>+</u>	17.3±			59.5±	9.1 <u>+</u>			80 ^a		
35.2	16.7			5.3	8.0					
70.7±	11.4±	58.6 <u>+</u>	83.9 <u>+</u> 8.8					81.8		4.2
15.9	7.4	12.3								
117.9 <u>+</u>	16.1 <u>+</u>					12.9 <u>+</u>	5.0 <u>+</u>	76.9	46.2	
9.2	26.2					5.5	4.1 ^b			
35.7 <u>+</u> 18	4.8 <u>+</u>							100		
	4.4									
40.2 <u>+</u>	9.9±			13.5 <u>+</u>	4.0 <u>+</u>	13 <u>+</u> 6	9±7			
25.6	7.2			8.6	3.5					
72.3±	7.2 <u>+</u>	63.2 <u>+</u>	86.6 <u>+</u> 3.4					95°		2.4
26.7ª	7.5ª	17.5								
71.2 <u>+</u>	7.6 <u>+</u>	67.5±	86.3 <u>+</u> 3.9					90 ^e		2.4
27.0ª	5.1ª	14.8						f		
61.6 <u>+</u>	9.2 <u>+</u> 8ª	75.9±	87.5±4.7					90'		
23.9ª		10.6						- 6		
75.3± 26.4ª	10.4 <u>+</u> 10.8 ^a	74.2 <u>+</u> 12	86.9 <u>+</u> 6.7					83.3'		
41.6 <u>+</u>	5.3±4	80.2 <u>+</u>	88.9±5			11.9 <u>+</u>	7±3	100 ^c		
19.2		9.7				7.3				
53.6 <u>+</u>	9.5±	80.9 <u>+</u>	94.1±3.5	38.7±	8.1 <u>+</u>	17 <u>+</u> 4.8	5.7±	90	50	
26.6	7.4	8.9		30.3	9.2		2.7			
38.3 <u>+</u>	6.5±			34.7±	5.4±	17.4±	0.8 <u>+</u>	100	52.9	<2
10.7	4.3			12.5	4.1	5.4	1.4			
26.8 <u>+</u>	12.3±					14.1 <u>+</u>	5.7±	70	40	
12.7	14.4					5.9	3.0			
56.8 <u>+</u>	8.1 <u>+</u> 7					11.6 <u>+</u>	7.7±	88	36	<7
16.5						2.8	1.3			
56.6 <u>+</u> 24	25.5±		83.1 <u>+</u> 5.8				7.5±	41.4		5-8
	20.6						4.7			
59.3±	10.9 <u>+</u>	73.4±	87.9 <u>+</u> 3.7			12.8 <u>+</u>	6.9 <u>+</u>	85.7	46.4	
14.5	3.3	10.8				2.8	2.5			

Study	Design	N	Age (years)	%	Follow-up	BMI (m	BMI (mean ±SD)		AHI (mean±SD)	
			(mean ± SD)	Male	(mean <u>+</u> SD)	Pre-op	Post- op	Pre-op	Post- op	
Subgroup 1. Sof	t palate lev	vel & to	ongue base leve	el						
Aynaci et al.	Retro	20	41.7±	85	6m			25.1 <u>+</u>	13.40 <u>+</u>	
2018			8.4					6.0	3.0	
		20	45.0±	80	6m			36.4±	10.0 <u>+</u>	
			7.1					4.9	1.9	
Babademez et	Retro	16	41.3±	100	6m	29.6 <u>+</u>	29.5±	20.1 <u>+</u>	8.9 <u>+</u> 6.5	
al. 2010	_		10.5			2.5	2.6	10.5		
Bostanci et al.	Retro	82	50.5±	92.7	6m	30.6 <u>+</u>		47.3±	19.9±	
2016			9.2			3.0		18.7	17.4	
Cambi et al.	Retro	20	55.6±	85	6m	30.1 <u>+</u>	28.9 <u>+</u>	49.3±	19.4 <u>+</u>	
2019 Commente et	Detre	10	9.1		. (2.3	2.4	18.5	10.1	
al 2017	Retro	10	58.4±		2011	26.8 <u>+</u>		34.0 <u>+</u>	22.9 <u>+</u>	
al. 2017		10	9.9		>6m	3./ 27.0±		25.64	0.6±0.2	
		10	11 /		20111	27.0 <u>+</u> 21		12.9	9.0-9.5	
		10	48.2+		>6m	28.8+		37.8+	13.5+	
			11.4		_0	2.6		21.6	7.8	
Ceylan et al.	Pro	26	46.3+	88.5	17	28.6+		29.6+	16.1±	
2009			3.9		-	3.8		7.8	3.9	
Chen et al.	Pro	22	40.5±	90.9	6m	29.1±	28.9±	66.4±	35.1±	
2019			6.8			3.5	3.6	17.0	18.5	
Chen et al.	Pro	24	42.3±	100	1у	27.5±		46.1 <u>+</u>	26.2 <u>+</u>	
2014			8.3			2.7		13.3	18.9	
		26	43±9.4	100	1у	26.6 <u>+</u>		51.8 <u>+</u>	25.2 <u>+</u>	
						2.4		14.7	7.9	
Chen et al.	RCT	45			6m			49.7±	27.0 <u>+</u>	
2018-group 2								7.4	4.0	
Chiffer et al.	Pro	18		83.3	6-24m	34.2 <u>+</u>	32.2 <u>+</u>	53.9±	19.8 <u>+</u>	
2015	Dre				(192	6.9	7.2	25.4	22.1	
Emara et al.	Pro	23			6111	27.5±		40./±	15.4 <u>±</u>	
Funetal 2008	Pro	66	44.7+	87.0	6m	27.6+	27.4+	22 0+	12.0+	
Lanctal. 2000	110	00	10.6	07.9	om	3.4	3.2	14 7 ^a	18.7 ^a	
Friedman et al.	Retro	143	47.0+	72.7	>6m	31.5+	5	43.9+	28.1+	
2003		15	11.7	7=-7		4.8		23.7	20.6	
Friedman et al.	Retro	122	42.2±	65.6	12.2 <u>+</u>	28.3±		23.2±	14.5±	
2007			11.4		4.2m	5.0		7.6	10.2	
Gunbey et al.	Pro	42	47.1 <u>+</u>	69	6m	32.6±	31.2 <u>+</u>	35.8 <u>+</u>	15.3 <u>+</u>	
2015			14.5			8.4	9.1	12.1	9.8	
Hendler et al.	Retro	33	47±10.5	84.8	6m	32.6 <u>+</u>		60.2 <u>+</u>	28.8 <u>+</u>	
2001	-					7.0		29.9 ^D	27.4 ^D	
Li et al. 2016	Retro	30	41.5±	90	6-8m	26.4 <u>+</u>	25.5±	48.4±	16.5 <u>+</u>	
Listal and	Detwo		9.4	0.0	6.0	3.0	3.0	16.9	11.2	
LI et al. 2016	Retro	25	42±9	80	6-8M	26.5 <u>+</u>	25.6 <u>+</u>	45./±	12.8 <u>+</u>	
listal 2012	Patro	45	40.2+	100	6m	3.0	2.9	21./	0.Z 8 0+F 0	
LI EL al. 2013	Relio	45	40. <u>3±</u>	100	0111	2/./± 26	27.4±	39.4±	0.9 <u>±</u> 5.9	
linetal 2010	Retro	13	39	95.3	6m	27.9+	28 O+	51 5+	23 /+	
2	Recio	υ	<i></i>	20.0	UIII	3,9	3.9	25.4	24.7	
Neruntarat et	Pro	72	35.8+	95.8	14.2+	28.8+	30.9+	35.6+	16.8+	
al. 2009		. –	10.9	, ,	1.8m	2.4	2.8	9.2	3.2	
Omur et al.	Retro	22	44.5±		14.0±	30.3 <u>+</u>	29.2 <u>+</u>	47.5±	17.3±	
2005			8.0		6.7m	3.8	3.3	15.7 ^b	14.2 ^b	

Table 2. Characteristics of studies on multilevel surgery

LSAT (m	ean <u>+</u> SD)	ODI (m	ean <u>+</u> SD)	ESS (me	ean <u>+</u> SD)	%	%	Day
Pre-op	Post-op	Pre-op	Post-op	Pre-op	Post-op	Success	Cure	(mean <u>+</u> SD)
80.3 <u>+</u>	91.9 <u>+</u>			19.8 <u>+</u>	11.1 <u>+</u>			1.3±0.4
6.0	1.7			2.5	1.5			
78.8 <u>+</u>	96.3 <u>+</u>			20.1 <u>+</u>	6.5 <u>+</u> 1.3			2.6 <u>+</u> 0.8
3.5	1.2 86.6+			1.7		62 F		
04.0 <u>+</u> 3.4	2.0					02.5		
75.7±	82.3±	44.8+	17.7±			74.4		
8.9	7.4	21.4	15.9					
69.5 <u>+</u>	80.0 <u>+</u>			12.7 <u>+</u>	7.7±4.5	60		5.2 <u>+</u> 0.9
9.9	7.4			4.3				
				12.3 <u>+</u>	8.5 <u>±</u> 5.4	50		6.7 <u>±</u> 1.3
				4.Z	4 0+2 0	00		7 1±1 E
				4.5	4.9_3.9	90		7.1-1.5
				10.4±	3.9 <u>+</u> 3.6	90		7.1 <u>+</u> 3.2
				2.5				
86.8 <u>+</u>	94.6±			10.8 <u>+</u>	8.2 <u>+</u> 2.7	53.8 ^c		
8.9	4.9			3.2				
61.9 <u>+</u>	67.8 <u>+</u>					63.6 ^u		
12.5	19.3							
60.3 <u>+</u>	76.9 <u>+</u>			13.0 <u>+</u>	8.5 <u>+</u> 2.0	64.4	11.1	
7.3	4.0			2.6		61		
						61		
78.9±	87.2±			14.2±	8.3+3.9	86.9		
12.6	11.1			2.3	0_07			
79.1 <u>+</u>	79.4±			11.4±	7.5±4.5	53.6	50	2
5.7 ^a	16.5 ^a			5.0				
81.4 <u>+</u>	85.9 <u>+</u>			15.2 <u>+</u>	8.3 <u>+</u> 3.9			
10.4	9.8			3.1	6 0+2 2	47.5		
00.9 <u>+</u> 4.8	90.4± 4.3			9.7±3.9	0.9 <u>±</u> 3.3	47.5		
,								
72.4±	80.4±							
15.2	12.3			10.01	0 7			
/6.4± 8 г	82.4 <u>+</u>			10.9±	8./±3.9	/3		
0.5 77.1+	5.4 83.3+			4.7 9.6+4.9	7.5+4.3	80		5.6+1.3
10.5	5.6			J.0 <u>-</u> J	/·J≟∓·J			5.05
66 <u>+</u> 16	83±5			12.9±	3.4±2.9	51.1	37.8	7
				4.9				
75.5±	82.1 <u>+</u>			12.8 <u>+</u>	10.0 <u>+</u>	60.5		
10.4	10.9			5.1	4.3	(
85.6± 2 4	88.2 <u>+</u>			14.2±	8.2 <u>+</u> 2.5	55.6		1
5.4	2.4			5.4 13.9+	5.4+4.3	81.8 ^e		3.8+1.6
				2.2	5.121.5			5

Study	Design	N	Age (years)	%	Follow-up	BMI (mean ±SD)		AHI (mean <u>+</u> SD)	
			(mean <u>+</u> SD)	Male	(mean <u>+</u> SD)	Pre-op	Post- op	Pre-op	Post- op
Plzak et al.	Retro	79	50.5±	78.5	6m	28.1 <u>+</u>	28.3±	28.7±	14.1 <u>+</u>
2013			9.1			3.1	3.5	17.1	18.2
Sezen et al.	Pro	12	48.3±	83.3	1у	30.9 <u>+</u>	30.6 <u>+</u> 2.7	28.8 <u>+</u>	15.3±
2011			8.8			2.8		10.7	11.1
Toh et al., 2014	Retro	20	47.1 <u>±</u>	80	8.2 <u>+</u>	26.9 <u>+</u>	26.2 <u>+</u>	41.3±	13.5 <u>+</u>
			11.4		3.2m	2.9	3.0	22.1	17.1
Tsou et al. 2018	Retro	36	40.2 <u>+</u>	88.9	1у	26.9 <u>+</u>	26.1 <u>+</u>	25.1±	17.5±
			9.1			2.9	2.9	17.5	18.9
Turhan et al.	Pro	90	48	91.1	6m	30.7		51.8 <u>+</u> 18.8	20.5 <u>+</u> 17.7
Vicente et al	Pro	54	47.2+	02.6	21/	20.6+	28.1+	52 8+	1/./
2006	110	54	47.3	92.0	Sy	1 8	<u> 1 8</u>	1/ 9	23.5
Vicini et al	Retro	12	49.6+	100	>6m	28.2+	27.0+	38.4+	19.8+
2014	Retro	12	11 3	100	20111	2 7	21	19.7	14 1
2014		12	54.2+	75	>6m	27.3+	26.1+	38 5+	9 9+8 6
			10.8	75	_0	2.0	2.0	14.3	<u></u>
Wang et al.	Retro	36	44	86.1	1v	29.2+	28.9+	59.8+	23.2+
2013		5-			.,	2.9	2.8	20.5	18.4
Yuksel et al.	Pro	14	41.4+	92.9	2V	30.8+		33.2+	18.0+
2016			8.9	,,	_,	3.7		18.9	11.3
Subgroup 2. Sof	t palate le	vel & h	yoid level						-
Benazzo et al.	Retro	109	51.3±	100	6m	28.2 <u>+</u>	27.7±	37.0±	18.7±
2008			9.4		_	3.1	2.9	19.1	16.0
El-Anwar et al.	Pro	20	47.1±		6-14m	33.4±		48.8 <u>+</u>	24.5±
2018			9.2			2.5		31.6	10.9
Tantawy et al.	Pro	32	46 <u>+</u> 4.7	43.8	6-14m	33.4±		68.4±	25.6 <u>+</u>
2018						2.0		25.3	9.5
Subgroup 3. Sof	t palate le	vel & to	ongue base leve	el & hyoi	d level				
Chen et al.	RCT	45			6m			52.3+	14.9±
2018-group 1								6.3	2.2
Cillo et al. 2013	Retro	13	43.0±	100	18 <u>+</u>			28.3±	12.1±
			2.4		3.6m			13.2	8.2
Neruntarat et	Retro	46	40.1±	82.6	3.3±	28.9±	31.1±	47.9±	18.6 <u>+</u>
al. 2003			4.2		0.5y	2.1	2.7	8.4 ^b	4.1 ^b
Sorrenti et al.	Retro	10	51.7±7	100	14.6m	31.0±	28.5±	54.7±	9.4±5.4
2006						2.5	2.4	11.5	
Sun et al. 2008	Pro	31	41±9.8	100	6m	28.5 <u>+</u>	28.4 <u>+</u>	65.9 <u>+</u>	28.6 <u>+</u>
						3.2	3.6	23.8	29.1
Yi et al. 2011	Pro	26	47	84.6	6m	29.3	28.0	65.6 <u>+</u>	30.1 <u>+</u>
								17.6	23.1

Table 2. continued

AHI, apnea-hypopnea index (events/h); BMI, body mass index (kg/m²); Day, days in hospital; ESS, Epworth sleepiness scale; LSAT, lowest oxygen saturation (%); m, months; N, number of patients; Post-op, postoperative; Pre-op, preoperative; Pro, prospective; RCT, randomized controlled trial; Retro, retrospective; y, years.

^a The number of patients was 58.

^b Respiratory disturbance index (RDI) in this study was extracted as AHI.

 $^{\rm c}$ This study defined surgical success as an AHI < 20 events/h with \geq 50% reduction in postoperative AHI and a postoperative ESS score < 10.

^d This study defined surgical success as \geq 50% reduction in postoperative AHI.

^e This study defined surgical success as a RDI < 20 events/h with \geq 50% reduction in postoperative RDI.

^f This study defined surgical success as an AHI < 15 events/h with \geq 50% reduction in postoperative AHI.

 $^{\rm g}$ This study defined surgical success as an AHI < 20 events/h with \geq 50% reduction in postoperative AHI and a postoperative ESS score < 11.

^h This study defined surgical success as an AHI < 20 events/h with significant clinical improvement reported by patients.

LSAT (m	ean <u>+</u> SD)	ODI (m	ean <u>+</u> SD)	D) ESS (mean <u>+</u> SD)		%	%	Day
Pre-op	Post-op	Pre-op	Post-op	Pre-op	Post-op	Success	Cure	(mean <u>+</u> SD)
		15.1 <u>+</u> 8.2	10.3 <u>+</u> 7.9	10.6 <u>+</u> 3.8	7.3 <u>+</u> 3.2	51.7		3
				14.8 <u>+</u> 2.5	7.6 <u>+</u> 3.2	50 ^f		
72.9 <u>+</u> 19.3	84.5± 7.1			13.0 <u>+</u> 2.8	5.6 <u>+</u> 4.4	55	35	4.1 <u>+</u> 0.7
				11.9 <u>+</u> 4.3	10.2 <u>+</u> 4.3	66.7		
75.6 <u>+</u> 9.3	82.4 <u>+</u> 6.6	48.0 <u>+</u> 19.5	18.2 <u>+</u> 15.5			74.4		
76.2 <u>+</u> 12.4	82.2 <u>+</u> 11.2			12.2 <u>+</u> 3.3	8.2 <u>+</u> 6.1	78 ^g		
				13.75±4	7.6 <u>+</u> 4.4	33.3		8.3±2.4
				12 <u>+</u> 4.9	4.4 <u>+</u> 4.1	83.3		7.3±1.5
70.5 <u>+</u> 12.4	85.6 <u>+</u> 10.0			12.2 <u>+</u> 5.8	5.5 <u>+</u> 3.6	66.7		
		30.3 <u>+</u> 16.9	15.5 <u>+</u> 13.2	11.9 <u>+</u> 7.0	5.0 <u>+</u> 4.4	57.1		
				10.5 <u>+</u> 3.1	7.2 <u>+</u> 2.3	61.5		
73.5± 14.8	84 <u>±</u> 5.3			12.6 <u>+</u> 5.6	4.1 <u>±</u> 2.7			
66.8 <u>+</u> 11.3	83.2 <u>+</u> 2.9			13.8 <u>+</u> 5.4	5.2 <u>+</u> 1.6			
-0 -1	86.01			12.9	60112	84.4	11 1	
58.7± 8.3	5.4			12.8 <u>+</u> 2.2	0.0±1.3	04.4	11.1	
				15.2 <u>+</u> 3.0	6.3 <u>+</u> 3.9			
81.2 <u>+</u> 2.9	87.2 <u>+</u> 3.1			15.9 <u>+</u> 2.7	7.3 <u>+</u> 2.7	65.2 ^e		
77 <u>+</u> 6.2	90.7±3			14.3	5.3	100		16 <u>+</u> 2
72.7 <u>+</u> 11.9	75.0 <u>+</u> 12.5			17.1 <u>+</u> 4.1	8.9 <u>+</u> 4.9	64.5 ^h		
74 <u>+</u> 28	82.8			13.5 <u>+</u> 5.9	6.8 <u>+</u> 5.2	46.2		

Variable		I	Pre-op	P	ost-op		Change		P ^b
		Ν	Weighted mean <u>+</u> SD	Ν	Weighted mean <u>+</u> SD	WMD	95% CI	P ^a	
Age, years	MMA	504	42.9 ± 11.3						
	MLS	1313	45.5 <u>+</u> 10.8						
BMI, kg/m²	MMA	359	28.6 <u>+</u> 6.6	185	29.4 <u>+</u> 6.2				
	MLS	1420	29.1 <u>+</u> 4.2	878	28.4 ± 4.1				
AHI, events/h	MMA	393	57.3 ± 26.6	393	10.4 ± 11.2	-46.2	[-52.4, -39.9]	<0.001	<0.001
	MLS	1639	42.2 <u>+</u> 21.0	1639	19.0 <u>+</u> 16.4	-24.7	[-28.1, -21.4]	<0.001	
LSAT, %	MMA	203	74.4 ± 12.9	203	88.1 <u>+</u> 5.5	13.5	[10.5, 16.5]	<0.001	0.014
	MLS	1164	76.7 <u>+</u> 12.5	1164	84.2 ± 9.5	8.7	[6.2, 11.1]	<0.001	
ODI, events/h	MMA	78	35.1 <u>+</u> 22.8	78	6.3 <u>+</u> 6.4	-30.3	[-46.3, -14.2]	<0.001	0.322
	MLS	265	36.3 <u>+</u> 22.5	265	15.5±14.1	-19.1	[-34.2, -4.0]	0.010	
ESS	MMA	164	14.1 ± 5.4	164	4.8 ± 4.1	-8.5	[-12.2, -4.9]	<0.001	0.143
	MLS	1309	12.6 ± 4.4	1309	7.3 ± 3.9	-5.8	[-6.6, -5.0]	<0.001	
Success rate, %	MMA	340				85.0	[76.4, 91.9]	<0.001	<0.001
	MLS	1339				65.1	[60.6, 69.5]	<0.001	
Cure rate, %	MMA	130				46.3	[38.0, 54.7]	<0.001	0.135
	MLS	221				28.1	[13.2, 46.1]	<0.001	

 Table 3. Summary of weighted data for studies on maxillomandibular advancement surgery and multilevel surgery

AHI, apnea-hypopnea index; BMI, body mass index; CI, confidence interval; ESS, Epwoth sleepiness scale; LSAT, lowest oxygen saturation; MLS, multilevel surgery; MMA, maxillomandibular advancement; N, number of patients; Post-op, postoperative; Pre-op, preoperative; SD, standard deviation; WMD, weighted mean difference.

^a Z-test for overall effect size.

^b Z-test for comparison the difference between two estimates.

Variable	Pre-op		I	Post-op	Change			
	Ν	Weighted mean <u>+</u> SD	Ν	Weighted mean <u>+</u> SD	WMD	95% CI	P ^a	
Subgroup 1. Soft palate le	evel & t	ongue base lev	/el					
Age, years	1052	45.2 ± 11.2						
BMI, kg/m²	1172	29.0 ± 4.3	682	28.4 ± 4.4				
AHI, events/h	1307	40.4 ± 20.3	1307	18.7 <u>+</u> 16.6	-22.7	[-25.7, -19.7]	<0.001	
LSAT, %	980	77.9 <u>+</u> 12.1	980	84.2 <u>+</u> 9.8	7.2	[5.0, 9.3]	<0.001	
ODI, events/h	265	36.3 <u>+</u> 22.5	265	15.5 <u>+</u> 14.1	-19.1	[-34.2, -4.0]	0.010	
ESS	987	12.4 ± 4.3	987	7.5 ± 4.1	-5.2	[-6.1, -4.4]	<0.001	
Success rate, %	1072				64.2	[59.3, 68.9]	<0.001	
Cure rate, %	176				33.0	[16.1, 52.5]	<0.001	
Subgroup 2. Soft palate le	evel & l	nyoid level						
Age, years	161	49.7±8.9						
BMI, kg/m²	161	29.9±3.7	109	27.7 ± 2.9				
AHI, events/h	161	44.7±25.4	161	20.8 <u>+</u> 14.6	-28.4	[-45.2, -11.5]	0.001	
LSAT, %	52	69.4±13.0	52	83.5±3.9	14.1	[8.5, 19.8]	<0.001	
ODI, events/h								
ESS	161	11.4 ± 4.2	161	6.4 <u>+</u> 2.5	-6.7	[-10.8, -2.5]	0.002	
Success rate, %	109				61.5			
Cure rate, %								
Subgroup 3. Soft palate le	evel & t	ongue base le	vel & hy	oid level				
Age, years	100	41.9 ± 7.3						
BMI, kg/m²	87	29.0 ± 2.7	87	29.8 <u>+</u> 3.3				
AHI, events/h	171	54.0±17.4	171	20.1 <u>+</u> 17.0	-33.4	[-39.7, -27.1]	<0.001	
LSAT, %	132	71.2 ± 12.4	132	84.2 <u>+</u> 8.8	12.4	[0.6, 24.3]	0.040	
ODI, events/h								
ESS	161	14.8 ± 3.9	161	7.1 ± 3.7	-7.8	[-8.9, -6.7]	<0.001	
Success rate, %	158				72.4	[55.3, 86.7]	<0.001	
Cure rate, %	45				11.1			

 Table 4. Summary of weighed data for studies on multilevel surgery – three subgroups according to the different target levels of obstructive sites addressed by surgery

AHI, apnea-hypopnea index; BMI, body mass index; CI, confidence interval; ESS, Epworth sleepiness scale; LSAT, lowest oxygen saturation; N, number of patients; Post-op, postoperative; Pre-op, preoperative; SD, standard deviation; WMD, weighted mean difference.

^a Z-test for overall effect size.

Variable			Pre-op		Post-op		Change		Pb
		N	Weighted mean <u>+</u> SD	N	Weighted mean <u>+</u> SD	WMD	95% CI	Pa	_
Baseline AHI le	ss than 4	lo ever	nts/h						
Age, years	MMA	60	39.4 ± 12.4						
	MLS	706	45.2 <u>+</u> 11.7						
BMI, kg/m²	MMA	50	25.8 ± 4.9	34	25.5 ± 4.3				
	MLS	693	28.5 ± 4.2	501	28.4 ± 4.2				
AHI, events/h	MMA	60	35.7±13.7	60	7.0 ± 7.3	-27.1	[-36.0, -18.2]	<0.001	0.030
	MLS	706	30.7 ± 15.6	706	15.1 ± 13.3	-16.7	[-19.9, -13.4]	<0.001	
LSAT, %	MMA								
	MLS	347	83.0 <u>+</u> 10.5	347	87.0±9.3	4.4	[1.9, 6.8]	0.001	
ODI, events/h	MMA	34	34.7 ± 12.5	34	5.4 ± 4.1	-29.3	[-33.7, -24.9]	<0.001	
	MLS	93	17.4 ± 11.3	93	11.1 <u>+</u> 9.0	-8.2	[-17.6, 1.1]	0.080	
ESS	MMA	44	16.7 <u>+</u> 5.6	44	1.9 <u>+</u> 2.8	-12.7	[-20.8, -4.7]	0.002	0.076
	MLS	648	11.5 ± 4.7	648	7.1 <u>+</u> 3.6	-5.4	[-6.6, -4.2]	<0.001	
Success rate, %	MMA	60				94.0	[74.3, 99.9]	<0.001	<0.001
	MLS	651				57.1	[51.7, 62.5]	<0.001	
Cure rate, %	MMA	44				50.0	[35.7, 64.2]	<0.001	0.579
	MLS	111				44.7	[33.2, 56.4]	<0.001	
Baseline AHI fre	om 40 to	70 eve	ents/h						
Age, years	MMA	215	44.3±10.6						
	MLS	607	45.8 ± 9.7						
BMI, kg/m²	MMA	233	27.9 <u>+</u> 6.0	75	28.3 <u>+</u> 5.3				
	MLS	727	29.7 <u>±</u> 4.1	377	28.4 <u>+</u> 4.1				
AHI, events/h	MMA	257	55.7 <u>+</u> 23.0	257	11.4 ± 11.4	-44.1	[-47.8, -40.4]	<0.001	<0.001
	MLS	933	51.0 <u>+</u> 20.3	933	22.0 <u>+</u> 17.9	-30.7	[-34.0, -27.5]	<0.001	
LSAT, %	MMA	140	77.6 <u>+</u> 10.7	140	89.1 <u>+</u> 5.2	11.6	[9.4, 13.8]	<0.001	0.387
	MLS	817	74.1 ± 12.3	817	82.9 ± 9.4	9.9	[6.9, 13.0]	<0.001	
ODI, events/h	MMA	44	35.4 ± 28.5	44	7.0 <u>+</u> 7.7	-30.4	[-57.6, -3.1]	0.030	0.900
	MLS	172	46.5 <u>+</u> 20.4	172	18.0 <u>+</u> 15.6	-28.6	[-32.4, -24.8]	<0.001	
ESS	MMA	107	13.2 <u>+</u> 5.1	107	6.0 <u>+</u> 4.0	-7.0	[-10.7, -3.4]	<0.001	0.633
	MLS	661	13.6 <u>+</u> 4.2	661	7.5 ± 4.1	-6.1	[-7.1, -5.2]	<0.001	
Success rate, %	MMA	204				82.3	[69.1, 92.5]	<0.001	0.061
	MLS	688				70.5	[65.4, 75.3]	<0.001	
Cure rate, %	MMA	73				44.0	[33.1, 55.3]	<0.001	0.020
	MLS	110				17.4	[7.1, 31.0]	<0.001	

Table 5. Summary of weighted data for studies on maxillomandibular advancement surgery and multilevel surgery in OSA patients with baseline AHI less than 40, from 40 to 70, and greater than 70 events/h

Variable			Pre-op		Post-op		Change		Pb
		Ν	Weighted mean <u>+</u> SD	Ν	Weighted mean <u>+</u> SD	WMD	95% CI	P ^a	
Baseline AHI gr	eater tha	ın 70 e	vents/h						
Age, years	MMA	76	44.1 <u>+</u> 16.4						
BMI, kg/m²	MMA	76	32.7±7.7	76	32.4 <u>+</u> 6.6				
AHI, events/h	MMA	76	79.8 <u>+</u> 28.9	76	10.0 <u>+</u> 12.6	-71.8	[-88.4, -55.2]	<0.001	
LSAT, %	MMA	63	67.2 <u>+</u> 14.5	63	86.0 <u>+</u> 5.6	18.7	[12.7, 24.6]	<0.001	
ODI, events/h									
ESS	MMA	13	12.9 <u>+</u> 5.5	13	5.0 ± 4.1	-7.9	[-11.6, -4.2]	<0.001	
Success rate, %	MMA	76				84.2	[75.5, 91.3]	<0.001	
Cure rate, %	MMA	13				46.2			

Table 5. continued

AHI, apnea-hypopnea index; BMI, body mass index; CI, confidence interval; ESS, Epworth sleepiness scale; LSAT, lowest oxygen saturation; MLS, multilevel surgery; MMA, maxillomandibular advancement; N, number of patients; Post-op, postoperative; Pre-op, preoperative; SD, standard deviation; WMD, weighted mean difference.

^a Z-test for overall effect size.

 $^{\rm b}$ Z-test for comparison the difference between two estimates.

SUPPLEMENTARY MATERIAL

	Ovid MEDLINE(R) ALL <1946 to May 6, 2020>	
Step	Search	Result
1	exp Sleep Apnea Syndromes/ or Snoring/	34066
2	((sleep adj3 (apnea or apnoea or hypopnea or hypopnoea)) or (upper adj airway adj resistance) or (sleep adj disordered adj breathing) or snore or snoring).ti,ab,kf,ot.	39581
3	1 or 2	45774
4	(uvulopalatopharyngoplasty or H-UPPP or HUPPP or UPPP or palatopharyngealplasty or uvulopalatoplasty or uvuloplasty or uvuloflap or uvulopalatal-flap or Z-palatoplasty or palatoplasty or (ablation adj2 palate) or palatal-stiffening or pharyngoplasty or tonsillectomy or ((pillar or palatal) adj implant) or "palatal pillar" or ((uvula or palat* or pharynx or pharyngeal) adj3 (remove or removal or ablation or surgery or surgical or remodel* or resection))). ti,ab,kw.	12813
5	(((midline glossectomy or "genioglossus advancement" or (hypoglossal adj nerve- stimulation) or (transoral adj2 surger*) or (hypogloss* or epiglott* or tongue)) adj5 (surgery or surgical or remove or removal or remodel* or resection or reduction or suspension or coblation or ablation)) or "tongue stabilization" or tonsillectomy or epiglottidectomy or epiglottoplasty or hyoepiglottoplasty).ti,ab,kf.	11418
6	(((hyoid or thyrohyoid) adj (suspension or myotomy or advancement)) or hyoidopexy).ti,ab,kf.	135
7	4 and 5	8555
8	4 and 6	74
9	5 and 6	63
10	7 or 8 or 9	8594
11	(mma or maxillomandibular advancement or bimaxillary surgery or maxillary osteotomy or mandibular advancement or orthognathic surgery).ti,ab,kf.	6451
12	(multilevel or multi-level).ti,ab,kf.	31482
13	10 or 11 or 12	46384
14	3 and 13	2302
15	(case reports or review).pt.	4410761
16	exp animals/ not humans/	4582720
17	15 or 16	8814783
18	14 not 17	1737

Table S1a. Search strategy in MEDLINE database

Table S1b. Search strategy in EMBASE database

	Embase Classic+Embase <1947 to May 6, 2020>	
Step	Search	Result
1	(uvulopalatopharyngoplasty or H-UPPP or HUPPP or UPPP or palatopharyngealplasty or uvulopalatoplasty or uvuloplasty or uvuloflap or uvulopalatal-flap or Z-palatoplasty or palatoplasty or (ablation adj2 palate) or palatal-stiffening or pharyngoplasty or tonsillectomy or ((pillar or palatal) adj implant) or "palatal pillar" or ((uvula or palat* or pharynx or pharyngeal) adj3 (remove or removal or ablation or surgery or surgical or remodel* or resection))).ti,ab,kw.	17201
2	(((midline glossectomy or "genioglossus advancement" or (hypoglossal adj nerve- stimulation) or (transoral adj2 surger*) or (hypogloss* or epiglott* or tongue)) adj5 (surgery or surgical or remove or removal or remodel* or resection or reduction or suspension or coblation or ablation)) or "tongue stabilization" or tonsillectomy or epiglottidectomy or epiglottoplasty or hyoepiglottoplasty).ti,ab,kw.	15411
3	(((hyoid or thyrohyoid) adj (suspension or myotomy or advancement)) or hyoidopexy). ti,ab,kw.	174
4	1 and 2	11764
5	1 and 3	90
6	2 and 3	79
7	4 or 5 or 6	11815
8	(mma or maxillomandibular advancement or bimaxillary surgery or maxillary osteotomy or mandibular advancement or orthognathic surgery).ti,ab,kw.	8418
9	(multilevel or multi-level).ti,ab,kw.	36392
10	7 or 8 or 9	56398
11	exp 'snoring'/ or exp 'sleep disordered breathing'/ or (sleep adj3 (apnea or apnoea or hypopnea or hypopnoea)).ti,ab. or 'upper airway resistance'.ti,ab. or 'sleep disordered breathing'.ti,ab. or snor*.ti,ab.	78264
12	10 and 11	3391
13	(case report or review).pt.	2490279
14	(exp experimental organism/ or animal tissue/ or animal cell/ or exp animal disease/ or exp carnivore disease/ or exp bird/ or exp experimental animal welfare/ or exp animal husbandry/ or animal behavior/ or exp animal cell culture/ or exp mammalian disease/ or exp mammal/ or exp marine species/ or nonhuman/ or animal.hw.) not human/	7089673
15	13 or 14	9367753
16	12 not 15	2980
17	limit 16 to embase	1713
18	limit 6 to conference abstracts	834
19	17 or 18	2547

	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Q9	Q10	Q11	Q12	Total score	Quality
			C	Quasi-	expe	rimer	ıtal st	udy						
Wu et al. 2019	2	2	0	2	1	2	0	0	0	2	0	2	13	Fair
					Coho	rt stu	dy							
Bettega et al. 2000	2	2	0	2	1	2	2	0					11	Fair
Bianchi et al. 2014	2	2	0	2	1	2	0	0					9	Fair
Boyd et al. 2015	2	2	2	2	1	2	0	2					13	High
Conradt et al. 1997	2	2	0	2	1	2	2	0					11	Fair
Gerbino et al. 2013	2	2	2	2	1	2	2	0					13	High
Goh et al. 2013	2	2	2	2	0	2	2	0					12	Fair
Goodday et al. 2016	2	2	0	2	0	2	0	0					8	Fair
Hsieh et al. 2014	2	0	2	2	1	2	0	0					9	Fair
Kastoer et al. 2019	2	0	2	2	1	2	2	0					11	Fair
Li et al. 1999	0	2	0	2	0	2	2	0					8	Fair
Li et al. 2000	2	2	0	2	0	2	0	0					8	Fair
Li et al. 2001	2	2	0	2	0	2	0	0					8	Fair
Li et al. 2002	2	1	2	2	0	2	0	0					9	Fair
Liao et al. 2015	2	2	2	2	1	2	0	0					11	Fair
Liu et al. 2015	2	2	0	2	1	2	0	0					9	Fair
Rubio-Bueno et al. 2017	2	2	2	2	1	2	0	0					11	Fair
Veys et al. 2015	2	2	2	2	0	2	0	0					10	Fair
Vigneron et al. 2016	2	2	0	2	1	2	0	0					9	Fair

Table S2a. Methodological appraisal of the individual studies according to MINORS assessment tool – maxillomandibular advancement surgery

Q1, a clear study aim; Q2, inclusion of consecutive patients; Q3, prospective collection of data; Q4, endpoint appropriate to the aim of the study; Q5, unbiased assessment of the study; Q6, follow-up period appropriate to the aim of the study endpoint; Q7, loss of follow-up less than 5%; Q8, prospective calculation of the study size; Q9, an adequate control group; Q10, contemporary group; Q11, baseline equivalent of groups; Q12, adequate statistical analysis.

	Q1	Q2	Q3	Q 4	Q5	Q6	Q7	Q 8	Q9	Q10	Q11	Q12	Total score	Quality
				Qua	si-exp	erim	ental	study	/					
Aynaci et al. 2018	2	0	0	2	0	2	0	0	0	0	0	2	8	Low
Cammaroto et al. 2017	2	0	0	2	1	2	0	0	0	0	2	2	11	Fair
Ceylan et al. 2009	2	2	2	2	0	2	2	0	2	2	2	2	20	High
Chen et al. 2014	2	2	2	2	1	2	0	0	0	2	1	1	15	Fair
El-Anwar et al. 2018	2	2	2	2	0	2	0	0	0	2	0	2	14	Fair
Friedman et al. 2003	2	2	0	2	1	2	0	0	0	0	1	2	12	Fair
Li et al. 2013	2	2	0	2	1	2	0	0	0	2	2	2	15	Fair
Li et al. 2016	2	2	0	2	1	2	0	2	0	2	2	2	17	High
Sezen et al. 2011	2	2	2	2	1	2	0	0	0	2	0	1	13	Fair
Vicini et al. 2014	2	2	0	2	1	2	0	0	0	0	2	2	13	Fair
Yuksel et al. 2016	2	2	2	2	1	2	2	0	0	2	1	2	18	High
					Col	nort s	tudy							
Babademez et al. 2010	2	2	0	2	1	2	2	0					11	Fair
Benazzo et al. 2008	2	2	0	2	1	2	0	0					9	Fair
Bostanci et al. 2016	2	2	0	2	1	2	0	0					9	Fair
Cambi et al. 2019	2	2	0	2	1	2	2	0					11	Fair
Chen et al. 2019	2	2	2	2	1	2	2	0					13	High
Chiffer et al. 2015	2	0	2	2	0	2	2	0					10	Fair
Cillo et al. 2013	2	2	0	2	0	2	0	2					10	Fair
Emara et al. 2011	2	2	2	2	1	2	0	0					11	Fair
Eun et al. 2008	2	0	2	2	1	2	0	0					9	Fair
Friedman et al. 2007	2	2	2*	2	0	2	0	0					10	Fair
Gunbey et al. 2015	2	0	2	2	1	2	0	0					9	Fair
Hendler et al. 2001	2	0	0	2	0	2	0	0					6	Low
Li et al. 2016	2	2	0	2	1	2	0	0					9	Fair
Lin et al. 2010	2	2	2*	2	1	2	0	0					11	Fair
Neruntarat et al. 2003	2	2	0	2	1	2	0	0					9	Fair
Neruntarat et al. 2009	2	2	2	2	1	2	0	0					11	Fair
Omur et al. 2005	2	2	2*	2	1	2	2	0					13	High
Plzak et al. 2013	2	2	0	2	1	2	2	0	0	0	2	2	15	Fair
Sorrenti et al. 2006	2	2	0	2	1	2	2	0					11	Fair
Sun et al. 2008	2	2	0	2	1	2	2	0					11	Fair
Tantawy et al. 2018	2	2	2	2	0	2	0	0					10	Fair
Toh et al. 2014	2	2	2*	2	1	2	0	0					11	Fair
Tsou et al. 2018	2	2	0	2	2	2	0	0					10	Fair
Turhan et al. 2015	2	2	2	2	1	2	2	0					13	High
Vicente et al. 2006	2	2	2	2	0	2	2	0					12	Fair
Wang et al. 2013	2	2	0	2	1	2	2	0					11	Fair
Yi et al. 2011	2	2	2	2	1	2	2	0					13	High

Table S2b. Methodological appraisal of the individual studies according to MINORS assessment tool – multilevel surgery

Q1, a clear study aim; Q2, inclusion of consecutive patients; Q3, prospective collection of data; Q4, endpoint appropriate to the aim of the study; Q5, unbiased assessment of the study; Q6, follow-up period appropriate to the aim of the study endpoint; Q7, loss of follow-up less than 5%; Q8, prospective calculation of the study size; Q9, an adequate control group; Q10, contemporary group; Q11, baseline equivalent of groups; Q12, adequate statistical analysis.

* A retrospective study of prospectively collected data.

5



Fig. S1. PRISMA flow diagram of the study selection process



Fig. S2. Risk of bias assessment of randomized controlled trials using the Cochrane Collaboration "Risk of bias" tool

		ar-min	n .	1-1	6-mm	•		mean Difference	Mean Unterence
study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
1.1.1 Baseline AHI less t	han 40/h	b)							
Isieh 2014 [31]	4.8	4.4	16	35.7	18	16	5.6%	-30.90 [-39.98, -21.82]	
Rubio-Bueno 2017 [39]	8.45	4.33	34	38.3	10.7	34	6.1%	-31 85 [-35 73, -27 97]	-
/eys 2017 [40]	12.3	14.4	10	26.8	12.7	10	5.1%	-14.50 [-26.40, -2.60]	
Subtotal (95% CI)			60			60	16.8%	-27.12 [-36.04, -18.21]	•
Heterogeneity: Tau# = 44	24; Chi*	= 7.40	, df = 2	(P=0.0	02); F=	: 73%			
fest for overall effect: Z =	5.96 (P	< 0.00	001)						
1.1.2 Baseline AHI from	40/h to 7	0/h							
3ettega 2000 (24)	11.1	8.9	20	59.3	29	20	4.9%	-48.20 [-61.49, -34.91]	
Jianchi 2014 (25)	12.3	5.5	10	56.8	5.2	10	6.1%	-44.50 (-49.19, -39.81)	-
3oyd 2015 [26]	8	10.7	14	50	20	14	5.1%	-42.00 [-53.88, -30.12]	
Conradt 1997 [27]	8.5	9.4	15	51.4	16.9	15	5.4%	-42.90 [-52.69, -33.11]	
Jerbino 2014 [28]	17.3	16.7	10	69.8	35.2	10	3.3%	-52.50 [-76.65, -28.35]	
Castoer 2019 [32]	9.9	72	14	40.2	25.6	14	4.8%	-30.30 [-44.23, -16.37]	
1 2001 [36]	9.2	8	52	61.6	23.9	52	5.0%	-52.40 [-59.25, -45.55]	-
Jao 2015 [37]	5.3	4	20	41.6	19.2	20	5.6%	-36.30 [-44.90, -27.70]	-
Ju 2016 (38)	9.5	7.4	20	53.6	26.6	20	5.1%	-44 10 [-56.20, -32.00]	
/icini 2010 [41]	8.1	7	25	56.8	16.5	25	5.8%	-48.70 [-55.73, -41.67]	-
/igneron 2017 [42]	25.5	20.6	29	56.6	24	29	5.2%	-31 10 [-42.61, -19.59]	
Nu 2019 [43]	10.9	3.3	28	59.3	14.5	28	6.0%	-48.40 [-53.91, -42.89]	-
Subtotal (95% CI)			257			257	63.2%	44.07 [47.75, 40.38]	•
leterogeneity: Tau [#] = 18.	61; Chi#	= 21.9	1, df=	11 (P=	0.03);	= 50°	%		
fest for overall effect: Z=	23.43 (F	° ≺ 0.0	0001)						
I.1.3 Baseline AHI great	er than 7	70/h							
3oh 2003 (29)	11.4	7.4	11	70.7	15.9	11	5.4%	-59.30 [-69.66, -48.94]	
3oodday 2016 [30]	16.1	26.2	13	117.9	9.2	13	4.6%	-101.80 [-116.89, -86.71]	
1 2000 [34]	7.6	5.1	40	71.2	27	40	5.6%	-63.60 [-72.12, -55.08]	-
1 2002 [33]	10.4	10.8	12	75.3	26.4	12	4.4%	-64.90 [-81.04, -48.76]	
Subtotal (95% CI)			76			76	20.0%	-71.77 [-88.37, -55.18]	•
teterogeneity: Tau² = 24 Test for overall effect: Z =	4.81; Ch 8.48 (P	r = 23 < 0.00	13, df= 001)	= 3 (P <	0.000), i *= {	37%		
Total (95% CI)			393			393	100.0%	-46.15 [-52.43, -39.88]	•
Heterogeneity: Tau ² = 16 Test for overall effect. Z =	3.03; Ch 14.42 (F	r = 18 < 0.0	2.62, df 0001)	(= 18 (F	• « 0.01	0001); (*= 90%		-100 -50 0 50

Test for subaroup differences: Chi² = 24.04. df = 2 (P < 0.00001), P = 91.7%

b

	Pos	t-MM	A	Pr	e-MM/	4		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
1.2.1 Baseline AHI fr	om 40/h	to 70	/h						
Bettega 2000 [24]	90	7	20	82	11	20	11.7%	8.00 [2.29, 13.71]	
Li 2001 [36]	87.5	4.7	52	75.9	10.6	52	16.6%	11.60 [8.45, 14.75]	-
Liao 2015 (37)	88.9	5	20	80.2	9.7	20	13.4%	8.70 [3.92, 13.48]	
Liu 2016 [38]	94.1	3.5	20	80.9	8.9	20	14.5%	13.20 [9.01, 17.39]	
Wu 2019 [43] Subtotal (95% CI)	87.9	3.7	28 140	73.4	10.8	28 140	14.5% 70.7%	14.50 [10.27, 18.73] 11.58 [9.39, 13.77]	•
Heterogeneity: Tau ² =	1.53; C	hi² = {	5.30, df	= 4 (P =	= 0.26)	= 24	5%		
Test for overall effect	Z=10.3	17 (P	< 0.000	01)					
1.2.2 Baseline AHI gr	eater th	an 70)/h						
Goh 2003 [29]	83.9	8.8	11	58.6	12.3	11	7.2%	25.30 [16.36, 34.24]	
Li 2000 (34)	86.3	3.9	40	67.5	14.8	40	13.5%	18.80 [14.06, 23.54]	
LI 2002 [33]	85.9	6.7	12	74.2	12	12	8.6%	12.70 [4.92, 20.48]	
Subtotal (95% CI)			63			63	29.3%	18.65 [12.67, 24.62]	•
Heterogeneity: Tau ² =	15.17,1	Chi#=	4.38, 0	f= 2 (P	= 0.11); * = 5	54%		
Test for overall effect	Z= 6.13	! (P <	0.0000	1)					
Total (95% CI)			203			203	100.0%	13.50 [10.50, 16.50]	•
Heterogeneity: Tau ² =	11.51;	Chi#=	20.32.	df = 7 (P = 0.0	105); P	= 66%		ten te te co
Test for overall effect	Z = 8.83	(P <	0.0000	1)					-50 -25 0 25 50
Test for subgroup diff	lerences	Chi	= 4.74	df = 1	(P = 0)	03), I*=	78.9%		roatmine rie-wine

а



d

	Pos	st-MM	A	Pre	-MM	A		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
1.4.1 Baseline AHI less t	han 40/h	É.							
Rubio-Bueno 2017 (39)	0.79	1.41	34	17.4	5.4	34	13.2%	-16.61 [-18.49, -14.73]	-
Veys 2017 [40] Subtotal (95% CI)	5.7	3	10	14.1	5.9	10 44	11.7% 24.9%	-8.40 [-12.50, -4.30] -12.72 [-20.75, -4.68]	-
Heterogeneity: Tau ² = 31	05; Chi*	= 12.7	2, df =	1 (P=0	.000	4); f ² = 1	92%		22-21
Test for overall effect Z=	3.10 (P=	= 0.00	2)						
1.4.2 Baseline AHI from	40/h to 7	0/h							
Kastoer 2019 [32]	9	7	14	13	6	14	11.0%	-4.00 [-8.83, 0.83]	
Liao 2015 [37]	7	3	20	11.9	7.3	20	12.2%	-4.90 [-8.36, -1.44]	
Liu 2016 [38]	5.7	2.7	20	17	4.8	20	12.9%	+11.30 [-13.71, -8.89]	
Vicini 2010 [41]	77	1.3	25	11.6	2.8	25	13.5%	-3.90 [-5.11, -2.69]	+
Wu 2019 [43]	2.5	2.5	28	12.8	2.8	28	13.5%	-10.30 [-11.69, -8.91]	-
Subtotal (95% CI)			107			107	63.1%	-7.04 [-10.70, -3.38]	•
Heterogeneity: Tau ² = 15.	.35; Chi*	= 61.9	4, df=	4 (P < 0	.000	01); P=	94%		0.4401
Test for overall effect Z=	3.77 (P =	= 0.00	02)						
1.4.3 Baseline AHI great	er than 7	0/h							
Goodday 2016 [30]	5	4.1	13	12.9	5.5	13	12.0%	-7.90 [-11.63, -4.17]	-
Subtotal (95% CI)			13			13	12.0%	-7.90[-11.03, -4.17]	-
Heterogeneity: Not applic	elde								
Test for overall effect Z=	4.15 (P	< 0.00	01)						
Total (95% CI)			164			164	100.0%	-8.54 [-12.15, -4.92]	•
Heterogeneity: Tau ² = 24	81; Chi#	= 146	51, df:	= 7 (P <	0.00	001); P	= 95%	10 U U U	-10 10 0 10 20
Test for overall effect: Z=	4.63 (P	< 0.00	001)						Post MMA Pro MMA
Test for subgroup differe	nces: Ch	1=1.	59, df=	2(P = 1)	0.45).	1= 09			i sachums Fiernings

Fig. S3. Pre- and post-MMA mean difference for apnea-hypopnea index (a), lowest oxygen saturation (b), oxygen desaturation index (c), and Epworth sleepiness scale (d). CI, confidence interval; MMA, maxillomandibular advancement; SD, standard deviation.

	P	ost-MLS	1	P	re-MLS			Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% C
1.1 Baseline AHI less than 40/h									
Babademez 2010 [45]	8.9	6.5	16	20.1	10.5	16	2.6%	-11.20 [-17.25, -5.15]	
Benazzo 2008 (46)	18.7	16	109	37	19.1	109	2.7%	-18.30 [-22.98, -13.62]	
Cammaroto 2017 (Group 1) [49]	22.92	13.3	10	34.04	14.03	10	2.1%	-11.12[-23.10, 0.86]	
Cammaroto 2017 (Group 2) [49]	9.63	9.25	10	35.59	13.87	10	2.2%	-25.96 [-38.29, -15.63]	
Cammaroto 2017 (Group 3) [49]	13,53	7.76	10	37.84	21.6	10	1.9%	-24.31 [-38.54, -10.06]	
Ceylan 2009 (50)	16.1	3.9	26	29.6	7.8	26	2.7%	-13.50 [-16.85, -10.15]	
Cillo 2013 [55]	12.1	8.2	13	28.3	13.2	13	2.4%	-16.20 [-24.65, -7.75]	
Eun 2008 (58)	13.9	18.7	66	22.9	14.7	66	2.6%	-9.00 [-14.74, -3.26]	
Friedman 2007 (60)	14.5	10.2	122	23.2	7.6	122	2.8%	-8.70 [-10.96, -6.44]	
Junbey 2015 [61]	15.3	9.8	42	35.8	12.1	42	2.7%	-20.50 [-25.21, -15.79]	
J 2013 [65]	8.91	5.9	45	39.42	17.82	45	2.6%	-30.51 [-35.99, -25.03]	
Veruntarat 2009 (22)	16.8	3.2	72	35.6	9.2	72	2.8%	-18.80 [-21.05, -16.55]	-
Pizak 2013 (69)	14.1	18.2	79	28.7	17.1	79	2.6%	-14.60 [-20.11, -9.09]	
Sezen 2011 [70]	15.34	11.05	12	28.78	10.69	12	2.4%	-13.44 [-22.14, -4.74]	
ľsou 2018 (75)	17.51	18.92	36	25.14	17.53	36	2.4%	-7.63[-16.06, 0.80]	
vicini 2014 (Group 1) [78]	19.8	14.1	12	38.4	19.7	12	1.9%	-18.60 [-32.31, -4.89]	
/icini 2014 (Group 2) [78]	9.9	8.6	12	38.5	14.3	12	2.3%	-28.60 [-38.04, -19.16]	
/uksel 2016 (81)	18	11.3	14	33.2	18.9	14	2.1%	-15.20 [-26.73, -3.67]	
Subtotal (95% CI)			706			706	43.5%	-16.65 [-19.85, -13.44]	•
leterogeneity: Tau ² = 33.67; Chi ² :	= 102.85	i, df = 17	(P < 0	00001)	, F= 83	%			
est for overall effect Z = 10.17 (P	< 0.000	01)							
.1.2 Baseline AHI from 40/h to 70	D/h								
3ostanci 2016 [47]	19.9	17.4	82	47.3	18.7	82	2.6%	-27.40 [-32.93, -21.87]	
Cambi 2019 [48]	19.4	10.1	20	49.3	18.5	20	2.3%	-29 90 [-39 14, -20 66]	
chen 2014 (Group 1) [52]	35.08	18.5	22	66.38	17.03	22	2.2%	-31.30 [-41.81, -20.79]	
chen 2014 (Group 2) [52]	26.17	18.85	24	46.1	13.26	24	2.3%	-19.93 [-29.15, -10.71]	
then 2018 (Group 1) [53]	25,21	7.85	26	51.78	14.65	26	2.5%	-26 57 -32 96, -20 18	
Chen 2018 (Group 2) [53]	14.87	217	45	52.34	6.29	45	2.8%	-37 47 [-39.41, -35.53]	-
Chen 2019 (51)	27.03	4.01	45	49.67	7.43	45	2.8%	-22.64 [-25.11, -20.17]	-
Chiffer 2015 [54]	19.8	22.1	18	53.9	25.4	18	1.7%	-34.10 [-49.65, -18.55]	
El-Anwar 2018 (56)	24.5	10.9	20	48.8	31.6	20	1.8%	-24 30 -38.95, -9.651	
Emara 2011 (57)	15.4	10.7	23	40.7	17.4	23	2.4%	-25.30 [-33.65, -16.95]	
riedman 2003 (59)	28.1	20.6	143	43.9	23.7	143	2.6%	-15.80 [-20.95, -10.65]	
1 2016 [63]	16.5	11.2	30	48.4	16.9	30	2.5%	-31.90 [-39.15, -24.65]	
1 2016 (64)	12.8	8.2	25	45.7	21.7	25	2.3%	-32.90 [-41.99, -23.81]	
in 2010 (66)	23.4	24.7	43	51.5	25.4	43	2.2%	-28.10 (-38.69, -17.51)	
Veruntarat 2003 (67)	18.6	4.1	46	47.9	8.4	46	2.8%	-29.301-32.00, -26.601	-
2mur 2005 (68)	17.31	14.17	22	47.5	1574	22	2.3%	-30 19 1-39 04 -21 341	
Sorrenti 2006 (71)	9.4	5.4	10	54.7	11.5	10	2.4%	-45.30 [-53.17, -37.43]	
Sun 2008 (72)	28.58	29.11	31	65.93	23.83	31	1.9%	-37 35 [-50.59, -24 11]	
Cantawy 2018 (73)	25.6	9.52	37	68.4	25.3	37	23%	-42 80 1-52 17 -33 431	
Tah 2014 (74)	13.5	17.1	20	41.3	22.1	20	2.0%	-27 80 1-40 05 -15 551	
Jurhan 2015 (76)	20.46	17.73	90	51 81	18.84	90	2.6%	-31 35 1-36 69 -26 011	
Acente 2006 (77)	141	22.5	54	52.0	14.9	54	2.5%	-38 70 1-46 12 -31 201	
Nana 2013 [79]	73.7	18.4	36	50.0	20.5	36	2 3 96	-36 60 L45 60 -27 60	
0 2011 1901	30.1	22.1	26	65.6	17.6	26	2 1 96	-35 50 L46 66 -24 341	
Subtotal (95% CI)	30.1	43.1	933	00.0		933	56.5%	-30.72 [-33.99, -27.46]	•
Heterogeneity: Tau ^a = 47.88; Chi ^a :	= 159.86	i, df = 23	(P < 0	00001)	; l ^a = 86	96		1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 -	No. Contract
Test for overall effect Z = 18.45 (P	< 0.000	01)							
Total (95% CI)			1639			1639	100.0%	-24.74 [-28.06, -21.42]	•
Heterogeneity Tau ² = 101.91; Chi ²	= 642.6	7, df = 4	1 (P +	0.0000	1); (= 9	4%			50 35 0
Tant for ownell offert 7 - 11 Et (D	- 0.000	041							-50 -25 0

Test for overall effect: Z = 14.81 (P < 0.00001) Test for subgroup differences: ChP = 36.37. df = 1 (P < 0.00001). P = 97.3%

b

	P	ost-MLS		P	re-MLS			Mean Difference		Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI		IV, Random, 95% Cl
2.2.1 Baseline AHI less that	in 40/h	10/C=C								
Babademez 2010 [45]	86.6	2	16	84.6	3.4	16	4.5%	2.00 [0.07, 3.93]		-
Cevian 2009 (50)	94.6	4.9	26	86.8	8.9	26	4.1%	7.80 [3.89, 11.71]		
Eun 2008 (58)	79.4	16.5	66	79.1	5.7	66	4.1%	0.30 (-3.91, 4.51)		
Friedman 2007 (60)	90.4	4.3	122	88.9	4.8	122	4.6%	1.50 [0.36, 2.64]		-
1 2013 (65)	83	5	45	66	16	45	3.9%	17.00 [12.10, 21.90]		
Neruntarat 2009 (22)	88.2	2.4	72	85.6	34	72	4.6%	2.60 [1.64, 3.56]		-
Subtotal (95% CI)			347			347	25.7%	4.35 [1.91, 6.79]		•
Heterogeneity Tau* = 7.08;	Chi#= 4	5.03, df	= 5 (P	< 0.000	01); I*=	89%				211
Test for overall effect: $Z = 3$	50 (P = 6	0.0005)	1997-0		221211					
2.2.2 Baseline AHI from 40	/h to 70/	h								
Bostanci 2016 [47]	82.3	7.4	82	75.7	8.9	82	4.4%	6.60 [4.09, 9.11]		+
Cambi 2019 [48]	80	7.4	20	69.5	9.9	20	3.8%	10.50 (5.08, 15.92)		
Chen 2018 (Group 1) (53)	86.03	5.44	45	58.67	8.33	45	4.3%	27.36 [24.45, 30.27]		
Chen 2018 (Group 2) [53]	76.88	4.02	45	60.26	7.25	45	4.4%	16.62 [14.20, 19.04]		-
Chen 2019 (51)	67.84	19.3	22	61.89	12.54	22	2.7%	5.95 [-3.67, 15.57]		
El-Anwar 2018 [56]	84	5.3	20	73.5	14.8	20	3.4%	10.50 [3.61, 17.39]		
Emara 2011 (57)	87.2	11.1	23	78.9	12.6	23	3.4%	8.30 [1.44, 15.16]		
Friedman 2003 (59)	85.9	9.8	143	81.4	10.4	143	4.4%	4.50 [2.16, 6.84]		-
Li 2016 (63)	82.4	5.4	30	76.4	8.5	30	4.2%	6.00 [2.40, 9.60]		-
1. 2016 [64]	83.3	5.6	25	77.1	10.5	25	3.9%	6.20 [1.54, 10.86]		
Lin 2010 (66)	82.1	10.9	43	75.5	10.4	43	4.0%	6.60 [2.10, 11.10]		
Neruntarat 2003 [67]	87.2	3.1	46	81.2	2.9	46	4.6%	6.00 [4.77, 7.23]		-
Sorrenti 2006 (71)	90.7	3	10	77	6.2	10	4.0%	13.70 [9.43, 17.97]		
Sun 2008 (72)	74.97	12.45	31	72.65	11.93	31	3.6%	2.32 [-3.75, 8.39]		
antawy 2018 [73]	83.2	2.86	32	66.8	11.3	32	4.1%	16.40 [12.36, 20.44]		
oh 2014 (74)	84.5	7.1	20	72.9	19.3	20	2.8%	11.60 [2.59, 20.61]		
Furhan 2015 [76]	82.38	6.57	90	75.63	9.3	90	4.4%	6.75 [4.40, 9.10]		-
/icente 2006 [77]	82.2	11.2	54	76.2	12.4	54	4.0%	6.00 [1.54, 10.46]		
Wang 2013 (79)	85.6	10	36	70.5	12.4	36	3.8%	15.10 (9.90, 20.30)		
Subtotal (95% CI)			817			817	74.3%	9.93 [6.90, 12.96]		•
leterogeneity: Tau ² = 39.33	3; Chi#=	271.04,	df=18	B (P = 0.	00001);	P# = 93	%			1.124
Test for overall effect $Z = 6$	42 (P < (0.00001)							
Fotal (95% CI)			1164			1164	100.0%	8.66 [6.24, 11.07]		•
Heterogeneity: Tau ^a = 32.84	4; Chi²=	488.70,	df = 24	1 (P < 0.	00001),	I#= 95	%		-50	-25 0 25
Test for overall effect: $Z = 7$.	.02 (P < 0)	0.00001)							Post-MLS Pre-MLS

Test for subgroup differences: Chi# = 7.90, df = 1 (P = 0.005), I# = 87.3%

С



_	
~	
n	
u	

	Po	st-ML	s	PI	e-MLS	S		Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl
2.4.1 Baseline AHI less than 40/h									
Benazzo 2008 (46)	7.2	23	109	10.5	3.1	109	3.8%	-3.30 [-4.02, -2.58]	-
Cammaroto 2017 (Group 1) [49]	8.5	5.42	10	12.3	4.24	10	1.9%	-3.60 1-8.07. 0.471	
Cammaroto 2017 (Group 2) [49]	4.9	3.87	10	13	4.49	10	2.2%	-8.10 -11.774.43	
Cammaroto 2017 (Group 3) [49]	3.9	3.57	10	10.4	2.5	10	2.7%	-6 50 1-9 20, -3 801	
Cevtan 2009 [50]	8.2	27	26	10.8	10.8	26	1.9%	-2.60 [-6.88, 1.68]	
Cilla 2013 (55)	6.3	3.9	13	15.2	3	13	2.8%	-8.90 (-11.57, -6.23)	
Eun 2008 (58)	7.5	4.5	66	11.4	5	66	3.4%	-3.90 [-5.52, -2.28]	
Friedman 2007 (60)	6.9	3.3	122	9.7	3.9	122	3.8%	-2.80 [-3.71, -1.89]	-
Li 2013 (65)	3.44	2.86	45	12.91	4.89	45	3.4%	-9.47 [-11.13, -7.81]	
Neruntarat 2009 (22)	8.2	2.5	72	14.2	3.4	72	3.7%	-6.00 [-6.97, -5.03]	-
Pizak 2013 (69)	7.3	3.2	79	10.6	3.8	79	3.7%	-3.30 [-4.40, -2.20]	-
Sezen 2011 (70)	7.58	3.15	12	14.83	2.52	12	3.0%	-7.25 [-9.53, -4.97]	
Tsou 2018 (75)	10.2	4.33	36	11.86	4.31	36	3.2%	-1.66 [-3.66, 0.34]	
vicini 2014 (Group 1) [78]	7.6	4.4	12	13.75	4	12	2.3%	-6.15 -9.51, -2.79]	
Vicini 2014 (Group 2) [78]	4.4	4.1	12	12	4.9	12	2.2%	-7.60 [-11.21, -3.99]	
Yuksel 2016 (81)	5	4.4	14	11.9	7	14	1.8%	-6.90 [-11.23, -2.57]	
Subtotal (95% CI)			648			648	45.8%	-5.37 [-6.55, -4.19]	•
Heterogeneity: Tau ^a = 4.16, Chi ^a = Test for overall effect: Z = 8.92 (P <	107.84, 0.0000	df=15 1)	5 (P < (00001); (* = 8	16%			
2.4.2 Baseline AHI from 40/h to 70)/h								
Cambi 2019 (48)	7.7	4.5	20	12.7	4.3	20	2.7%	-5.00 [-7.73, -2.27]	
Chen 2018 (Group 1) [53]	6.01	1.27	45	12.76	2.24	45	3.8%	-6.75 [-7.50, -6.00]	-
Chen 2018 (Group 2) (53)	8.49	2.02	45	13.01	2.59	45	3.8%	-4.52 [-5.48, -3.56]	-
El-Anwar 2018 (56)	4.1	27	20	12.6	5.6	20	2.7%	-8.50 [-11.22, -5.78]	
Emara 2011 (57)	8.3	3.9	23	14.2	2.3	23	3.3%	-5.90 [-7.75, -4.05]	
Friedman 2003 (59)	8.3	3.9	143	15.2	3.1	143	3.8%	-6.90 [-7.72, -6.08]	-
Li 2016 (63)	8.7	3.9	30	10.9	4.7	30	3.1%	-2.20 [-4.39, -0.01]	
L) 2016 [64]	7.5	4.3	25	9.6	4.9	25	2.8%	-2.10 [-4.66, 0.46]	
in 2010 [66]	10	4.3	43	12.8	5,1	43	3.2%	-2.80 [-4.79, -0.81]	
Neruntarat 2003 (67)	7.3	27	46	15.9	2.7	46	3.7%	-8.60 [-9.70, -7.50]	
Omur 2005 (68)	5.4	4.27	22	13.9	2.15	22	3.2%	-8.50 [-10.50, -6.50]	
Sun 2008 (72)	8.9	4.9	31	17.1	4.1	31	3.0%	-8.20 [-10.45, -5.95]	
Tantawy 2018 (73)	5.2	1.6	32	13.8	5.4	32	3.2%	-8.60 [-10.55, -6.65]	
Toh 2014 (74)	5.6	4.4	20	13	2.8	20	3.0%	-7.40 [-9.69, -5.11]	
/icente 2006 [77]	8.2	6.1	54	12.2	3.3	54	3.3%	-4.00 [-5.85, -2.15]	
Wang 2013 (79)	5.5	3.6	36	12.2	5.8	36	3.0%	-6.70 [-8.93, -4.47]	
YI 2011 [80]	6.8	5.2	26	13.5	5.9	26	2.5%	-6.70 [-9.72, -3.68]	
Subtotal (95% CI)			661			661	54.2%	-6.12 [-7.06, -5.18]	•
Heterogeneity: Tau [#] = 2.90; Chi [#] = Test for overall effect: Z = 12.76 (P	92.29, d < 0.000	ff=16 01)	(P < 0.1	00001);	l#= 83	196			
Total (95% CI)			1309			1309	100.0%	-5.77 [-6.57, -4.96]	•
Heterogeneity: Tau ² = 4.25; Chi ² = Test for overall effect Z = 14.06 (P Test for subgroup differences: Chi	255.51, < 0.000 = 0.95	df = 32 01) . df = 1	2 (P ≤ 0 (P = 0.	.00001) 33), F=); i ^o = 8 0%	7%			-20 -10 0 10 Post-MLS Pre-MLS

Fig. S4. Pre- and post-MLS mean difference for apnea-hypopnea index (a), lowest oxygen saturation (b), oxygen desaturation index (c), and Epworth sleepiness scale (d). CI, confidence interval; MLS, multilevel surgery; SD, standard deviation.

1	
-	

	Pe	ost-MLS		P	re-MLS			Mean Difference	Mean Difference
tudy or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% C
3.1.1 Soft palate level & fongue b	ase leve	4		1.000.000				The second second second second second	
Babademez 2010 (45)	8.9	6.5	16	20.1	10.5	16	2.6%	-11.20 [-17.25, -5.15]	
Bostanci 2016 [47]	19.9	17.4	82	47.3	18.7	82	2.6%	-27.40 [-32.93, -21.87]	
Cambi 2019 (48)	19.4	10.1	20	49.3	18.5	20	2.3%	-29.90 [-39.14, -20.66]	
Cammaroto 2017 (Group 1) [49]	22.92	13.3	10	34.04	14.03	10	2.1%	-11.12 [-23.10, 0.86]	
Cammaroto 2017 (Group 2) [49]	9.63	9.25	10	35.59	13.87	10	2.2%	-25.96 (-36.29, -15.63)	
Cammaroto 2017 (Group 3) [49]	13.53	7.76	10	37.84	21.6	10	1.9%	-24.31 [-38.54, -10.08]	
Ceylan 2009 (50)	16.1	3.9	26	29.6	7.8	26	2.7%	-13.50 [-16.85, -10.15]	-
Chen 2014 (Group 1) [52]	26.17	18.85	24	46.1	13.26	24	2.3%	-19 93 [-29 15, -10.71]	
Chen 2014 (Group 2) [52]	25.21	7.85	26	51.78	14.65	26	2.5%	-26.57 [-32.96, -20.18]	
Chen 2018 (Group 2) [53]	27.03	4.01	45	49.67	7.43	45	2.8%	-22.64 [-25.11, -20.17]	-
Chen 2019 (51)	35.08	18.5	22	66.38	17.03	22	2.2%	-31.30 [-41.81, -20.79]	
Chiffer 2015 [54]	19.8	22.1	18	53.9	25.4	18	1.7%	-34.10 [-49.65, -18.55]	
Emara 2011 (57)	15.4	10.7	23	40.7	17.4	23	2.4%	-25.30 [-33.65, -16.95]	
Eun 2008 (58)	13.9	18.7	66	22.9	14.7	66	2.6%	-9.00 [-14.74, -3.26]	
Friedman 2003 (59)	28.1	20.6	143	43.9	237	143	2.6%	-15 80 J-20.95, -10.651	
Friedman 2007 (60)	14.5	10.2	122	23.2	7.6	122	2.8%	-8 70 1-10 96 -6.44	~
Gunbey 2015 (61)	153	9.8	42	35.8	121	42	2.7%	-20 50 1-25 21 -15 791	
1 2013 [65]	8.91	5.9	45	39.42	17.82	45	2.6%	-30 51 (-35 99 -25 03)	
1 2016 (63)	16.5	11.2	30	48.4	16.9	30	2.5%	-31.90 (-39 15 -24 65)	
1 2016 (64)	17.9	82	25	45.7	21.7	25	7 3%	-32 90 641 99 -23 911	
Lin 2010 (66)	23.4	247	43	51 5	25 4	43	2 2%	-28 10 1-38 69 -17 511	
Neruntarat 2009 (22)	15.9	32	72	35.6	97	73	2.9%	-18 80 1 21 05 -16 561	-
Omur 2006 (60)	17.51	1417	22	47.5	15.74	22	2.0%	20 10 1 20 04 21 24	
Dirak 2012 (60)	14.1	10.2	70	97.0	13.74	70	2.370	14 801 20 11 0 001	
Peren 2011 [70]	16.34	11.00	13	20.1	10.60	10	7 4 96	-19.00 [-20.11, -0.00]	
Sezen zori (roj	10.04	17.00	20	41.2	22.4	20	2.4.30	27 001 40 06 16 661	
Ten: 2010 (76)	13.5	10.00	20	91.3	17.63	20	2.0%	-27.80 [-40.05, -15.55]	
1990 2018 (/ 0) Turbas 2016 (70)	17.51	18.92	30	20.14	17.03	30	2.470	-/.03 [-10.00, 0.80]	
Tuman 2015 [76]	20.40	17.73	90	51.81	10.04	90	2.0%	-31.35 [-30.69, -20.01]	
vicente 2006 [77]	14.1	23.5	- 54	52.8	14.9	04	2.5%	-38,70 [-46,12,-31,28]	
vicini 2014 (Group 1) [78]	19.8	14.1	12	38.4	19.7	12	1.9%	-18 60 [-32 31, -4 89]	
Vicini 2014 (Group 2) [78]	9.9	8.6	12	38.5	14.3	12	2.3%	-28.00 [-38.04, -19.16]	
wang 2013 [79]	23.2	18.4	36	59.8	20.5	36	2.3%	-36.60 [-45.60, -27.60]	
Yuksel 2016 (B1)	18	11.3	14	33.2	18.9	14	21%	-15.20[-26.73, -3.67]	
Subtotal (95% Ct)		1945.0-24	1307	122222		1307	18.8%	-22.12 [-25,14, -19.09]	•
Heterogeneity: Tau* = 61.36; Chi* Test for overall effect: Z = 14.70 (P	= 279.91	, df = 32 01)	(P < 0	00001)); (* = 89	196			
3.1.2 Soft palate level & hvoid lev	/el								
Bena770 2008 (46)	187	16	109	37	191	109	27%	-18 30 1-22 98 -13 621	
El-Anwar 2018 (56)	24.5	10.9	20	48.9	31.6	20	1.8%	-24 30 1-38 95 -9 651	
Tantawy 2018 (73)	25.6	0.52	33	68.4	25.9	32	2.3%	-42.80 L52 17 -32 421	
Subtotal (95% CI)	20.0	19.00 k	161	00.3	20.3	161	6.8%	-28.36 [-45.18, -11.54]	
Heterogeneity: Tau ^a = 194.66; Chi Test for overall effect: Z = 3.30 (P :	P = 21.06	, df= 2	(P < 0.)	0001), P	= 91%				
	112221								
3.1.3 Soft palate level & tongue b	ase leve	& hyoi	d level						
Chen 2018 (Group 1) [53]	14.87	2.17	45	52.34	6.29	45	2.8%	-37.47 [-39.41, -35.53]	-
Cille 2013 [55]	12.1	8.2	13	28.3	13.2	13	2.4%	-16 20 [-24 65, -7 75]	
Neruntarat 2003 (67)	18.6	4.1	46	47.9	8.4	46	2.8%	-29.30 [-32.00, -26.60]	-
Sorrenti 2006 (71)	9.4	5.4	10	54.7	11.5	10	2.4%	-45.30 [-53.17, -37.43]	
Sun 2008 [72]	28.58	29.11	31	65.93	23.83	31	1.9%	-37.35 [-50.59, -24.11]	
YI 2011 [80]	30.1	23.1	26	65.6	17.6	26	2.1%	-35.50 [-46.68, -24.34]	
Subtotal (95% CI)			171			171	14.4%	-33.36 [-39.66, -27.06]	•
Helereneneity Tout - 40 40: Ohit	= 48.66,	df = 5 (F	e < 0.00	0001); P	= 90%				
Test for overall effect: Z = 10.38 (P	< 0.000	01)							
Test for overall effect Z = 10.38 (P	< 0.000	01)	1630			1630	100.05	24 74 1 20 06 24 423	
Total (95% CI)	° < 0.000	01) 7. w - 1	1639	0.0000		1639	100.0%	-24.74 [-28.06, -21.42]	•

Test for overall effect: Z = 14.61 (P < 0.00001) Test for subgroup differences: Chi² = 9.08. df = 2 (P = 0.01), P = 78.0%

b

	P	st MLS		P	re-MLS			Mean Difference		Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI		IV, Random, 95% CI
3.2.1 Soft palate level & to	ngue bas	se level	1				1100			
Babademez 2010 (45)	86.6	2	16	84.6	3.4	16	4.5%	2.00 (0.07, 3.93)		-
Bostanci 2016 [47]	82.3	7.4	82	75.7	8.9	82	4.4%	6.60 (4.09, 9.11)		-
Cambi 2019 (48)	80	7.4	20	69.5	9.9	20	3.8%	10.50 (5.08, 15.92)		
Cevian 2009 (50)	94.6	4.9	26	86.8	8.9	25	4.1%	7.80 (3.89, 11,71)		
Chen 2018 (Group 2) (53)	76.88	4.02	45	60.26	7 25	45	4.4%	16.62 [14 20, 19.04]		-
Chen 2019 (51)	67.84	19.3	22	61.89	12.54	22	2.7%	5 95 1-3 67, 15 571		
Emara 2011 (57)	87.2	11.1	23	78.9	12.6	23	3.4%	8 30 11 44 15 16		
Eun 2008 (58)	79.4	16.5	66	79.1	57	66	4.1%	0.30 [-3.91, 4.51]		
Friedman 2003 (59)	85.9	9.8	143	81.4	10.4	143	4.4%	4.50 (2.16, 6.84)		-
Friedman 2007 (60)	90.4	43	122	88.9	4.8	122	4.6%	1 50 (0 38, 2 64)		-
1 2013 (65)	83	5	45	66	16	45	3.9%	17 00 (12 10 21 90)		
1 2016 [63]	87.4	5.4	30	76.4	85	30	4 7%	6 00 12 40 9 601		
2015 (64)	83.3	5.6	25	77.1	10.5	25	3.0%	6 20 11 54 10 861		
Lin 2010 (66)	82.1	10.9	43	75.5	10.4	43	4.0%	6 60 12 10, 11 101		
Neruntarat 2009 (22)	88 2	2.4	72	85.6	34	72	4.6%	2.60 (1.64. 3.66)		-
Tob 2014 (74)	84.5	7.1	20	72.9	19.3	20	2.8%	11 60 12 59 20 611		
Turban 2015 (76)	82.38	6.57	90	75.63	93	90	4 4 %	6 75 (4 40 9 10)		-
Vicente 2006 (77)	02.30	11.2	64	76.2	12.4	54	4 0.06	6 00 11 64 10 461		
Wang 2013 (79)	95.6	10	35	70.5	174	36	3.8%	15 10 19 90 20 301		
Subtotal (95% CI)	05.0	10	980	10.5	14.4	980	76.0%	7.16 (5.00, 9.32)		•
3.2.2 Soft palate level & hy	oid level									
El-Anwar 2018 (56)	84	53	20	73.5	14.8	20	3 4 %	10 50 13 61 17 391		
Tantawy 2018 (73)	83.2	2.86	32	66.8	113	32	4 1 %	16 40 112 36 20 441		
Subtotal (95% CI)	0.012		52	00.0	1.1.1.1.1.1.1.1	52	7.5%	14.14 [8.51, 19.76]		•
Heterogeneity: Tau ² = 9.10, Test for overall effect: Z = 4	Chi¥ = 2 93 (P < (.10, df=).00001	1 (P=	0.15); P	= 52%					
3.2.3 Soft palate level & to	ngue ba:	se level	& hyoi	d level						
Chen 2018 (Group 1) [53]	86.03	5.44	45	58.67	8.33	45	4.3%	27.36 [24.45, 30.27]		-
Neruntarat 2003 [67]	87.2	3,1	46	81.2	2.9	48	4.6%	6.00 [4.77, 7.23]		-
Sorrenti 2006 (71)	90.7	3	10	77	6.2	10	4.0%	13.70 [9.43, 17.97]		
Sun 2008 [72]	74.97	12.45	31	72.65	11.93	31	3.6%	2.32 [-3.75, 8.39]		
Subtotal (95% CI)			132			132	16.5%	12.44 [0.59, 24.30]		
Heterogeneity: Tau² = 142. Fest for overall effect: Z = 2	13; Chi#= .06 (P= (= 185.24 0.04)	l, df = 3	(P < 0.)	00001);	°= 98	%			
			1164			1164	100.0%	8.66 [6.24, 11.07]		•
Total (95% CI)			1144							
Total (95% CI) Heterogeneity: Tau# = 32.8	4; Chi#=	488.70,	df= 24	(P < 0.	00001);	1= 95	96		1	te la te
Heterogeneity Tau ^a = 32.8	4; Chi ² =	488.70,	df= 24	(P < 0.	00001),	I* = 95	96		-50	-25 0 25





	Post-MLS			Pre-MLS				Mean Difference	Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI
4.1 Soft palate level & tongue b	ase leve	4							
ambi 2019 [48]	7.7	4.5	20	12.7	4.3	20	2.7%	-5.00 [-7.73, -2.27]	
ammaroto 2017 (Group 1) [49]	8.5	5.42	10	12.3	4.24	10	1.9%	-3.80 [-8.07, 0.47]	
ammaroto 2017 (Group 2) [49]	4.9	3.87	10	13	4.49	10	2.2%	-8.10 [-11.77, -4.43]	
ammaroto 2017 (Group 3) [49]	3.9	3.57	10	10.4	2.5	10	2.7%	-6.50 [-9.20, -3.80]	
eytan 2009 (50)	8.2	2.7	26	10.8	10.8	26	1.9%	-2.60 [-6.89, 1.68]	
hen 2018 (Group 2) [53]	8,49	2.02	45	13.01	2.59	45	3.8%	-4.52 [-5.48, -3.56]	
mara 2011 (57)	8,3	3.9	23	14.2	2.3	23	3.3%	-5.90 [-7.75, -4.05]	
un 2008 (58)	7.5	4.5	66	11.4	5	66	3.4%	-3.90 [-5.52, -2.28]	
rledman 2003 (59)	8.3	3.9	143	15.2	3.1	143	3.8%	-6.90 [-7.72, -6.08]	-
iedman 2007 (60)	6.9	3.3	122	9.7	3.9	122	3,8%	-2.80 [-3.71, -1.89]	
2013 (65)	3.44	2.86	45	12.91	4.89	45	3.4%	-9.47 [-11.13, -7.81]	
2016 (63)	8.7	3.9	30	10.9	4.7	30	3.1%	-2.20 [-4.39, -0.01]	
2016 (64)	7.5	4.3	25	9.6	4.9	25	2.8%	-2.10 [-4.66, 0.46]	
n 2010 (66)	10	4.3	43	12.8	5.1	43	3.2%	-2.80 [-4.79, -0.81]	
eruntarat 2009 [22]	8.2	2.5	72	14.2	3.4	72	3.7%	-6.00 [-6.97, -5.03]	-
mur 2005 (68)	5.4	4.27	22	13.9	2.15	22	3.2%	-8.50 [-10.50, -6.50]	
zak 2013 (69)	7.3	3.2	79	10.6	3.8	79	3.7%	-3.30 [-4.40, -2.20]	-
zen 2011 (70)	7.58	3.15	12	14.83	2.52	12	3.0%	-7.25 [-9.53, -4.97]	
uh 2014 (74)	5.6	4.4	20	13	2.8	20	3.0%	-7.40 [-9.69, -5.11]	
ou 2018 [75]	10.2	4.33	36	11.86	4.31	36	3.2%	-1.66 [-3.66, 0.34]	
ente 2006 (77)	8.2	6.1	54	12.2	3.3	54	3.3%	-4.00 [-5.85, -2.15]	
ini 2014 (Group 1) [78]	7.6	4.4	12	13,75	- 4	12	2.3%	-6.15 [-9.51, -2.79]	
ini 2014 (Group 2) [78]	4.4	4.1	12	12	4.9	12	2.2%	-7.60 [-11.21, -3.99]	
ing 2013 (79)	5.5	3.6	36	12.2	5.8	36	3.0%	-6.70 [-8.93, -4.47]	
ksel 2016 (81)	5	4.4	14	11.9	7	14	1.8%	-6.90 [-11.23, -2.57]	
ibtotal (95% CI)			987			987	74.4%	-5.23 [-6.12, -4.35]	♦
terogeneity: Tau* = 3.72; Chi* = st for overall effect: Z = 11.58 (P	146.52, < 0.000	df = 24 91)	(P < 0	00001)	; i* = 8	4%			
4.2 Soft palate level & hvoid lev	el								
anazzo 2008 (46)	7 2	22	100	10.5	34	100	3.8%	-3 30 64 07 -2 50	+
Anwar 2018 [56]	41	27	20	12.6	58	20	2.7%	-8 50 1-11 22 -5 701	
antawy 2018 (73)	5.7	1.6	32	13.8	5.4	32	3.2%	-8 60 1-10 55 -6 651	
ibtotal (95% CD	0.4	1.0	161	15.0	4.4	161	9.8%	-6.68 [-10.84, -2.53]	
eterogeneity: Tau² = 12.49, Chi² st for overall effect Z = 3.15 (P =	= 35.02, = 0.002)	df= 2	(P < 0.0	00001);	l*= 94	%			
4.3 Soft palate level & tongue b	ase leve	& hy	oid leve	zi					
ten 2018 (Group 1) (53)	6.01	1.27	45	12.76	7.74	45	3.8%	-6 75 1 7 50 -6 001	+
lin 2013 (55)	6.3	10	13	15.7	1	13	2.8%	-8 90 611 57 -6 231	
eruntarat 2003 (67)	73	27	46	15.9	27	46	3.7%	-8 60 1-9 70 -7 501	-
an 2008 (72)	8.9	4.9	31	17.1	4.1	31	3.0%	-8.201-10.455.951	
2011 (80)	6.8	52	26	13.5	5.9	26	2.5%	-6 70 1-9 72 -3 681	
ibtotal (95% CI)	0.0	4.6	161			161	15.8%	-7.75 [-8.86, -6.65]	•
terogeneity: Tau" = 0.78; Chi" = st for overall effect: Z = 13.78 (P	9.29, df < 0.000	= 4 (P 01)	= 0.05)	; P= 57	%	0.075		2010 • 110 • 110 •	
		28							1225
stal (05% CI)			4300			4300	400.0**	E 7718 E7 4001	▲ I

Fig. S5. Pre- and post-MLS mean difference for apnea-hypopnea index (a), lowest oxygen saturation (b), oxygen desaturation index (c), and Epworth sleepiness scale (d) – three subgroups according to the different target levels of obstructive sites addressed by surgery. CI, confidence interval; MLS, multilevel surgery; SD, standard deviation.

Begg's funnel plot with pseudo 95% confidence limits GWM -50 -100 Ó s.e. of: WMD

b



Fig. S6. Begg's funnel plot (a) and Egger's publication bias plot (b) for all maxillomandibular advancement surgery studies in meta-analysis. s.e., standard error; WMD, weighted mean difference.





Fig. 57. Begg's funnel plot (a) and Egger's publication bias plot (b) for all multilevel surgery studies in metaanalysis. s.e., standard error; WMD, weighted mean difference.



Fig. S8. Sensitivity analysis of AHI in meta-analysis for maxillomandibular advancement surgery studies. CI, confidence interval.



Fig. S9. Sensitivity analysis of AHI in meta-analysis for multilevel surgery studies. CI, confidence interval.

REFERENCES

- 1. Malhotra A, White DP. Obstructive sleep apnoea. Lancet. 2002;360:237-45.
- 2. Senaratna CV, Perret JL, Lodge CJ, Lowe AJ, Campbell BE, Matheson MC, et al. Prevalence of obstructive sleep apnea in the general population: a systematic review. *Sleep Med Rev.* 2017;34:70-81.
- 3. Epstein LJ, Kristo D, Strollo PJ Jr, Friedman N, Malhotra A, Patil SP, et al. Clinical guideline for the evaluation, management and long-term care of obstructive sleep apnea in adults. J Clin Sleep Med. 2009;5:263-76.
- 4. Ravesloot MJ, de Vries N. Reliable calculation of the efficacy of non-surgical and surgical treatment of obstructive sleep apnea revisited. *Sleep.* 2011;34:105-10.
- 5. Carberry JC, Amatoury J, Eckert DJ. Personalized Management Approach for OSA. Chest. 2018; 153:744-55.
- 6. Sharifian MR, Zarrinkamar M, Alimardani MS, Bakhshaee M, Asadpour H, Morovatdar N, et al. Drug induced sleep endoscopy in obstructive sleep apnea. *Tanaffos.* 2018;17:122-6.
- Riley RW, Powell NB, Guilleminault C, Nino-Murcia G. Maxillary, mandibular, and hyoid advancement: an alternative to tracheostomy in obstructive sleep apnea syndrome. *Otolaryngol Head Neck Surg.* 1986;94:584-8.
- 8. Butterfield KJ, Marks PL, McLean L, Newton J. Linear and volumetric airway changes after maxillomandibular advancement for obstructive sleep apnea. J Oral Maxillofac Surg. 2015;73:1133-42.
- 9. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.
- 10. Caples SM, Rowley JA, Prinsell JR. Surgical modifications of the upper airway for obstructive sleep apnea in adults: a systematic review and meta-analysis. *Sleep.* 2010;33:1396-407.
- 11. Berry R, Brooks R, Gamaldo C, Harding S, Lloyd R, Marcus C, et al. *The AASM Manual for the Scoring of Sleep and Associated Events: Rules, Terminology and Technical Specifications, version* 2.0.2. Darien, IL: American Academy of Sleep Medicine; 2013.
- 12. Sher AE, Schechtman KB, Piccirillo JF. The efficacy of surgical modifications of the upper airway in adults with obstructive sleep apnea syndrome. *Sleep*. 1996;19:156-77.
- 13. Elshaug AG, Moss JR, Southcott AM, Hiller JE. Redefining success in airway surgery for obstructive sleep apnea: a meta analysis and synthesis of the evidence. *Sleep.* 2007;30:461-7.
- 14. Higgins JP, Altman DG, Gøtzsche PC, Jüni P, Moher D, Oxman AD, et al. The Cochrane Collaboration's tool for assessing risk of bias in randomised trials. *BMJ*. 2011;343:d5928.
- 15. Moosavian SP, Arab A, Paknahad Z, Moradi S. The effects of garlic supplementation on oxidative stress markers: A systematic review and meta-analysis of randomized controlled trials. *Complement Ther Med.* 2020;50:102385.
- 16. Slim K, Nini E, Forestier D, Kwiatkowski F, Panis Y, Chipponi J. Methodological index for non-randomized studies (minors): development and validation of a new instrument. *ANZ J Surg.* 2003;73:712-6.
- 17. Shah A, Memon M, Kay J, Wood TJ, Tushinski DM, Khanna V, et al. Preoperative patient factors affecting length of stay following total knee arthroplasty: a systematic review and meta-analysis. J Arthroplasty 2019;34:2124-65.e1.
- 18. Altman DG, Machin D, Bryant TN and Gardner MJ. In: *Statistics with Confidence: confidence intervals and statistical guidelines, 2nded*. BMJ Books; 2000. p. 28-31.
- 19. Cohen J. In: Statistical Power Analysis for the Behavioral Sciences, 2nd edition. Hillsdale, NJ: Lawrence Erlbaum Associates; 1988.
- 20. Higgins JP, Thomas J. In: Cochrane handbook for systematic reviews of interventions. John Wiley & Sons, Ltd; 2019.
- 21. Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *BMJ*. 2003;327:557-60.
- 22. Neruntarat C, Chantapant S. Radiofrequency surgery for the treatment of obstructive sleep apnea: short-term and long-term results. *Otolaryngol Head Neck Surg.* 2009;141:722-6.
- 23. Altman DG, Bland JM. Interaction revisited: the difference between two estimates. BMJ. 2003; 326:219.
- 24. Bettega G, Pépin JL, Veale D, Deschaux C, Raphaël B, Lévy P. Obstructive sleep apnea syndrome: fifty-one consecutive patients treated by maxillofacial surgery. *Am J Respir Crit Care Med*. 2000;162(2 Pt 1):641-9.

- 25. Bianchi A, Betti E, Tarsitano A, Morselli-Labate AM, Lancellotti L, Marchetti C. Volumetric threedimensional computed tomographic evaluation of the upper airway in patients with obstructive sleep apnoea syndrome treated by maxillomandibular advancement. Br J Oral Maxillofac Surg. 2014;52:831-7.
- 26. Boyd SB, Walters AS, Waite P, Harding SM, Song Y. Long-Term effectiveness and safety of maxillomandibular advancement for treatment of obstructive sleep apnea. J Clin Sleep Med. 2015;11:699-708.
- 27. Conradt R, Hochban W, Brandenburg U, Heitmann J, Peter JH. Long-term follow-up after surgical treatment of obstructive sleep apnoea by maxillomandibular advancement. *Eur Respir J*. 1997;10:123-8.
- 28. Gerbino G, Bianchi FA, Verzé L, Ramieri G. Soft tissue changes after maxillo-mandibular advancement in OSAS patients: a three-dimensional study. *Journal of Cranio-Maxillofacial Surgery*. 2014;42:66-72.
- 29. Goh YH, Lim KA. Modified maxillomandibular advancement for the treatment of obstructive sleep apnea: a preliminary report. *Laryngoscope*. 2003;113:1577-82.
- 30. Goodday RH, Bourque SE, Edwards PB. Objective and subjective outcomes following maxillomandibular advancement surgery for treatment of patients with extremely severe obstructive sleep apnea (Apnea-hypopnea index >100). J Oral Maxillofac Surg. 2016;74:583-9.
- Hsieh YJ, Liao YF, Chen NH, Chen YR. Changes in the calibre of the upper airway and the surrounding structures after maxillomandibular advancement for obstructive sleep apnoea. Br J Oral Maxillofac Surg. 2014;52:445-51.
- 32. Kastoer C, Op de Beeck S, Dom M, Neirinckx T, Verbraecken J, Braem MJ, et al. Drug-induced sleep endoscopy upper airway collapse patterns and maxillomandibular advancement. *Laryngoscope*. 2020;130:E268-E274.
- 33. Li KK, Guilleminault C, Riley RW, Powell NB. Obstructive sleep apnea and maxillomandibular advancement: an assessment of airway changes using radiographic and nasopharyngoscopic examinations. J Oral Maxillofac Surg. 2002;60:526-30; discussion 531.
- 34. Li KK, Powell NB, Riley RW, Troell RJ, Guilleminault C. Long term results of maxillomandibular advancement surgery. *Sleep Breath*. 2000;4:137-40.
- 35. Li KK, Riley RW, Powell NB, Troell R, Guilleminault C. Overview of phase II surgery for obstructive sleep apnea syndrome. *Ear Nose Throat J.* 1999;78:851, 854-7.
- 36. Li KK, Troell RJ, Riley RW, Powell NB, Koester U, Guilleminault C. Uvulopalatopharyngoplasty, maxillomandibular advancement, and the velopharynx. *Laryngoscope*. 2001;111:1075-8.
- 37. Liao YF, Chiu YT, Lin CH, Chen YA, Chen NH, Chen YR. Modified maxillomandibular advancement for obstructive sleep apnoea: towards a better outcome for Asians. *Int J Oral Maxillofac Surg.* 2015;44:189-94.
- Liu SY, Huon LK, Iwasaki T, Yoon A, Riley R, Powell N, et al. Efficacy of maxillomandibular advancement examined with drug-Induced sleep endoscopy and computational fluid dynamics airflow modeling. Otolaryngol Head Neck Surg. 2016;154:189-95.
- Rubio-Bueno P, Landete P, Ardanza B, Vázquez L, Soriano JB, Wix R, et al. Maxillomandibular advancement as the initial treatment of obstructive sleep apnoea: Is the mandibular occlusal plane the key? Int J Oral Maxillofac Surg. 2017;46: 1363-71.
- 40. Veys B, Pottel L, Mollemans W, Abeloos J, Swennen G, Neyt N. Three-dimensional volumetric changes in the upper airway after maxillomandibular advancement in obstructive sleep apnoea patients and the impact on quality of life. *Int J Oral Maxillofac Surg.* 2017;46:1525-32.
- 41. Vicini C, Dallan I, Campanini A, De Vito A, Barbanti F, Giorgiomarrano G, et al. Surgery vs ventilation in adult severe obstructive sleep apnea syndrome. *Am J Otolaryngol.* 2010;31:14-20.
- 42. Vigneron A, Tamisier R, Orset E, Pepin JL, Bettega G. Maxillomandibular advancement for obstructive sleep apnea syndrome treatment: long-term results. J Craniomaxillofac Surg. 2017;45:183-91.
- 43. Wu Q, Wang Y, Wang P, Xiang Z, Ye B, Li j. The inverted-L ramus osteotomy versus sagittal split ramus osteotomy in maxillomandibular advancement for the treatment of obstructive sleep apnea patients: a retrospective study. J Craniomaxillofac Surg. 2019;47:1839-47.
- 44. Aynacı E, Karaman M, Kerşin B, Fındık MO. Comparison of radiofrequency and transoral robotic surgery in obstructive sleep apnea syndrome treatment. *Acta Otolaryngol.* 2018;138: 502-6.
- 45. Babademez MA, Ciftci B, Acar B, Yurekli MF, Karabulut H, Yilmaz A, et al. Low-temperature bipolar radiofrequency ablation (coblation) of the tongue base for supine-position-associated obstructive sleep apnea. ORLJ Otorhinolaryngol Relat Spec. 2010;72:51-5.
- 46. Benazzo M, Pagella F, Matti E, Zorzi S, Campanini A, Frassineti S, et al. Hyoidthyroidpexia as a treatment in multilevel surgery for obstructive sleep apnea. *Acta Otolaryngol.* 2008;128: 680-4.

- 47. Bostanci A, Bozkurt S, Turhan M. The relationship between the duration of obstructive respiratory events and outcomes of multilevel upper airway surgery in patients with obstructive sleep apnea. *Eur Arch Otorhinolaryngol.* 2016;273:2651-7.
- 48. Cambi J, Chiri ZM, Santis SD, Franci E, Frusoni F, Ciabatti PG, et al. Outcomes in single-stage multilevel surgery for obstructive sleep apnea: transoral robotic surgery, expansion sphincter pharyngoplasty and septoplasty. *Int J Med Robot*. 2019;15:e2034.
- 49. Cammaroto G, Montevecchi F, D'Agostino G, Zeccardo E, Bellini C, Meccariello G, et al. Palatal surgery in a transoral robotic setting (TORS): preliminary results of a retrospective comparison between uvulopalatopharyngoplasty (UPPP), expansion sphincter pharyngoplasty (ESP) and barbed repositioning pharyngoplasty (BRP). Acta Otorhinolaryngol Ital. 2017;37:406-9.
- 50. Ceylan K, Emir H, Kizilkaya Z, Tastan E, Yavanoglu A, Uzunkulaoglu H, et al. First-choice treatment in mild to moderate obstructive sleep apnea: single-stage, multilevel, temperature-controlled radiofrequency tissue volume reduction or nasal continuous positive airway pressure. Arch Otolaryngol Head Neck Surg. 2009;135:915-9.
- 51. Chen H, Wang J, Huang X, Huang Y, Lu J, Li X. Z-palatopharyngoplasty combined with 70-degree endoscopy-assisted coblator partial medial glossectomy on severe obstructive sleep apnea. *Acta Otolaryngol.* 2019;139:902-7.
- 52. Chen S, Shi S, Xia Y, Liu F, Chen D, Zhu M. Changes in sleep characteristics and airway obstruction in OSAHS patients with multi-level obstruction following simple UPPP, UPPP-GA, or UPPP-TBA: a prospective, single-center, parallel group study. ORL J Otorhinolaryngol Relat Spec. 2014;76:179-88.
- 53. Chen WJ, Xiong GF, Xiang GZ. Combined hyoid suspension with Repose system, uvulopalatopharyngoplasty and radiofrequency ablation in treatment of moderate and severe obstructive sleep apnea hypopnea syndrome. *Int J Clin Exp Med*. 2018;11:1003-8.
- 54. Chiffer RC, Schwab RJ, Keenan BT, Borek RC, Thaler ER. Volumetric MRI analysis pre- and post-Transoral robotic surgery for obstructive sleep apnea. *Laryngoscope.* 2015;125:1988-95.
- 55. Cillo JE Jr, Dalton PS, Dattilo DJ. Combined elliptical window genioglossus advancement, hyoid bone suspension, and uvulopalatopharyngoplasty decrease apnea hypopnea index and subjective daytime sleepiness in obstructive sleep apnea. J Oral Maxillofac Surg. 2013;71:1729-32.
- 56. El-Anwar MW, Amer HS, Askar SM, Elsobki A, Awad A. Could nasal surgery affect multilevel surgery results for obstructive sleep apnea? *J Craniofac Surg.* 2018;29:1897-9.
- 57. Emara TA, Omara TA, Shouman WM. Modified genioglossus advancement and uvulopalatopharyngoplasty in patients with obstructive sleep apnea. *Otolaryngol Head Neck Surg.* 2011;145:865-71.
- Eun YG, Kim SW, Kwon KH, Byun JY, Lee KH. Single-session radiofrequency tongue base reduction combined with uvulopalatopharyngoplasty for obstructive sleep apnea syndrome. *Eur Arch Otorhinolaryngol.* 2008;265:1495-500.
- 59. Friedman M, Ibrahim H, Lee G, Joseph NJ. Combined uvulopalatopharyngoplasty and radiofrequency tongue base reduction for treatment of obstructive sleep apnea/hypopnea syndrome. *Otolaryngol Head Neck Surg.* 2003;129:611-21.
- 60. Friedman M, Lin HC, Gurpinar B, Joseph NJ. Minimally invasive single-stage multilevel treatment for obstructive sleep apnea/hypopnea syndrome. *Laryngoscope*. 2007;117:1859-63.
- 61. Günbey E, Karabulut I, Karabulut H, Zaim M. Impact of multilevel surgical treatment on mean platelet volume in patients with obstructive sleep apnea syndrome. J Craniofac Surg. 2015;26:1287-9.
- 62. Hendler BH, Costello BJ, Silverstein K, Yen D, Goldberg A. A protocol for uvulopalatopharyngoplasty, mortised genioplasty, and maxillomandibular advancement in patients with obstructive sleep apnea: an analysis of 40 cases. J Oral Maxillofac Surg. 2001;59:892-7; discussion 898-9.
- 63. Li HY, Lee LA, Kezirian EJ. Efficacy of coblation endoscopic lingual lightening in multilevel surgery for obstructive sleep apnea. JAMA Otolaryngol Head Neck Surg. 2016;142:438-43.
- 64. Li HY, Lee LA, Kezirian EJ. Coblation endoscopic lingual lightening (CELL) for obstructive sleep apnea. *Eur* Arch Otorhinolaryngol. 2016;273:231-6.
- 65. Li S, Wu D, Shi H. Treatment of obstructive sleep apnea hypopnea syndrome caused by glossoptosis with tongue-base suspension. *Eur Arch Otorhinolaryngol.* 2013;270:2915-20.
- 66. Lin HC, Friedman M, Chang HW, Su MC, Wilson M. Z-palatopharyngoplasty plus radiofrequency tongue base reduction for moderate/severe obstructive sleep apnea/hypopnea syndrome. *Acta Otolaryngol.* 2010;130:1070-6.

- 67. Neruntarat, C. Genioglossus advancement and hyoid myotomy: short-term and long-term results. J Laryngol Otol. 2003;117:482-6.
- 68. Omur M, Ozturan D, Elez F, Unver C, Derman S. Tongue base suspension combined with UPPP in severe OSA patients. *Otolaryngol Head Neck Surg.* 2005;133:218-23.
- 69. Plzak J, Zabrodsky M, Kastner J, Betka J, Klozar J. Combined bipolar radiofrequency surgery of the tongue base and uvulopalatopharyngoplasty for obstructive sleep apnea. *Arch Med Sci.* 2013;9:1097-101.
- Sezen OS, Aydin E, Eraslan G, Haytoglu S, Coskuner T, Unver S. Modified tongue base suspension for multilevel or single level obstructions in sleep apnea: clinical and radiologic results. *Auris Nasus Larynx*. 2011;38:487-94.
- Sorrenti G, Piccin O, Mondini S, Ceroni AR. One-phase management of severe obstructive sleep apnea: tongue base reduction with hyoepiglottoplasty plus uvulopalatopharyngoplasty. Otolaryngol Head Neck Surg. 2006;135:906-10.
- 72. Sun X, Yi H, Cao Z, Yin S. Reorganization of sleep architecture after surgery for OSAHS. *Acta Otolaryngol.* 2008;128:1242-7.
- 73. Tantawy AA, Askar SM, Amer HS, Awad A, El-Anwar MW. Hyoid bone suspension as a part of multilevel surgery for obstructive sleep apnea syndrome. *Int Arch Otorhinolaryngol*. 2018;22:266-70.
- 74. Toh ST, Han HJ, Tay HN, Kiong KL. Transoral robotic surgery for obstructive sleep apnea in asian patients. JAMA Otolaryngol Head Neck Surg. 2014;140:624-9.
- 75. Tsou YA, Huang CW, Wu TF, Hung LW, Chang WD. The effect of tongue base suspension with uvulopalatopharyngoplasty on sleep quality in obstructive sleep apnea. *Sci Rep.* 2018;8: 8788.
- 76. Turhan M, Bostanci A, Bozkurt S. Predicting the outcome of modified tongue base suspension combined with uvulopalatopharyngoplasty. *Eur Arch Otorhinolaryngol.* 2015;272:3411-6.
- 77. Vicente E, Marín JM, Carrizo S, Naya MJ. Tongue-base suspension in conjunction with uvulopalatopharyngoplasty for treatment of severe obstructive sleep apnea: long-term follow-up results. *Laryngoscope.* 2006;116:1223-7.
- 78. Vicini C, Montevecchi F, Pang K, Bahgat A, Dallan I, Frassineti S, et al. Combined transoral robotic tongue base surgery and palate surgery in obstructive sleep apnea-hypopnea syndrome: expansion sphincter pharyngoplasty versus uvulopalatopharyngoplasty. *Head Neck*. 2014;36:77-83.
- 79. Wang L, Liu JX, Yang XL, Yang CW, Qin YX. Z-palatoplasty and tongue radiofrequency for patients with small tonsils. *Otolaryngol Head Neck Surg.* 2013;148:873-7.
- 80. Yi HL, Sun XQ, Chen B, Zhang YJ, Guan J, Wu HM, et al. Z-palatopharyngoplasty plus genioglossus advancement and hyoid suspension for obstructive sleep apnea hypopnea syndrome. *Otolaryngol Head Neck Surg.* 2011;144:469-73.
- 81. Yüksel A, Ugur KS, Kizilbulut G, Ark N, Kurtaran H, Kaya M, et al. Long-term results of one staged multilevel surgery with tongue suspension surgery or one level palatal surgery for treatment of moderate and severe obstructive sleep apnea. *Eur Arch Otorhinolaryngol.* 2016;273:1227-34.
- 82. Rangabashyam M, Huang W, Hao Y, Han HJ, Loh S, Toh ST. State of the art transoral robotic surgery for obstructive sleep apnea-hypopnea syndrome. *Robot Surg.* 2016;3:13-28.
- Kezirian EJ, Hohenhorst W, de Vries N. Drug-induced sleep endoscopy: the VOTE classification. Eur Arch Otorhinolaryngol. 2011;268:1233-36.
- 84. Rashwan MS, Montevecchi F, Cammaroto G, Badr El Deen M, Iskander N, El Hennawi D, et al. Evolution of soft palate surgery techniques for obstructive sleep apnea patients: A comparative study for single-level palatal surgeries. *Clin Otolaryngol.* 2018;43:584-90.
- 85. Strollo PJ Jr, Soose RJ, Maurer JT, de Vries N, Cornelius J, Froymovich O, et al. Upper-airway stimulation for obstructive sleep apnea. N Engl J Med. 2014;370:139-49.
- 86. Costantino A, Rinaldi V, Moffa A, Luccarelli V, Bressi F, Cassano M, et al. Hypoglossal nerve stimulation long-term clinical outcomes: a systematic review and meta-analysis. *Sleep Breath*. 2020;24:399-411.
- Pottel L, Neyt N, Hertegonne K, Pevernagie D, Veys B, Abeloos J, et al., Long-term quality of life outcomes of maxillomandibular advancement osteotomy in patients with obstructive sleep apnoea syndrome. *Int J Oral Maxillofac Surg.* 2019;48:332-40.
- 88. Lin HC, Lee CY, Friedman M, Wang PC, Salapatas AM, Lin MC, et al., Effects of minimally invasive surgery for patients with OSA on quality of life. *Ann Otol Rhinol Laryngol*. 2018;127:118-23.

- Camacho M, Noller MW, Del Do M, Wei JM, Gouveia CJ, Zaghi S, et al. Long-term results for maxillomandibular advancement to treat obstructive sleep apnea: a meta-analysis. Otolaryngol Head Neck Surg. 2019;160:580-93.
- 90. Hou J, Yan J, Wang B, Zhu K, Sheng Y, Zheng G, et al. Treatment of obstructive sleep apnea-hypopnea syndrome with combined uvulopalatopharyngoplasty and midline glossectomy: outcomes from a 5-year study. *Respir Care.* 2012;57:2104-10.
- 91. Andsberg U, Jessen M. Eight years of follow-up--uvulopalatopharyngoplasty combined with midline glossectomy as a treatment for obstructive sleep apnoea syndrome. *Acta Otolaryngol Suppl.* 2000;543:175-8.
- 92. Riley RW, Powell NB, Guilleminault C, Pelayo R, Troell RJ, Li KK. Obstructive sleep apnea surgery: risk management and complications. *Otolaryngol Head Neck Surg.* 1997;117:648-52.
- Van Sickels JE, Hatch JP, Dolce C, Bays RA, Rugh JD. Effects of age, amount of advancement, and genioplasty on neurosensory disturbance after a bilateral sagittal split osteotomy. J Oral Maxillofac Surg. 2002;60:1012-7.
- 94. Pang KP, Siow JK, Tseng P. Safety of multilevel surgery in obstructive sleep apnea: a review of 487 cases. Arch Otolaryngol Head Neck Surg. 2012;138:353-7.
- 95. Pang KP. Identifying patients who need close monitoring during and after upper airway surgery for obstructive sleep apnoea. J Laryngol Otol. 2006;120:655-60.
- 96. Hogan PW, Argalious M. Total airway obstruction after maxillomandibular advancement surgery for obstructive sleep apnea. *Anesth Analg.* 2006;103:1267-9.


CHAPTER

Maxillomandibular advancement and upper airway stimulation for treatment of obstructive sleep apnea: A systematic review

Ning Zhou, Jean-Pierre T.F. Ho, René Spijker, Ghizlane Aarab, Nico de Vries, Madeline J.L. Ravesloot, Jan de Lange

Published in revised form: J Clin Med, 2022. doi: 10.3390/jcm11226782

ABSTRACT

This systematic review aimed to comparatively evaluate the efficacy and safety of maxillomandibular advancement (MMA) and upper airway stimulation (UAS) in obstructive sleep apnea (OSA) treatment. A MEDLINE and Embase databases search of articles on MMA and/or UAS for OSA was conducted. Twenty-one MMA studies and nine UAS studies were included. All the MMA studies demonstrated a reduction in apnea hypopnea index (AHI) postoperatively and success rates ranged from 41.1% to 100%. Ten MMA studies reported pre- and postoperative Epworth sleepiness scale (ESS), and all but one studies demonstrated a reduction in ESS. In the UAS studies, all but one demonstrated a reduction in AHI and success rates ranged from 26.7% to 77.8%. In the eight UAS studies reporting pre- and postoperative ESS, an ESS reduction was demonstrated. No studies reported any deaths related to MMA or UAS. The most common postoperative complication after MMA and UAS was facial paresthesia in mandibular area and discomfort due to electrical stimulation, respectively. This systematic review suggests that both MMA and UAS are effective and generally safe therapies for OSA. However, due to the limitations of the included studies, there is no evidence yet to directly compare these two procedures in OSA treatment.

Keywords: Obstructive sleep apnea; Therapy; Maxillo-mandibular surgery; Hypoglossal nerve; Systematic review

INTRODUCTION

Obstructive sleep apnea (OSA) is a prevalent sleep-related breathing disorder characterized by recurrent upper airway obstruction during sleep¹, and its overall prevalence ranges from 9% to 38% in the general adult population². OSA is associated with considerable health risks, such as cardiovascular and cerebrovascular disease^{3.4}. Continuous positive airway pressure (CPAP) is accepted as the first-line therapy for moderate to severe OSA, but poor compliance and suboptimal use of CPAP drive OSA patients to seek alternative therapies, including other non-invasive therapies and surgical treatment^{5, 6}.

Moderate to severe OSA is usually caused by multilevel obstructions of the upper airway, which highlights the need for surgical therapies able to resolve multilevel upper airway collapse⁷. One such therapy that has existed for many decades is maxillomandibular advancement (MMA)^{8, 9}. MMA is a multilevel skeletal surgery in which the maxilla and mandible are advanced by a combination of a Le Fort I osteotomy of the maxilla and a bilateral sagittal split osteotomy of the mandible^{8, 9}. By expanding the skeletal framework attached with the pharyngeal soft tissues, MMA enlarges the velo-orohypopharyngeal airway¹⁰ and increases the tension of the pharyngeal soft tissues, decreasing the collapsibility of the upper airway¹¹. MMA is currently considered as the most effective surgical treatment modality for moderate to severe OSA in adults aside from tracheostomy.

A more contemporary therapy is hypoglossal nerve stimulation (HNS), which works by electrically stimulating the branches of the hypoglossal nerve that innervate muscles responsible for protruding the tongue and thus maintaining upper airway patency during sleep¹². Currently, there are three different systems for HNS therapy, including the Aura6000 Targeted Hypoglossal Neurostimulation system (LivaNova PLC, London, England, UK), the GenioTM system (Nyxoah SA, Mont-Saint-Guibert, Belgium), and the Inspire II upper airway stimulation (UAS) system (Inspire Medical Systems, Maple Grove, MN, USA)¹³. Given that the Inspire UAS system is the most widely used system having Food and Drug Administration (FDA) approval for clinical use¹⁴, this review only focused on the UAS therapy (Inspire[®] system). Over the past decade, UAS has emerged as an effective therapy and therefore has become an increasingly popular treatment option for moderate to severe OSA^{15, 16}. Chapter 6

Currently, the main indications for MMA are moderate to severe OSA, and mild OSA in patients presenting with a dentofacial deformity¹⁷. UAS therapy is generally indicated for patients with the following characteristics: moderate to severe OSA (apnea hypopnea index [AHI] 15-65 events/h with <25% central or mixed apneas), positive airway pressure (PAP) therapy failure, and absence of complete concentric velum collapse (CCCp) on drug-induced sleep endoscopy (DISE)¹⁸. When no generally accepted indicative results are found during clinical, laboratory, or endoscopic examinations (e.g., significant skeletal-dental deformity, AHI > 65 events/h, CCCp on DISE), patients with moderate to severe OSA may be expected to benefit from MMA as well as UAS therapy. Although MMA and UAS have both demonstrated efficacy and safety for patients, there is a paucity of evidence on comparison of these two treatment options¹⁷.

Therefore, the purpose of this systematic review was to comprehensively evaluate and compare the efficacy of MMA and UAS for moderate to severe OSA, through the assessment of AHI and Epworth sleepiness score (ESS) as primary outcomes. Secondly, the postoperative complications of these two therapies were investigated.

METHODS

This systematic review was performed in accordance with the preferred reporting items for systematic review and meta-analysis (PRISMA) statement¹⁹. The protocol for this system-atic review was registered at PROSPERO (PROSPERO ID: CRD42021261394; https://www.crd.york.ac.uk/PROSPERO/display_record. php?ID=CRD42021261394).

Selection criteria

The inclusion criteria were as follows: (1) adult patients (> 18 years old) with moderate to severe OSA diagnosed by polysomnography (PSG; AHI \geq 15 events/h); (2) patients who underwent MMA or UAS for OSA; (3) studies that reported pre- and postoperative PSG data; (4) studies with a follow-up \geq 6 months; (5) study designs: randomized controlled trials (RCTs), quasi-experimental studies, and cohort studies; and (6) English language.

The exclusion criteria were the following: (1) sample size < 10 patients; (2) patients who underwent other adjunctive surgical procedures (e.g., uvulopalatopharyngoplasty) at the time of MMA or UAS; and (3) preliminary studies in which the findings had been nested in other studies with larger sample size and/or longer follow-up.

Literature search

A literature search was performed with the help of an information specialist (RS) using MEDLINE and Embase databases on Dec 14, 2021. Search terms and search strategies used for each database are available in supplementary materials (**Table S1a**).

Study selection

After removal of duplicate articles, the remaining results were screened based on title and abstract by two independent reviewers (NZ and JH). The full texts of potentially relevant articles were retrieved and further evaluated by NZ and JH independently for compliance of studies with the eligibility criteria. Discrepancies were resolved by discussion. Reference lists of eligible studies were checked for additional studies.

Data extraction

The extracted data included: article title, year of publication, first author, study design, specific surgical technique, length of follow-up, sample size, age, gender, body mass index (BMI), preoperative and postoperative PSG data (AHI, respiratory disturbance index [RDI], and oxygen desaturation index [ODI]), preoperative and postoperative ESS score, preoperative and postoperative data on quality of life (QoL), surgical success rate and cure rate, and postoperative complications. According to the accordion severity grading system of surgical complications²⁰, the postoperative complications were classified as major and minor, depending on the needs for endoscopic or interventional radiologic procedures or re-operation as well as failure of one or more organ systems.

Data were extracted by NZ and JH independently. Discrepancies were resolved through discussion. If RDI is reported by a study, it would be extracted as AHI, since these two respiratory parameters have been consolidated based on the 2013 American Academy of Sleep Medicine's manual for the scoring of sleep and associated events²¹. If there were multiple follow-up data in a study, the data with longest follow-up time were included. Surgical success was defined as "a postoperative AHI < 20 events/h

and at least 50% reduction in AHI after surgery"²², and surgical cure was defined as "a postoperative AHI < 5 events/h"²³.

Quality assessment

Methodologic quality assessment of each study was performed by NZ and JH independently, and any discrepancies were resolved by discussion.

The Methodological Index for Non-Randomized Studies (MINORS) quality assessment tool, a validated tool for the methodological assessment of nonrandomized surgical studies²⁴, was used to assess the methodological quality of the included studies. The MINORS tool is composed of eight items applicable to all nonrandomized studies and four additional items specifically for comparative studies. Each item was scored as 0 (not reported), 1 (reported but inadequate), or 2 (reported and adequate), giving a global ideal score of 24 for comparative studies and 16 for non-comparative studies. For comparative studies, the categorizations are as follows: 0-6, very low quality; 7-10, low quality; 11-15 fair quality; and \geq 16, high quality. For non-comparative studies, the categorizations are as follows: 5-7, low quality; 8-12, fair quality; and \geq 13, high quality²⁵.

Statistical analysis

The collected parameters (age, BMI, AHI, ODI, and ESS) were pooled by weighted average and weighted standard deviation²⁶. When there were RCTs or comparative studies between MMA and UAS, meta-analyses were performed to compare the overall effect of MMA and UAS in treating OSA. Heterogeneity of the studies was assessed by I² statistic with cut-off of 25% (low), 50% (moderate) and 75% (high)²⁷. When moderate to high heterogeneity was present, a random-effects model was adopted; otherwise, a fixed-effects model was used. Because some patients may report multiple complications, the complication rate of each study was calculated by dividing the number of events by the number of patients.

RESULTS

Search results

The flow diagram of study selection progress is summarized in **Figure 1**. A total of 2952 studies were screened after deduplication, 212 were retrieved for full-text review.

MMA group Twenty-one studies^{11, 28-47} were identified, producing a pooled data set of 581 patients (male 78.5%) with weighted age of 42.2 ± 11.5 years and weighted BMI of 28.1 \pm 6.4 kg/m². The mean follow-up period from surgery to final postoperative PSG was 25.9 months (range, 6 months-12.5 years). One study³⁹ was excluded from the analyses for clinical efficacy, because the data of a subset of the patients with longer follow-up period were nested in another included study³⁸. The characteristics of these studies are shown in **Table 1**.

UAS group Nine studies^{15, 48-55} were identified, yielding a total of 1029 patients (male 96.2%) with weighted age of 55.1 \pm 10.1 years and weighted BMI of 29.1 \pm 4.2 kg/m². The mean follow-up period was 18.8 months (range, 6 months-5 years). The characteristics are summarized in **Table 2**.

Because there was no RCT or comparative study of MMA and UAS in treating OSA, a meta-analysis could not be performed to compare their overall effect sizes on OSA.

Quality assessment

MMA group One of the included studies was an RCT of MMA and autotitrating positive airway pressure (APAP), one was a retrospective quasi-experimental study, ten were prospective cohort studies, and nine were retrospective cohort studies. As only MMA cohort of the RCT was included in the analyses, after omitting the unrequired APAP cohort, this study was regarded as a single-arm trial. The quality of the RCT was therefore assessed using MINORS tool as per the other included studies. Of these studies, three studies were classified as "high quality", and the others were classified as "fair quality" (Supplementary **Table Sza**).

UAS group Six prospective studies and three retrospective studies were included. Of these, one study was classified as "high quality", and eight studies as "fair quality" (Supplementary **Table S2b**).

Respiratory parameters

MMA group Fifteen MMA studies^{11,28-31,33-37,41,42,44,45,47} reported a significant reduction in AHI postoperatively (P < 0.05), the others^{32,38,40,43,46} reported an AHI reduction but did not report a *P*-value. All the studies^{11,28-38,40-47} totaling 446 patients demonstrated a weighted baseline AHI of 54.6 ± 27.4 events/h and a weighted postoperative AHI of 10.1 ± 10.8 events/h. In four studies^{11,32,36,43} (n = 78) reporting pre- and postoperative ODI, two demonstrated a significant reduction in ODI after MMA (P < 0.05), and the other two also reported a ODI reduction but without a P-value. The weighted pre- and postoperative ODI was 35.1 ± 22.8 events/h and 6.3 ± 6.4 events/h, respectively.

UAS group Of the selected studies, the study form Bachour et al.⁵⁵ did not show a significant reduction in AHI postoperatively, five studies^{48-51, 54} demonstrated a significant reduction in AHI postoperatively (P < 0.05), and three studies^{15, 52, 53} showed a AHI reduction but did not report a *P*-value. The weighted pre- and postoperative AHI in 1003 patients was 35.2 ± 14.7 events/h and 15.0 ± 16.1 events/h, respectively.

Of six studies^{15, 49-52, 55} reporting pre- and postoperative ODI, the study from Bachour et al. ⁵⁵ did not found a significant improvement in ODI postoperatively, while the others^{15, 49-52} reported a reduction in ODI after surgery, of which two studies did not report a *P*-value. The weighted pre- and postoperative ODI was 26.5 ± 16.0 events/h and 14.6 ± 18.5 events/h (n = 180), respectively.

Subjective parameters

MMA group Of nine studies^{11, 34, 36, 41-45, 47} (n = 217) reporting pre- and postoperative ESS, the study from Lin et al. did not show an improvement in ESS after MMA, one study demonstrated a reduction in ESS but without a *P*-value, and the others reported a significant reduction in ESS (P < 0.05). The weighted pre- and postoperative ESS was 13.1 ± 5.5 and 6.7 ± 4.8, respectively.

Three studies^{30, 42, 44} assessed pre- and postoperative QoL. Boyd et al. found that after MMA there was a significant improvement in Functional Outcomes of Sleep Questionnaire (FOSQ) (P < 0.05)³⁰. Veys et al. assessed subjective outcome of MMA using OSA QoL questionnaire. They found that there was an improvement in all of the following six symptoms after MMA: daytime sleepiness, snoring, concentration, waking up at night, headache, and high blood pressure, while the influence of MMA on nocturia and sexual activity was variable⁴⁴. Lin et al. found that there was no significant improvement in Short-Form 36 Quality of Life (SF-36) after MMA⁴².

UAS group Of eight studies^{15, 49-55} reporting pre- and postoperative ESS, seven demonstrated a significant reduction in ESS postoperatively (P < 0.05), and one reported a ESS reduction but did not report a *P*-value. The weighted pre- and postoperative ESS was 11.4 ± 5.4 (n = 1006) and 7.0 ± 4.6 (n = 1001), respectively.

Two studies reported pre- and post-UAS FOSQ score. The STAR trial cohort demonstrated an increase of FOSQ score five years after surgery (14.3 \pm 3.3 to 18.0 \pm 2.2). Van de Heyning et al. also found a significant improvement in FOSQ score postoperatively (89.1 \pm 23.5 to 100.8 \pm 16.9, *P* < 0.05).

Surgical success and cure

MMA group Surgical success rate of MMA was available in 15 studies^{11, 28, 32-35, 37, 38, 40, 41, 43-47}, which ranged from 41.1% to 100%. Surgical cure rate of MMA was reported in seven studies^{11, 34, 42-45, 47}, which ranged from 36% to 67.9%.

UAS group Surgical success rate of UAS was available in six studies^{15, 50-52, 54, 55}, ranging from 26.5% to 77.8%. Surgical cure rate was reported in four studies^{15, 50, 51, 55}, which ranged from 6.7% to 44%.

Long-term follow-up outcomes

MMA group Five studies^{30, 31, 38, 42, 46} reported long-term follow-up (\geq 2 years) data in 151 patients with weighted baseline AHI of 51.7 ± 28.2 events/h. At a mean follow-up of 5.0 years, the weighted postoperative AHI was 11.1 ± 13.0 events/h. Only one study⁴² with 53 patients reported long-term follow-up ESS (10.8 ± 5.0 to 10.2 ± 5.1, *P* > 0.05). Boyd et al.³⁰ reported a long-term improvement in FOSQ score after MMA. Surgical success rate was reported in two studies^{38, 46} (90% and 41.4%, respectively), and surgical cure rate was only available in one study⁴² (67.9%).

UAS group Three studies^{15, 50, 51} reported long-term follow-up (\geq 2 years) data in 127 patients with weighted baseline AHI of 29.7±11.0 events/h. At a mean follow-up of 4.2 years, the weighted postoperative AHI was 12.3±14.8 events/h. These three studies^{15, 50, 51} also reported a long-term improvement in ODI and ESS after UAS therapy. One study¹⁵ reported a long-term (five years follow-up) improvement in FOSQ score. Surgical success and cure rates were reported in all three studies^{15, 50, 51} (success rate: 77.8%, 71.1%, and 74.6%, respectively; cure rate: 33.3%, 35%, and 44%, respectively).

Safety

There were no studies reporting any deaths related to MMA or UAS surgery.

MMA group Of the included studies, ten reported participants' complications after MMA (n = 428)^{28, 30, 33, 39, 42-47}. The rate of major complication ranged from 0 to 18%. Five

studies reported the major compilations after MMA, which included reoperations for removal of osteosynthesis screws and plates $(n = 8)^{30, 33, 46}$, reoperations for maxillary non-union $(n = 2)^{28, 46}$, and acute dyspnea $(n = 1)^{45}$.

The most common minor complication reported was facial paresthesia caused by the impairment of inferior alveolar nerve^{30, 33, 39, 43, 45-47}. Four studies^{39, 45-47} reported both the rates of transient and persistent paresthesia in mandibular area, which were 100% and 13% (n = 175), 100% and 28% (n = 25), 90% and 60% (n = 34), and 32% and 0% (n = 28), respectively. Besides, one study⁴³ (n = 34) reported only the rate of transient paresthesia in mandibular area – 75%; one study³³ (n = 11) reported only the rate of the persistent symptom – 27%. In the long-term follow-up study from Boyd et al.³⁰ (n = 30), although no patients exhibited such facial anesthesia as measured objectively, 40% of patients perceived a decrease in sensation subjectively. Facial paresthesia in infraorbital area was reported by two studies^{45, 46}. In the study by Vicini et al.⁴⁵ (n = 25), the rates of transient and persistent paresthesia in infraorbital area was reported by two studies^{45, 46}. In the study by Vicini et al.⁴⁵ (n = 25), the rates of transient and persistent paresthesia in infraorbital area % reported by two studies^{45, 46}. In the study by Vicini et al.⁴⁵ (n = 25), the rates of transient and persistent paresthesia in infraorbital area % reported by two studies^{45, 46}. In the study by Vicini et al.⁴⁵ (n = 34), respectively; in the study by Vigneron et al.⁴⁶ (n = 34), they were 37% and 30%, respectively.

Excluding facial paresthesia, the other reported minor complications consisted of developed malocclusion^{30, 45-47} (n = 13), temporomandibular disorders^{46, 47} (n = 11), local infection^{28, 30, 47} (n = 6), minor postoperative wound pain³³ (n = 2), and others (n = 5)^{28, 44, 47}. Of ten studies^{28, 30, 32, 41-47} that investigated patients' perception of their facial appearance after MMA, two studies^{30, 46} reported that there were 13% (4/30) and 15% (5/34) patients who perceived worsening of their facial appearance after MMA, respectively; the others^{28, 32, 41-45, 47} reported that the perception of facial appearance was positive or neutral in all the patients after MMA.

UAS group Of the five studies reporting patients' complications (n = 2051)^{15, 49, 51, 52, 54}, the rate of serious device-related adverse events range from 0 to 7%. Four studies^{15, 51, 52, 54} reported a total of 50 serious device-related adverse events requiring surgical repositioning or replacement of the neurostimulator or implanted leads. In addition, in the study from Suurna et al.⁵⁴ (n = 1849), 0.4% of the patients reported intraoperative serious adverse events including but not limited to hematoma (n = 8), infection (n = 2), extra implant procedure (n = 1), intraoperative arrest (n = 1), and pneumothorax (n = 1).

Since one study⁵⁴ did not report the count of minor complications, the safety outcomes of a subset of the study population (ADHERE cohort) reported in a previous study⁵⁶ were used for analyzing the minor complication rate. In that study⁵⁶, the rates of minor surgery-related and device-related complication 137 ± 77 days after UAS implant were 6% (18/313) and 22% (69/313), respectively; 386 ± 136 days after UAS implant were 4% (8/217) and 24% (53/217), respectively. In STAR trial cohort¹⁵ consisting of 126 participants, the rates of minor surgery-related and device-related complication were both 136% (171/126) at the first year; at the fifth year, they were decreased to 1% (1/126) and 16% (20/126), respectively. Van de Heyning et al.⁵² reported only minor surgery-related adverse events in their population, which yielded a minor complication rate of 57% (16/28). Philip et al.⁴⁹ and Steffen et al.⁵¹ did not report any minor complications in their study populations. The most common minor surgery-related and device-related complication was incision discomfort^{15, 51, 56}, respectively.

DISCUSSION

This is the first systematic review aiming to comparatively evaluate MMA and UAS therapy in treating OSA. We reviewed 21 studies on MMA and nine studies on UAS in treating OSA. Due to the fact that there is no RCT or comparative study of MMA and UAS, a meta-analysis cannot be performed to directly compare these two interventions. Separate analyses of studies on MMA and UAS were utilized for this review. It should be noted that UAS therapy has stricter inclusion criteria (e.g., $15 \le AHI \le 65$ events/h, absence of CCCp during DISE)^{14, 17} for patients than MMA. There is therefore discrepancy of patients' baseline characteristics between the MMA cohort and UAS cohort. In this review, MMA cohort has younger age and higher baseline AHI compared to UAS cohort. To obtain definitive results on the comparison of MMA and UAS, future studies should include comparative studies of these two therapies where participants would have comparable baseline characteristics and be qualified for both therapies.

Objective outcomes

Based on the separate analysis of studies on MMA and UAS, we reported that these two procedures are both effective treatment modalities for OSA. However, compared to UAS, MMA seems to be more effective in treating OSA with a more significant decrease

in AHI and higher success rate. Through different mechanisms, MMA and UAS have been proven to be able to address multiple sites of collapse simultaneously^{11, 36}. MMA enlarges the entire pharynx and reduces the collapsibility of the upper airway by advancing the maxillomandibular complex and anterior pharyngeal tissues attached to the maxilla, mandible, and hyoid bone³⁹. The mechanism by which UAS resolves multilevel collapse, is enlargement of the retropalatal airway associated with tongue protrusion, which is so called "palatoglossus coupling" phenomenon⁴⁸. Safiruddin et al. found that the retropalatal enlargement in response to UAS was statistically significant only in the responders, while the responders and non-responders had similar degrees of retro-lingual opening to stimulation⁵⁷. Therefore, we are of opinion that the superiority of MMA over UAS in OSA treatment may be associated with the ability of MMA to enlarge the retropalatal airway more significantly. To improve patient selection for MMA and UAS, the mechanism of action of these two surgical procedure and the role of pathogenesis of OSA on the outcome of both surgeries need to be further clarified in future studies.

Subjective outcomes

It is interesting to note that several studies^{42, 55} reported a discordance between objective outcome measures (e.g., AHI) and patient-reported outcome measures, which highlights the importance of subjective outcomes evaluation for OSA patients. In contrast to published ESS data, there is a scarcity of evidence related to other subjective outcomes of surgical treatment for OSA. Boyd et al.³⁰ evaluated the impact of MMA on guality of life (QoL) by Functional Outcomes of Sleep Questionnaire (FOSQ). At two years after MMA, a significant improvement in mean FOSQ scores of 4.7 was observed. In a study by Woodson et al.¹⁵, the improvements in mean FOSQ scores following UAS were 3.0 at 1 year and 3.7 at 5 years, respectively. In addition to the daytime sleepiness and QoL, patient satisfaction – an important measure of therapy quality – should be noted when evaluating treatment options for OSA. Currently only a few studies have evaluated the patient satisfaction with MMA or UAS for the management of OSA^{56, 58-62}. In a study by Butterfield et al.⁵⁹, 95.5% of patients were satisfied with MMA surgery for OSA, 90.9% would repeat the procedure, and 86.4% would recommend MMA to others for OSA treatment. In the ADHERE registry, 94% of patients reported that they were satisfied with UAS therapy and would undergo UAS again, and 93% reported that they would recommend UAS to others⁵⁶. According to the available evidence, both MMA and UAS could significantly improve the perception for OSA patients, with high level of patient satisfaction. However,

the comparison of improvement in patient-perceived measures between the two therapies needs to be addressed in future studies.

Long-term outcomes

The long-term follow-up period of the included MMA studies ranges from 2 years to 12.5 years. Because of the small sample size, one study by Pottel et al.⁶³ reporting the longest follow-up result of MMA was excluded. In that study, the short term (within 2 years) success rate was 66.67% (8/12), and the long-term (range 14-20 years) success rate of MMA was 44.44% (4/9). In a study of 29 OSA patients treated by MMA, Vigneron et al.⁴⁶ concluded that the success rate was 85.7% in the immediate postoperative period and 41.1% at 12.5 years. Besides, they concluded that the good candidates for long-term success of MMA were the young patients (< 45 years old), with BMI < 25kg/m², AHI < 45 events/h, SNB angle < 75° , narrow retrolingual space (< 8 mm), and preoperative orthodontics, and without co-morbidity. It has been suggested that long-term failure of MMA might be attributed to weight gain^{38, 63,} ⁶⁴, skeletal relapse⁶⁴, and ageing⁶³. Given that UAS is an innovative therapy for OSA during the last decade, the longest follow-up period of the UAS studies was 5 years from STAR trial¹⁵. The success rates of UAS in STAR trial cohort were 66% (83/126), 74% (73/98), 75% (53/71) at 1, 3, and 5 years, respectively. In UAS therapy for OSA treatment, patients' adherence is necessary to guarantee the clinical efficacy⁶⁵. The STAR trial revealed a high adherence to UAS therapy in long-term, with the patient self-reported nightly device use of 80% at 5 years, which might partially explain the stability of treatment effect. In addition, lower baseline ODI was found to be predictive of 5-year response to UAS therapy. It is therefore concluded that both MMA and UAS were relatively stable treatment for patients with moderate to severe OSA. In order to maintain the clinical efficacy, more effort is needed to provide continuous follow-up for OSA patients and to ascertain the factors associated with long-term stability of outcomes.

Safety

In terms of treatment safety, this systematic review revealed that both MMA and UAS were generally safe surgical procedure for OSA, with a relatively low rate of major complication. In the included MMA studies, all but one of the major complications were reoperation for removal of hardware. Age has been shown to be a risk factor for increased need for hardware removal⁶⁶. In addition, Passeri et al. found that patients who were active smokers or have a history of smoking had higher risk of

complications, among which included removal of hardware⁶⁷. The most common minor complication of MMA detailed in literatures was paresthesia of the lower lip and chin. It has been suggested that age at the time of surgery and addition of a genioplasty increase the risk of facial paresthesia, and large degree of advancement further increase the risk in older patients^{68, 69}. In the STAR cohort (n = 126), the rates of major complication requiring device explanation, reposition or replacement were 4% at 4 years and 9.5% at 5 years, indicating that the reoperations after UAS may occur more often during the late time frame. The STAR cohort also suggested that the majority of minor complications after UAS were gradually resolved. Of note, Withrow et al. evaluated the impact of age on safety of UAS and found no significant difference between younger and older cohorts in complication rates⁷⁰. The current evidence suggests that both MMA and UAS appear to be a safe approach in OSA treatment, while compared to MMA treating OSA with UAS may lead to less complications for older patients.

Clinical relevance

In patients with moderate to severe OSA and failure of CPAP treatment, a portion of them could qualify for both MMA and UAS therapy. The current evidence shows that MMA may have superior efficacy in OSA treatment. However, MMA is a more invasive intervention exposing patients to longer recovery time and higher risk of postoperative complications. An overnight admission of intensive care unit is required for OSA patients following MMA surgery, and the length of hospitalization after MMA reported previously ranged from <2 days to 5-8 days⁶⁹. Additionally, MMA surgery often involves time-consuming preoperative and/or postoperative orthodontic work. One notable potential problem with MMA has been the accompanying alteration in facial appearance; however, most of patients undergoing MMA for OSA view the change in facial appearance as neutral even positive^{30, 32, 46}. In comparison to MMA, UAS surgery is less invasive and more patient-friendly and does not require extended recovery. The majority of patients are discharged the same day or one day after UAS surgery⁷¹. In addition to the information regarding treatment efficacy and safety, cost of treatment options is important in assisting decision-making in OSA treatment. It has been indicated that UAS is cost effective, with a lifetime incremental cost effectiveness ratio (ICER) of USD 39,471 per quality-adjusted life year (QALY) in the United States healthcare system⁷² and EUR 44,446 per QALY in a European setting⁷³, respectively. However, to our knowledge, no study has assessed the cost-effectiveness of MMA, which precludes the comparison of cost-effectiveness between these two therapies. Hence, to further assist decision making in OSA treatment, there is a need to assess and compare the costs and cost-effectiveness of each intervention.

While the primary target patient population differs between MMA and UAS, it has been proposed that these two procedures might be considered as complementary therapies¹⁷. For example, the UAS may be considered when a patient fails to respond to MMA. It is interesting to note that in a recent study⁷⁴, Sarber et al. evaluated the efficacy of UAS therapy in 18 OSA patients who did not meet all FDA criteria for UAS and found a promising treatment outcome. They suggested that future studies need to consider the expansion of current FDA criteria for UAS, particularly in BMI and AHI criteria. Thus, to optimize surgical outcome, reduce rate of mortality and morbidity, and improve quality of life and other subjective outcomes, further investigation is essential to clarify indications of each therapy for OSA.

Limitations

There are several limitations of the present review. Firstly, because of the inherent difficulty of randomizing patients to different surgical interventions or sham surgery⁷⁵, except for one RCT and one quasi-experimental trail, all the included studies were cohort studies, the majority of which demonstrated fair quality according to the MINORS tool. Due to the lack of RCT and comparative studies of MMA and UAS for OSA, a meta-analysis cannot be performed to directly compare these two procedures. Besides, meta-analyses were not conducted to separately assess overall effect sizes of MMA and UAS therapy on OSA, as mean and SD of the difference between pre- and postoperative measures were absent in majority of the selected studies. In this review we performed separate analyses for MMA and UAS studies, combined with noticeable differences between the two cohorts in age and OSA severity, which prevented us from generating a solid conclusion on the comparison of these two procedures. Due to the fact that some patients may fall between two stools, comparison of the two procedures is important. Future studies should include quasi-experimental trial and comparative cohort study comparing MMA and UAS to better clarify which modality is superior in OSA treatment. These studies can be part of a future large international consortium, which is more likely to generate solid conclusions. Secondly, due to the implemented inclusion criteria, which included the presence of both preoperative and postoperative PSG data, some well-conducted studies reporting on only subjective outcomes and/or safety were excluded for this study. Therefore, the present analysis of subjective outcomes and safety may not be entirely representative of the population undergoing MMA or UAS in current literature. Lastly, our review is exclusively based on studies published in English, which can introduce a language bias⁷⁶.

CONCLUSION

The results presented in this review suggest that both MMA and UAS are effective and generally safe surgical treatment modalities for patients with moderate to severe OSA. However, within the limitation of the selected studies, there is currently no evidence on the comparison of MMA and UAS in the treatment of OSA.



Figure 1. PRISMA flow diagram of the study selection process

Study	Design	Ν	Age (years)	% Male	Degree of adv (mm) (me	Follow-up (mean <u>+</u> SD)	
			(mean <u>+</u> SD)		Max	Mand	
Bettega et al. 2000	Retro	20	44.4 <u>+</u> 10.6	90	11.8 <u>+</u> 0.5	11.8 <u>+</u> 0.5	6m
Bianchi et al. 2014	Retro	10	45±14	100	10	10	6m
Boyd et al. 2015	Pro	14			7.0 <u>±</u> 2.3	9.2 <u>+</u> 3.3	6.6 <u>+</u> 2.8y
Conradt et al. 1997	Retro	15	44±12	93.3			>2y
Gerbino et al. 2014	Pro	10	44.9		9.2 <u>+</u> 1.2	10.4 <u>+</u> 2.2	6m
Goh et al. 2003	Pro	11	42.8 <u>+</u> 8.2	100	10	10	7.7m
Goodday et al. 2016	Retro	13	37.8 <u>+</u> 8.6	84.6			9.6m
Hsieh et al. 2014	Pro	16	33±7.9	75			12 <u>+</u> 8m
Kastoer et al. 2019	Pro	14	51.1 <u>+</u> 7.3	57.1			6m
Li et al. 1999	Retro	175	43.5±11.5	83			6m
Li et al. 2000	Retro	40	45.6 <u>+</u> 20.7	82.5	10.8 <u>+</u> 2.7	10.8 <u>+</u> 2.7	4.2±2.7y
Li et al. 2001	Retro	52	46.6 <u>+</u> 6.7	82.7	10.5 <u>+</u> 1.5		6m
Li et al. 2002	Pro	12	47.3±9.8	75	10.5 <u>+</u> 1.2	10.5 <u>+</u> 1.2	6m
Liao et al. 2015	Pro	20	33.4±6.5	85			14±9.3m
Lin et al. 2020	Pro	53	35.7±11.7	75.7	4.3±2.9	13.3±3.8	24m
Liu et al. 2016	Retro	20	44 <u>+</u> 12	85	7±1.4		6m
Rubio-Bueno et al. 2017	Pro	34	40.8 <u>±</u> 13.9	41.2	4.9 <u>±</u> 3.2	10.4±3.9	6m
Veys et al. 2017	Pro	10	44.7±9.5	80	4.8 <u>+</u> 2.8	8.3±2.3	6m
Vicini et al. 2010	RCT	25	49.1 <u>±</u> 9.1	92		11	13±2.5m
Vigneron et al. 2017	Retro	29	40.7 <u>+</u> 12.6		8.4 <u>+</u> 4.1	11.7±5.1	12.5±3.5y
Wu et al. 2019	Retro	28	37.2 <u>+</u> 11.8	53.6	2.0 <u>+</u> 3.1	8.8 <u>+</u> 3.7	>1y

Table 1. Characteristics of studies on maxillomandibular advancement

AHI, apnea-hypopnea index (events/h); BMI, body mass index (kg/m²); ESS, Epworth sleepiness scale; m, months; Max, maxilla; Mand, mandible; N, number of patients; ODI, oxygen desaturation index (events/h); Post-op, postoperative; Pre-op, preoperative; Pro, prospective; RCT, randomized controlled trial; Retro, retrospective; y, years.

^a Respiratory disturbance index (RDI) in this study was extracted as AHI.

^b The number of patients was 9.

^c This study defined surgical success as an AHI < 15 events/h with \geq 50% reduction in postoperative AHI.

^d This study did not define the criteria of surgical success.

 $^{\rm e}$ This study defined surgical success as a RDI < 15 events/h with \ge 50% reduction in postoperative RDI.

^f This study defined surgical success as a postoperative RDI < 20 events/h.

BA (mear	ИI n <u>+</u> SD)	Al (meal	HI 1 <u>+</u> SD)	OI (mean	DI 1 <u>+</u> SD)	ES (mear	S 1 <u>+</u> SD)	% Success	% Cure
Pre-op	Post- op	Pre-op	Post- op	Pre-op	Post- op	Pre-op	Post- op		
26.9 <u>+</u> 4.3	25.4±3.3	59.3±29.0	11.1 <u>+</u> 8.9					75 ^c	
		56.8 <u>+</u> 5.2	12.3 <u>+</u> 5.5						
		50.0 <u>+</u> 20.0	8.0 <u>+</u> 10.7						
28.3 <u>+</u> 3.4		51.4 <u>+</u> 16.9	8.5±9.4						
31.6 <u>+</u> 5.5	28 <u>+</u> 1.4	69.8 <u>+</u> 35.2	17.3 <u>+</u> 16.7	59.5±5.3	9.1 <u>+</u> 8			80 ^d	
29.4 <u>+</u> 4.6	27.2 <u>+</u> 3.3	70.7±15.9	11.4±7.4					81.8	
38.8 <u>+</u> 10.9	37.3 <u>+</u> 8.0	117.9 <u>+</u> 9.2	16.1 <u>+</u> 26.2			12.9 <u>±</u> 5.5 ^b	5.0±4.1 ^b	76.9	46.2
22 <u>+</u> 3.3		35.7 <u>+</u> 18	4.8 <u>±</u> 4.4					100	
25.7±3.7		40.2 <u>+</u> 25.6	9.9 <u>+</u> 7.2	13.5 <u>+</u> 18.6	4.0 <u>±</u> 3.5	13 <u>+</u> 6	9±7		
		72.3 <u>+</u> 26.7 ^a	7.2±7.5 ^a					95 ^e	
31.4 <u>+</u> 6.7	32.2 <u>+</u> 6.3	71.2 <u>+</u> 27.0 ^a	7.6 <u>±</u> 5.1 ^a					90 ^e	
32.0 <u>+</u> 6.0		61.6 <u>+</u> 23.9 ^a	9.2 <u>+</u> 8 ^a					90 ^f	
33.5 <u>+</u> 6.2	32.3±4.1	75.3±26.4 ^a	10.4±10.8 ^a					83.3 ^f	
22.4±3.4		41.6 <u>+</u> 19.2	5.3±4			11.9±7.3	7±3	100 ^c	
24.8 <u>+</u> 3.3	23.9±4.7	34.8 <u>+</u> 26.0	7.4 <u>+</u> 6.7			10.8 <u>+</u> 5	10.2 <u>+</u> 5.1		67.9
27 <u>+</u> 4.6	27.4 <u>+</u> 4.6	53.6 <u>+</u> 26.6	9.5±7.4	38.7±30.3	8.1 <u>+</u> 9.2	17.0 <u>+</u> 4.8	5.7±2.7	90	50
27.6 <u>+</u> 4.5	25.5±4.3	38.3±10.7	6.5±4.3	34.7±12.5	5.4±4.1	17.4±5.4	0.8 <u>+</u> 1.4	100	52.9
		26.8 <u>+</u> 12.7	12.3±14.4			14.1 <u>+</u> 5.9	5.7±3.0	70	40
32.7±5.8	31.4 <u>+</u> 6.5	56.8 <u>+</u> 16.5	8.1±7			11.6 <u>+</u> 2.8	7.7±1.3	88	36
24.6 <u>+</u> 4		56.6 <u>+</u> 24	25.5 <u>+</u> 20.6				7.5±4.7	41.4	
24.2 <u>+</u> 5.1		59.3±14.5	10.9±3.3			12.8 <u>+</u> 2.8	6.9 <u>+</u> 2.5	85.7	46.4

Study	Design	N	Age (years)	% Male	Follow-	BMI (mean±SD)	
			(mean±SD)		up (month)	Pre-op	Post-op
Bachour et al. 2021	Retro	15	52.9 <u>+</u> 6.6	86.7	18 <u>+</u> 9.6	29.1 <u>+</u> 3.3	30.1 <u>+</u> 4.5
Heiser et al. 2017	Pro	20	57±12	100	12	28.1 <u>+</u> 13.1	
Philip et al. 2018	Pro	10	52.0 <u>+</u> 9.4	100	6	28.8 <u>+</u> 3.3	
Steffen et al. 2019	Retro	18	51.5		24	27.9±4.5	28.0±4.7
Steffen et al. 2020	Pro	38	58.0 <u>+</u> 10.0	97.4	36	29.1 <u>+</u> 3.9	28.6 <u>+</u> 3.3
Suurna et al. 2021	Pro	782			14.3 <u>+</u> 7.0	29.2 <u>+</u> 4	
Van de Heyning et al. 2012	Pro	28	55.1 <u>+</u> 9.2	96.4	6	29.5±2.5	
Vanderveken et al. 2013	Retro	21	55 <u>+</u> 11	95.2	6	28 <u>+</u> 2	
Woodson et al. 2018	Pro	97	54.4 <u>+</u> 10.3		60	28.6 <u>+</u> 2.5	

Table 2. Characteristics of studies on upper airway stimulation

AHI, apnea-hypopnea index (events/h); BMI, body mass index (kg/m²); ESS, Epworth sleepiness scale; N, number of patients; ODI, oxygen desaturation index (events/h); Post-op, postoperative; Pre-op, preoperative; Pro, prospective; Retro, retrospective.

^a The number of patients was 18.

^b The number of patients was 71.

^c The number of patients was 92.

AHI (me	ean±SD)	ODI (me	an±SD)	ESS (me	ean±SD)	%	%
Pre-op	Post- op	Pre-op	Post-op	Pre-op	Post-op	Success	Cure
33.0 <u>+</u> 16.5	36.5 <u>+</u> 23.8	25.3 <u>+</u> 18.3	30.3 <u>+</u> 21.1	11.5 <u>+</u> 3.8	8.1 <u>+</u> 4.5	26.7	6.7
28.9 <u>+</u> 7.6	6.6 <u>+</u> 5.1						
46.7 <u>+</u> 12.2	14.5 <u>+</u> 8.9	38.1 <u>+</u> 21.1	10.5±9.9	15.9±3.5	10.0 <u>+</u> 6.1		
26.3±10.6	10.4 <u>+</u> 10.1	12.8 <u>+</u> 10.2	10.1 <u>+</u> 12.0	12.7 <u>+</u> 5.2	5.1 <u>+</u> 3.8	77.8	33.3
30.0 <u>+</u> 13.7	13.1 <u>+</u> 14.1	25.8 <u>+</u> 16.7	11.6 <u>+</u> 14.0	12.1 <u>+</u> 5.8	6.0 <u>+</u> 3.2	62	35
35.8±15.0	14.5±14.9			11.4±5.5	7.1 <u>+</u> 4.6	69.7	
42.3±16.4	32.6 <u>+</u> 29.1	30.7 <u>+</u> 21.6	26.7 <u>+</u> 27.0	11.0 <u>+</u> 5.0	7.6 <u>+</u> 4.3	50	
38.5 <u>+</u> 11.8	20. <u>3+</u> 20.6			8.2 ± 5.0 ^a	6.4 ± 4.3^{a}	62	
30.4±9.4 ^b	12.4 <u>+</u> 16.3	27.2 <u>+</u> 10.0 ^b	9.9±14.5	11.3 <u>+</u> 5.2	6.9±4.7 ^c	74.6 ^b	44

SUPPLEMENTARY MATERIAL

	Ovid MEDLINE(R) ALL <1946 to Dec 14, 2021>								
Step	Search	Result							
1	exp Sleep Apnea Syndromes/ or Snoring/ or ((sleep adj3 (apnea or apnoea or hypopnea or hypopnoea)) or (upper adj airway adj resistance) or (sleep adj disordered adj breathing) or snore or snoring).ti,ab,kf,ot.	54852							
2	(mma or ((maxillomandibular or mandibular) adj2 advancement) or ((bimaxillar or orthognathic) adj2 surgery) or maxillary-osteomy or (multilevel or multi-level)).ti,ab,kf.	51520							
3	'Electric Stimulation Therapy'/ or 'Electric Stimulation'/ or 'implantable Neurostimulators'/ or (((hypoglossal-nerve* or nervus-hypoglossus or cranial-nerve* or (XII adj nerve*)) adj2 (stimulat* or surgery or therap*)) or (upper-airway adj stimulat*) or Neurostimulat* or (implantable-nerve adj stimulat*) or electrical-stimulat*).ti,ab,kf.	52389							
4	2 or 3	103811							
5	1 and 4	1972							

Table S1a. Search strategy in MEDLINE database

Table S1b. Search strategy in Embase database

Embase Classic+Embase <1947 to Dec 14, 2021>								
Step	Search	Result						
1	exp 'snoring'/ or exp 'sleep disordered breathing'/ or (sleep adj3 (apnea or apnoea or hypopnea or hypopnea)).ti,ab. or 'upper airway resistance'.ti,ab. or 'sleep disordered breathing'.ti,ab. or snor*.ti,ab.	97390						
2	(mma or ((maxillomandibular or mandibula) adj2 advancement) or bimaxillar-surgery or maxillary-osteomy or orthognathic-surgery).ti,ab,kw.	8129						
3	(multilevel or multi-level).ti,ab,kw.	49506						
4	exp electrostimulation/ or exp 'nerve stimulator'/	101629						
5	(((hypoglossal-nerve* or nervus-hypoglossus or cranial-nerve* or (XII adj nerve*)) adj2 (stimulat* or surgery or therap*)) or (upper-airway adj stimulat*) or Neurostimulat* or (implantable-nerve adj stimulat*) or electrical-stimulat*).ti,ab,kw.	71405						
6	2 or 3 or 4 or 5	190441						
7	1 and 6	2037						
8	(exp experimental organism/ or animal tissue/ or animal cell/ or exp animal disease/ or exp carnivore disease/ or exp bird/ or exp experimental animal welfare/ or exp animal husbandry/ or animal behavior/ or exp animal cell culture/ or exp mammalian disease/ or exp mammal/ or exp marine species/ or nonhuman/ or animal.hw.) not human/	7698484						
9	7 not 8	1960						
10	limit 9 to (conference abstracts or embase)	1775						

	Q1	Q2	Q3	Q 4	Q5	Q6	Q7	Q8	Q9	Q10	Q11	Q12	Total score	Quality
Bettega et al. 2000	2	2	0	2	1	2	2	0					11	Fair
Bianchi et al. 2014	2	2	0	2	1	2	0	0					9	Fair
Boyd et al. 2015	2	2	2	2	1	2	0	2					13	High
Conradt et al. 1997	2	2	0	2	1	2	2	0					11	Fair
Gerbino et al. 2013	2	2	2	2	1	2	2	0					13	High
Goh et al. 2013	2	2	2	2	0	2	2	0					12	Fair
Goodday et al. 2016	2	2	0	2	0	2	0	0					8	Fair
Hsieh et al. 2014	2	0	2	2	1	2	0	0					9	Fair
Kastoer et al. 2019	2	0	2	2	1	2	2	0					11	Fair
Li et al. 1999	0	2	0	2	0	2	2	0					8	Fair
Li et al. 2000	2	2	0	2	0	2	0	0					8	Fair
Li et al. 2001	2	2	0	2	0	2	0	0					8	Fair
Li et al. 2002	2	1	2	2	0	2	0	0					9	Fair
Liao et al. 2015	2	2	2	2	1	2	0	0					11	Fair
Lin et al. 2020	2	2	2	2	1	2	0	0					11	Fair
Liu et al. 2016	2	2	0	2	1	2	0	0					9	Fair
Rubio-Bueno et al. 2017	2	2	2	2	1	2	0	0					11	Fair
Veys et al. 2017	2	2	2	2	0	2	0	0					10	Fair
Vicini et al. 2010	2	2	2	2	1	2	2	0					13	High
Vigneron et al. 2017	2	2	0	2	1	2	0	0					9	Fair
Wu et al. 2019	2	2	0	2	1	2	0	0	0	2	0	2	13	Fair

Table Sza. Methodological appraisal of the individual studies according to MINORS assessment tool – maxillomandibular advancement surgery

Q1, a clear study aim; Q2, inclusion of consecutive patients; Q3, prospective collection of data; Q4, endpoint appropriate to the aim of the study; Q5, unbiased assessment of the study; Q6, follow-up period appropriate to the aim of the study endpoint; Q7, loss of follow-up less than 5%; Q8, prospective calculation of the study size; Q9, an adequate control group; Q10, contemporary group; Q11, baseline equivalent of groups; Q12, adequate statistical analysis.

	Q1	Q2	Q3	Q4	Q5	Q6	Q7	Q8	Total score	Quality
Bachour et al. 2021	2	2	0	2	0	2	2	0	10	Fair
Heiser et al. 2017	2	2	2	2	0	2	0	0	10	Fair
Philip et al. 2018	2	0	2	2	1	2	0	0	9	Fair
Steffen et al. 2019	2	2	0	2	0	2	0	0	8	Fair
Steffen et al. 2020	2	2	2	2	0	2	0	0	8	Fair
Suurna et al. 2021	2	2	2	2	0	2	0	0	10	Fair
Van de Heyning et al. 2012	2	2	2	2	1	2	2	0	13	High
Vanderveken et al. 2013	2	0	2	2	1	2	0	0	9	Fair
Woodson et al. 2018	2	0	2	2	1	2	0	0	9	Fair

Table S2b. Methodological appraisal of the individual studies according to MINORS assessment tool – upper airway stimulation

Q1, a clear study aim; Q2, inclusion of consecutive patients; Q3, prospective collection of data; Q4, endpoint appropriate to the aim of the study; Q5, unbiased assessment of the study; Q6, follow-up period appropriate to the aim of the study endpoint; Q7, loss of follow-up less than 5%; Q8, prospective calculation of the study size.

REFERENCES

- 1. Jordan AS, McSharry DG, Malhotra A. Adult obstructive sleep apnoea. Lancet. 2014;383:736-47.
- Senaratna CV, Perret JL, Lodge CJ, et al. Prevalence of obstructive sleep apnea in the general population: A systematic review. Sleep Med Rev. 2017;34:70-81.
- 3. Bagai K. Obstructive sleep apnea, stroke, and cardiovascular diseases. *Neurologist.* 2010;16:329-39.
- 4. Knauert M, Naik S, Gillespie MB, Kryger M. Clinical consequences and economic costs of untreated obstructive sleep apnea syndrome. *World J Otorhinolaryngol Head Neck Surg*. 2015;1:17-27
- Gottlieb DJ, Punjabi NM. Diagnosis and Management of Obstructive Sleep Apnea: A Review. JAMA. 2020;323:1389-400.
- 6. Rotenberg BW, Vicini C, Pang EB, Pang KP. Reconsidering first-line treatment for obstructive sleep apnea: a systematic review of the literature. J Otolaryngol Head Neck Surg. 2016;45:23.
- 7. Sharifian MR, Zarrinkamar M, Alimardani MS, et al. Drug Induced Sleep Endoscopy in Obstructive Sleep Apnea. *Tanaffos.* 2018;17:122-6.
- 8. Li KK. Surgical management of obstructive sleep apnea. Clin Chest Med. 2003;24:365-70.
- Riley RW, Powell NB, Guilleminault C, Nino-Murcia G. Maxillary, mandibular, and hyoid advancement: an alternative to tracheostomy in obstructive sleep apnea syndrome. Otolaryngol Head Neck Surg. 1986;94:584-8.
- Cokce SM, Gorgulu S, Gokce HS, Bengi AO, Karacayli U, Ors F. Evaluation of pharyngeal airway space changes after bimaxillary orthognathic surgery with a 3-dimensional simulation and modeling program. *Am J Orthod Dentofacial Orthop.* 2014;146:477-92.
- 11. Liu SY, Huon LK, Iwasaki T, et al. Efficacy of Maxillomandibular Advancement Examined with Drug-Induced Sleep Endoscopy and Computational Fluid Dynamics Airflow Modeling. *Otolaryngol Head Neck Surg.* 2016;154:189-95.
- 12. Strollo PJ, Soose RJ, Maurer JT, et al. Upper-Airway Stimulation for Obstructive Sleep Apnea. N Engl J Med. 2014;370:139-49.
- 13. Steffen A, Heiser C, Galetke W, et al. Hypoglossal nerve stimulation for obstructive sleep apnea: updated position paper of the German Society of Oto-Rhino-Laryngology, Head and Neck Surgery. *Eur Arch Otorhinolaryngol.* 2022;279:61-6.
- 14. Vanderveken OM, Beyers J, Op de Beeck S, et al. Development of a Clinical Pathway and Technical Aspects of Upper Airway Stimulation Therapy for Obstructive Sleep Apnea. *Front Neurosci.* 2017;11:523.
- 15. Woodson BT, Strohl KP, Soose RJ, et al. Upper Airway Stimulation for Obstructive Sleep Apnea: 5-Year Outcomes. Otolaryngol Head Neck Surg. 2018;159:194-202.
- 16. Dedhia RC, Strollo PJ, Soose RJ. Upper Airway Stimulation for Obstructive Sleep Apnea: Past, Present, and Future. *Sleep.* 2015;38:899-906.
- 17. Yu MS, Ibrahim B, Riley RW, Liu SY. Maxillomandibular Advancement and Upper Airway Stimulation: Extrapharyngeal Surgery for Obstructive Sleep Apnea. *Clin Exp Otorhinolaryngol.* 2020;13:225-33.
- 18. Mashaqi S, Patel SI, Combs D, et al. The Hypoglossal Nerve Stimulation as a Novel Therapy for Treating Obstructive Sleep Apnea—A Literature Review. Int J Environ Res Public Health. 2021;18:1642.
- 19. Page MJ, Moher D, Bossuyt PM, et al. PRISMA 2020 explanation and elaboration: updated guidance and exemplars for reporting systematic reviews. *BMJ*. 2021;372:n160.
- 20. Strasberg SM, Linehan DC, Hawkins WG. The accordion severity grading system of surgical complications. *Ann Surg.* 2009;250:177-86.
- 21. Berry RB, Brooks R, Gamaldo CE, et al. The AASM Manual for the Scoring of Sleep and Associated Events: Rules, Terminology and Technical Specifications, Version 2.0.2. Darien, IL: American Academy of Sleep Medicine; 2013.
- 22. Sher AE, Schechtman KB, Piccirillo JF. The efficacy of surgical modifications of the upper airway in adults with obstructive sleep apnea syndrome. *Sleep*. 1996;19:156-77.
- 23. Elshaug AG, Moss JR, Southcott AM, Hiller JE. Redefining success in airway surgery for obstructive sleep apnea: a meta analysis and synthesis of the evidence. *Sleep*. 2007;30:461-7.
- 24. Slim K, Nini E, Forestier D, Kwiatkowski F, Panis Y, Chipponi J. Methodological index for non-randomized studies (minors): development and validation of a new instrument. *ANZJ Surg.* 2003;73:712-6.
- Shah A, Memon M, Kay J, Wood TJ, Tushinski DM, Khanna V. Preoperative Patient Factors Affecting Length of Stay following Total Knee Arthroplasty: A Systematic Review and Meta-Analysis. J Arthroplasty. 2019;34:2124-65.e1.
- 26. Altman D, Machin D, Bryant T, Gardner M. Statistics with confidence: confidence intervals and statistical guidelines. John Wiley & Sons; 2013.

- 27. Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *BMJ*. 2003;327:557-60.
- 28. Bettega G, Pepin JL, Veale D, Deschaux C, Raphael B, Levy P. Obstructive sleep apnea syndrome. fifty-one consecutive patients treated by maxillofacial surgery. *Am J Respir Crit Care Med*. 2000;162:641-9.
- 29. Bianchi A, Betti E, Tarsitano A, Morselli-Labate AM, Lancellotti L, Marchetti C. Volumetric threedimensional computed tomographic evaluation of the upper airway in patients with obstructive sleep apnoea syndrome treated by maxillomandibular advancement. *Br J Oral Maxillofac Surg.* 2014;52:831-7.
- 30. Boyd SB, Walters AS, Waite P, Harding SM, Song Y. Long-Term Effectiveness and Safety of Maxillomandibular Advancement for Treatment of Obstructive Sleep Apnea. J Clin Sleep Med. 2015;11:699-708.
- 31. Conradt R, Hochban W, Brandenburg U, Heitmann J, Peter JH. Long-term follow-up after surgical treatment of obstructive sleep apnoea by maxillomandibular advancement. *Eur Respir J*. 1997;10:123-8.
- 32. Gerbino G, Bianchi FA, Verzé L, Ramieri G. Soft tissue changes after maxillo-mandibular advancement in OSAS patients: A three-dimensional study. J Craniomaxillofac Surg. 2014;42:66-72.
- 33. Goh YH, Lim KA. Modified maxillomandibular advancement for the treatment of obstructive sleep apnea: a preliminary report. *Laryngoscope*. 2003;113:1577-82.
- 34. Goodday RH, Bourque SE, Edwards PB. Objective and Subjective Outcomes Following Maxillomandibular Advancement Surgery for Treatment of Patients With Extremely Severe Obstructive Sleep Apnea (Apnea-Hypopnea Index >100). J Oral Maxillofac Surg. 2016;74:583-9.
- Hsieh YJ, Liao YF, Chen NH, Chen YR. Changes in the calibre of the upper airway and the surrounding structures after maxillomandibular advancement for obstructive sleep apnoea. Br J Oral Maxillofac Surg. 2014;52:445-51.
- 36. Kastoer C, Op de Beeck S, Dom M, et al. Drug-Induced Sleep Endoscopy Upper Airway Collapse Patterns and Maxillomandibular Advancement. *Laryngoscope*. 2020;130:E268-E274.
- 37. LiKK, Guilleminault C, Riley RW, Powell NB. Obstructive sleep apnea and maxillomandibular advancement: an assessment of airway changes using radiographic and nasopharyngoscopic examinations. J Oral Maxillofac Surg. 2002;60:526-30; discussion 531.
- 38. Li KK, Powell NB, Riley RW, Troell R, Guilleminault C. Long Term Results of Maxillomandibular Advancement Surgery. Sleep Breath. 2000;4:137-40.
- 39. Li KK, Riley RW, Powell NB, Troell R, Guilleminault C. Overview of phase II surgery for obstructive sleep apnea syndrome. *Ear Nose Throat J.* 1999;78:851, 854-7.
- 40. Li KK, Troell RJ, Riley RW, Powell NB, Koester U, Guilleminault C. Uvulopalatopharyngoplasty, maxillomandibular advancement, and the velopharynx. *Laryngoscope*. 2001;111:1075-8.
- 41. Liao YF, Chiu YT, Lin CH, Chen YA, Chen NH, Chen YR. Modified maxillomandibular advancement for obstructive sleep apnoea: towards a better outcome for Asians. *Int J Oral Maxillofac Surg.* 2015;44:189-94.
- 42. Lin CH, Chin WC, Huang YS, et al. Objective and subjective long term outcome of maxillomandibular advancement in obstructive sleep apnea. *Sleep Med.* 2020;74:289-96.
- Rubio-Bueno P, Landete P, Ardanza B, et al. Maxillomandibular advancement as the initial treatment of obstructive sleep apnoea: Is the mandibular occlusal plane the key? Int J Oral Maxillofac Surg. 2017;46:1363-71.
- 44. Veys B, Pottel L, Mollemans W, Abeloos J, Swennen G, Neyt N. Three-dimensional volumetric changes in the upper airway after maxillomandibular advancement in obstructive sleep apnoea patients and the impact on quality of life. *Int J Oral Maxillofac Surg.* 2017;46:1525-32.
- 45. Vicini C, Dallan I, Campanini A, et al. Surgery vs ventilation in adult severe obstructive sleep apnea syndrome. *Am J Otolaryngol*. 2010;31:14-20.
- 46. Vigneron A, Tamisier R, Orset E, Pepin JL, Bettega G. Maxillomandibular advancement for obstructive sleep apnea syndrome treatment: Long-term results. J Craniomaxillofac Surg. 2017;45:183-91.
- 47. Wu Q, Wang Y, Wang P, Xiang Z, Ye B, Li J. The inverted-L ramus osteotomy versus sagittal split ramus osteotomy in maxillomandibular advancement for the treatment of obstructive sleep apnea patients: A retrospective study. J Craniomaxillofac Surg. 2019;47:1839-47.
- 48. Heiser C, Edenharter G, Bas M, Wirth M, Hofauer B. Palatoglossus coupling in selective upper airway stimulation. *Laryngoscope*. 2017;127:E378-E383.
- 49. Philip P, Heiser C, Bioulac S, et al. Hypoglossal nerve stimulation on sleep and level of alertness in OSA: A preliminary study. *Neurology*. 2018;91:e615-e619.
- 50. Steffen A, Abrams N, Suurna MV, Wollenberg B, Hasselbacher K. Upper-Airway Stimulation Before, After, or Without Uvulopalatopharyngoplasty: A Two-Year Perspective. *Laryngoscope*. 2019;129:514-8.
- 51. Steffen A, Sommer UJ, Maurer JT, Abrams N, Hofauer B, Heiser C. Long-term follow-up of the German postmarket study for upper airway stimulation for obstructive sleep apnea. *Sleep Breath*. 2020;24:979-84.

- 52. Van de Heyning PH, Badr MS, Baskin JZ, et al. Implanted upper airway stimulation device for obstructive sleep apnea. *Laryngoscope*. 2012;122:1626-33.
- 53. Vanderveken OM, Maurer JT, Hohenhorst W, et al. Evaluation of drug-induced sleep endoscopy as a patient selection tool for implanted upper airway stimulation for obstructive sleep apnea. J Clin Sleep Med. 2013;9:433-8.
- 54. Suurna MV, Steffen A, Boon M, et al. Impact of Body Mass Index and Discomfort on Upper Airway Stimulation: ADHERE Registry 2020 Update. *Laryngoscope*. 2021;131:2616-24.
- 55. Bachour A, Bäck L, Pietarinen P. No changes in nocturnal respiration with hypoglossal neurostimulation therapy for obstructive sleep apnoea. *Clin Respir J*. 2021;15:329-35.
- 56. Heiser C, Steffen A, Boon M, et al. Post-approval upper airway stimulation predictors of treatment effectiveness in the ADHERE registry. Eur Respir J 2019;53:1801405. doi:10.1183/13993003.01405-2018
- 57. Safiruddin F, Vanderveken OM, de Vries N, et al. Effect of upper-airway stimulation for obstructive sleep apnoea on airway dimensions. *Eur Respir J.* 2015;45:129-38.
- 58. Li KK, Riley RW, Powell NB, Gervacio L, Troell RJ, Guilleminault C. Obstructive sleep apnea surgery: patient perspective and polysomnographic results. *Otolaryngol Head Neck Surg.* 2000;123:572-5.
- 59. Butterfield KJ, Marks PL, McLean L, Newton J. Quality of life assessment after maxillomandibular advancement surgery for obstructive sleep apnea. J Oral Maxillofac Surg. 2016;74:1228-37.
- 60. Beranger T, Garreau E, Ferri J, Raoul G. Morphological impact on patients of maxillomandibular advancement surgery for the treatment of obstructive sleep apnea-hypopnea syndrome. *Int Orthod.* 2017;15:40-53.
- 61. Cillo JE, Robertson N, Dattilo DJ. Maxillomandibular Advancement for Obstructive Sleep Apnea Is Associated With Very Long-Term Overall Sleep-Related Quality-of-Life Improvement. J Oral Maxillofac Surg. 2020;78:109-17.
- 62. Mehra R, Steffen A, Heiser C, et al. Upper Airway Stimulation versus Untreated Comparators in Positive Airway Pressure Treatment-Refractory Obstructive Sleep Apnea. *Ann Am Thorac Soc.* 2020;17:1610-9.
- 63. Pottel L, Neyt N, Hertegonne K, et al. Long-term quality of life outcomes of maxillomandibular advancement osteotomy in patients with obstructive sleep apnoea syndrome. *Int J Oral Maxillofac Surg.* 2019;48:332-40.
- 64. Riley RW, Powell NB, Li KK, Troell RJ, Guilleminault C. Surgery and obstructive sleep apnea: Long-term clinical outcomes. *Otolaryngol Head Neck Surg.* 2000;122:415-21.
- 65. Hofauer B, Steffen A, Knopf A, Hasselbacher K, Heiser C. Patient experience with upper airway stimulation in the treatment of obstructive sleep apnea. *Sleep Breath*. 2019;23:235-41.
- 66. Peacock ZS, Lee CC, Klein KP, Kaban LB. Orthognathic surgery in patients over 40 years of age: indications and special considerations. J Oral Maxillofac Surg. 2014;72:1995-2004.
- 67. Passeri LA, Choi JG, Kaban LB, Lahey ET. Morbidity and Mortality Rates After Maxillomandibular Advancement for Treatment of Obstructive Sleep Apnea. J Oral Maxillofac Surg. 2016;74:2033-43.
- 68. Van Sickels JE, Hatch JP, Dolce C, Bays RA, Rugh JD. Effects of age, amount of advancement, and genioplasty on neurosensory disturbance after a bilateral sagittal split osteotomy. *J Oral Maxillofac Surg.* 2002;60:1012-7.
- 69. Zhou N, Ho JTF, Huang Z, et al. Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis. *Sleep Med Rev.* 2021;57:101471.
- 70. Withrow K, Evans S, Harwick J, Kezirian E, Strollo P. Upper Airway Stimulation Response in Older Adults with Moderate to Severe Obstructive Sleep Apnea. *Otolaryngol Head Neck Surg*. 2019;161:714-9.
- 71. Murphey AW, Baker AB, Soose RJ, et al. Upper airway stimulation for obstructive sleep apnea: The surgical learning curve. *Laryngoscope*. 2016;126:501-6.
- 72. Pietzsch JB, Liu S, Garner AM, Kezirian EJ, Strollo PJ. Long-Term Cost-Effectiveness of Upper Airway Stimulation for the Treatment of Obstructive Sleep Apnea: A Model-Based Projection Based on the STAR Trial. *Sleep*. 2015;38:735-44.
- 73. Pietzsch JB, Richter A-K, Randerath W, et al. Clinical and economic benefits of upper airway stimulation for obstructive sleep apnea in a European setting. *Respiration*. 2019;98:38-47. doi: 10.1159/000497101
- 74. Sarber KM, Chang KW, Ishman SL, Epperson MV, Dhanda Patil R. Hypoglossal Nerve Stimulator Outcomes for Patients Outside the U.S. FDA Recommendations. *Laryngoscope*. 2020;130:866-72.
- 75. Cook JA. The challenges faced in the design, conduct and analysis of surgical randomised controlled trials. *Trials.* 2009;10:9.
- 76. Higgins JP, Thomas J, Chandler J, et al. Cochrane handbook for systematic reviews of interventions. John Wiley & Sons; 2019.



Intra-individual variation of upper airway measurements based on computed tomography

Ning Zhou, Jean-Pierre T.F. Ho, Cornelis Klop, Ruud Schreurs, Ludo F. M. Beenen, Ghizlane Aarab, Jan de Lange

Published in: PLoS One, 2021. doi: 10.1371/journal.pone.0259739

ABSTRACT

The aims of this study were (1) to quantify the intra-individual variation in the upper airway measurements on supine computed tomography (CT) scans at two different time points; and (2) to identify the most stable parameters of the upper airway measurements over time. Ten subjects with paired CT datasets (3-6 months interval) were studied, using computer software to segment and measure the upper airway. The minimum cross-sectional area of the total airway and all its segments (velopharynx, oropharynx, tongue base, and epiglottis) generally had the largest variation, while the length of the total airway had the lowest variation. Sphericity was the only parameter that was stable over time (relative difference < 15%), both in the total airway and each subregion. There was considerable intra-individual variation in CT measurements of the upper airway, with the same patient instruction protocol for image acquisitions. The length of the total airway, and the sphericity of the total upper airway and each segment were stable over time. Hence, such intra-individual variation should be taken into account when interpreting and comparing upper airway evaluation parameters on CT in order to quantify treatment results or disease progress.

Keywords: Intra-individual variation; Repeatability; Upper airway; Computed tomography; Measurement

INTRODUCTION

Over the past decades growing awareness of the detrimental effects of obstructive sleep apnea (OSA) has increasingly raised interest in morphometric evaluation of the upper airway¹⁻³. Traditionally, upper airway morphology imaging consisted of a twodimensional (2D) lateral cephalogram^{4, 5}. However, due to the technical advancement of computed tomography (CT), this imaging modality has gained increasing popularity^{5, 6}. Compared with a 2D lateral cephalogram, CT exhibits the capacity to analyze the upper airway three-dimensionally^{7, 8}. Three-dimensional (3D) analysis has been widely used to assess the upper airway, which has given rise to the proposal and usage of multiple methods^{3, 9, 10}. Volumetric, areal, and linear measurements, the parameters commonly used for upper airway evaluation, have been shown to have good to excellent inter-operator and intra-operator reliability in previous studies⁹⁻¹².

The rationale behind upper airway measurements may be to compare results of an individual to a reference group, or, more likely, to quantify changes within the airway between different time points. While the previous studies quantify the variation and precision of the measurement method itself, measurement on different time points could yield variation in repeated airway measurements as well. It has been proven that the upper airway dimension is influenced by an individual's body position, head and neck posture, mandibular movement, tongue position, and breathing stage^{5,13-16}. It is a challenge to standardize all these interfering factors during CT scan acquisition^{11,14}. Therefore, even if no airway-influencing intervention has been performed, it is suspected that considerable intra-individual variation in CT volumetric, areal and linear measurements of the upper airway at different time points exists. This variation between time points may hamper adequate evaluation of upper airway changes after surgical and orthodontic procedures, even if a validated measurement method is used. Understanding the degree of intra-individual variation in the upper airway measurements is thus imperative for clinical evaluation and research.

The intra-individual variation of the upper airway measurements has been studied only scarcely. In the study by Obelenis Ryan et al.¹¹, different volumetric readings of the upper airway were found in the context of different CBCT examinations with identical scanning and patient positioning protocols. However, in their study, the CBCT scans were taken in an upright position. It is well known that upper airway dimension is different between the upright and supine positions^{17, 18}. For this reason, a new study under controlled conditions with the patient in supine position during the image acquisition is relevant. Hence, the primary aim of this study was to quantify the natural intra-individual variation in the upper airway measurements on supine CT scans at two different time points. The secondary aim was to identify the most stable parameters of the upper airway measurements over time, by which accurate evaluation and comparison of the upper airway before and after intervention may be achieved in the future.

METHODS

Due to the retrospective nature of the study and de-identifying patient data prior to conducting the study, the Medical Ethical Committee of the Amsterdam UMC decided that the Medical Research Human Subjects Act was not applicable to this study (Ref. NoW20_261).

Study population

The population consisted of 10 subjects selected from a patient database of the Department of Oral and Maxillofacial Surgery (5 males and 5 females; mean age 50.3 \pm 10.3 years, range 34-68 years), which had two CT datasets (To and T1) of the head and neck region acquired in the Amsterdam UMC. They were scanned for various indications, viz., maxillary/mandibular granuloma and palatal fistula, with a 3-6 months' time interval between scans (mean 4.8 \pm 1.2 months). The inclusion criteria were the following: (1) adequate scan quality; (2) sufficient field of view (sella/nasion to epiglottis base); and (3) time interval between the scans of 3 to 6 months. The exclusion criteria were as follows: (1) patients younger than 18 years; (2) cases with intubation or other potential airway-influencing interventions during or between scans; (3) patients with previous upper airway surgery; and (4) patients with suspected or diagnosed OSA, of whom the upper airway may alter during the progression of OSA disease.

CT image acquisition

The included spiral CT scans of head and neck were acquired between 2018 and 2019 using the following scanning protocol (SOMATOM Force, Siemens Medical Solutions, Erlangen, Germany): 120 kV, 380 mAs, max. FOV 300 mm, pitch 0.85, slice thickness 1.0 mm, slice increment 1.0 mm, image matrix 512×512, window W1600/ L400, hard-tissue kernel H60s. During the imaging procedure, the patients were in supine position and were instructed to remain still with maximum intercuspation, to breathe gently, and not to swallow.

CT measurements

Reference frame The Digital Imaging and Communications in Medicine (DICOM) files of the CT were imported in Maxilim software (version 2.3.0, Medicim NV, Mechelen, Belgium) for measurements. A hard-tissue reconstruction was created at 300 Hounsfield units (HU) and a soft-tissue reconstruction at -400 HU. To standardize the measurements and minimize the measurement error, the Frankfort Horizontal (FH) plane was constructed for reorientation of the 3D images at T0¹⁹. The T1 dataset was superimposed on the T0 dataset, using voxel-based matching on the structures of the cranial base^{20, 21}.

Landmarks After re-orientation and superimposition of the paired CT scans, four anatomical landmarks (Figure 1) were identified for segmentation of the regions of interest: posterior nasal spine (PNS), tip of uvula (TUV), tip of epiglottis (TEP), and base of epiglottis (BEP). The reliability of these landmarks has been validated in a previous study⁹. Based on TUV and TEP, the midpoint between them (MUE) was then calculated and localized (Figure 1). Because PNS is a bony landmark and thus an unaltered position between scans, it was localized only once for the To scan and reused for the T1 scan; the other four landmarks were identified on both scans.

Boundary The soft-tissue model was imported into Blender software (version 2.81, Blender Foundation, Amsterdam, The Netherlands) for further analysis. The superior boundary of the upper airway was defined as the plane through the PNS parallel to the FH plane^{9, 22}. The inferior boundary was the plane through the BEP parallel to the FH plane^{9, 22}. The lateral and posterior boundaries consisted of the pharyngeal walls and the anterior boundary was composed of the soft palate, base of tongue, and anterior wall of the pharynx, with a cut-off at PNS point^{10, 23}.

Segmentation Based on the identified landmarks, the upper airway was segmented into four distinct regions (**Figure 1**): velopharynx region (between PNS and TUV), oropharynx region (between TUV and MUE), tongue base region (between MUE and TEP), and epiglottis region (between TEP and BEP). Cutting planes were parallel to the FH plane.

Upper airway parameters One operator (CK), with extensive experience with Blender, performed the measurements in all 20 datasets. The operator was blinded to the measurement results of To scans during the measurement for T1 scans. To quantify the inter-operator reliability, a second operator (RS) repeated the entire

measurement protocol in five randomly selected datasets. The operators were blinded to each other's results. The upper airway parameters of interest were volume, length, surface area, minimum cross-sectional area (MCA), and lateral dimension (LAT) and anteroposterior dimension (AP) of the MCA. These parameters were measured for the total airway and for the individual segments (**Table 1**). Before measuring the MCA, the "islands" (loose air parts) and "dead space" (space in mouth and space between tongue base and epiglottis) were removed from the upper airway model (**Figure 2**).

Derived airway parameters Based on these parameters, the following derived parameters were calculated: mean cross-sectional area (meanCSA)²⁴ for the size of the total airway and each segment; LAT/AP ratio in MCA, airway uniformity¹⁰, and sphericity¹⁰ for the shape of the total airway and of each segment separately (**Table 1**).

Outcome variables The following outcome variables were derived in this study:

- Intra-individual variation (number of patients = 10; number of CT datasets = 20): the relative difference in the measurements between two scans (To and T1) of an individual by operator 1.
- Intra-individual repeatability (number of patients = 10; number of CT datasets = 20): the intra-class correlation coefficient (ICC) for the measurements between two scans (To and T1) of an individual by operator 1.
- Inter-operator variation (number of CT datasets = 5): the relative difference between the measurements by operator 1 and operator 2 at To/T1.
- Inter-operator reliability (number of CT datasets = 5): the ICC for the measurements by operator 1 and operator 2 at To/T1.
- Agreement and smallest detectable difference (SDD) in the measurements between two scans (To and T1) of an individual (number of patients = 10; number of CT datasets = 20) by operator 1.

Statistical analysis

All data were analyzed using SPSS software (version 26, IBM Corp., Armonk, NY, USA). Descriptive statistical analysis was performed for all demographic and outcome variables.
The intra-individual repeatability and inter-operator reliability of upper airway measurements were evaluated using ICC²⁵. Values of ICC less than 0.40, between 0.40 and 0.75, and greater than 0.75 are indicative of poor, fair to good, and excellent reliability, respectively²⁵. The relative difference was used to estimate the intra-individual variation and inter-operator variation, which was calculated with the formula: (absolute difference/mean)*100%. Bland-Altman analysis was used to determine the agreement of the airway measurements between two different scans and to obtain the precise confidence interval for paired difference²⁶. Based on Bland-Altman's method, SDD in the airway measurements between two scans of an individual was calculated with the formula: (1.96*SD_{To-To}).

RESULTS

Descriptive statistics of all measurements, intra-individual variation estimated by relative difference, intra-individual repeatability estimated by ICC, inter-operator variation estimated by relative difference, and inter-operator reliability estimated by ICC are presented in Table 2. Of the 50 upper airway parameters, the ICC values of intra-individual repeatability were greater than 0.75 for 26, between 0.40 to 0.75 for 19, and less than 0.40 for 5. For the inter-operator reliability estimated by the ICC, all the parameters showed excellent reliability (ICC 0.832-0.999). As for the intraindividual variation in the total airway, the mean relative difference was maximum in MCA (35.5%) and minimum in length (4.9%). Regarding the different airway subregions, the mean relative differences between two scans were exceedingly large (> 25%) in: volume at the oropharynx (34.4%), tongue base (29.8%), and epiglottis (25.4%); LAT of MCA at the epiglottis (25.4%); AP of MCA at the velopharynx (28.4%) and tongue base (26.5%); meanCSA at the oropharynx (25.3%), tongue base (26.9%), and epiglottis (25.1%); LAT/AP ratio of MCA at the tongue base (26.3%); and MCA at all levels. The relative differences of the sphericity between two scans in the total airway and each segment were all below 15%.

Table 3 shows the results of Bland-Altman analysis of differences between the paired scans (To-T1; mean, SD, and 95% limits of agreement), as well as the absolute value of differences (|To-T1|; mean and SD) and SDD values.

DISCUSSION

This is the first study to evaluate the intra-individual variation of linear, areal, and volumetric measurements of the upper airway in CT scans acquired at two different time points. Because of the short time interval between To and T1 (3-6 months), the absence of airway-influencing intervention during or between scans, no airway-influencing pathology or disease present in the patient, and the same position protocol between CT acquisitions, no airway alteration was expected within the scan pairs in our study population. Nevertheless, our findings suggest that different degree of variation exists in each segment of the upper airway between To and T1. Although patients with an airway-altering disease (i.e., OSA) or intervention were excluded, this finding may be especially important for evaluating change in these patients as a method to quantify diseases progress or treatment effects.

Regarding the intra-individual variation of the upper airway measurements between To and T1 (see **Table 2**), we found that the MCA of the total airway and of each segment separately generally showed the largest variation, with a relative difference of approximately 30%. Such variation could have two causes. Firstly, the location of MCA is not always constant during the dynamic upper airway movement due to breathing. Secondly, errors or variation in determining the location of the MCA may exist. Although several studies have found that MCA is the most important characteristic of the upper airway that may contribute to distinguishing OSA cases from non-OSA cases^{27, 28}, caution is thus warranted in interpreting this finding or applying it in clinical practice due to the natural variation found for MCA in the present study.

A significant limitation in CT analysis of the upper airway is differentiating the boundaries of soft tissues and empty spaces (air) by using limited difference in grey levels between them. However, the measurement of upper airway length is not affected by this as it is determined by a user-generated plane. Increased airway length has been suggested to be correlated with the presence and severity of OSA¹⁰. For consistency and reproducibility, we used a bony landmark having shown excellent reliability in previous studies – PNS – to define the superior boundary of the upper airway^{9, 29}. In our study, the length of the total upper airway showed the least variation (relative difference: 4.9%) and it may therefore be regarded as a stable evaluation parameter for the upper airway.

Airway shape may contribute to the development of OSA^{1, 10}. Recently, a derived variable, that is sphericity of the upper airway, was suggested and investigated^{10, 30}.

Klazen et al. found that less sphericity was the main predictor for OSA in patients with craniofacial macrosomia³⁰. It is interesting to note that sphericity had low ICC values for intra-individual repeatability; however, it also showed low variation between To and T1 in both the total airway and each segment, all the relative differences being below 15%. This may be explained by the fact that ICC is a ratio between inter-unit variability and total variability (intra-unit and inter-unit)³¹. In this study, minor inter-unit variabilities of the sphericity measurements were indicated by the extremely low SDs, which could explain the low ICC values. Therefore, this parameter should not be disregarded based on ICC value alone.

The mean relative differences between two CT scans of the volumes of the total airway, velopharynx, oropharynx, tongue base, and epiglottis were 21.3%, 15.9%, 34.4%, 29.8%, and 25.4%, respectively. Obelenis Ryan et al.¹¹ evaluated CBCT scans of 27 patients obtained at two time points and reported that the mean relative differences of the volumes of the nasopharynx, oropharynx, and hypopharynx were 9.8%, 17.8%, and 12.0%, respectively. However, care should be taken in comparing the results between the two studies because of the different methodology in the upper airway segmentation. Moreover, differences between CT and CBCT evaluation of the upper airway should be noted. CT are performed when the patient is in the supine position, while most CBCT units acquire images with the patient in the upright position³². Soft tissue contrast resolution on CBCT imaging is inferior to CT imaging and therefore segmentation results are different³³.

There are several studies describing the morphometric evaluation of the upper airway^{23, 24}. To date, however, there is no methodological standardization in 3D analysis of the upper airway³⁴. Chen et al.⁹ proposed a method of landmark localization for 3D upper airway measurements, which showed excellent intra- and inter-operator reliability. In the present study, four of the six landmarks proposed by Chen et al. were utilized: PNS, TUV, TEP, and BEP. Additionally, a derived landmark, MUE, was localized at the midpoint between TUV and TEP. Through the experience of over 8,000 drug-induced sleep endoscopy (DISE) examinations, Kezirian et al.³⁵ found four structures, namely velum, oropharyngeal lateral wall, tongue base, and epiglottis, which play a prominent role in upper airway obstruction. Accordingly, they proposed the VOTE classification system, which has been widely used for characterizing DISE findings. In 3D evaluation of the upper airway, various subregion definitions of the airway have been used in previous studies^{11, 22, 23}. However, structure-based assessment for the upper airway cannot be achieved in these methods. Therefore, based on the work by Kezirian et al.³⁵, the upper airway was divided into four subregions corresponding to the VOTE classification system. Because PNS, TUV, TEP, and BEP demonstrated excellent intra- and inter-operator reliability in the study of Chen et al.⁹, the segmentation of the upper airway based on these landmarks may be considered reliable.

In the current study, all the parameters showed excellent inter-operator reliability. Zimmerman et al. conducted a study to assess the reliability of upper airway analysis with CBCT³⁴. Interestingly, in contrast to our results, they found that the MCA and total airway volume showed poor inter-operator reliability. It needs to be noted that in Zimmerman et al.' study, six examiners of varying levels of education and clinical experience separately performed the upper airway analysis, and the reliability improved with the examiner education and experience. In our study, the measurement protocol was conducted by two experienced examiners, which may explain the discrepancy of reliability between the two studies. In addition, unlike their study, we used a fixed threshold for the selection of the upper airway. In this way, the operator's subjectivity in the threshold sensitivity selection was eliminated. Since it is generally accepted that the inter-operator reliability of the airway measurements is lower than the intra-operator reliability³⁴, it was decided to evaluate only the interoperator reliability. Given that the measurement method of the upper airway used in this study is considered to be reliable, it was possible to evaluate the variation of upper airway measurements between repeated CT scans.

For the upper airway analysis, the primary confounding factors during 3D radiographic image acquisition include the individual's body, head, jaw, and tongue position, as well as the respiratory phase^{5, 14, 15}. A systematic review on the effect of head and tongue posture on the dimensions and morphology of the pharyngeal airway concluded that altered head, body, and jaw position had a significant effect on the upper airway dimensions, particularly on the retro-palatal and retro-glossal regions of the oropharynx¹⁴. In another study by Gurani et al.⁵, five sagittal MRI scans from ten subjects in different head and tongue positions were measured. They found that with the head in supine neutral position, the retropalatal, oropharyngeal, and total volumes increased significantly when the tongue was altered from a resting position to the tip of the tongue in contact with the posterior edge of the hard palate ($P \le 0.05$). Schwab et al.¹⁵ investigated the effects of respiration on the upper airway size using cine-CT in 15 normal subjects, 14 snorer/mildly apneic subjects, and 13 patients with OSA, all of whom were scanned in the supine position during awake

nasal breathing. In all three groups, there were significant dimensional changes at all anatomic levels of the upper airway during the respiratory cycle, especially in the OSA groups. Therefore, 3D assessment of the upper airway cannot be considered reliable unless all the above confounding factors are controlled during image acquisition. In this study, even with the same patient instruction during CT acquisition, different upper airway readings were found between two repeated CT scans within the same individual, which emphasizes the need for a more standardized patient instruction in terms of posture and breathing phase during image acquisition for evaluation. This needs to be developed and validated in future studies. As recommended by the American Association of Orthodontists White Paper³⁶, three-dimensional imaging of the airway is a snapshot of a specific moment of the breathing cycle and such technique currently does not represent a proper and reliable risk assessment tool for OSA. The results of the current study reinforce this recommendation.

This study can provide better insight into the real effects of potentially airwayaltering procedures on airway size and morphology, such as orthognathic surgery and orthodontics treatment. The differences in the upper airway measurements caused by orthognathic surgery, such as maxillomandibular advancement for OSA treatment, are probably larger than those between two distinct CT scans in our study. However, minor differences in the upper airway measurements should be interpreted cautiously, in particular when quantifying the effect of treatment on the upper airway parameters in a single individual. The SDD provides the amount of potential variation that should be taken into account when interpreting the measurement changes over time at individual level (see **Table 3**). For example, a SDD of the MCA at the total airway of 61.3 mm² was found in the present study. This suggests that a change in MCA can only be considered to represent a real change if it is larger than 61.3 mm².

Our study has several limitations. First, the sample size might be considered limited. However, it should be mentioned that the sample size is sufficient to demonstrate the considerable intra-individual variation in upper airway measurements. This variation is not expected to decrease with a larger sample size; only its estimate will be more precise³⁷. Second, although patients were provided with standardized instructions during CT acquisition, the retrospective nature of the data collection makes it impossible to verify this. While in theory this study could have been performed prospectively, using an enlarged field-of-view, this would have exposed patients who do not need imaging of the complete airway to a larger radiation dose, including vital structures, raising ethical objections to a prospective set-up. This is the reason why we tried to make use of this set of existing radiographic examinations. The fact that most of published studies on 3D evaluation of the upper airway are retrospective studies with various patient instruction protocols, emphasize the difficulty of this issue. Our study highlights that caution should be taken when interpreting the results of upper airway comparison and evaluation using CT, and that a strict protocol is required for repeated measurements and subsequent imaging sessions. Further studies with a larger sample size should be performed to re-determine the natural intra-individual variation of the airway between two CT scans acquired at different time points, using a standardized patient instruction protocol.

CONCLUSION

Our study demonstrates that the dimensions and morphology of the upper airway in CT scans can vary considerably within an individual at different time points, even if the same patient instruction protocol for image acquisition is used. The MCA of the total airway and all its segments generally had the largest intra-individual variation, with relative differences of approximately 30%. The length of the total airway had the lowest intra-individual variation, with relative difference of 4.9%. The relative differences of the sphericity between two scans in the total airway and each segment were all below 15%. The length of the total upper airway, and the sphericity of the total airway and each segment were stable over time. Therefore, such intra-individual variation should be considered when interpreting the results of upper airway comparison and evaluation using CT, and the smallest detectable difference is necessary to detect true differences in upper airway measurements over time at individual level.



Figure 1. Location of the anatomic landmarks on the midsagittal plane and upper airway subregions of interest defined according to the landmarks. Landmarks: PNS, posterior nasal spine; TUV, tip of the uvula; MUE, midpoint between tip of the uvula and tip of the epiglottis; TEP, tip of the epiglottis; and BEP, base of epiglottis. Subregions: V, velopharynx region; O, oropharynx region; T, tongue base region; and E, epiglottis region.



Figure 2. Measurement of minimum cross-sectional area (MCA), using Blender software. (A) "Dead space" (yellow shadow) of the evaluated airway. (B) "Islands" (red shadow) of the evaluated airway. (C) Anteroposterior dimension (AP) and lateral dimension (LAT) of MCA.

Table 1. Definition of airway parameters

Airway parameter	Unit	Definition
Volume	mm ³	Volume of upper airway
Length	mm	Length perpendicular to Frankfort Horizontal (FH) plane of upper airway
Surface area	mm²	Surface area of upper airway without the top and bottom
Minimum cross-sectional area (MCA)	mm²	At axial view, the minimum cross-sectional area of upper airway after removal of "islands" and "dead space"
Lateral dimension of MCA (LAT of MCA)	mm	At MCA, the maximum lateral dimension in an orientation perpendicular to the midsagittal plane
Anteroposterior dimension of MCA (AP of MCA)	mm	At MCA, the anteroposterior dimension on the midsagittal plane
Mean cross-sectional area (meanCSA)	mm²	Equal to the ratio of volume to length (V/L)
LAT/AP of MCA	dimensionless (ratio)	Ratio of LAT to AP of MCA
Airway uniformity	dimensionless (ratio)	Uniformity of upper airway, equal to the ratio of MCA to meanCSA (MCA/meanCSA)
Airway sphericity	dimensionless (ratio)	Mathematical measure of sphericity (how round an object is). A flat object has a sphericity of o, and a sphere has a sphericity of 1 ⁹ . Sphericity = $[\pi^{1/3} (6 \times V)^{2/3}]/SA'$ Note: the closed surface area with top and bottom (SA') was used for calculating the airway sphericity.

mer operator valla				Intra-individual					
		-			Variation				ICC
	Mean	SD	Mean	SD	Mean	SD	Min	Max	
Volume (mm ³) in t	the region	of				_			
Total airway	12603.3	5057.8	12455.9	5553.7	21.3	15.8	3.1	52.5	0.836
Velopharynx	5795.6	2122.3	5720.0	2337.1	15.9	15.0	0.8	47.7	0.913
Oropharynx	1707.3	1301.1	1679.7	1512.3	34.4	16.4	11.8	65.8	0.895
Tongue base	1613.3	1089.1	1572.6	1218.2	29.8	17.9	8.5	67.9	0.878
Epiglottis	3487.1	1650.2	3483.6	1514.9	25.4	29.4	2.6	82.6	0.665
Length (mm) in th	e region o	f							
Total airway	66.3	12.4	66.0	12.4	4.9	4.3	0.3	14.1	0.944
Velopharynx	33.5	6.3	32.4	4.6	6.9	6.2	1.8	18.8	0.850
Oropharynx	8.4	3.8	8.5	3.9	16.9	13.3	0.0	46.8	0.905
Tongue base	8.4	3.8	8.5	3.9	16.9	13.3	0.0	46.8	0.905
Epiglottis	16.0	3.0	16.5	2.6	10.2	14.1	2.9	49.4	0.668
Surface area (mm	²) in the re	gion of							
Total airway	5572.1	1865.1	5512.9	1916.7	16.0	13.2	3.1	36.6	0.829
Velopharynx	2661.5	695.2	2558.2	730.0	16.6	15.5	0.9	42.0	0.669
Oropharynx	607.0	471.9	617.5	487.2	22.5	12.6	6.4	47.2	0.966
Tongue base	525.3	287.6	538.6	303.5	19.0	14.2	2.2	43.4	0.913
Epiglottis	1778.3	707.0	1798.6	710.0	19.6	14.0	3.4	44.3	0.832
minCSA (mm²)in	the region	of							
Total airway	80.6	54.1	81.1	47.3	35.5	20.3	8.0	69.0	0.845
Velopharynx	97.5	75.4	96.6	62.4	30.2	22.4	5.7	69.0	0.912
Oropharynx	176.6	83.6	163.8	74.6	31.0	21.2	3.1	73.1	0.740
Tongue base	176.0	88.3	156.4	65.6	30.2	18.2	11.2	58.2	0.788
Epiglottis	126.0	63.1	108.2	53.6	36.9	26.1	8.0	81.2	0.669
LAT of minCSA (m	m) in the I	region of							
Total airway	15.7	6.7	16.3	7.4	22.4	17.2	2.9	44.2	0.845
Velopharynx	16.7	6.3	17.2	6.5	17.6	14.7	5.1	44.2	0.864
Oropharynx	21.9	8.2	23.1	7.2	19.1	19.3	0.5	51.6	0.809
Tongue base	21.7	6.7	21.2	5.6	12.5	15.4	1.2	49.4	0.822
Epiglottis	19.7	6.2	18.6	6.1	25.4	22.9	6.2	63.2	0.499
AP of minCSA (mi	n) in the re	egion of							
Total airway	6.5	3.3	6.1	2.1	27.7	13.3	9.4	51.7	0.781
Velopharynx	5.5	2.8	6.0	3.3	28.4	14.0	8.0	52.4	0.852
Oropharynx	10.9	2.6	10.4	2.6	16.9	10.3	2.7	30.9	0.713
Tongue base	11.5	3.1	10.6	2.9	26.5	20.1	3.7	69.4	0.552
Epiglottis	8.3	3.2	6.9	2.1	23.2	23.3	5.0	75.0	0.658
meanCSA (mm²) i	n the regio	on of							
Total airway	194.3	79.5	188.0	69.5	24.4	19.8	2.0	59.4	0.728
Velopharynx	181.4	72.7	180.5	82.8	22.5	17.5	2.7	59.1	0.861

Table 2. Descriptive statistics of variables (N = 10), intra-individual variation and repeatability (N = 10), and inter-operator variation and reliability (n = 5)

Inter-operator								
	Varia	ation		ICC				
Mean	SD	Min	Max					
2.8	2.1	1.4	6.5	0.997				
7.4	4.5	2.8	15.0	0.959				
1.8	1.3	0.1	3.4	0.999				
2.7	0.9	1.9	3.9	0.999				
4.9	4.0	1.1	11.0	0.989				
1.3	0.9	0.5	2.9	0.997				
3.2	2.1	0.7	6.6	0.980				
4.8	6.0	0	15.2	0.994				
4.8	6.0	0	15.2	0.994				
4.5	2.3	2.0	8.1	0.949				
3.8	3.9	0.2	9.7	0.993				
5.4	4.1	1.1	12.2	0.968				
2.8	2.0	1.3	6.0	0.999				
3.3	1.7	1.7	5.4	0.998				
6.9	10.9	0.6	26.3	0.979				
3.3	7.0	0	16.0	0.980				
3.3	7.1	0	16.0	0.980				
1.0	1.0	0	2.7	0.999				
2.9	3.5	0	7.2	0.996				
1.4	1.6	0.2	3.6	0.999				
1.2	1.9	0	4.5	0.996				
2.3	2.8	0	6.1	0.990				
3.2	2.9	0	6.9	0.988				
1.8	1.0	0.6	3.2	0.995				
2.3	3.9	0.4	9.3	0.982				
3.2	4.5	0	10.9	0.929				
3.2	4.5	0	10.9	0.960				
3.6	3.7	1.4	10.2	0.962				
3.5	3.2	0	7.6	0.893				
6.9	10.3	0	25.2	0.948				
2.2	1.5	0.2	3.6	0.998				
4.2	3.0	0.3	8.5	0.986				

	То		T1		Intra-individual				
						Variation			
	Mean	SD	Mean	SD	Mean	SD	Min	Max	
Oropharynx	196.0	87.4	181.9	72.3	30.3	21.3	2.4	59.3	0.712
Tongue base	193.6	89.4	175.3	61.1	26.9	18.3	1.0	57.0	0.757
Epiglottis	214.9	91.3	208.8	82.4	25.1	28.0	2.0	86.7	0.605
LAT/AP of MCA in t	he region	of							
Total airway	2.64	0.85	2.62	0.79	22.7	17.9	2.4	65.4	0.540
Velopharynx	3.29	1.08	3.40	2.03	24.2	10.9	12.5	40.9	0.734
Oropharynx	2.00	0.56	2.22	0.51	18.3	17.7	0	62.4	0.614
Tongue base	2.01	0.83	2.18	1.03	26.3	29.5	6.7	94.7	0.126
Epiglottis	2.46	0.46	2.87	1.09	18.0	18.6	0.5	57.0	0.569
Airway uniformity	in the reg	ion of							
Total airway	0.41	0.16	0.43	0.15	17.8	15.8	0.1	44.3	0.819
Velopharynx	0.51	0.20	0.52	0.18	18.9	15.2	0.8	41.6	0.809
Oropharynx	0.90	0.07	0.89	0.11	8.3	6.4	1.5	19.8	0.533
Tongue base	0.90	0.06	0.88	0.10	5.0	4.2	0.4	12.2	0.775
Epiglottis	0.58	0.13	0.52	0.14	21.9	17.1	5.8	59.9	0.372
Sphericity in the re	gion of								
Total airway	0.42	0.04	0.41	0.05	11.1	7.5	2.6	24.9	0.128
Velopharynx	0.47	0.05	0.48	0.07	11.4	8.7	1.3	28.8	0.279
Oropharynx	0.69	0.08	0.68	0.06	7.0	5.1	0.2	16.6	0.633
Tongue base	0.70	0.06	0.70	0.06	5.0	4.3	0.3	11.1	0.691
Epiglottis	0.49	0.04	0.49	0.05	8.2	6.5	1.6	18.6	0.304

Table 2. continued

AP, anteroposterior dimension; LAT, lateral dimension; Max, maximum; MCA, minimum cross-sectional area; meanCSA, mean cross-sectional area; Min, minimum; N, number of patients; n; number of CT datasets; SD, standard deviation.

	In	ter-operat	tor	
	ICC			
Mean	SD	Min	Max	
3.0	6.3	0	14.2	0.986
3.3	5.7	0.2	13.3	0.975
6.9	10.3	0	25.2	0.997
4.3	6.3	0	15.3	0.983
5.5	6.5	0	15.3	0.984
5.4	7.1	0.6	17.0	0.919
4.8	3.2	0.6	9.6	0.862
5.2	6.2	0.7	16.0	0.940
4.8	6.0	0.2	15.3	0.970
5.7	4.3	0.2	10.9	0.978
3.4	5.7	0.1	13.5	0.950
5.4	5.0	0.2	13.1	0.869
4.5	4.3	0.2	10.3	0.832
3.2	3.6	0	7.9	0.929
3.0	1.8	0.2	4.6	0.945
0.3	0.3	0.1	0.8	0.999
0.5	0.3	0.1	0.9	0.991
3.4	5.8	0.1	13.8	0.851

Table 3. Bland-Altman analysis of difference between two scans (To and T1) (N = 10)						
	To-T1	To-T1	95% CI			

	То	D-T1	To-T1		95% CI		SDD	
	Mean	SD	Mean	SD	Upper	Lower		
Volume (mm ³) in t	the region of							
Total airway	147.5	3037.6	2379.9	1719.9	6101.2	-5806.3	5953.8	
Velopharynx	75.6	928.9	703.0	565.7	1896.1	-1744.9	1820.5	
Oropharynx	27.7	645.0	514.0	351.3	1291.9	-1236.6	1264.3	
Tongue base	40.7	570.3	445.6	326.4	1158.5	-1077.1	1117.8	
Epiglottis	3.5	1296.8	857.5	929.8	2545.2	-2538.1	2541.7	
Length (mm) in th	e region of							
Total airway	0.3	4.2	3.1	2.6	8.4	-7.8	8.1	
Velopharynx	1.2	3.0	2.3	2.2	7.1	-4.8	5.9	
Oropharynx	-0.1	1.7	1.3	1.0	3.1	-3.4	3.3	
Tongue base	-0.1	1.7	1.3	1.0	3.1	-3.4	3.3	
Epiglottis	-0.6	2.3	1.5	1.8	4.0	-5.1	4.5	
Surface area (mm	²) in the regio	on of						
Total airway	59.2	1105.8	847.9	654.2	2226.7	-2108.3	2167.5	
Velopharynx	103.3	579.6	415.7	394.6	1239.2	-1032.6	1135.9	
Oropharynx	-10.5	126.0	111.0	48.0	236.4	-257.4	246.9	
Tongue base	-13.3	123.0	96.7	70.3	227.8	-254.4	241.1	
Epiglottis	-20.3	399.4	323.1	209.8	762.5	-803.2	782.8	
MCA (mm²) in the	region of							
Total airway	-0.6	28.3	24.3	12.1	54.9	-56	55.4	
Velopharynx	0.9	29.1	23.2	15.8	57.8	-56.1	57.0	
Oropharynx	12.8	57.1	48.1	29.6	124.8	-99.2	112.0	
Tongue base	19.6	50.6	45.7	26	118.9	-79.6	99.2	
Epiglottis	17.8	47.7	40.2	28.8	111.2	-75.7	93.5	
LAT of MCA (mm)	in the region	of						
Total airway	-0.6	3.9	3.1	2.2	7.1	-8.3	7.7	
Velopharynx	-0.5	3.3	2.6	1.9	6.0	-7.0	6.5	
Oropharynx	-1.2	4.8	3.6	3.2	8.2	-10.6	9.4	
Tongue base	0.5	3.7	2.5	2.7	7.8	-6.7	7.2	
Epiglottis	1.1	6.2	4.6	4.0	13.2	-11.0	12.1	
AP of MCA (mm) i	n the region o	of						
Total airway	0.3	1.8	1.6	0.7	3.9	-3.2	3.6	
Velopharynx	-0.5	1.7	1.5	0.8	2.8	-3.7	3.3	
Oropharynx	0.6	2.0	1.7	1.0	4.4	-3.3	3.9	
Tongue base	0.9	2.8	2.6	1.4	6.5	-4.6	5.6	
Epiglottis	1.4	2.2	1.8	1.9	5.8	-3.0	4.4	
meanCSA (mm²) i	n the region (of						
Total airway	6.3	55.0	41.9	33.5	114.1	-101.5	107.8	
Velopharvnx	0.9	43.5	33.3	25.8	86.3	-84.4	85.3	
Oropharynx	14.2	60.9	52.3	29.8	133.5	-105.2	119.3	
Tongue base	18.3	53.4	46.4	28.9	123.0	-86.3	104.7	
Fniglottis	61	77 3	52.1	54.8	157.6	-145 4	151 5	

	To-T1		To-	-T1	95%	95% CI		
	Mean	SD	Mean	SD	Upper	Lower		
LAT/AP of MCA in t	he region of							
Total airway	0.02	0.78	0.60	0.47	1.55	-1.51	1.53	
Velopharynx	-0.11	1.18	0.87	0.76	2.20	-2.42	2.32	
Oropharynx	-0.22	0.47	0.38	0.35	0.70	-1.14	0.92	
Tongue base	-0.17	1.24	0.70	1.01	2.26	-2.60	2.42	
Epiglottis	-0.41	0.78	0.54	0.68	1.12	-1.94	1.52	
Airway uniformity in the region of								
Total airway	-0.01	0.09	0.07	0.06	0.18	-0.19	0.18	
Velopharynx	-0.01	0.12	0.09	0.08	0.22	-0.24	0.23	
Oropharynx	0.01	0.09	0.07	0.05	0.18	-0.16	0.17	
Tongue base	0.01	0.05	0.04	0.03	0.12	-0.09	0.10	
Epiglottis	0.07	0.15	0.12	0.10	0.36	-0.23	0.29	
Sphericity in the re	gion of							
Total airway	0.01	0.06	0.05	0.03	0.12	-0.10	0.11	
Velopharynx	-0.01	0.07	0.06	0.05	0.13	-0.16	0.15	
Oropharynx	0.01	0.06	0.05	0.03	0.12	-0.11	0.12	
Tongue base	0	0.05	0.03	0.03	0.09	-0.09	0.09	
Epiglottis	0	0.06	0.04	0.03	0.11	-0.11	0.11	

Table 3. continued

AP, anteroposterior dimension; CI, confidence interval; LAT, lateral dimension; MCA, minimum cross-sectional area; meanCSA, mean cross-sectional area; N, number of patients; SD, standard deviation; SDD, smallest detectable difference.

REFERENCES

- 1. Eckert DJ, Malhotra A. Pathophysiology of adult obstructive sleep apnea. Proc Am Thorac Soc. 2008;5:144-53.
- 2. Schendel SA, Jacobson R, Khalessi S. Airway growth and development: a computerized 3-dimensional analysis. J Oral Maxillofac Surg. 2012;70:2174-83.
- 3. Gandedkar NH, Chng CK, Por YC, Yeow VKL, Ow ATC, Seah TE. Influence of bimaxillary surgery on pharyngeal airway in Class III deformities and effect on sleep apnea: A STOP-BANG questionnaire and cone-beam computed tomography study. J Oral Maxillofac Surg. 2017;75:2411-21.
- 4. Major MP, Flores-Mir C, Major PW. Assessment of lateral cephalometric diagnosis of adenoid hypertrophy and posterior upper airway obstruction: A systematic review. *Am J Orthod Dentofacial Orthop.* 2006;130:700-8.
- Gurani SF, Cattaneo PM, Rafaelsen SR, Pedersen MR, Thorn JJ, Pinholt EM. The effect of altered head and tongue posture on upper airway volume based on a validated upper airway analysis: An MRI pilot study. Orthod Craniofac Res. 2020;23:102-9.
- 6. Chen H, Aarab G, de Ruiter MH, de Lange J, Lobbezoo F, van der Stelt PF. Three-dimensional imaging of the upper airway anatomy in obstructive sleep apnea: a systematic review. *Sleep Med*. 2016;21:19-27.
- 7. Aboudara C, Nielsen I, Huang JC, Maki K, Miller AJ, Hatcher D. Comparison of airway space with conventional lateral headfilms and 3-dimensional reconstruction from cone-beam computed tomography. *Am J Orthod Dentofacial Orthop.* 2009;135:468-79.
- 8. El H, Palomo JM. Measuring the airway in 3 dimensions: A reliability and accuracy study. Am J Orthod Dentofacial Orthop. 2010;137:S50.e1-S52.
- 9. Chen H, Aarab G, Parsa A, de Lange J, van der Stelt PF, Lobbezoo F. Reliability of three-dimensional measurements of the upper airway on cone beam computed tomography images. *Oral Surg Oral Med Oral Pathol Oral Radiol.* 2016;122:104-10.
- 10. Abramson Z, Susarla S, August M, Troulis M, Kaban, L. Three-dimensional computed tomographic analysis of airway anatomy in patients with obstructive sleep apnea. J Oral Maxillofac Surg. 2010;68:354-62.
- 11. Obelenis Ryan DP, Bianchi J, Ignácio J, Wolford LM, Gonçalves JR. Cone-beam computed tomography airway measurements: Can we trust them? *Am J Orthod Dentofacial Orthop.* 2019;156:53-60.
- 12. Kaur S, Rai S, Kaur M. Comparison of reliability of lateral cephalogram and computed tomography for assessment of airway space. *Niger J Clin Pract.* 2014;17:629-36.
- 13. Liang Q, Auvenshine R. Pharyngeal airway dimension in patients before and after treatment of myofascial pain syndrome. *Cranio.* 2019;1-8.
- 14. Gurani SF, Di Carlo G, Cattaneo PM, Thorn JJ, Pinholt EM. Effect of head and tongue posture on the pharyngeal airway dimensions and morphology in three-dimensional imaging: A systematic review. J Oral Maxillofac Res. 2016;7:e1.
- Schwab RJ, Gefter WB, Hoffman EA, Gupta KB, Pack AI. Dynamic upper airway imaging during awake respiration in normal subjects and patients with sleep disordered breathing. *Am Rev Respir Dis.* 1993;148:1385-400.
- 16. Ferguson KA, Love LL, Ryan CF. Effect of mandibular and tongue protrusion on upper airway size during wakefulness. *Am J Respir Crit Care Med.* 1997;155:1748-54.
- 17. Van Holsbeke CS, Verhulst SL, Vos WG, De Backer JW, Vinchurkar SC, Verdonck PR, et al. Change in upper airway geometry between upright and supine position during tidal nasal breathing. J Aerosol Med Pulm Drug Deliv. 2014;27:51-7.
- Camacho M, Capasso R, Schendel S. Airway changes in obstructive sleep apnoea patients associated with a supine versus an upright position examined using cone beam computed tomography. J Laryngol Otol. 2014;128:824-30.
- 19. Guijarro-Martínez R, Swennen GR. Three-dimensional cone beam computed tomography definition of the anatomical subregions of the upper airway: A validation study. *Int J Oral Maxillofac Surg.* 2013;42:1140-9.
- 20. Ponce-Garcia C, Lagravere-Vich M, Cevidanes LHS, de Olivera Ruellas AC, Carey J, Flores-Mir C. Reliability of three-dimensional anterior cranial base superimposition methods for assessment of overall hard tissue changes: A systematic review. *Angle Orthod.* 2018;88:233-45.

- 21. Maes F, Collignon A, Vandermeulen D, Marchal G, Suetens P. Multimodality image registration by maximization of mutual information. *IEEE Trans Med Imaging*. 1997;16:187-98.
- 22. Hsieh YJ, Liao YF, Chen NH, Chen YR. Changes in the calibre of the upper airway and the surrounding structures after maxillomandibular advancement for obstructive sleep apnoea. *Br J Oral Maxillofac Surg.* 2014;52:445-51.
- 23. Abramson Z, Susarla SM, Lawler M, Bouchard C, Troulis M, Kaban LB. Three-dimensional computed tomographic airway analysis of patients with obstructive sleep apnea treated by maxillomandibular advancement. J Oral Maxillofac Surg. 2011;69:677-86.
- 24. Butterfield KJ, Marks PL, McLean L, Newton J. Linear and volumetric airway changes after maxillom and ibular advancement for obstructive sleep apnea. J Oral Maxillofac Surg. 2015;73:1133-42.
- 25. Fleiss JL. Analysis of Covariance and the Study of Change. In: *The Design and Analysis of Clinical Experiments*. New York: John Wiley & Sons; 1999. p. 186-219.
- 26. Bland JM, Altman DG. Statistical methods for assessing agreement between two methods of clinical measurement. *Lancet.* 1986;1:307-10.
- 27. Ogawa T, Enciso R, Shintaku WH, Clark GT. Evaluation of cross-section airway configuration of obstructive sleep apnea. Oral Surg Oral Med Oral Pathol Oral Radiol Endod. 2007;103:102-8.
- 28. Ciscar MA, Juan G, Martínez V, Ramón M, Lloret T, Mínguez J, et al. Magnetic resonance imaging of the pharynx in OSA patients and healthy subjects. *Eur Respir J*. 2001;17:79-86.
- 29. Hassan B, Nijkamp P, Verheij H, Tairie J, Vink C, van der Stelt P, et al. Precision of identifying cephalometric landmarks with cone beam computed tomography in vivo. *Eur J Orthod*. 2013;35:38-44.
- Klazen YP, Caron CJJM, Schaal SC, Borghi A, Van der Schroeff MP, Dunaway DJ, et al. What are the characteristics of the upper airway in patients with craniofacial microsomia? J Oral Maxillofac Surg. 2019;77:1869-81.
- 31. Luiz RR, Szklo M. More than one statistical strategy to assess agreement of quantitative measurements may usefully be reported. J Clin Epidemiol. 2005;58:215-6.
- 32. Alsufyani NA, Noga ML, Witmans M, Major PW. Upper airway imaging in sleep-disordered breathing: role of cone-beam computed tomography. *Oral Radiol.* 2017;33;161–9.
- Miracle AC, Mukherji SK. Conebeam CT of the head and neck, part 1: physical principles. AJNR Am J Neuroradiol. 2009;30:1088-95.
- 34. Zimmerman JN, Vora SR, Pliska BT. Reliability of upper airway assessment using CBCT. *Eur J Orthod.* 2019;41:101-8.
- 35. Kezirian EJ, Hohenhorst W, de Vries N. Drug-induced sleep endoscopy: The VOTE classification. *Eur Arch Otorhinolaryngol.* 2011;268:1233-6.
- 36. Behrents RG, Shelgikar AV, Conley RS, Flores-Mir C, Hans M, Levine M, et al. Obstructive sleep apnea and orthodontics: An American Association of Orthodontists White Paper. *Am J Orthod Dentofacial Orthop.* 2019;156:13-28.e1.
- 37. Hackshaw A. Small studies: strengths and limitations. Eur Respir J. 2008;32:1141-3.





Obstructive sleep apnea caused by acromegaly: Case report

Ning Zhou, Jean-Pierre T.F. Ho , Nico de Vries, Jan de Lange

Pulished in: Cranio, 2022. doi: 10.1080/08869634.2020.1776530

ABSTRACT

Background: Acromegaly is an uncommon syndrome caused by growth hormone-producing pituitary adenoma or pituitary gland hypertrophy. Acromegaly is known to be characterized by progressive somatic disfigurement and a wide range of systematic manifestations. This case study describes a rare case of severe obstructive sleep apnea (OSA) caused by acromegaly.

Clinical presentation: A female patient presented to the consultant clinic with the chief complaint of progressively worsening sleep and was diagnosed with severe OSA. Because of a peculiar facial appearance of the patient, acromegaly was suspected and confirmed by the findings of hormonal analysis and magnetic resonance imaging (MRI). After transsphenoidal resection of the pituitary adenoma, her OSA was almost cured, with residual apnea-hypopnea index of 5.5 events/h.

Conclusion: This case highlights the importance of a comprehensive clinical examination of OSA patients. In every sleep-related breathing disorder case, sleep clinicians should be aware of alternate problems that could cause upper airway obstruction.

Keywords: Obstructive sleep apnea; Acromegaly; Pituitary adenoma

INTRODUCTION

Acromegaly is a rare syndrome which affects both sexes equally, with an estimated annual incidence of three to four cases per million¹. It is characterized by excessive secretion of growth hormone (GH) and insulin-like growth factor type 1 (IGF-1), largely owing to a hyperfunctioning pituitary adenoma². It may present with a variety of clinical manifestations, the most common being acral and soft tissue overgrowth, diabetes mellitus, hypertension, and heart and respiratory comorbidities³. Currently, there is considerable evidence that acromegaly is associated with an increased risk of sleep apnea (SA), given that acromegaly alters the structure, elasticity, and function of the entire respiratory system⁴.

This paper reported a rare case of severe OSA caused by acromegaly. The patient's OSA was almost cured following transsphenoidal resection of the pituitary adenoma.

CASE PRESENTATION

In April 2013, a 50-year old woman who complained of poor sleep was diagnosed with moderately severe OSA at the Department of Otorhinolaryngology (ENT) and Head and Neck Surgery. Her polysomnogram (PSG) showed an apnea-hypopnea index (AHI) of 23.8 events/h (**Table 1**). A mandibular advancement device (MAD) was prescribed. As shown in **Table 1**, a follow-up PSG 15 months later, performed with the MAD in situ, revealed a residual mild positional OSA with an AHI of 8.7 events/h and an AHI supine of 14.3 events/h.

In November 2017, she presented to the Department of ENT and Head and Neck Surgery again, due to increasing complaints of poor sleep, snoring, apneas, choking and not being refreshed after a night's rest, in spite of compliant use of the MAD. On physical examination, she weighed 71 kg, height was 168 cm, BMI was 25.2 kg/m², and neck circumference was 34 cm. A PSG, the results of which were shown in **Table 1**, confirmed severe OSA (AHI = 74.1 events/h). A drug-induced sleep endoscopy (DISE) exhibited a total obstruction at velum and oropharynx levels, together with partial obstruction at tongue base and epiglottis levels. When the jaw thrust maneuver was applied, only the obstruction at tongue base level disappeared. Continuous positive airway pressure (CPAP) therapy was proposed and advocated by the ENT surgeon. However, the patient refused CPAP therapy.

The patient was referred to the Department of Oral and Maxillofacial Surgery (OMFS) for a maxillomandibular advancement (MMA). At this point, both the ENT surgeon and the maxillofacial surgeon noticed a peculiar facial appearance, e.g. thickened skin, widened nose and pronounced chin. The patient was therefore also referred to the Department of Internal Medicine.

A thorough workup at the Department of Internal Medicine, including hormonal analysis and magnetic resonance imaging (MRI) (**Figure 1**), revealed the diagnosis "acromegaly" due to a pituitary macroadenoma. Thereafter the patient was referred to the Department of Neurosurgery for resection of the pituitary macroadenoma.

In September 2018, the patient underwent an endoscopic transsphenoidal resection of the pituitary macroadenoma. The histopathology confirmed a plurihormonal pituitary adenoma. The postoperative course was uneventful.

A follow-up PSG 7 months after surgery demonstrated a dramatic improvement of OSA (AHI = 5.5 events/h), as shown in **Table 1**. The patient reported significant improvement of sleep quality and did not show any symptoms of residual OSA. Her IGF-1 level remained normal in hormonal analysis at 13 months after surgery (81 nmol/L pre-surgery vs 25 nmol/L post-surgery; reference range: 10–27 nmol/L).

However, at 13 months after surgery, the clinical examination showed a malocclusion which had not been present at the initial consult at the Department of OMFS. A conebeam computed tomography (CBCT) scan showed a significant condylar hyperplasia on the right side and a skeletal asymmetry. Different treatment options were discussed with the patient for creating an optimal occlusion and skeletal symmetry. However, the patient declined orthodontic treatment and/or orthognathic surgery, due to the fact that she did not want to have another operation. She was prescribed an Essix retainer to prevent further malocclusion and she had regular checkups to evaluate possible further progression of asymmetry.

DISCUSSION

Acromegaly is a rare disease which can lead to multi-systemic disorder. Patients with acromegaly are at a high risk of developing SA, specifically OSA. In a review by Attal et al., PSG-diagnosed OSA was found in an average of 69% of patients with active acromegaly in 11 studies (n = 239)⁵.

The pathophysiology of nocturnal airway obstruction in acromegaly is not yet understood, but it is thought to be multifactorial^{4, 6}. The persistent excessive GH and IGF-1 in acromegaly could cause anatomical changes, affecting the craniofacial bones and soft tissues, respiratory mucosa and cartilage, as well as the activity of the respiratory muscles, thus facilitating collapse or obstruction of the upper and middle oropharyngeal space during sleep.

This case, to the authors' knowledge, is the first reported acromegalic case where DISE was performed, by means of which dynamic evaluation of the upper airway during sleep was obtained. DISE revealed total obstruction during inspiration at the level of the soft palate and oropharynx with partial narrowing at the base of the tongue and epiglottis, which was similar to the outcome of the Muller maneuver in the study of Pelttari et al.⁷. In that study, significant narrowing during forced inspiration at the level of the soft palate was observed in 11 patients with acromegaly, while little if any narrowing occurred at the base of the tongue.

The effect of treated acromegaly on OSA is inconsistent. Tasbakan et al. found that OSA commonly persisted in well-controlled acromegaly patients, despite normal levels of IGF-1 and GH after adenomectomy⁶. In another study, it was demonstrated that surgical treatment of acromegaly had no significant effect on OSA⁸. In contrast, Buyse et al. reported the cases of three acromegalic patients with severe OSA who demonstrated a manifest improvement in apnea after treatment of acromegaly⁹.

In this case, after adenomectomy, severe OSA was dramatically reduced, as was proved by the postoperative PSG. Therefore, the cure or control of acromegaly could be associated with alleviation of OSA, although further studies are needed to investigate this relationship. Swelling of soft tissue, owing to direct stimulation of the epithelial sodium channel by the high GH and IGF-1 levels, is considered to play a major role in the onset of OSA for patients with acromegaly. Therefore, the reduction of soft tissue swelling after the treatment of acromegaly, possibly leading to better upper airway patency, may be the main explanation for the patient's relief from OSA^{5.10}.

This patient was satisfied with the final treatment outcome and believed that all her concerns about sleep quality and daily energy had been addressed. She did not want any further orthodontic and/or orthognathic treatment for her facial asymmetry and malocclusion. Long-time follow-up is therefore needed to monitor the possible progression of her facial abnormalities and malocclusion, together with her OSA status.

CONCLUSION

Acromegaly, as a rare risk factor for OSA, is often detected late owing to its insidious onset and slow progression. This case highlights the importance of clinical examination and diagnostic suspicion in OSA. Given the complex interplay of multiple etiologies in OSA, the assessment of patients with suspected OSA should take into consideration all the possible risk factors.



Figure 1. Preoperative magnetic resonance imaging in the axial (A), coronal (B) and sagittal (C) planes demonstrated the large pituitary mass (indicated by arrows)

	1 st PSG	2 nd PSG with MAD	3 rd PSG	4 th PSG	
Parameters	April 2013	August 2014	February 2018	April 2019	
AHI, events/h	23.8	8.7	74.1	5.5	
Al, events/h	14.8	5.7	55.5	2.4	
HI, events/h	9	2.9	18.6	3.2	
AHI supine, events/h	43.5	14.3	62.2	5.4	
AHI non-supine, events/h	17.6	3.5	81.6	5.9	
Mean O² saturation, %	95	96	93	95	
Minimum O² saturation, %	80	77	67	86	
3% ODI, events/h	18.6	14.8	75.9	16	
REM sleep rate, % TST	21.2	20.6	15.7	27.2	

Table 1. Results of polysomnogram

PSG, polysomnogram; MAD, mandibular advancement device; AHI, apnea-hypopnea index; AI, apnea index; HI, hypopnea index; ODI, oxygen desaturation index; REM, rapid eye movement; TST, total sleep time.

REFERENCES

- 1. Alexander L, Appleton D, Hall R, Ross WM, Wilkinson R. Epidemiology of acromegaly in the Newcastle region. *Clin Endocrinol*. 1980;12:71-9.
- 2. Ben-Shlomo A, Melmed S. Acromegaly. Endocrinol Metab Clin North Am. 2008;37:101-22.
- 3. AlDallal S. Acromegaly: a challenging condition to diagnose. Int] Gen Med. 2018;11:337-43.
- 4. Colao A, Ferone D, Marzullo P, Lombardi G. Systemic complications of acromegaly: epidemiology, pathogenesis, and management. *Endocr Rev.* 2004;25:102-52.
- 5. Attal P, Chanson P. Endocrine aspects of obstructive sleep apnea. J Clin Endocrinol Metab. 2010;95:483-95.
- Taşbakan MS, Kaçmaz Başoğlu Ö, Yıldırım Şimşir I, Saygılı F. Obstructive sleep apnea characteristics in patients with well-controlled acromegaly and their compliance with positive airway pressure therapy. *Turk* J Med Sci. 2017;47:1348-53.
- 7. Pelttari L, Polo O, Rauhala E, Vuoriluoto J, Aitasalo K, Hyyppä MT, et al. Nocturnal breathing abnormalities in acromegaly after adenomectomy. *Clin Endocrinol (Oxf)*. 1995;43:175-82.
- 8. Chemla D, Attal P, Maione L, Veyer AS, Mroue G, Baud D, et al. Impact of successful treatment of acromegaly on overnight heart rate variability and sleep apnea. J Clin Endocrinol Metab. 2014;99:2925-31.
- 9. Buyse B, Michiels E, Bouillon R, Bobbaers H, Demedts M. Relief of sleep apnoea after treatment of acromegaly: report of three cases and review of the literature. *Eur Respir J.* 1997;10:1401-4.
- 10. Imbert-Teboul M, Doucet A, Chanson P, Lombes M. Epithelial sodium channel is a key mediator of growth hormone-induced sodium retention in acromegaly. *Endocrinology*. 2008;149:3294-305.



General discussion

General discussion

The studies presented in this thesis aim to extend knowledge on the role of maxillomandibular advancement (MMA) for the treatment of obstructive sleep apnea (OSA), and more specifically the effects of MMA in OSA patients with and without anteroposterior maxillomandibular deficiency; predictors of surgical response/non-response to MMA; and comparison between MMA and other surgical interventions for OSA (i.e., multilevel surgery [MLS], upper airway stimulation [UAS]). Furthermore, the thesis aims to examine the intra-individual variation of upper airway measurements using computed tomography (CT). Additionally, a rare case with severe OSA caused by acromegaly is presented.

In this chapter, the main findings of the studies included in this thesis are presented and discussed. Where appropriate, suggestions for future research are made. This chapter ends by giving a general conclusion.

MAIN FINDINGS

Indications for MMA

Recognizing that maxillary and/or mandibular deficiency contribute to the development of OSA, MMA has been advocated in the management of OSA since the mid-1980s¹. In the early stage, MMA was the second phase of a two-phase airway reconstructive protocol for the treatment of OSA, where patients underwent MMA for persistent OSA in cases of incomplete response to phase I surgery (e.g., palatal surgery, genioglossus advancement, and/or hyoid myotomy)². As investigators have gained more insight into the role of MMA in OSA management, various surgical protocols for MMA have been proposed. However, to date, there has been no consistent indication for MMA. In the most current American Academy of Sleep Medicine (AASM) practice guidelines, it is recommended that "MMA is indicated for surgical treatment of severe OSA in patients who cannot tolerate or who are unwilling to adhere to positive airway pressure therapy, or in whom oral appliances, which are more often appropriate in mild and moderate OSA patients, have been considered and found ineffective or undesirable (Option)"³. Of note, this recommendation is given as an "Option" as it is drawn from low-quality evidence. Currently, the general indications for MMA are: moderate to severe OSA, and OSA patients with concomitant dentofacial deformity⁴. While the advancement of both jaws is functionally and esthetically beneficial to OSA patients with anteroposterior maxillomandibular deficiency (maxillary and mandibular retrognathia), there is a concern about the effect of MMA for OSA patients without such deficiency. In this thesis, we found no significant difference in the effects of MMA on respiratory function and facial esthetics between OSA patients with and without maxillomandibular deficiency, which supports the notion that MMA can be considered as an appropriate treatment for OSA patients without such deficiency (**chapter 2**).

Both complete concentric collapse at the palate (CCCp) and complete collapse at lateral pharyngeal wall observed during drug induced sleep endoscopy (DISE) have been suggested to be a negative predictor of treatment success of some non-continuous positive airway pressure (non-CPAP) therapies, such as upper airway surgery, and mandibular advancement device (MAD)⁵⁻⁷. CCCp is also defined as an absolute contraindication to unilateral UAS⁸. Most recently, Liu et al. suggested that MMA could be considered as a first-line treatment in OSA patients with those two specific airway collapse patterns⁹. However, limited evidence is available on this topic^{10,11}. The finding in **chapter 4**, that those two DISE phenotypes are not associated with surgical response to MMA, further supports this indication for MMA.

To conclude, current evidence supports that the indications for MMA could be expanded, which include OSA patients with coexisting dentofacial deformity; moderate to severe OSA patients who do not accept or have failed other forms of therapy, either with or without coexisting dentofacial deformity; presence of complete concentric collapse at the palate; and presence of complete collapse at lateral pharyngeal wall. In addition, especially younger people gain from a MMA procedure because earlier intervention could enhance life expectancy and besides that, MMA shows better results in younger patients¹². Nevertheless, since the precise indications and staging protocols for MMA remain undefined, large-scale prospective studies are necessary to further explore the role of MMA in OSA management and define the indications for MMA.

Predictors of response and non-response to MMA

MMA has been proven to be a highly effective surgical procedure for OSA, with a success rate of approximately 85% (**chapter 5**)¹², but there is still room for improvement. Identifying predictors of MMA surgical response can help offer adequate treatment plans upfront based on predicted therapeutic response. A meta-analysis by Holty et al. showed that younger age, lower preoperative weight and

apnea hypopnea index, as well as greater degree of maxillary advancement were predictive of increased surgical success¹². Several other potential predictors were also identified previously, such as a smaller neck circumference¹³ and mandibular retrognathia¹⁴.

In this thesis, we further surveyed the potential predictors from the most common clinically available data (patient-related, polysomnographic, cephalometric, and surgical data) and DISE findings, respectively (**chapter 3** and **4**). In **chapter 3**, we found that the existence of cardiovascular disease, higher central apnea index, and larger superior posterior airway space were the independent predictors of non-response to MMA (**chapter 3**).

In terms of the pattern of upper airway collapse during DISE, we found that complete anteroposterior epiglottic collapse was a negative predictor of response to MMA (**chapter 4**). It should be noted that **chapter 4** did not distinguish primary and secondary epiglottis collapse. Future prospective studies with larger study groups are necessary to investigate if the predictive value of primary and secondary epiglottic collapse to MMA is different. In addition, multilevel collapse of the upper airway is prevalent in patients with moderate to severe OSA (**chapter 4**)¹⁵. It may be the case that certain combinations of collapse levels are associated with the surgical response to MMA. Future research is needed to answer this question.

Taken together, given the absence of consistent findings and limited evidence, it is still a challenge for clinicians to precisely identify responders or nonresponders to MMA. More research is necessary to investigate which parameters can reliably predict the surgical outcome and thus should be taken into consideration in the patient selection procedure. In addition, because patients' satisfaction is also an important outcome measurement in the management of OSA, the predictors of patients' satisfaction in MMA surgery should be identified in future studies.

MMA versus other surgical approaches

Moderate to severe OSA is usually characterized by multilevel collapse of the upper airway¹⁵, which highlights the need for surgical therapies able to resolve multilevel collapse. MMA, MLS, and UAS are all multilevel approaches^{8, 11, 16}. MMA, as stated above, is generally indicated for patients with moderate to severe OSA, and OSA patients with concomitant dentofacial deformity⁴. In terms of MLS, AASM practice

Chapter 9

guidelines suggest that it is acceptable in patients with narrowing of multiple sites in the upper airway, particularly if they have failed UPPP as a sole treatment³. In contrast to MMA and MLS, UAS therapy has stricter indication criteria, viz., moderate to severe OSA (AHI 15-65 events/h) with less than 25% central and/or mixed apneas, intolerance or failure of PAP treatment, and absence of CCCp during DISE¹⁷.

Despite that the indications for the three types of surgery are not exactly same, a subset of patients may be expected to benefit from all of them when no generally accepted indicative factors for treatment failure are found (e.g., significant dentofacial deformity, CCCp during DISE). These approaches have shown favorable treatment outcomes for OSA^{12, 17-19}, but there is a paucity of evidence on the comparison between them. Consequently, there is no adequate evidence for final decisionmaking regarding the choice of surgery types. On the basis of the meta-analyses in chapter 5, it was concluded that MMA might offer greater improvements in AHI compared to MLS, but the complication rate of MMA is higher. It should be noted that the conclusion of **chapter 5** is drawn from the comparison between MMA and MLS by separately pooling results from studies on each type of surgery. To compare the efficacy and safety of MMA and UAS in the treatment of OSA, a systematic review was performed including 21 studies on MMA and 9 studies on UAS (chapter 6). Due to the noticeable differences between the MMA cohort and UAS cohort in age and baseline AHI, it is not feasible to generate a solid conclusion on the comparison of efficacy and safety of these two procedures. Notably, in addition to treatment efficacy and safety, cost of the therapy option is important in assisting decision-making in OSA management. To our knowledge, few studies have assessed the cost-effectiveness of MLS²⁰ or UAS^{21, 22}, but no study has assessed the such information about MMA. This precludes the comparison of cost-effectiveness among MMA, MLS, and UAS.

Future research should entail well designed comparative studies among MMA, MLS, and UAS with larger sample size and long-term follow-up, in which thorough assessment of objective respiratory and sleep parameters, subjective outcomes, quality of life, morbidity and mortality, and cost-effectiveness is performed. Such findings will help optimize shared decision-making between clinicians and patients for OSA treatment.

Three-dimensional evaluation of the upper airway

Three-dimensional (3D) evaluation of the upper airway has been widely used to investigate the role of upper airway anatomy in the pathogenesis of OSA and in the treatment outcome of therapies for OSA (e.g., MMA, MAD)²³⁻²⁶. It has been suggested that upper airway dimension is affected by multiple factors, such as body position, mandibular movement, tongue position, breathing stage, wakefulness versus sleep, and sleep stage²⁷⁻³⁰. In the clinic, it is still a challenge to standardize all these interfering factors during imaging acquisition³¹. This may result in considerable intra-individual variation in 3D upper airway measurements, which hampers accurate upper airway assessment.

Obelenis Ryan et al. evaluated the differences in upper airway volume in the same patients from their consecutive cone-beam computed tomography (CBCT) scans. They found that volumetric measurements of the upper airway differed between different CBCT scans with identical scanning and patient positioning protocols³¹. In addition to volumetric parameters of the upper airway, our study in **chapter 7** also guantified the intra-individual variation in the areal and linear parameters on the supine CT scans at two different time points (3 to 6 months interval). It was found that there was considerable intra-individual variation in CT volumetric, areal, and linear measurements of the upper airway, even if a same patient instruction protocol for image acquisition was used. As recommended by the American Association of Orthodontists White Paper, 3D imaging of the airway is a snapshot of a specific moment of the breathing cycle, and such technique currently does not represent a proper and reliable risk assessment tool for OSA³². The finding in **chapter 7** reinforces this recommendation. Such intra-individual variation should be taken into account when interpreting and comparing upper airway measurements on CT, which may also apply to other 3D imaging techniques (e.g., CBCT, magnetic resonance imaging [MRI]). Despite the many advantages of CT, the main downside is the inevitable radiation exposure. Future research should be performed to evaluate such variation on different airway imaging techniques. Moreover, a standardized and validated patient instruction protocol in imaging acquisition should be developed in future research.

Personalized treatment for OSA

The heterogeneity of OSA is reflected by various risk factors^{33, 34}, pathophysiological causes³⁵, clinical presentations³⁶⁻³⁸, and consequences^{38, 39}. Recognition of this heterogeneity is imperative, because the treatment for OSA can be personalized based on individual characteristics.

Risk factors

In addition to the most common risk factors, such as obesity, increased age, and male gender, some endocrine and metabolic disorders are also associated with OSA^{33, 34}. This thesis presented a patient who was referred for consultation of MMA for severe OSA but was subsequently diagnosed with acromegaly. The patient's OSA was almost completely resolved after transsphenoidal resection of a pituitary adenoma (**chapter 8**). Current evidence has shown a variable response of OSA to the treatment of acromegaly^{40, 41}, hence clinicians who treat patients with acromegaly should not assume that OSA will recover by curing the underlying endocrine disorder. Nevertheless, our case still highlights that the assessment of patients with suspected OSA should involve all possible risk factors for the purpose of personalized and effective treatment.

Pathophysiologic phenotypes

Available evidence has indicated that phenotyping is the prerequisite for developing personalized medicine in OSA⁴²⁻⁴⁴. Eckert et al. proposed a three-point (Passive critical closing pressure of the upper airway, Arousal threshold, Loop gain, and Muscle responsiveness [PALM]) scale to categorize OSA patients according to pathophysiologic phenotypes and thus select the appropriate therapy for the individual patient⁴⁵. Briefly, in patients who had severe and moderate pharyngeal anatomical impairment (i.e., category 1 and 2a), there is an indication for CPAP or other anatomic intervention (e.g., MAD, upper airway surgery, positional therapy). In the category 2b and 3, nonanatomic traits are likely to contribute importantly to OSA pathogenesis, and single or combined therapy that targets on nonanatomic traits may be required⁴⁵. For example, for the patients with impaired muscle function, hypoglossal nerve simulation¹⁸, muscle training⁴⁶, and pharmacologic therapy⁴⁷ may be helpful, while oxygen therapy⁴⁸ and carbonic anhydrase inhibitors⁴⁹ may be beneficial in the patients with high loop gain. However, it should be highlighted that implementing this concept into the clinical setting is difficult now, as it is complicated to measure the PALM variables. A simplified phenotyping tool should therefore be developed for routine clinical use.

Clinical phenotypes

In addition to the pathophysiologic phenotypes, different OSA phenotypes based on clinical features and polysomnographic data have been proposed as well, in order to assess disease severity, OSA consequence, and predict treatment outcome⁵⁰⁻⁵². Ye et

al. firstly applied cluster analysis to identify clinical phenotypes in an Icelandic Sleep Apnea Cohort (822 patients with moderate to severe OSA). Three main clusters were identified, which were "disturbed sleep group", "minimally symptomatic group", and "excessive daytime sleepiness group"⁵¹. Among these three clusters, the probabilities of having comorbid hypertension, diabetes, and cardiovascular disease were highest in the "minimally symptomatic group" and lowest in the "excessive daytime sleepiness group"⁵¹. Currently, there is accumulating evidence on the association between clinical phenotypes and response to treatment for OSA such as CPAP⁵³, MAD⁵⁴, and MMA (**chapter 3** and **4**). This thesis revealed that the presence of cardiovascular disease, higher baseline central apnea index, larger superior posterior airway space (**chapter 3**), and the presence of complete anteroposterior epiglottic collapse on DISE (**chapter 4**) might be unfavorable clinical phenotypes for surgical response to MMA.

Genotypes

Assessment of differences in molecular signatures between individuals now plays an important role in developing personalized medicine. However, neither biomarkers nor "-omics" analyses (e.g., genomics, proteomics) have been extensively used in the study on OSA^{55, 56}. To truly achieve a personalized OSA treatment, a combination of physiological phenotypes, clinical phenotypes, and genotypes, integrated with patients' preference, may be necessary in the future.

CONCLUSION

The following can be concluded from this thesis:

- There is no significant difference in the effect of MMA on respiratory function and facial esthetics between OSA patients with and without maxillomandibular deficiency (chapter 2).
- 2. The presence of cardiovascular disease, higher preoperative central apnea index, and larger preoperative superior posterior airway space are independently associated with non-response to MMA for OSA (**chapter 3**).
- DISE can be a promising tool in order to identify patients who will or will not respond to MMA for treating OSA. Patients with complete anteroposterior epiglottic collapse may be less suitable candidates for MMA (chapter 4).

- 4. Based on a systematic review and meta-analysis, it was concluded that both MMA and MLS are effective treatment options for OSA to improve respiratory parameters and patient-reported outcomes. Compared to MLS, MMA may be more effective in improving OSA. However, the complication rate of MMA is higher (chapter 5).
- 5. Based on a systematic review, it was concluded that both MMA and UAS are effective and generally safe therapies for OSA. However, due to the limitations of the included studies, there is no evidence yet to directly compare these two procedures (**chapter 6**).
- 6. The dimensions and morphology of the upper airway in CT scans may vary considerably within an individual at different time points, even if the same patient instruction protocol for image acquisition is used. Such intra-individual variation should be considered when interpreting the results of upper airway comparison and evaluation using CT, and the smallest detectable difference is necessary to detect true differences in upper airway measurements over time at the individual level (**chapter 7**).
- 7. A comprehensive clinical examination of OSA patients is important. Given the complex interplay of multiple etiologies in OSA, the assessment of patients with suspected OSA should take into consideration all possible risk factors (chapter 8).

REFERENCES

- Riley RW, Powell NB, Guilleminault C, Nino-Murcia G. Maxillary, mandibular, and hyoid advancement: an alternative to tracheostomy in obstructive sleep apnea syndrome. *Otolaryngol Head Neck Surg.* 1986;94:584-8.
- 2. Li KK, Riley RW, Powell NB, Guilleminault C. Maxillomandibular advancement for persistent obstructive sleep apnea after phase I surgery in patients without maxillomandibular deficiency. *Laryngoscope*. 2000;110:1684-8.
- 3. Aurora RN, Casey KR, Kristo D, et al. Practice parameters for the surgical modifications of the upper airway for obstructive sleep apnea in adults. *Sleep*. 2010;33:1408-13.
- 4. Yu MS, Ibrahim B, Riley RW, Liu SY. Maxillomandibular Advancement and Upper Airway Stimulation: Extrapharyngeal Surgery for Obstructive Sleep Apnea. *Clin Exp Otorhinolaryngol.* 2020;13:225-33.
- Petri N, Christensen IJ, Svanholt P, Sonnesen L, Wildschiødtz G, Berg S. Mandibular advancement device therapy for obstructive sleep apnea: a prospective study on predictors of treatment success. *Sleep Med.* 2019;54:187-94.
- 6. Soares D, Sinawe H, Folbe A), et al. Lateral oropharyngeal wall and supraglottic airway collapse associated with failure in sleep apnea surgery. *Laryngoscope*. 2012;122:473-9.
- 7. Iwanaga K, Hasegawa K, Shibata N, et al. Endoscopic examination of obstructive sleep apnea syndrome patients during drug-induced sleep. *Acta Otolaryngol Suppl.* 2003;(550):36-40.
- 8. Ong AA, Murphey AW, Nguyen SA, et al. Efficacy of Upper Airway Stimulation on Collapse Patterns Observed during Drug-Induced Sedation Endoscopy. *Otolaryngol Head Neck Surg.* 2016;154:970-7.
- 9. Liu SY, Awad M, Riley RW. Maxillomandibular advancement: contemporary approach at Stanford. *Atlas Oral Maxillofac Surg Clin North Am.* 2019;27:29-36.
- 10. Kastoer C, Op de Beeck S, Dom M, et al. Drug-Induced Sleep Endoscopy Upper Airway Collapse Patterns and Maxillomandibular Advancement. *Laryngoscope*. 2020;130:E268-E274.
- 11. Liu SY, Huon L-K, Iwasaki T, et al. Efficacy of maxillomandibular advancement examined with druginduced sleep endoscopy and computational fluid dynamics airflow modeling. *Otolaryngol Head Neck Surg.* 2016;154:189-95.
- 12. Holty JE, Guilleminault C. Maxillomandibular advancement for the treatment of obstructive sleep apnea: a systematic review and meta-analysis. *Sleep Med Rev.* 2010;14:287-97.
- 13. de Ruiter MHT, Apperloo RC, Milstein DMJ, de Lange J. Assessment of obstructive sleep apnoea treatment success or failure after maxillomandibular advancement. *Int J Oral Maxillofac Surg.* 2017;46:1357-62.
- Makovey I, Shelgikar AV, Stanley JJ, Robinson A, Aronovich S. Maxillomandibular Advancement Surgery for Patients Who Are Refractory to Continuous Positive Airway Pressure: Are There Predictors of Success? J Oral Maxillofac Surg. 2017;75:363-70.
- 15. Sharifian MR, Zarrinkamar M, Alimardani MS, et al. Drug Induced Sleep Endoscopy in Obstructive Sleep Apnea. *Tanaffos.* 2018;17:122-6.
- 16. Lin HC, Weaver EM, Lin HS, Friedman M. Multilevel Obstructive Sleep Apnea Surgery. *Adv Otorhinolaryngol.* 2017;80:109-15.
- 17. Suurna MV, Steffen A, Boon M, et al. Impact of Body Mass Index and Discomfort on Upper Airway Stimulation: ADHERE Registry 2020 Update. *Laryngoscope*. 2021;131:2616-24.
- 18. Woodson BT, Strohl KP, Soose RJ, et al. Upper Airway Stimulation for Obstructive Sleep Apnea: 5-Year Outcomes. Otolaryngol Head Neck Surg. 2018;159:194-202.
- 19. Su YY, Lin PW, Lin HC, et al. Systematic review and updated meta-analysis of multi-level surgery for patients with OSA. *Auris Nasus Larynx*. 2022;49:421-30.
- 20. Tan KB, Toh ST, Guilleminault C, Holty JE. A Cost-Effectiveness Analysis of Surgery for Middle-Aged Men with Severe Obstructive Sleep Apnea Intolerant of CPAP. J Clin Sleep Med. 2015;11:525-35.
- 21. Pietzsch JB, Liu S, Garner AM, Kezirian EJ, Strollo PJ. Long-Term Cost-Effectiveness of Upper Airway Stimulation for the Treatment of Obstructive Sleep Apnea: A Model-Based Projection Based on the STAR Trial. *Sleep.* 2015;38:735-44.
- 22. Pietzsch JB, Richter AK, Randerath W, et al. Clinical and Economic Benefits of Upper Airway Stimulation for Obstructive Sleep Apnea in a European Setting. *Respiration*. 2019;98:38-47.

- 23. Schwab RJ, Gupta KB, Gefter WB, Metzger LJ, Hoffman EA, Pack AI. Upper airway and soft tissue anatomy in normal subjects and patients with sleep-disordered breathing. Significance of the lateral pharyngeal walls. *Am J Respir Crit Care Med*. 1995;152(5 Pt 1):1673-89.
- 24. Barbero M, Flores-Mir C, Blanco JC, et al. Tridimensional upper airway assessment in male patients with OSA using oral advancement devices modifying their vertical dimension. J Clin Sleep Med. 2020;16:1721-9.
- 25. Schendel SA, Broujerdi JA, Jacobson RL. Three-dimensional upper-airway changes with maxillomandibular advancement for obstructive sleep apnea treatment. *Am J Orthod Dentofacial Orthop*. 2014;146:385-93.
- 26. Ryan CF, Lowe AA, Li D, Fleetham JA. Three-dimensional upper airway computed tomography in obstructive sleep apnea. A prospective study in patients treated by uvulopalatopharyngoplasty. *Am Rev Respir Dis.* 1991;144:428-32.
- 27. Gurani SF, Di Carlo G, Cattaneo PM, Thorn JJ, Pinholt EM. Effect of Head and Tongue Posture on the Pharyngeal Airway Dimensions and Morphology in Three-Dimensional Imaging: a Systematic Review. J Oral Maxillofac Res. 2016;7:e1.
- Schwab RJ, Gefter WB, Hoffman EA, Gupta KB, Pack AI. Dynamic upper airway imaging during awake respiration in normal subjects and patients with sleep disordered breathing. *Am Rev Respir Dis.* 1993;148:1385-400.
- 29. Ferguson KA, Love LL, Ryan CF. Effect of mandibular and tongue protrusion on upper airway size during wakefulness. *Am J Respir Crit Care Med.* 1997;155:1748-54.
- Trudo FJ, Gefter WB, Welch KC, Gupta KB, Maislin G, Schwab RJ. State-related changes in upper airway caliber and surrounding soft-tissue structures in normal subjects. *Am J Respir Crit Care Med.* 1998;158:1259-70.
- 31. Obelenis Ryan DP, Bianchi J, Ignácio J, Wolford LM, Gonçalves JR. Cone-beam computed tomography airway measurements: Can we trust them? *Am J Orthod Dentofacial Orthop*. 2019;156:53-60.
- 32. Behrents RG, Shelgikar AV, Conley RS, et al. Obstructive sleep apnea and orthodontics: An American Association of Orthodontists White Paper. *Am J Orthod Dentofacial Orthop.* 2019;156:13-28.e1.
- 33. Young T, Skatrud J, Peppard PE. Risk factors for obstructive sleep apnea in adults. JAMA. 2004;291:2013-6.
- 34. Al Lawati NM, Patel SR, Ayas NT. Epidemiology, risk factors, and consequences of obstructive sleep apnea and short sleep duration. *Prog Cardiovasc Dis.* 2009;51:285-93.
- 35. Eckert DJ, Malhotra A. Pathophysiology of adult obstructive sleep apnea. Proc Am Thorac Soc. 2008;5:144-53.
- 36. Arnardottir ES, Bjornsdottir E, Olafsdottir KA, Benediktsdottir B, Gislason T. Obstructive sleep apnoea in the general population: highly prevalent but minimal symptoms. *Eur Respir J*. 2016;47:194-202.
- Bjorvatn B, Lehmann S, Gulati S, Aurlien H, Pallesen S, Saxvig IW. Prevalence of excessive sleepiness is higher whereas insomnia is lower with greater severity of obstructive sleep apnea. Sleep Breath. 2015;19:1387-93.
- 38. Mazzotti DR, Keenan BT, Lim DC, Gottlieb DJ, Kim J, Pack AI. Symptom Subtypes of Obstructive Sleep Apnea Predict Incidence of Cardiovascular Outcomes. *Am J Respir Crit Care Med*. 2019;200:493-506.
- 39. Roca GQ, Redline S, Claggett B, et al. Sex-specific association of sleep apnea severity with subclinical myocardial injury, ventricular hypertrophy, and heart failure risk in a community-dwelling cohort: the Atherosclerosis Risk in Communities–Sleep Heart Health Study. *Circulation*. 2015;132:1329-37.
- 40. Davi MV, Dalle Carbonare L, Giustina A, et al. Sleep apnoea syndrome is highly prevalent in acromegaly and only partially reversible after biochemical control of the disease. *Eur J Endocrinol*. 2008;159:533-40.
- 41. Rosenow F, Reuter S, Deuss U, et al. Sleep apnoea in treated acromegaly: relative frequency and predisposing factors. *Clin Endocrinol (Oxf)*. 1996;45:563-9.
- 42. Zinchuk AV, Gentry MJ, Concato J, Yaggi HK. Phenotypes in obstructive sleep apnea: a definition, examples and evolution of approaches. *Sleep Med Rev.* 2017;35:113-23.
- Op de Beeck S, Dieltjens M, Verbruggen AE, et al. Phenotypic labelling using drug-induced sleep endoscopy improves patient selection for mandibular advancement device outcome: a prospective study. J Clin Sleep Med. 2019;15:1089-99.
- 44. Gray EL, McKenzie DK, Eckert DJ. Obstructive sleep apnea without obesity is common and difficult to treat: evidence for a distinct pathophysiological phenotype. J Clin Sleep Med. 2017;13:81-8.
- 45. Eckert DJ, White DP, Jordan AS, Malhotra A, Wellman A. Defining phenotypic causes of obstructive sleep apnea. Identification of novel therapeutic targets. *Am J Respir Crit Care Med.* 2013;188:996-1004.
- 46. Camacho M, Certal V, Abdullatif J, et al. Myofunctional Therapy to Treat Obstructive Sleep Apnea: A Systematic Review and Meta-analysis. *Sleep*. 2015;38:669-75.
- 47. Taranto-Montemurro L, Sands SA, Edwards BA, et al. Desipramine improves upper airway collapsibility and reduces OSA severity in patients with minimal muscle compensation. *Eur Respir J.* 2016;48:1340-50.
- 48. Wellman A, Malhotra A, Jordan AS, Stevenson KE, Gautam S, White DP. Effect of oxygen in obstructive sleep apnea: role of loop gain. *Respir Physiol Neurobiol*. 2008;162:144-51.
- 49. Edwards BA, Sands SA, Eckert DJ, et al. Acetazolamide improves loop gain but not the other physiological traits causing obstructive sleep apnoea. J Physiol. 2012;590:1199-211.
- 50. Lacedonia D, Carpagnano GE, Sabato R, et al. Characterization of obstructive sleep apnea-hypopnea syndrome (OSA) population by means of cluster analysis. *J Sleep Res.* 2016;25:724-30.
- 51. Ye L, Pien GW, Ratcliffe SJ, et al. The different clinical faces of obstructive sleep apnoea: a cluster analysis. *Eur Respir J.* 2014;44:1600-7.
- 52. Zinchuk AV, Jeon S, Koo BB, et al. Polysomnographic phenotypes and their cardiovascular implications in obstructive sleep apnoea. *Thorax*. 2018;73:472.
- 53. Gagnadoux F, Le Vaillant M, Paris A, et al. Relationship between OSA clinical phenotypes and CPAP treatment outcomes. *Chest.* 2016;149:288-90.
- 54. Sutherland K, Takaya H, Qian J, Petocz P, Ng AT, Cistulli PA. Oral Appliance Treatment Response and Polysomnographic Phenotypes of Obstructive Sleep Apnea. J Clin Sleep Med. 2015;11:861-8.
- 55. Pack Al. Application of Personalized, Predictive, Preventative, and Participatory (P4) Medicine to Obstructive Sleep Apnea. A Roadmap for Improving Care? *Ann Am Thorac Soc.* 2016;13:1456-67.
- 56. Khalyfa A, Gileles-Hillel A, Gozal D. The Challenges of Precision Medicine in Obstructive Sleep Apnea. *Sleep Med Clin.* 2016;11:213-26.



Summary in English and Dutch

SUMMARY

Obstructive sleep apnea (OSA) is a common sleep-related breathing disorder, which is characterized by repetitive episodes of complete or partial upper airway collapse during sleep. Undiagnosed and untreated OSA can potentially lead to serious medical issues and substantial economic costs. Although continuous positive airway pressure (CPAP) is the gold standard treatment for OSA, there is a need for other treatment modalities as the efficacy of CPAP is often hampered by low tolerance and poor compliance.

Maxillomandibular advancement (MMA) has been considered as a highly effective surgical therapy for moderate to severe OSA. The position of MMA within the arsenal of treatment options for OSA, however, is still not fully understood. The main purpose of this thesis was to expand the body of knowledge concerning the role of MMA in OSA treatment, which may contribute to optimizing surgical management of OSA. Furthermore, intra-individual variation of upper airway measurements using computed tomography (CT) was studied. Additionally, treatment of a severe OSA case caused by acromegaly was described.

In **chapter 2**, we compared the effect of MMA on respiratory function between OSA patients with and without anteroposterior maxillomandibular deficiency based on respiratory parameters measured by polysomnography (PSG) and patient satisfaction with postoperative breathing. Also, the effect of MMA on facial esthetics was compared between the two groups based on cephalometric measurements and patient satisfaction with postoperative esthetics. We found that there was no significant difference in the effects of MMA on respiratory function and facial esthetics between OSA patients with and without such deficiency. This supports the view that MMA can also be considered as an appropriate treatment for OSA patients without maxillomandibular deficiency.

In **chapter 3**, the potential predictors of surgical response to MMA were explored from the most common clinically available data, i.e., patient-related, polysomnographic, cephalometric, and surgical variables. In this retrospective study, one hundred patients were included. Surgical response was achieved in 66 patients (66%). Multivariate logistic regression revealed that the presence of cardiovascular disease, higher baseline central apnea index, and larger superior posterior airway space were independently associated with non-response to MMA. If confirmed in future research, these predictors may guide patient selection for MMA.

Drug induced sleep endoscopy (DISE) is a unique tool for dynamic visualization of upper airway collapse. **Chapter 4** presents a retrospective study aiming to investigate if DISE findings were predictive of surgical response to MMA. Furthermore, the predictive value of jaw thrust maneuver during DISE in terms of MMA outcome was explored. A total of 64 patients were included. Thirty-nine patients were responders, and 25 were nonresponders. After adjusting for baseline characteristics and surgical characteristics, the presence of complete anteroposterior epiglottic collapse was independently associated with non-response to MMA. No significant association was found between the effect of jaw thrust maneuver during DISE on upper airway patency and response to MMA. It was concluded that DISE could be a promising tool for predicting MMA surgical outcome, and patients with complete anteroposterior epiglottic collapse might be less suitable candidates for MMA.

Chapter 5 describes a systematic review and meta-analysis on the comparison of clinical efficacy and safety between MMA and multilevel surgery (MLS) for OSA. In total, twenty studies on MMA and 39 studies on MLS were included. We found that regardless of disease severity (i.e., baseline apnea hypopnea index [AHI]), MMA might be a more effective therapy compared to MLS in improving OSA, demonstrating a significantly higher AHI reduction and success rate. However, the rates of major and minor complications of MMA are higher than those of MLS.

Chapter 6 presents a systematic review aiming to compare the clinical efficacy and safety of MMA and upper airway stimulation (UAS) in the treatment of OSA. Twenty-one studies on MMA and nine studies on UAS were included. Current evidence suggests that both MMA and UAS are effective and generally safe therapies for OSA. However, due to the noticeable differences between MMA cohort and UAS cohort in age and baseline AHI, a solid conclusion cannot be drawn about the comparison between these two therapies.

Chapter 7 focuses on the intra-individual variation in the upper airway measurements on supine computed tomography (CT) scans at two different time points. This is relevant, due to the fact that three-dimensional upper airway measurements are commonly used to evaluate the role of upper airway anatomy in pathogenesis of OSA and to assess the treatment effect or disease progress. In addition to a reliable measurement method, understanding the degree of intra-individual variation of the upper airway on different scans is imperative to achieve accurate evaluation and comparison of the upper airway. Therefore, ten subjects with paired CT datasets (3-6 months interval) were studied. There was considerable intra-individual variation in CT measurements of the upper airway, with the same patient instruction protocol for image acquisition. The minimum cross-sectional area of the total airway and all its segments generally had the largest variation, while the length of the total airway had the lowest variation. Sphericity was the only parameter that was stable over time both in the total airway and each segment. Our results suggested that such variation should be considered when interpreting the results of upper airway evaluation and comparison using CT.

Chapter 8 presents a female patient of middle age whose severe OSA was caused by an uncommon syndrome – acromegaly. The patient presented to the Department of Otorhinolaryngology due to her progressively worsening sleep and was diagnosed with severe OSA (AHI = 74.1 events/h). She was referred to the Department of Oral and Maxillofacial Surgery for consultation of MMA surgery for OSA. Due to her peculiar facial appearance, acromegaly was suspected and then confirmed by hormonal analysis and magnetic resonance imaging. After transsphenoidal resection of her pituitary adenoma, the patient's OSA was almost resolved (AHI = 5.5 events/h). This case highlights the importance of a comprehensive clinical examination for OSA patients.

10

SAMENVATTING

Obstructief Slaap Apneu (OSA) is een veel voorkomende slaap gerelateerde ademhalingsstoornis, die wordt gekenmerkt door herhaalde episodes van volledige of gedeeltelijke collaps van de bovenste luchtwegen tijdens het slapen. Niet-gediagnosticeerde en onbehandelde OSA kan leiden tot potentieel ernstige medische en economisch kostbare gevolgen. Hoewel continue positieve luchtwegdruk (CPAP) de gouden standaardbehandeling voor OSA is, is er behoefte aan andere behandelingsmodaliteiten, aangezien de werkzaamheid van CPAP vaak wordt belemmerd door lage tolerantie en slechte therapietrouw.

Maxillomandibular advancement (MMA) wordt beschouwd als een zeer effectieve chirurgische therapie voor matige tot ernstige OSA. De positie van MMA binnen het arsenaal aan behandelingsopties voor OSA is echter nog steeds niet volledig begrepen. Het belangrijkste doel van dit proefschrift was het uitbreiden van de kennis over de rol van MMA in de behandeling van OSA, wat kan bijdragen aan de optimalisatie van de chirurgische behandeling van OSA. Verder werd de intraindividuele variatie van metingen van de bovenste luchtwegen met behulp van computertomografie (CT) bestudeerd. Daarnaast werd de behandeling van een ernstig geval van OSA, veroorzaakt door acromegalie, beschreven.

In **hoofdstuk 2** vergeleken we het effect van MMA op de ademhalingsfunctie tussen OSA-patiënten met en zonder anteroposterieure maxillomandibulaire deficiëntie op basis van respiratoire parameters gemeten door polysomnografie (PSG) en patiënttevredenheid voor wat betreft postoperatieve ademhaling. Ook werd het effect van MMA op de esthetiek van het gezicht vergeleken tussen de twee groepen op basis van cephalometrische metingen en de patiënttevredenheid over zijn of haar postoperatieve esthetiek. We vonden dat er geen significant verschil was in de effecten van MMA op de ademhalingsfunctie en gezichtsesthetiek tussen OSApatiënten met en zonder een dergelijke deficiëntie. Dit ondersteunt de opvatting dat MMA ook kan worden beschouwd als een geschikte behandeling voor OSA-patiënten zonder maxillomandibulaire deficiëntie.

In **hoofdstuk 3** werden mogelijke voorspellers van chirurgische respons op MMA onderzocht op basis van de meest voorkomende klinisch beschikbare gegevens, d.w.z. patiëntgerelateerde, polysomnografische, cephalometrische en chirurgische variabelen. In deze retrospectieve studie werden honderd patiënten geïncludeerd. Een chirurgische respons werd bereikt bij 66 patiënten (66%). Multivariate logistische regressie toonde aan dat de aanwezigheid van hart- en vaatziekten, een hogere centrale apneu-index bij aanvang en een grotere ruimte van de achterste luchtwegen onafhankelijk waren geassocieerd met non-respons op MMA. Indien bevestigd in toekomstig onderzoek, kunnen deze voorspellers helpen bij de selectie van patiënten voor MMA.

Drug induced sleep endoscopy (DISE) is een unieke hulpmiddel voor dynamische visualisatie van de ineenstorting van de bovenste luchtwegen. Hoofdstuk 4 presenteert een retrospectieve studie met als doel te onderzoeken of de bevindingen tijdens DISE voorspellend waren voor de chirurgische respons op MMA. Verder werd de voorspellende waarde van naar ventraal plaatsen van de onderkaak tijdens DISE in termen van MMA-uitkomst onderzocht. In totaal werden 64 patiënten geïncludeerd. Negenendertig patiënten waren responders en 25 waren nonresponders. Na correctie voor baseline kenmerken en chirurgische kenmerken, was de aanwezigheid van volledige anteroposterior epiglottische collaps onafhankelijk geassocieerd met non-respons op MMA. Er werd geen significant verband gevonden tussen het effect van het naar ventraal plaatsen van de onderkaak tijdens de DISE op de doorgankelijkheid van de bovenste luchtwegen en de respons op MMA. Er werd geconcludeerd dat DISE een veelbelovend hulpmiddel zou kunnen zijn voor het voorspellen van de uitkomst van MMA-chirurgie, en dat patiënten met een volledige anteroposterior epiglottische collaps mogelijk minder geschikte kandidaten zijn voor MMA.

Hoofdstuk 5 beschrijft een systematische review en meta-analyse waarin de klinische werkzaamheid en veiligheid tussen MMA en multilevel chirurgie (MLS) voor OSA werd vergeleken. In totaal werden twintig onderzoeken naar MMA en negenendertig onderzoeken naar MLS geïncludeerd. We ontdekten dat ongeacht de ernst van de ziekte (d.w.z. baseline apneu hypopneu index [AHI]), MMA een effectievere therapie zou kunnen zijn in vergelijking met MLS bij het verbeteren van OSA, wat een significant hogere AHI-reductie en succespercentage aantoont. Het aantal grote en kleine complicaties van MMA is echter hoger dan die van MLS.

Hoofdstuk 6 presenteert een systematische review met als doel de klinische werkzaamheid en veiligheid van MMA en upper airway stimulation (UAS) bij de

Chapter 10

behandeling van OSA te vergelijken. Eenentwintig studies over MMA en negen studies over UAS werden opgenomen. Het huidige bewijs suggereert dat zowel MMA als UAS effectieve en over het algemeen veilige therapieën voor OSA zijn. Vanwege de merkbare verschillen tussen MMA-cohort en UAS-cohort in leeftijd en baseline AHI, kan er echter geen solide conclusie worden getrokken over de vergelijking tussen deze twee therapieën.

Hoofdstuk 7 richt zich op de intra-individuele variatie van de bovenste luchtweg gemeten bij computertomografie (CT) scans in liggende positie van eenzelfde patient op twee verschillende tijdstippen. Dit is relevant vanwege het feit dat driedimensionale metingen van de bovenste luchtwegen vaak worden gebruikt om de rol van de anatomie van de bovenste luchtwegen in de pathogenese van OSA te evalueren en om het behandeleffect of ziekteverloop te beoordelen. Naast een betrouwbare meetmethode is het noodzakelijk om de mate van intra-individuele variatie van de bovenste luchtwegen op verschillende scans te weten voor een nauwkeurige evaluatie en vergelijking van de bovenste luchtwegen. Daarom werden tien proefpersonen met twee CT-datasets (3-6 maanden interval) bestudeerd. Er was aanzienlijke intraindividuele variatie in CT-metingen van de bovenste luchtwegen, met hetzelfde patiëntinstructieprotocol voor beeldacquisitie. De minimale dwarsdoorsnede van de totale luchtweg en al zijn segmenten vertoonde over het algemeen de grootste variatie, terwijl de lengte van de totale luchtweg de laagste variatie vertoonde. Sfericiteit was de enige parameter die stabiel was in de tijd, zowel in de totale luchtweg als in elk segment. Onze resultaten suggereerden dat een dergelijke variatie in overweging moet worden genomen bij het interpreteren van de resultaten van evaluatie en vergelijking van de bovenste luchtwegen met behulp van CT.

Hoofdstuk 8 presenteert een vrouwelijke patiënt van middelbare leeftijd wiens ernstige OSA werd veroorzaakt door een ongewoon syndroom, namelijk acromegalie. De patiënte meldde zich op de afdeling KNO vanwege haar steeds slechter wordende slaap en werd gediagnosticeerd met ernstige OSA (AHI = 74.1 episodes/uur). Ze werd verwezen naar de afdeling Mondzieken, Kaa- en Aangezichtschirurgie om te beoordelen of MMA een chirurgische therapeutische optie zou kunnen zijn. Vanwege haar bijzonder gelaatsuitdrukking werd acromegalie vermoed en vervolgens bevestigd doorhormoon analyse en Magnetic Resonance Imaging (MRI). Na transsfenoïdale resectie van haar hypofyseadenoom was haar OSA zo goed als verdwenen (AHI = 5.5 gebeurtenissen/uur). Deze casus benadrukt het belang van een uitgebreid klinisch onderzoek voor OSA-patiënten.



APPENDICES

- Author contributions
- About the author
- PhD portfolio
- Acknowledgements

AUTHOR CONTRIBUTIONS

CHAPTER 2 Effects of maxillomandibular advancement on respiratory function and facial esthetics in obstructive sleep apnea patients with versus without maxillomandibular deficiency *Ning Zhou, Jean-Pierre T.F. Ho, Frank Lobbezoo, Ghizlane Aarab, Nico de Vries, Jan de Lange* Conception and design of study: NZ, JH, JL Acquisition of data: NZ Analysis of data: NZ Drafting of article and/or critical revision: NZ, JH, FL, GA, NV, JL

CHAPTER 3 | Maxillomandibular advancement for obstructive sleep apnea: A retrospective prognostic factor study for surgical response Ning Zhou, Jean-Pierre T.F. Ho, Wouter P. Visscher, Naichuan Su, Frank Lobbezoo, Jan de Lange Conception and design of study: NZ, JH, JL Acquisition of data: NZ, JH Analysis of data: NZ, JH, NS Drafting of article and/or critical revision: NZ, JH, WV, NS, FL, JL

CHAPTER 4 | Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement *Ning Zhou, Jean-Pierre T.F. Ho, Nico de Vries, Pien F.N. Bosschieter, Madeline J.L. Ravesloot, Jan de Lange* Conception and design of study: NZ, JH, JL Acquisition of data: NZ, PB Analysis of data: NZ, JH Drafting of article and/or critical revision: NZ, JH, NV, PB, MR, JL

CHAPTER 5 | Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis Ning Zhou, Jean-Pierre T.F. Ho, Zhengfei Huang, René Spijker, Nico de Vries, Ghizlane Aarab, Frank Lobbezoo, Madeline J.L. Ravesloot, Jan de Lange Conception and design of study: NZ, JH, NV, GA, FL, MR, JL Acquisition of data: NZ, ZH, RS Analysis of data: NZ Drafting of article and/or critical revision: NZ, JH, ZH, RS, NV, GA, FL, MR, JL **CHAPTER 6** Maxillomandibular advancement and upper airway stimulation for treatment of obstructive sleep apnea: A systematic review *Ning Zhou, Jean-Pierre T.F. Ho, René Spijker, Ghizlane Aarab, Nico de Vries, Madeline J.L. Ravesloot, Jan de Lange* Conception and design of study: NZ, JH, JL Acquisition of data: NZ, JH Analysis of data: NZ, JH Drafting of article and/or critical revision: NZ, JH, RS, GA, NV, MR, JL

CHAPTER 7 | Intra-individual variation of upper airway measurements based on computed tomography *Ning Zhou, Jean-Pierre T.F. Ho, Cornelis Klop, Ruud Schreurs, Ludo F. M. Beenen, Ghizlane Aarab, Jan de Lange* Conception and design of study: NZ, JH, CK, RS, JL Acquisition of data: NZ, CK, RS Analysis of data: NZ Drafting of article and/or critical revision: NZ, JH, CK, RS, LB, GA, JL

CHAPTER 8 Obstructive sleep apnea caused by acromegaly: Case report *Ning Zhou, Jean-Pierre T.F. Ho, Nico de Vries, Jan de Lange* Conception and design of study: N/A Acquisition of data: NZ, NV Analysis of data: N/A Drafting of article and/or critical revision: NZ, JH, NV, JL

ABOUT THE AUTHOR

Ning Zhou was born in 1992 in Rongcheng, China. In 2010, she graduated from high school at the No. 2 high school, Rongcheng, China. In the same year, she started her dentistry study in Shandong University, where she met her other half. After graduating as a dentist in 2015, she was exempted from admission exam and joined the 3-year master program of orthodontics in Shandong University. In 2018, she graduated as an orthodontist and went to Amsterdam with her partner for their PhD. During her PhD, she conducted research at the department of Oral and Maxillofacial Surgery at the Amsterdam UMC and Academic Centre for Dentistry Amsterdam and the department of Orofacial Pain and Dysfunction at the Academic Centre for Dentistry Amsterdam, under the supervision of Prof. dr. Jan de Lange, Prof. dr. Frank Lobbezoo, Prof. dr. Nico de Vries, and Prof. dr. Ghizlane Aarab.

PHD PORTFOLIO

Peer-reviewed publications

Included in this thesis

- Zhou N, Ho JPTF, Lobbezoo F, Aarab G, de Vries N, de Lange J. Effects of maxillomandibular advancement on respiratory function and facial aesthetics in obstructive sleep apnoea patients with versus without maxillomandibular deficiency. *Int J Oral Maxillofac Surg.* 2022;S0901-5027(22)00327-7. doi:10.1016/j. ijom.2022.08.012. Epub ahead of print.
- **Zhou N**, Ho JPTF, Visscher WP, Su N, Lobbezoo F, de Lange J. Maxillomandibular advancement for obstructive sleep apnea: A retrospective prognostic factor study for surgical response. *Sleep Breath*. 2022. doi:10.1007/s11325-022-02731-x. Epub ahead of print.
- Zhou N, Ho JPTF, de Vries N, Bosschieter PFN, Ravesloot MJL, de Lange J. Evaluation of drug-induced sleep endoscopy as a tool for selecting patients with obstructive sleep apnea for maxillomandibular advancement. J Clin Sleep Med. 2022;18(4):1073-1081. doi:10.5664/jcsm.9802
- Zhou N, Ho JPTF, Huang Z, Spijker R, de Vries N, Aarab G, Lobbezoo F, Ravesloot MJL, de Lange J. Maxillomandibular advancement versus multilevel surgery for treatment of obstructive sleep apnea: A systematic review and meta-analysis. Sleep Med Rev. 2021;57:101471. doi:10.1016/j.smrv.2021.101471
- **Zhou N**, Ho JPTF, Spijker R, Aarab G, de Vries N, Ravesloot MJL, de Lange J. Maxillomandibular advancement and upper airway stimulation for treatment of obstructive sleep apnea: A systematic review. J Clin Med. 2022;11(22):6782. doi:10.3390/jcm11226782
- **Zhou N**, Ho JPTF, Klop C, Schreurs R, Beenen LFM, Aarab G, de Lange J. Intraindividual variation of upper airway measurements based on computed tomography. *PLoS One*. 2021;16(11):e0259739. doi:10.1371/journal.pone.0259739
- **Zhou N**, Ho JPTF, De Vries N, De Lange J. Obstructive sleep apnea caused by acromegaly: Case report. *Cranio.* 2022;40(5):451-453. doi:10.1080/08869634.2020 .1776530

Other publications

 Ho JPTF, Zhou N, Verbraecken J, de Vries N, de Lange J. Central and mixed sleep apnea related to patients treated with maxillomandibular advancement for obstructive sleep apnea: A retrospective cohort study. J Craniomaxillofac Surg. 2022;50(7):537-542. doi:10.1016/j.jcms.2022.06.005

- Ho JPTF, Donders HCM, Zhou N, Schipper K, Su N, de Lange J. Association between the degree of obstructive sleep apnea and the severity of COVID-19: An explorative retrospective cross-sectional study. *PLoS One.* 2021;16(9):e0257483. doi:10.1371/journal.pone.0257483
- Huang Z, Aarab G, Ravesloot MJL, Zhou N, Bosschieter PFN, van Selms MKA, den Haan C, de Vries N, Lobbezoo F, Hilgevoord AAJ. Prediction of the obstruction sites in the upper airway in sleep-disordered breathing based on snoring sound parameters: a systematic review. *Sleep Med.* 2021;88:116-133. doi:10.1016/j. sleep.2021.10.015
- Huang Z, Zhou N, Lobbezoo F, Almeida FR, Cistulli PA, Dieltjens M, Huynh NT, Kato T, Lavigne GJ, Masse JF, Pliska BT, van de Rijt L, Sutherland K, Thymi M, Vanderveken OM, de Vries R, Aarab G. Dental sleep-related conditions and the role of oral healthcare providers: a scoping review. *Sleep Med Rev.* 2022;67:101721. doi: 10.1016/j.smrv.2022.101721

Congress & oral presentations

- 2020 Virtual American Association of Oral and Maxillofacial Surgeons Annual Meeting. Presentation: "Maxillomandibular Advancement versus Multilevel Surgery for Treatment of Obstructive Sleep Apnea: A Systematic Review and Meta-Analysis".
- 2021 25th Congress of the European Association for Cranio Maxillo Facial Surgery. Presentation: "Maxillomandibular Advancement versus Multilevel Surgery for Treatment of Obstructive Sleep Apnea: A Systematic Review and Meta-Analysis".
- 2021 Virtual American Academy of Dental Sleep Medicine Annual Meeting. Presentation: "Evaluation of drug-induced sleep endoscopy as a patient selection tool of maxillomandibular advancement for OSA".
- 2022 26th Congress of the European Association for Cranio Maxillo Facial Surgery. Presentation: "Comparison of the effects of maxillomandibular advancement on respiration function and facial esthetics between obstructive sleep apnea patients with and without maxillomandibular deficiency".

List of awards

- 2021 American Academy of Dental Sleep Medicine Student Research Award
- 2022 American Academy of Dental Sleep Medicine Student Research Award

Courses

- Statistics and Methodology, Academisch Centrum Tandheelkunde Amsterdam (ACTA), 2020
- Scientific Integrity, ACTA, 2020
- Oral Biology, ACTA, 2021
- Scientific Writing and English Presenting, ACTA, 2021

ACKNOWLEDGEMENTS

During my PhD journey, I have received a great deal of support, help, and encouragement. I could never have completed this dissertation alone. I would like to express my sincere gratitude to my promoters, participants in my research, colleagues, friends, and family.

To **Prof. dr. Jan de Lange**: Dear Jan, I deeply appreciate you for offering me the precious opportunity to do research at AMC. As an oral surgeon and director of department, you always have a full schedule, but whenever I asked for help, you was on my side and helped me out efficiently. I deeply appreciate your constant trust, support, encouragement, and guidance for my PhD projects. Also, thank you for creating an open, participatory, and independent work atmosphere for the team, which provides everyone with sufficient space for personal development and individual contribution. I appreciate those a lot!

To **Prof. dr. Frank Lobbezoo**: Dear Frank, the moment I asked Ghizlane and you in Jinan if it is possible to do a PhD program in ACTA still appears vividly before me. I am very grateful for your trust and for offering me the opportunity to do research at ACTA! Thank you for your continued concern and support not only in my research but also in my life. Also, thank you for your guidance and contributions to my projects. Your feedback on my manuscripts were detailed and insightful, from which I benefit a lot!

To **Prof. dr. Chizlane Aarab**: Dear Ghizlane, thank you for your trust and encouragement in my application to the doctoral program at ACTA. Without your encouragement, I would never have the courage to do research in a new country. Thank you for your concern and support in both my research and life. Also, I appreciate all your efforts and guidance in my projects. I am impressed by your critical thinking and rigorous academic attitude. I have learned a lot from your feedback on my manuscripts. Thank you!

To **Prof. dr. Nico de Vires**: Dear Nico, I deeply appreciate all your effort, support, encouragement, and guidance for my research. Whenever I needed your help, you always support me as much as you can. Thank you for helping me in the ethical approval application, so that I could proceed with my projects smoothly. Thank you for your patience in answering all my questions and even offering me a valuable

chance to visit the DISE room. Thank you for your prompt feedback and constructive comments on my manuscripts. Also, thank you for your trust, and I am so grateful to have been involved in a chapter of your book. Thank you, Nico!

I would like to express my profound gratitude to the **participants in my research**. I could not have conducted the projects without your participation.

To **Jean-Pierre T.F. Ho**: Dear JP, working with you in the last four years is a pleasance! You are warm, approachable, and helpful. Thank you for your tremendous and continued effort for my research. Whenever I met problems, you always tried your best to help me, even sometimes you were already quite busy with your own work. Your support made my PhD journey easier and more enjoyable. We often discuss our projects together, your encouragement and affirmation help me become a confident and independent researcher. Also, I appreciate Krista and you for your concern for my life in the Netherlands!

To **Dr. Madeline J.L. Ravesloot**: Dear Madeline, to me, is a pleasure to work with you! What impressed me most is your insightful and constructive feedback and comments on my manuscripts, which brought the papers to a higher level. I deeply appreciate your support and precious input in my projects.

To **Cornelis Klop**: Dear Niels, thank you for all your effort in my project. You helped me with participants inclusion, CT measurements, figures, etc. Also, you taught me how to use software to do CT measurements. Whenever I met problems during the measurements, you always answered my questions and helped me solve the problems efficiently. I appreciate those a lot!

To **Dr. Ruud Schreurs**: Dear Ruud, I am grateful for your suggestions to my project, which helped me reconsider the research question. Thank you for your efforts in revising my manuscript again and again. Also, thank you for making a nice picture for my case report!

To **René Spijker**: Dear René, thank you for your contribution to the two systematic reviews. I appreciate that you not only assisted me - a novice at systematic review - with the literature search, but also gave me lots of valuable advices and guidance on how to do a systematic review in a standard way.

To **Dr. Naichuan Su**: Dear Naichuan, thank you for the support with the statistical analysis of my projects. Your suggestions and feedback on the statistics were always detailed and prompt. Also I would like to thank you for sharing your study and work experience with Zhengfei and me.

To **Anja Van de Velde-Muusers**: Dear Anja, I am very thankful for your help with the ethical approvals of my PhD projects. I could not have finalized data collection timely without your help. I do appreciate your support!

To **Dr. Pien F.N. Bosschieter**: Dear Pien, thank you for showing me the procedure of DISE. I still remember that you tried your best to explain DISE findings to me, which is really helpful to my research. Also, I really appreciate your feedback and revision of the manuscript!

To **Dr. Yuelian Liu**: Dear Yuelian, thank you for all your support to all Chinese students at ACTA including me. No matter what difficulties we meet, either in study or in life, you always try your best to help us. Thank you!

To **my other colleagues in the department of Oral and Maxillofacial Surgery, AMC**: Dear Simone, Tom, Renée, Sophie, Karel, Misha, and many others. Thank you for your help with my research and your kindness. Although I didn't meet you very often, I do enjoy the pleasant atmosphere of this big family!

I also would like to thank **all my colleagues in the department of Orofacial Pain and Dysfunction, ACTA**.

To **Prof. dr. Jing Guo**: Dear Jing, I feel so lucky to be your student! Thank you for your guidance during my three-year master program. I have learned a lot from you, not only in terms of professional knowledge but also in terms of conducting oneself. You have always been my role model. Thank you for encouraging and supporting Zhengfei and me to start the PhD program in the Netherlands. You are the strong back for us!

Lastly, I want to express my love and the deepest gratitude to **my family** for your selfless love and constant support: 致我亲爱的家人们,感谢你们一直以来对我的关爱、支持和鼓励,感谢你们对我每个人生决定的尊重、理解和信任! ♥